# The regulation of plasminogen activator inhibitor-1 (PAI-1) and other uPA system members in the gastric epithelium by *Helicobacter pylori*

Thesis submitted in accordance with the requirements of the University of Liverpool

for the degree of Doctor in Philosophy by

Susan Helen Kenny

April 2008

This thesis is dedicated to my loving parents, especially to you mum. You always believed in me and gave me the strength and determination to carry on. I love and miss you both deeply and am so proud to be your daughter. Thank you from the bottom of my heart for everything *xxxx* 

## **Acknowledgements**

I would like to say a very big thank you to my primary supervisor Professor Andrea Varro for all of her guidance, help and support throughout my PhD, it was much appreciated.

I would also like to thank my second supervisor, Professor Graham Dockray for great scientific advice and many useful discussions. A big thanks to Professor Rod Dimaline for his wise words of wisdom, real-time PCR and molecular biology help, and making of the plasmids used in this thesis. I would also like to thank Dr Mark Pritchard for his help with animal work, friendly advice, human biopsy samples, and for my Tuesday morning lessons, teaching me about the world of Gastroenterology and some of the pathologies associated with it! I am very grateful to Dr Islay Steele for many things including teaching me lots of techniques over the past four years and having the patience of a Saint, thank you very much. Thank you to Cédric Duval, for help and assistance in many areas. The help of Debbie Sales for culturing *H. pylori* was very much appreciated, as was that of the Biomedical Service Unit staff, particularly Joanne Sanders and Katharine Gittins, thank you. I would like to acknowledge John Atherton and Richard Argent for the making and provision of *H. pylori* mutant strains. Thank you also to other members of Green Block (past and present) for your friendship, entertainment and continued encouragement.

I would like to thank my family and friends for their continual support and help in keeping me sane. My thesis is dedicated to my Mum for so many reasons, mainly because I really could never have done this without her; she was definitely one in a million. To my sister Jane and Brizer, you have been amazing,

Π

thanks for always being there for me. My Aunty Angela, there are not many people in the world like you, you truly are a star and I will be eternally grateful for all of your love and acts of kindness you have shown to me. I would also like to thank my brother David and Aunty Cathy for everything you do, even the little things. Finally, thank you (in too many ways to mention) to my boyfriend Mike, you have been my rock and I am lucky to have you in my life, 'Rwy'n dy garu di'.

# **Contents**

| Dedication                     | I    |
|--------------------------------|------|
| Acknowledgements               | II.  |
| Contents                       | IV   |
| List of figures                | IX   |
| List of tables                 | XI   |
| Abbreviations                  | XII  |
| Publications and presentations | XVI  |
| Abstract                       | XVII |

| <u>CH</u> | CHAPTER 1: INTRODUCTION1  |    |
|-----------|---|----|
| 1.1       | Overview  | 2  |
| 1.2       | Gastric architecture  | 3  |
|           | 1.2.1 Organisation of the gastric mucosa                              | 3  |
|           | 1.2.2 Gastric epithelium  | 3  |
|           | 1.2.3 Role of gastric epithelial cells in acid secretion              | 6  |
|           | 1.2.4 Gastric submucosa   | 7  |
|           | 1.2.5 Role of gastrin in gastric epithelium function                  | 8  |
|           | 1.2.6 Gastric mucosal protection                                      | 10 |
| 1.3       | Helicobacter pylori infection of the gastric epithelium               | 14 |
|           | 1.3.1 Epidemiology  | 15 |
|           | 1.3.2 Gastric epithelium remodelling and gastric cancer               | 16 |
|           | 1.3.3 Helicobacter pylori virulence factors                           | 19 |
|           | 1.3.3.1 Outer membrane proteins                                       | 20 |
|           | 1.3.3.2 <i>cag</i> pathogenicity island                               | 21 |
|           | 1.3.3.3 CagA  | 23 |
|           | 1.3.3.4 VacA  | 25 |
|           | 1.3.3.5 Helicobacter felis  | 26 |
|           | 1.3.4 Environmental factors   | 27 |
|           | 1.3.5 <i>Helicobacter</i> and gastrin                                 | 28 |
|           | 1.3.6 Host response to <i>H. pylori</i> infection                     | 29 |
|           | 1.3.7 Animal models of <i>Helicobacter</i> -induced gastric carcinoma | 31 |
|           | 1.3.7.1 INS-GAS mice  | 32 |

| 1.3.7.2 <i>Helicobacter</i> vaccination                | 33 |
|--|----|
| 1.4 Epithelial-mesenchymal interactions and signalling |    |
| 1.5 Urokinase plasminogen activator (uPA) system       |    |
| 1.5.1 Urokinase plasminogen activator                  |    |
| 1.5.2 Urokinase plasminogen activator receptor (uPAR)  | 41 |
| 1.5.2.1 Cell signalling via uPAR                       |    |
| 1.5.3 Plasminogen activator inhibitors (PAIs)          | 44 |
| 1.5.4 Plasminogen activator inhibitor-1 (PAI-1)        |    |
| 1.5.4.1 PAI-1 in disease states                        | 47 |
| 1.5.5 Pathophysiological functions of the uPA system   | 49 |
| 1.5.5.1 Cell motility                                  |    |
| 1.5.5.2 Cancer   |    |
| 1.6 Overall aims and objectives                        |    |
| 1.6.1 Objective 1                                      |    |
| 1.6.2 Objective 2                                      |    |
| 1.6.3 Objective 3                                      |    |
|  |    |
| <u>CHAPTER 2</u> : MATERIALS AND METHODS               | 57 |
| 2.1 Materials and methods                              |    |
| 2.1.1 Tissue culture, plasmids and transfection        |    |
| 2.1.2 Growth factors                                   |    |
| 2 1 3 Bacteria   | 59 |

| 2.1 Materials and methods   | 58 |
|---|----|
| 2.1.1 Tissue culture, plasmids and transfection                     | 58 |
| 2.1.2 Growth factors  | 58 |
| 2.1.3 Bacteria  |    |
| 2.1.4 Antibodies and antisera                                       | 59 |
| 2.2 Animals   | 62 |
| 2.3 Patients and ethical approval                                   | 63 |
| 2.4 Tissue culture  | 64 |
| 2.4.1 Culture of AGS2 cells   | 64 |
| 2.4.2 Human primary gastric gland isolation and culture             |    |
| 2.4.3 Murine gastric gland isolation and culture                    |    |
| 2.5 Transient transfection and luciferase assays                    | 66 |
| 2.5.1 AGS2 cell transfection  | 66 |
| 2.5.2 Human primary gastric gland transfection and luciferase assay | 67 |
| 2.6 BrdU incorporation and antisense oligonucleotide treatment      | 67 |
| 2.7 Immunocytochemistry of adherent gastric glands                  | 69 |

| 2.8 Immunohistochemistry of paraffin embedded tissue | 70 |
|--|----|
| 2.9 Western blot analysis                            | 71 |
| 2.10 Bacterial re-transformation with plasmid DNA    | 72 |
| 2.11 Preparation of plasmid DNA                      | 72 |
| 2.12 Real-time PCR                                   | 73 |
| 2.12.1 RNA extraction and cDNA preparation           | 73 |
| 2.12.2 Quantitative real-time PCR                    | 74 |
| 2.13 Haemorrhagic lesion model76                     |    |
| 2.14 Statistical analysis76                          |    |

#### <u>CHAPTER 3</u>: STIMULATION OF PAI-1 EXPRESSION IN THE GASTRIC EPITHELIUM AND ITS SIGNIFICANCE FOR EPITHELIAL CELL EUNCTION

| EP  | ITHELIAL CELL FUNCTION  | 77  |
|-----|---|-----|
| 3.1 | Introduction  | 78  |
|     | 3.1.1 Aims  | 80  |
| 3.2 | Materials and methods   | 81  |
|     | 3.2.1 Patients  | 81  |
|     | 3.2.2 Animals   | 81  |
|     | 3.2.3 PAI-1 protein expression in <i>H.pylori</i> infection and gastric cancer. | 81  |
|     | 3.2.4 Real-time PCR   | .82 |
|     | 3.2.5 Cellular localisation of endogenous PAI-1, uPA and uPAR                   | .82 |
|     | 3.2.6 Analysis of proliferation   | 83  |
| 3.3 | Results   | 84  |
|     | 3.3.1 PAI-1 mRNA abundance is increased in <i>H. pylori</i> infection and in    | l   |
|     | gastric cancer  | 84  |
|     | 3.3.2 <i>H. pylori</i> infection increases PAI-1 protein expression             | 84  |
|     | 3.3.3 PAI-1 protein is expressed in epithelial cells in gastric cancer          | 87  |
|     | 3.3.4 PAI-1 mRNA abundance is increased in a mouse model of                     |     |
|     | epithelial remodelling  | 88  |
|     | 3.3.5 Cultured gastric glands contain the same epithelial cell types as in      | 1   |
|     | vivo glands   | 90  |
|     | 3.3.6 Endogenous PAI-1 protein expression in primary gastric glands             | 92  |
|     | 3.3.7 Endogenous uPAR protein expression in primary gastric glands              | 92  |
|     | 3.3.8 Both PAI-1 and uPAR are expressed in gastric myofibroblasts               | 96  |

| 3.5 Conclusions   | 112  |
|---|------|
| 3.4 Discussion  | 106  |
| 3.3.12 PAI-1 ASO reduce the percentage of PAI-1 expressing cells          | 104  |
| 3.3.11 Exogenous uPA stimulates proliferation of epithelial cells         | 102  |
| increased by PAI-1 knock-down   | 100  |
| 3.3.10 <i>H. pylori</i> stimulates epithelial cell proliferation which is |      |
| 3.3.9 Endogenous uPA protein is expressed in gastric myofibroblast        | s_96 |

## <u>CHAPTER 4</u>: TRANSCRIPTIONAL REGULATION OF uPA, uPAR AND PAI-1 IN HUMAN GASTRIC EPITHELIAL CELLS

| uPAR AND PAI-1 IN HUMAN GASTRIC EPITHELIAL CELLS                          | 113   |
|---|-------|
| 4.1 Introduction  | 114   |
| 4.1.1 Aims  | 117   |
| 4.2 Materials and methods   | 118   |
| 4.2.1 Transient transfection and luciferase assays                        | _118  |
| 4.2.2 Cellular targeting of uPAR-, uPA- and PAI-1-luc in human            |       |
| primary adherent glands   | 118   |
| 4.3 Results   | 119   |
| 4.3.1 Gastrin stimulates PAI-1-, uPA- and uPAR-luc expression in          |       |
| AGS2 cells  | 119   |
| 4.3.2 Transient transfection of human gastric gland cells with PAI-1-,    |       |
| uPA- or uPAR-luc  | 119   |
| 4.3.3 PAI-1-luc is targeted to cell types which express endogenous        |       |
| PAI-1   | 123   |
| 4.3.4 uPA-luc is targeted to cell types which express endogenous uPA      | 123   |
| 4.3.5 uPAR-luc is targeted to cell types which express endogenous         |       |
| uPAR  | 124   |
| 4.3.6 PAI-2-luc is expressed by a variety of gastric epithelial cells     | 124   |
| 4.3.7 Induction of uPAR-, uPA- and PAI-1-luc by H. pylori in gastric      |       |
| epithelial cells  | 131   |
| 4.3.8 Mechanism of <i>H.pylori</i> induction of uPAR-, uPA- and PAI-1-luc | c_133 |
| 4.3.9 Regulation of PAI-1-luc by growth factors                           | 133   |
| 4.4 Discussion  | 137   |
| 4.5 Conclusions   | 141   |

| <u>CHAPTER 5</u> : CONTRIBUTION OF PAI-1 TO THE RESPONSE OF          |    |
|--|----|
| EPITHELIAL CELL DAMAGE USING uPA SYSTEM TRANSGENIC                   | t  |
| MICE14   | 12 |
| 5.1 Introduction14   | 13 |
| 5.1.1 Aims14   | 15 |
| 5.2 Materials and methods14  | 16 |
| 5.2.1 Induction of haemorrhagic lesions within the gastric mucosa14  | 16 |
| 5.2.2 Development and production of PAI-1 over-expressing mice14     | 16 |
| 5.2.2.1 Expression of PAI-114  | 16 |
| 5.2.2.2 Transfection of murine gastric gland cells14                 | 17 |
| 5.3 Results14  | 18 |
| 5.3.1 PAI-1 KO mice develop more severe haemorrhagic lesions than    |    |
| wild-type or uPAR KO mice14  | 18 |
| 5.3.2 PAI-1 KO mice display a more severe phenotype when challenged  |    |
| with ethanol than uPAR KO or wild-type mice15                        | 50 |
| 5.3.3 PAI-1 KO mice develop more severe haemorrhagic lesions with an |    |
| overall greater lesion length than wild- type or uPAR KO mice15      | 52 |
| 5.3.4 Generation of PAI-1 over-expressing transgenic mice15          | 54 |
| 5.3.5 PAI-1 over-expressing mice develop less severe haemorrhagic    |    |
| lesions after ethanol challenge than PAI-1 KO mice16                 | 51 |
| 5.3.6 uPA over-expressing mice develop a more severe phenotype when  |    |
| challenged with ethanol compared to wild-type mice16                 | 53 |
| 5.4 Discussion16   | 55 |
| 5.5 Conclusions16  | 58 |
|  |    |
| CHAPTER 6: CONCLUSIONS AND FUTURE WORK17                             | 70 |
| 6.1 Major findings17   | 71 |
| 6.2 Conceptual advances and future work17                            | 72 |
|  |    |
| CHAPTER 7: REFERENCES18  | 31 |

•

## List of Figures

- Figure 1.1 Schematic representation of the stomach, showing the major anatomical divisions.
- Figure 1.2 Histological steps and factors involved in progression to gastric adenocarcinoma.
- Figure 1.3 Schematic outline of the urokinase plasminogen activator (uPA) system.

Figure 1.4 Structure of uPA.

- Figure 1.5 Model depicting proposed events in PAI-1-induced cell detachment.
- Figure 2.1 Method outline for BrdU incorporation into human gastric glands.
- Figure 3.1 PAI-1 mRNA abundance is increased in *H. pylori* infection and gastric cancer.
- Figure 3.2 PAI-1 protein expression is increased in *H. pylori* infection.
- Figure 3.3 Increased PAI-1 staining in gastric cancer is localised to epithelial cells.
- Figure 3.4 PAI-1 mRNA abundance is increased in *H. felis* infected mice 6 months post infection.
- Figure 3.5 Schematic of the gastric corpus epithelium and the corresponding adherent cultured gland preparation.
- Figure 3.6 PAI-1 endogenous protein expression in human primary cultured gastric glands.
- Figure 3.7 uPAR endogenous expression in human primary cultured gastric glands.
- Figure 3.8 PAI-1 and uPAR endogenous protein expression in human primary cultured gastric myofibroblasts.
- Figure 3.9 uPA endogenous protein expression in human primary cultured gastric glands.
- **Figure 3.10** *H. pylori* stimulates proliferation of epithelial cells which is further increased by PAI-1 anti-sense oligonucleotides treatment.
- Figure 3.11 uPA stimulates proliferation of epithelial cells which is further increased by PAI-1 anti-sense oligonucleotides treatment.
- **Figure 3.12** PAI-1 anti-sense oligonucleotides reduce the number of PAI-1 expressing cells in human gastric glands.

IX

- **Figure 4.1** Simplified model of several epithelial and mesenchymal factors (and their respective receptors) which are known to play a role in epithelial-mesenchymal interactions within the gastric mucosa.
- Figure 4.2 Gastrin stimulates expression of uPAR-, uPA- and PAI-1-luc.
- Figure 4.3 Schematic protocol for transfection of human gastric glands with uPAR-, uPA- or PAI-1-luc.
- **Figure 4.4** 4.5kb of the human PAI-1 promoter targets expression of PAI-1-luc to specific cell types in human primary gastric glands.
- Figure 4.5 2.9kb of the human uPA promoter targets expression of uPA-luc to specific cell types in human primary gastric glands.
- **Figure 4.6** 2kb of the human uPAR promoter targets expression of uPAR-luc to specific cell types in human primary gastric glands.
- **Figure 4.7** 2.3kb of the human PAI-2 promoter targets expression of PAI-2-luc to specific cell types in human primary gastric glands.
- Figure 4.8 *H. pylori* stimulates uPAR-, uPA- and PAI-1-luc in human gastric gland cells.
- **Figure 4.9** uPA, but not uPAR or PAI-1 luciferase expression is abolished by a CagE null *H. pylori* mutant in human primary gastric epithelial cells.
- Figure 4.10 IGF-II, HGF and gastrin increase PAI-1-luc expression.
- **Figure 5.1** PAI-1 KO mice develop more severe haemorrhagic lesions than wildtype or uPAR KO mice after ethanol administration.
- **Figure 5.2** PAI-1 KO mice have a more severe phenotype when challenged with ethanol than uPAR KO or wild-type mice.
- Figure 5.3 PAI-1 KO mice have a greater average lesion length after challenge with 50% or 99.8% ethanol compared to uPAR KO or wild-type, C57BL/6 mice.
- **Figure 5.4** Schematic diagram of the stages and various expression plasmids involved in the generation of PAI-1 over-expressing mice.
- **Figure 5.5** Increased PAI-1 expression in primary epithelial cells transfected with pcDNA3-PAI-1 expression plasmid.
- Figure 5.6 Map of expression plasmids used to over-express PAI-1 or uPA in murine gastric parietal cells.
- **Figure 5.7** A higher percentage of parietal cells express PAI-1 in murine gastric glands transfected with HKB-PAI-1.

- **Figure 5.8** PAI-1 over-expressing mice show a less severe phenotype when challenged with ethanol compared to wild-type mice.
- **Figure 5.9** uPA over-expressing mice have a greater average lesion length after challenge with ethanol compared to wild-type mice.
- Figure 6.1 Schematic overview of major findings.

# **List of Tables**

- **Table 2.1** Antibodies used to identify main cell types within primary gastric glands.
- **Table 2.2** Primary uPA system antibodies and dilutions used inimmunocytochemistry to determine endogenous protein expression.
- **Table 2.3** Primer and probe volumes for real-time PCR of human and mouse cDNA samples.
- **Table 2.4** Human and mouse PAI-1 specific forward and reverse primer and probe sequences for real-time PCR.
- **Table 3.1** Proportion of PAI-1, uPAR and uPA expressing cells in human

   cultured gastric gland cells after stimulation with wild-type *H. pylori*.
- **Table 3.2** There is a decrease in PAI-1 expressing cells in human gastric glands treated with PAI-1 anti-sense oligonucleotides before and after the addition of *H. pylori*.
- **Table 4.1** Proportion of PAI-1-, uPA- and uPAR-luc expressing cells in humanprimary cultured gastric glands.

# **Abbreviations**

The abbreviations used in this thesis are listed below.

| α-SMA       | $\alpha$ -smooth muscle actin                |
|-------------|--|
| APC         | Activated protein C                          |
| ASO         | Anti-sense oligonucleotide                   |
| ATF         | Amino terminal fragment                      |
| BrdU        | 5'-bromo-2'-deoxyuridine                     |
| BSA         | Bovine serum albumin                         |
| C57BL/6     | C57Black/6 wild-type mouse strain            |
| CagA        | Cytotoxin A                                  |
| cag-PAI     | Cag Pathogenicity Island                     |
| CCK2R       | Cholecystokinin receptor 2                   |
| CGRP        | Calcitonin gene-related peptide              |
| COX (-1,-2) | Cyclooxygenase (-1, -2)                      |
| CSO         | Control scrambled antisense oligonucleotides |
| DMEM        | Dulbecco's Modified Eagles Medium            |
| DTT         | Dithiothreitol                               |
| EC          | Enterochromaffin cell                        |
| ECL         | Enterochromaffin-like-cell                   |
| ECM         | Extracellular matrix                         |
| EGF         | Epidermal growth factor                      |
| EGF-R       | Epidermal growth factor receptor             |
| ELISA       | Enzyme-linked immunosorbent assay            |
| ЕМТ         | Epithelial-mesenchymal transition            |
| ERK         | Extracellular signal-related kinase          |
| FBS         | Fetal bovine serum                           |
| FGF         | Fibroblast growth factor                     |
| FITC        | Fluorescein isothiocyanate                   |
| FM          | Full media                                   |
| FPRL-1      | FPR-like receptor/lipoxin A4 receptor        |
| FVB/N       | Background mouse strain for INS-GAS mice     |
| G-17        | Amidated heptadecapeptide gastrin            |
| GAS-KO      | Gastrin knock-out                            |

XII

| GFP                                     | Green fluorescent protein                                   |
|---|---|
| GI                                      | Gastrointestinal  |
| GMBF                                    | Gastric mucosal blood flow                                  |
| GPI                                     | Glycosyl-phosphatidyl-inositol moeity                       |
| $H_2$                                   | Histamine   |
| HB-EGF                                  | Heparin-binding epidermal growth factor                     |
| HBSS                                    | Hank's balanced salt solution                               |
| HCL                                     | Hydrochloric acid   |
| HCO <sub>3</sub> -                      | Bicarbonate   |
| HDC                                     | Histamine decarboxylase                                     |
| H. felis                                | Helicobacter felis  |
| H. pylori                               | Helicobacter pylori   |
| HGF                                     | Hepatocyte growth factor                                    |
| H <sup>+</sup> , K <sup>+</sup> -ATPase | Hydrogen-Potassium-ATPase                                   |
| HMW                                     | High molecular weight                                       |
| HRP                                     | Horseradish peroxidase                                      |
| IFN-a, -γ                               | Interferon -alpha, -gamma                                   |
| IGF-II                                  | Insulin-like growth factor II                               |
| IGF-BP5                                 | Insulin-like growth factor-binding protein 5                |
| IHC                                     | Immunohistochemistry  |
| IL                                      | Interleukin   |
| INS-GAS                                 | Transgenic for gastrin driven by insulin promoter           |
| JAK-STAT                                | Janus kinase/signal transducers activators of transcription |
| JNK                                     | Jun N-terminal kinase                                       |
| KGF                                     | Keratinocyte growth factor                                  |
| LDLR                                    | Low density lipoprotein receptor                            |
| LMW                                     | Low molecular weight  |
| LPS                                     | Lipopolysaccharide  |
| LRP                                     | Lipoprotein receptor-related protein                        |
| МАРК                                    | Mitogen-activated protein kinase                            |
| MI                                      | Myocardial infraction                                       |
| MMP                                     | Matrix metalloproteinase                                    |
| ΜΟΙ                                     | Multiplicity of infection                                   |

XIII

And Street of

| NaCl      | Sodium chloride                                       |
|-----------|---|
| ΝϜκΒ      | Nuclear factor kappa-B                                |
| NKA       | Neurokinin A  |
| NO        | Nitric-oxide  |
| NOCs      | N-nitroso compounds                                   |
| NOS       | Nitric-oxide synthase                                 |
| NSAIDs    | Non-steroid-anti-inflammatory drugs                   |
| oipA      | Outer inflammatory protein A                          |
| PAI-1     | Plasminogen activator inhibitor-1                     |
| PAI-2     | Plasminogen activator inhibitor-2                     |
| PAI-3     | Plasminogen activator inhibitor-3                     |
| PAI-1 KO  | Plasminogen activator inhibitor-1 gene knock-out mice |
| PBS       | Phosphate buffered saline solution                    |
| PCI       | Protein C inhibitor                                   |
| PCR       | Polymerase chain reaction                             |
| PDGF      | Platlet derived growth factor                         |
| PFA       | Paraformaldehyde                                      |
| PGs       | Prostaglandins  |
| PI3K      | Phosphoinositol 3-kinase                              |
| PMA       | Phorbol 12-myristate 13-acetate                       |
| PPI       | Proton pump inhibitor                                 |
| RCL       | Reactive centre loop                                  |
| RhoA      | Ras homologue gene family member A                    |
| ROS       | Reactive oxygen species                               |
| RT-PCR    | Reverse transcription polymerase chain reaction       |
| SCF       | Stem cell factor                                      |
| SFM       | Serum free media                                      |
| SMB       | Somatomedin B domain                                  |
| SP        | Substance P   |
| TBST      | Tween-Tris-Buffered saline                            |
| TFF-1/-2  | Trefoil factor family -1/-2                           |
| TFFS      | Type IV secretory system                              |
| TGF-α,- β | Transforming growth factor -alpha, -beta              |

| TIMP    | Tissue inhibitor of metalloproteinase                   |
|---------|---|
| TLR     | Toll-like receptor                                      |
| ΤΝΓ-α   | Tumour necrosis factor -alpha                           |
| TRed    | Texas red   |
| uPA     | Urokinase plasminogen activator                         |
| uPAR    | Urokinase plasminogen activator receptor                |
| uPAR KO | Urokinase plasminogen activator receptor gene knock-out |
| VacA    | Vacuolating cytotoxin A                                 |
| VEGF    | Vascular endothelial growth factor                      |
| VLDL    | Very low density lipoprotein                            |
| VMAT-2  | Vesicular monoamine transporter type-2                  |
| VN      | Vitronectin   |
| VR1     | Vanilliod receptors of type 1                           |
| WHO     | World Health Organisation                               |
| wt      | Wild-type   |

# **Publications and Presentations**

#### **<u>PUBLICATIONS</u>**:

Susan Kenny, Cedric Duval, Stephen J. Sammut, Islay Steele, Mark D. Pritchard, John Atherton, Richard H. Argent, Rod Dimaline, Graham J Dockray, and Andrea Varro. Increased expression of the urokinase activator system by *Helicobacter pylori* in gastric epithelial cells. (Submitted to Am J Physiol, March 2008)

Andrea Varro, <u>Susan Kenny</u>, Elaine Hemers, Catherine McCaig, Sabine Przemeck, Timothy. C. Wang, Keith Bodger, and D. Mark Pritchard. Increased gastric expression of MMP-7 in hypergastrinemia and significance for epithelial-mesenchymal signalling. *Am J Physiol Gastrointest Liver Physiol* 292: G1133-G1140, 2007. (Copy included at end of thesis).

#### **PRESENTATIONS**:

International Postgraduate Medical Conference, Czech Rebulic (30<sup>th</sup> November - 1<sup>st</sup> December, 2006) - The regulation of plasminogen activator inhibitor-1 (PAI-1) expression in the gastric epithelium by *Helicobacter pylori* (Oral presentation).

**Digestive Disease Week, Washington DC, USA (19<sup>th</sup> - 23<sup>rd</sup> May 2007)** - The regulation of the urokinase plasminogen activator system in the gastric epithelium by *Helicobacter pylori* (Poster Presentation).

National Cancer Research Institute, Birmingham, UK (8<sup>th</sup> - 11<sup>th</sup> October 2006) - The regulation of the urokinase plasminogen activator (uPA) system in the gastric epithelium by *Helicobacter pylori* (Poster Presentation).

Liverpool and Bristol Joint PhD Symposium, University of Liverpool (25<sup>th</sup> September 2006) – The regulation of plasminogen activator inhibitor-1 (PAI-1) expression in the gastric epithelium by *Helicobacter pylori* (Poster presentation).

#### <u>Abstract</u>

The gastric pathogen *Helicobacter pylori* is linked to gastritis, leading to gastric cancer but the mechanisms are unclear. Gastritis is characterised by remodelling of the gastric mucosa, which requires degradation of the extracellular matrix (ECM). Proteases such as matrix metalloproteinases (MMPs) and urokinase plasminogen activator (uPA), as well as proteases inhibitors such as plasminogen activator (nPAI)-1 play an important role in this process. Since members of the uPA system (namely PAI-2, uPA and its receptor, uPAR) have been shown to be increased by *H. pylori*, it was hypothesised that PAI-1 expression is also increased in the gastric mucosa during infection, which could play a role in the tissue remodelling observed during the development of gastritis. The aim of this study was to investigate the effect of *H. pylori* on PAI-1 expression in the gastric mucosa, to examine the consequences of this protease inhibitor on human primary gastric epithelial cell proliferation and to determine its initial contribution to gastric mucosal protection during epithelial injury.

Real-time quantitative polymerase chain reaction (Q-PCR) of biopsies from gastric corpus, but not antrum, showed significantly increased PAI-1 expression in *H. pylori* positive patients which was further increased in gastric cancer patients compared to controls. Using immunohistochemistry, it was found that *H. pylori* significantly increased the expression of PAI-1 in primary gastric epithelial cells, particularly in the acid-secreting parietal cells. Expression of PAI-1 in gastric cancer patients was also found in epithelial cells. A transgenic mouse model of gastric remodelling (INS-GAS) infected with *H. felis* showed an increase in PAI-1 expression six months post-infection.

Proliferation assays showed that both *H. pylori* and exogenous uPA stimulated human primary gastric epithelial cell proliferation, which was further increased after PAI-1 knockdown using anti-sense oligonucleotides treatment, consistent with PAI-1 inhibition of endogenous uPA.

Transfection of human primary gastric epithelial cells with uPA, PAI-1 or uPAR promoters, in luciferase reporter constructs, revealed expression of all three in  $H^+$ ,  $K^+$ -ATPase- and VMAT-2-expressing cells; uPA-luciferase was also expressed in pepsinogen-containing cells and uPAR in TFF-1-expressing cells. In each case, expression patterns of the luciferase reporter constructs were similar to that of the endogenously expressed proteins, and all were increased in response to *H. pylori*. However, the virulence factor CagE was required for the stimulation of uPA, but not PAI-1 or uPAR.

The production and use of uPA system transgenic mice in an acute model of gastric mucosal injury revealed that PAI-1 plays a protective role during gastric epithelial damage. Mice in which this gene is deleted displayed a more severe phenotype after challenge compared to wild-type mice, as did uPA over-expressing mice. However, this phenotype was reversed when PAI-1 was over-expressed, consistent with the idea that expression of PAI-1 is vital for the restraint of uPA in order for the initial formation and stabilisation of a haemostatic fibrin plug at the site of epithelial injury to occur.

Together these data indicate that the initial induction of PAI-1 expression in acute *H. pylori* infection is part of a defensive mechanism as it has been demonstrated that PAI-1 plays a pivotal role in the protection of the gastric epithelium after injury. However, PAI-1 also acts to restrain uPA, therefore

controlling epithelial cell proliferation which is a key characteristic of a preneoplastic, *H. pylori* infected epithelium. During prolonged *H. pylori* infection it can be suggested that sustained increased levels of anti-fibrinolytic PAI-1 contribute to the process of fibrosis, increased proliferation favours the acquisition of mutations which influences tissue remodelling leading to gastric cancer.

# Chapter 1

# Introduction

# 1.1 Overview

The Nobel Prize for Physiology or Medicine in 2005 was awarded to Marshall and Warren for their discovery in 1982 of the bacterium *Helicobacter pylori* (*H. pylori*) in patients suffering from inflammation of the gastric mucosa (gastritis) and duodenal ulcer (Marshall & Warren, 1984). It had previously been known that patients suffering from duodenal ulcer displayed an increase in gastric acid secretion (Henning, 1966). Subsequently it became clear that there was relationship between *H. pylori* infection, gastritis and gastric cancer. Gastric cancer is the second most common cause of cancer related death worldwide (Pisani *et al.*, 1999) and the estimated risk of dying from *H. pylori* related disease is 1:30 for men and 1:60 for woman (Axon & Forman, 1997). The mechanisms by which *H. pylori* causes gastritis and gastric cancer remain poorly understood.

This thesis examines the effects of *H. pylori* on several members of the urokinase plasminogen activator (uPA) system within the gastric epithelium. The uPA system consists of uPA itself, its cell surface receptor uPAR and the plasminogen activator inhibitors (PAI)-1, -2. Collectively these proteins play a part in a diverse range of processes including control of fibrinolysis and haemostasis, ovulation, wound healing, fibrosis, cancer cell invasion and metastasis (Andreasen *et al.*, 2000). The uPA system members are expressed in the stomach and have been shown to correlate with a worse prognosis in human gastric cancer (Kaneko *et al.*, 2003). Expression of PAI-2 in the stomach is increased by *H. pylori* (Varro *et al.*, 2004) therefore it is an interesting question to ask if expression of other members of the uPA system are also influenced by *H. pylori*, and if so what are the implications for the gastric mucosa?

# **1.2 Gastric architecture**

The primary function of the stomach is to temporarily store food and release it slowly into the duodenum. Through the release of acid and pepsin to initiate digestion, the stomach processes the food to a semi-solid chyme which enables better contact with the mucous membrane of the intestine, thereby facilitating absorption of nutrients. The acid environment of the stomach lumen inactivates most micro-organisms although *H. pylori* is a special case since this bacterium is adapted for survival in acid conditions. Several mucosal protective mechanisms exist within the stomach which limit tissue injury, contribute to gastric mucosal protection and inhibit auto-digestion of the stomach (discussed in section 1.2.6).

# **1.2.1** Organisation of the gastric mucosa

The wall of the stomach is subdivided into two functionally distinct regions. The body of the stomach, also known as the corpus is responsible for acid and pepsinogen secretion and through relaxation of its muscle layer it accommodates most ingested food. The lowermost region is the antrum which produces the hormone gastrin (this stimulates acid production) and grinds the food. In both cases an epithelium is the first/inner layer containing gastric glands; the submucosa is followed by a muscle layer, the serosa and the subserosa (Caletti *et al.*, 1984).

# 1.2.2 Gastric epithelium

The gastric epithelium is a single layer of cells that lines the interior face of the stomach, providing physical protection from the lumen to the inner layers of the

mucosa. The epithelium consists of a variety of specialised secretory epithelial cells that invaginate to form gastric glands. These are highly organised branching tubular structures that reach deep into the muscularis mucosa. Various differentiated epithelial cell types are found in characteristic locations in the gastric gland and arise from the proliferation and renewal of an undifferentiated, granule-free, multipotent stem cell. This is located in the isthmus or neck region of the gland and forms the precursor cells of the differentiated cell types (Hall, 1989).

The single multipotent stem cell is located in a physiological environment known as a "niche" (Karam & Leblond, 1993a; Dockray, 1999; Brittan & Wright, 2002) and from here cells migrate up or down the gland. Over a period of approximately 3 days, cells that migrate upwards of the neck region differentiate into surface mucous cells on reaching the pit region and secrete mucus to protect the epithelium from the mechanical strain of digestion and the low pH of lumen. At the surface of the gland, pit cells undergo necrosis/apoptosis, are phagocytosed by neighboring cells or are shed into the lumen (Karam & Leblond, 1993b). Cells migrating down the gland give rise to mucous neck cells and acid (HCL) secreting parietal cells forming the neck region of the gland. Histamine producing enterochromaffin-like (ECL) cells and pepsinogen secreting chief cells form the base region of the gland (Karam & Leblond, 1993d; Karam *et al.*, 2003). Chief cells are derived entirely from mucous neck cells as they migrate towards the base of the gland.

Gastric glands of the corpus are called oxyntic (from the Greek for acid, *oxys*) glands as they contain the acid secreting parietal cells (Ito, 1987). Figure 1.1

shows a schematic of the main regions of the stomach and a representation of an oxyntic gland which contains parietal, ECL, chief and somatostatin producing Dcells. Other endocrine cells present in the oxyntic glands other than ECL and Dcells are serotonin-releasing enterochromaffin cells (EC) and recently identified ghrelin (X) cells (Bordi *et al.*, 2000a; Rindi *et al.*, 2002). The pyloric glands of the antrum contain gastrin secreting G-cells, D and EC cells (Walsh, 1994; Bordi *et al.*, 2000b). The life span and function of gastric epithelial cells varies from cell to cell with cell types such as surface mucous cells surviving for 2-3 days. Parietal, ECL, G and D-cells are the four cell types crucial for acid secretion (Hou & Schubert, 2006).



## Figure 1.1

Schematic representation of the stomach showing the major anatomical divisions. An enlarged area of the oxyntic mucosa shows a representation of a corpus gastric gland (Adapted from Dimaline & Varro, 2007).

# 1.2.3 Role of gastric epithelial cells in acid secretion

Formed primarily from the differentiation of pre-parietal cells but also from preneck and pre-pit cell precursors (Karam, 1993), the parietal cell was first described by Heidenhain in 1870. It has an average life span in mouse of approximately 20 days (Kirton *et al.*, 2002), is solely found in the corpus mucosa and is responsible for the secretion of 100mM HCL. In addition to this, parietal cells also secrete intrinsic factor required for absorption of vitamin  $B_{12}$  (Okuda, 1999).

ECL cells are one of the main endocrine cell types within the stomach and are also confined to the corpus mucosa. They are responsible for the synthesis and secretion of the acid secretagogue histamine and they express the gastrin or cholecystokinin receptor (CCK2R; Schmitz *et al.*, 2001). Since its discovery in 1905 by Edkins, and isolation and purification in 1964, the hormone gastrin which is produced by and released from antral G-cells (Gregory & Tracy, 1964; McGuigan, 1968; Modlin *et al.*, 1997) is now recognised as a major stimulant of acid secretion in the gastric corpus.

Release of gastrin is primarily stimulated by amino acids and peptides in the stomach and acts through the CCK2R. However, although parietal cells express functional CCK2R (Reubi *et al.*, 1997) they play an indirect role in acid secretion (Sachs *et al.*, 1997). The effects of gastrin on parietal cell acid secretion are paracrine as ECL cells are the primary gastrin target which causes the release of histamine (Sandvik *et al.*, 1987).

Histamine released from ECL cells in response to gastrin acts on histamine-H<sub>2</sub> receptors expressed on parietal cells triggering a cascade that initiates the translocation of the proton pump,  $H^+$ ,  $K^+$ -ATPase, from cytoplasmic tubulovesicles to deep invaginations of the apical plasma membrane termed secretory canaliculi. This causes morphological transformation of these cells (Forte & Yao, 1996; Yao & Forte, 2003) and ultimately stimulation of acid secretion (Kitano *et al.*, 2000; Dockray *et al.*, 2001). G-cell release of gastrin is inhibited in a negative feedback mechanism by a decrease in luminal pH caused by acid secretion. This is sensed by antral D-cells which act as chemoreceptors and release somatostatin in response to a pH lower than 3.5 (Brand & Stone, 1988; Dimaline *et al.*, 1991).

## 1.2.4 Gastric submucosa

The submucosal layers located underneath the epithelium consist of mesenchymal cells (the main type being myofibroblasts), blood vessels, connective tissue, inflammatory cells and nerve fibers. Myofibroblasts were originally described as muscle-like fibroblasts (Gabbiani *et al.*, 1971) and are characterised by the expression of  $\alpha$ -smooth muscle actin ( $\alpha$ -SMA) and vimentin. Myofibroblasts derive from mesenchymal stem cells (Luria *et al.*, 1971; Friedenstein *et al.*, 1974; Ogawa *et al.*, 2006) through the actions of platelet-derived growth factor (PDGF) and stem cell factor (SCF) and also through transdifferentiation of fibroblasts and smooth muscle cells via PDGF and transforming growth factor (TGF)- $\beta$  (Powell *et al.*, 1999a; Powell *et al.*, 1999b). Myofibroblasts play an important role in epithelial maintenance including

repairing the epithelium in response to physical damage and bacterial infection through epithelial-mesenchymal interactions (discussed in section 1.4).

# 1.2.5 Role of gastrin in gastric epithelium function

In addition to the regulation of acid secretion (discussed in section 1.2.3) the physiological functions of gastrin include roles in the organisation and maintenance of the gastric epithelium (Dockray, 1999; Dockray *et al.*, 2001; Dimaline & Varro, 2007). In the last decade new roles for gastrin have emerged from studies in genetically modified mice in which the gastrin gene is deleted (GAS-KO; Koh *et al.*, 1997; Friis-Hansen *et al.*, 1998), over-expressed (INS-GAS; Wang *et al.*, 1996), or in mice in which the CCK2R gene through which gastrin acts is knocked out (CCK2R -/-; Nagata *et al.*, 1996).

INS-GAS mice develop gastric carcinoma or adenocarcinoma, particularly after *H. pylori* infection. This was demonstrated by *Wang et al* (2000) who showed that chronic hypergastrinaemia in these mice acts in synergy with *Helicobacter* infection, contributing to the rapid onset (approximately 6 months post-infection) of atrophy. These animals show marked hyperplasia, inflammation and had extensive parietal cell loss (67%) with an accelerated progression towards gastric cancer compared to uninfected INS-GAS mice that only showed 40% parietal cell loss with the onset of gastric cancer being much slower (requiring nearly the entire lifetime of the animal,18-20 months). Infection of wild-type animals in the same genetic background not over-expressing gastrin (FVB/N) resulted in moderate foveolar hyperplasia and inflammation, showing only 27% parietal cell loss (Wang *et al.*, 2000).

A marked increase in basal gastric pH and an approximate 10-fold elevation of plasma gastrin concentration is seen in CCK2R -/- mice that also show atrophy of the corpus mucosa associated with a 70% decrease in active parietal cells (Langhans *et al.*, 1997) and a reduced number of D-cells (Nagata et al., 1996). The ECL cells in these animals do not secrete histamine (Chen *et al.*, 2002) and they also have an increased number of gastrin immunoreactive G-cells (Nagata *et al.*, 1996). Hypochlorhydria is seen in GAS-KO mice (Koh *et al.*, 1997) which are a good model for studying tumorigenesis as their phenotype leads to bacterial overgrowth, gastric inflammation, metaplasia and tumour development (Zavros *et al.*, 2002; Friis-Hansen *et al.*, 2006a; Friis-Hansen *et al.*, 2006b; Tu *et al.*, 2007). Taken together, transgenic animal studies have shown that gastrin acting via CCK2R, is not essential for normal epithelial cell proliferation but is important for maintaining the normal cellular composition and function of the gastric mucosa.

Gastrin increases expression of a wide range of genes in gastric epithelial cells which are involved in proteolysis, proliferation, differentiation and migration (Varro *et al.*, 2002a). Among these are members of the uPA system such as PAI-2 and members of the matrix metalloproteinase (MMP) family such as MMP-7 and MMP-9, both of which are involved in tissue remodeling. Gastrin acting via MMP-9 stimulates invasiveness of gastric cancer cells (Wroblewski *et al.*, 2002) and MMP -7 plays an important role in migration of primary human gastric epithelial cells (Wroblewski *et al.*, 2003) and also in epithelial-mesenchymal interactions (discussed in section 1.4; McCaig *et al.*, 2006; Varro *et al.*, 2007). Other gastrin sensitive genes important in epithelial maintenance include trefoil family factor (TFF)- 1 which is required for restitution and repair of the

epithelium and is rapidly up-regulated in response to injury (Sands & Podolsky, 1996; Khan *et al.*, 2003) and the cytoskeleton-membrane linker protein, ezrin which is needed for parietal cell maturation (Pagliocca *et al.*, 2003).

Given its extensive range of effects there is little evidence suggesting that hypergastrinaemia alone predisposes humans to gastric atrophy (characterised by chronic inflammation of the gastric mucosa with decreased numbers of parietal cells and low levels of acid secretion), but it is thought to exert its actions by intensifying the effects of inflammatory conditions such as infection with *Helicobacter* which is also often associated with elevated plasma gastrin concentrations (Harris *et al.*, 2002; Calam *et al.*, 1997) and is discussed above in the context of INS-GAS mice and discussed further in section 1.3.7.1.

### **1.2.6 Gastric mucosal protection**

The stomach is under continuous exposure to potentially hazardous agents. Luminal acid and pepsin constitute a major threat to the gastric mucosa. Reflux of alkaline duodenal contents which contain bile and pancreatic enzymes are additional damaging factors. Infection of the stomach with *Helicobacter* (discussed in section 1.3) has been found to be the main cause of gastritis (inflammation of the gastric mucosa) and gastric ulcers, which is one of the common ailments affecting humans. However, alcohol, cigarette smoking, drugs, reduction in gastric mucosal blood flow (GMBF), stress and a regular intake of non-steroid-anti-inflammatory drugs (NSAIDS), particularly aspirin are examples of exogenous mucosal irritants that can also inflict mucosal injury (Konturek *et al.*, 1981a; Konturek *et al.*, 1981b). The stomach has the ability to defend itself against these noxious agents by having a multitude of structural and

physiological factors in place to reduce tissue damage. These include mucus and bicarbonate secreted by surface epithelial cells, prostaglandins (PGs), modulation of GMBF and a fast acting neural system releasing sensory neuropeptides.

An initial defense barrier is provided by mucus cells which constantly produce a viscid layer of secreted mucus and bicarbonate (Hollander, 1954; Flemstorm, 1977). Gastric mucus consists of a glycoprotein tetramer with overlapping molecules resulting in a gel formation (Allen, 1976; Allen, 1984). Bicarbonate (HCO<sub>3</sub><sup>-</sup>) is secreted at a basal rate amounting to 5-10 percent maximal acid output (Flemstorm, 1977) and together these protect the gastric epithelium from luminal acid and pepsin. An unstirred layer where mixing of H<sup>+</sup> and HCO<sub>3</sub><sup>-</sup> is limited, gastric diffusion through mucus gel is delayed, allowing HCO<sub>3</sub><sup>-</sup> to neutralise the pH at the epithelial surface (Crampton, 1988). Local paracrine agents, especially PGs are potent stimulators of bicarbonate secretion (Garner *et al.*, 1979).

Prostaglandins are the major product of arachidonate metabolism through cyclooxygenase (COX) activity and play a crucial role in protecting the gastric mucosa against injury (Robert *et al.*, 1979; Terano *et al.*, 1987). Two isoforms of COX have been identified (Kargman *et al.*, 1996); COX-1, a constitutive enzyme responsible for the production of PGs that are physiologically important for maintenance of mucosal integrity (Eberhart & Dubois, 1995), and COX-2 which is an inducible isoform (Wallace, 2006). Cyclooxygenase-2 is responsible for the rise in PG levels at sites of inflammation and disease and is induced in inflammatory cells such as fibroblasts and myofibroblasts in gastric ulcers (Takahashi *et al.*, 1998). The stomach can be damaged by NSAIDs principally as

a result of their inhibition of prostaglandin synthesis. This is mediated via abrogation of the secretion of mucus and bicarbonate and by reduction in GMBF (Brzozowski *et al.*, 2005).

If the "muco-bicarbonate layer" is disrupted then GMBF is one way in which the removal of acid is facilitated, preventing its accumulation in tissue. This is influenced by neuronal modulating processes stimulating the release of vasoactive mediators thus increasing GMBF and vascular permeability (Holzer & Sametz, 1986; Holzer *et al.*, 2001), processes crucial for the gastric mucosa to resist the continual onslaught of aggressive agents (Yonei *et al.*, 1990). A reduction in GMBF can cause degeneration of the epithelium by potentiating the effect of mucosal damaging agents, making it vulnerable to injury as it is unable to produce appropriate protective secretions.

The maintenance of gastric mucosal integrity after attack by noxious chemicals is highly dependent on a neural alarm system. Afferent sensory neurons of intrinsic and extrinsic origin (vagal and spinal afferents) are present within the stomach. Extrinsic afferents operate as the main emergency protective system (Wimalawansa *et al.*, 1990), with many of the homeostatic actions of spinal afferents being mediated by the release of neuropeptides from their peripherial nerve endings.

Many primary afferent neurons are sensitive to the excitotoxic action of capsaicin, the pungent ingredient of red pepper/ chillis (Holzer, 1991). This exerts gasoprotective effects and is a unique neuropharmacological tool (Kang *et al.*, 1995; Mozsik *et al.*, 2005). It acts specifically on vanilloid receptors of type 1 (VR1; Caterina *et al.*, 1997), which are expressed on vagal and spinal afferents

(Schicho *et al.*, 2004). Capsaicin is associated with increased GMBF in the gastric mucosa (Holzer, 1990; Holzer *et al.*, 1990) as it stimulates capsaicin–sensitive neurons which are present in the stomach and form a dense plexus around the gastric submucosal vessels (Green & Dockray, 1988).

The effects of capsaicin are mediated through the release of neuropeptide transmitters such as calcitonin gene-related peptide (CGRP) and tachykinins such as substance P (SP) and neurokinin A (NKA) from spinal afferent nerve endings in the gastric mucosa and submucosa (Holzer, 1988; Chen *et al.*, 1992; Yeoh *et al.*, 1995). Calcitonin gene-related peptide is a 37-amino acid, vasoactive neuropeptide which acts on CGRP receptors to enhance the resistance of the gastric mucosa to experimental injury (Holzer *et al.*, 1991). The expression of VR1 and CGRP has been shown to be significantly increased in the epithelial layer of the gastric mucosa in patients with *H. pylori* associated chronic gastritis (Domotor *et al.*, 2007).

The gastroprotective action of primary transmitters such as CGRP involves secondary messengers such as the powerful vasodilator nitric oxide (NO). Endothelial cells lining the gastric blood vessels release NO, which mediates vascular relaxation induced by vagal stimulation (Palmer *et al.*, 1987) and acts to drive and maintain the gastric microcirculation. The beneficial effects on the gastric mucosa of capsaicin, CGRP and NKA are suppressed by NO synthase inhibitors (Stroff *et al.*, 1996). These inhibitors have been shown to decrease GMBF (Pique *et al.*, 1989; Tepperman & Whittle, 1992) and intragastric administration of NO donors is known to prevent ethanol-induced gastric lesions

(MacNaughton *et al.*, 1989; Lopez-Belmonte *et al.*, 1993). Gastric mucus secretion can also be regulated by NO (Brown *et al.*, 1993).

Excessive ethanol ingestion can result in gastritis characterised by mucosal edema, subepithelial hemorrhages, cellular exfoliation and inflammatory cell infiltration (Guslandi, 1987). Alcohol-induced gastric superficial injury is limited to the epithelium and involves mostly the inter-foveolar epithelium and gastric pits which heals rapidly by epithelial cell migration through a process known as restitution (Lacy et al., 1993). Gastric myofibroblasts can also migrate through the basement membrane and help restitution (Wu et al., 1999). However, the deeper lesions involve intramucosal hemorrhage and vascular engorgement (Guth et al., 1984). Due to damage, microvessels leak inflammatory mediators and vasoconstriction of submucosal arteries results in ischemia which eventually enhances the formation of more severe necrotic mucosal injury. Gastric mucosal damage can be produced by direct application of ethanol and high concentrations elicit injury in parts of the stomach attributed to congestion of the mucosa caused by constrictions of collecting venules (Ohno et al., 1995; Ohno et al., 1999). Ethanol-induced injury due to constriction has been shown to be inhibited by CGRP (Ohno et al., 1995) and by prostaglandin E2 and prostaglandin E4 signaling (Hattori et al., 2008).

# 1.3 Helicobacter pylori infection of the gastric epithelium

After being identified as a separate genus in 1989, the originally named *Campylobacter pyloridis* was renamed *Helicobacter pylori* (*H. pylori*). It is a microaerophilic, S-shaped, gram-negative bacteria known to have colonised the

stomach since ancient times (Linz & Schuster, 2007). Although earlier work identified a spiral bacterium within the stomach, *H. pylori* was only first identified in culture in 1982, and it was shown to be associated with peptic ulcers (Marshall & Warren, 1984). It is now known that this bacterium is strongly associated with gastritis (inflammation of the gastric mucosa) and infected individuals have a susceptibility to either peptic ulcer or gastric cancer.

# 1.3.1 Epidemiology

Infection with *H. pylori* is usually acquired in early childhood and when left untreated generally persists for the host's entire lifetime (Everhart, 2000). This is in part due to its ability to evade the antimicrobial effects of the immune response (Gobert *et al.*, 2001). People in all geographical locations can carry *H. pylori* but the prevalence of infection rises from about 50% of the population in developed countries, to 90% of the third world (Correa, 1996; Pisani *et al.*, 1999; Ernst, 2000; Everhart, 2000). Over the past decade, progressively fewer children have been shown to carry *H. pylori*, with this decrease being accelerated by the wide-use of medications such as antibiotics and also improved hygiene. Risk factors for acquisition of the bacterium include low socioeconomic status, household crowding, country of origin and ethnicity (Everhart, 2000; Tindberg *et al.*, 2001). The principle reservoir of *H. pylori* is man and infection is probably passed from individual to individual via proposed routes of faecal-oral, oral-oral and gastric-oral. There have also been descriptions of infection spread by means of uncooked vegetables contaminated with sewage (Stone, 1999).

## **1.3.2** Gastric epithelium remodelling and gastric cancer

*H. pylori* is recognised as an established causative agent of chronic inflammation of the mucous membranes of the stomach and of peptic ulcer. Individuals colonised with *H. pylori* develop co-existing gastritis, however, most people are asymptomatic. A long-term result of infection leads to the development of chronic gastric inflammation, conferring a significantly increased risk of developing gastric cancer (Forman *et al.*, 1991; Karnes *et al.*, 1991; Parsonnet *et al.*, 1991; Sipponen *et al.*, 1992; Hansson *et al.*, 1993; Correa, 1996; Wu *et al.*, 2003).

Approximately 15% of infected subjects will develop gastric or duodenal ulcer disease and a small proportion (less than 1%) develop neoplasia (Kuipers *et al.*, 1995; Uemura *et al.*, 2001; Go, 2002) thus indicating the involvement of other factors in cancer progression. *H. pylori* associated duodenal ulcer or cancer development are mutually exclusive clinical outcomes and do not develop together. In patients with duodenal ulcer, *H. pylori* colonises the antrum but not the corpus and is associated with increased acid and gastrin secretion (Hansson *et al.*, 1996; McColl *et al.*, 2000a, McColl *et al.*, 2000b; Dockray *et al.*, 2001; Uemura *et al.*, 2001). However, in other patients *H. pylori* can also spread to the corpus of the stomach resulting in inhibition of acid secretion, destruction of parietal and chief cells and progression to gastric cancer (Hansson *et al.*, 1996; Dockray *et al.*, 2001; Uemura *et al.*, 2001).

People infected with *H. pylori* not only have an increased risk of developing gastric cancer but also MALToma and non-Hodgkin's lymphoma of the stomach (Ernst & Gold, 2000; Peek & Blaser, 2002). Although less information is known
regarding the specific factors involved in these two malignancies, the local inflammatory process initiated by *H. pylori* is thought to be of importance (Isaacson & Du, 2005).

Gastric adenocarcinoma is the second leading cause of cancer-related death worldwide (Peek & Blaser, 2002) and approximately 649 000 people die from this malignancy each year (Correa, 2003). Most gastric cancers are sporadic in nature and arise secondary to multiple interacting environmental factors. *H. pylori* is the single most common cause which led to it being classified as a class 1 carcinogen by the World Health Organisation (WHO) and the International Agency for Research on Cancer consensus group (IARC) in 1994 (Houghton *et al.*, 2002; WHO, 1994). These findings have been extended from human patients (Uemura *et al.*, 2001) and several animal models (discussed in section 1.3.7).

The muscularis propria of the stomach is typically invaded by gastric cancer before diagnosis and the 5-year survival rate in the United States is less than 15% (Correa, 2003). Two distinct histological variants of gastric adenocarcinoma have been described each having different epidemiological and pathophysiological features: diffuse-type and intestinal-type gastric cancer.

Intestinal-type gastric adenocarcinoma progresses through a series of distinct histological steps, usually occurs at a late age and predominates in men (Sipponen & Marshall, 2000). These steps are initiated by the transition from normal mucosa to chronic superficial gastritis, leading to atrophic gastritis, intestinal metaplasia and ultimately to dysplasia and adenocarcinoma (Correa, 1998; Peek & Blaser, 2002; Correa, 2004).

Gastritis is characterised by diffuse infiltration of the gastric mucosa by cells representing chronic inflammation such as lymphocytes and macrophages but does not show loss of glands (Dixon *et al.*, 1996), as in atrophic gastritis which is also associated with the introduction of fibrous stromal tissue. At the stage of intestinal metaplasia, original glands and foveolar epithelium are replaced by cells with an intestinal phenotype and mucin-filled goblet cells (Matsukura *et al.*, 1980). Dysplasia is characterized by atypical changes in nuclear morphology and formation of irregular tissue architecture, frequently forming closely packed tubular structures (adenomas) with irregular lumens. All of the atypical cells are confined within the tubular structure but if they go through the basal membrane, they become invasive carcinomas. There is general agreement that the dysplastic epithelium is neoplastic, therefore, dysplasia was also referred to in the past as intraepithelial neoplasia (Rugge *et al.*, 2000). Figure 1.2 depicts the histological stages involved in gastric remodelling (Correa & Houghton, 2007) and other factors thought to be involved in the progression to gastric cancer.

Histologically, intestinal-type gastric cancer consists of gland-like structures that mimic intestinal glands. Diffuse type gastric cancer more commonly affects younger people, affects men and woman equally and consists of individual infiltrating neoplastic cells that do not form glandular structures and are not associated with intestinal metaplasia (Houghton *et al.*, 2002). The differing outcomes of *H. pylori* infection involves specific choreographed interactions between pathogen and host which are dependent on differences between bacterial strains including strain-specific bacterial factors and/or inflammatory responses governed by genetic diversity, environmental influences or variations in the individual host (Blaser & Berg, 2001).

# 1.3.3 Helicobacter pylori bacterial virulence factors

*Helicobacter* has been postulated to induce growth alterations and malignant transformations of the gastric mucosa through mechanisms involving direct bacterial-host cell contact and/or bacterial secreted products (Mobley, 1997). Since only a relatively small percentage of infected individuals' will progress to neoplasia, it has been suggested that this inconsistency might be in part due to bacterial strain-specific virulence factors (Blaser *et al.*, 1995; Danesh, 1999).





Virulence factors are related to the ability of a microbe to induce disease and many are known to be involved in the pathology of *H. pylori* infection. Following colonisation of the gastric epithelium via the involvement of bacterial outer membrane proteins, among the most important virulence factors are the cytotoxin associated antigen pathogenicity island (*cag*-PAI), cytotoxin A (CagA)

and vacuolating cytotoxin A (VacA). On this basis, *H. pylori* strains have been divided into two groups: Type I strains which are *cag*-PAI positive (*cag*+) and *vacA* positive, and Type II which are *cagA*<sup>-</sup> and do not produce functional VacA.

*H. pylori* is able to invade and colonise the harsh environment within the gastric epithelium using several mechanisms including the use of *H. pylori* urease to catalyse the degradation of urea to carbon dioxide and ammonia enabling the survival of *H. pylori* in the low-pH acid environment of the stomach (Mobley *et al.*, 1991). The possession of flagella allows the bacteria to freely move around the gastric epithelium and to colonise the layer between the mucus gel and epithelial cells (Peek, 2005).

### 1.3.3.1 Outer membrane proteins

Approximately 20% of *H. pylori* in the stomach is found adhered to the surface of mucus epithelial cells (Hessey *et al.*, 1990) involving specialised interactions between host receptors and bacterial adhesins. The bacterium possesses over 30 genes dedicated to the expression of outer membrane proteins, several of which have been classed as adhesins. These have been shown to contribute to the host inflammatory response to *H. pylori* and are likely to be involved in disease progression (Prinz *et al.*, 2001; Mahdavi *et al.*, 2002; Peterson *et al.*, 2006).

One such protein is outer inflammatory protein A (OipA) which can promote interleukin (IL)-8 production and is also associated with bacterial density, gastric inflammation and risk for clinical sequelae such as peptic ulceration (Yamaoka *et al.*, 2002). The adhesin BabA binds to the host fucosylated blood group antigen Lewis-b (Ilver *et al.*, 1998) and SabA adheres to sialylated glycoproteins,

specifically to sialyl- Lewis-X (Mahdavi *et al.*, 2002) causing activation of neutrophils (Unemo *et al.*, 2005; Petersson *et al.*, 2006). It has also been shown that *H. pylori* binds to trefoil factor (TFF)-1 on the epithelial cell surface with the pattern of colonisation mirroring the distribution of TFF-1 (Clyne *et al.*, 2004).

#### 1.3.3.2 cag pathogenicity island

The *cag* pathogenicity island (*cag*-PAI) is a well-characterised *H. pylori* virulence factor present in approximately 60% of Western *H. pylori* strains (N.B. PAI in the context of *cag*-PAI bears no relationship to plasminogen activator inhibitor (PAI) described later in this thesis). It is 40-kb encoding a group of approximately 30 genes inserted into the *H. pylori* glutamate racemase gene. Several *cag*-PAI genes encode a type IV secretory system (TFSS) which acts as a molecular syringe, injecting the CagA protein and other bacterial factors into eukaryotic cells (Censini *et al.*, 1996; Tomb *et al.*, 1997; Akopyants *et al.*, 1998; Alm *et al.*, 1999). Its presence is associated with an increased risk of severe gastritis (Peek *et al.*, 1995; Yamaoka *et al.*, 1998), peptic ulcer (Crabtree *et al.*, 1991; Covacci *et al.*, 1993; Cover *et al.*, 1995), atrophic gastritis (Kuipers *et al.*, 1995; Webb *et al.*, 1999) and distal gastric cancer (Blaser *et al.*, 1995; Parsonnet *et al.*, 1997).

A consequence of *cag*-PAI mediated epithelial cell contact is the induction of the neutrophil activating pro-inflammatory chemokine IL-8. Many *cag* island genes including *cagE*, *cagG*, *cag*H and *cagL* but not *cag*A have been found to be necessary for nuclear factor- $\kappa$ B (NF- $\kappa$ B) mediated IL-8 production (Glocker *et al.*, 1998; Crabtree *et al.*, 1999). Although IL-8 expression does not require CagA delivery, it does require a functional TFSS as disruption of the *cagE* gene

abrogates this response (Tummuru *et al.*, 1995; Sharma *et al.*, 1998; Nozawa *et al.*, 2002; Selbach *et al.*, 2002). It has recently been shown that Nod1, an intracytoplasmic pathogen-recognition molecule is responsible for the activation of NF- $\kappa$ B as it senses *H. pylori* peptidoglycan injected into host cells via the TFSS (Viala *et al.*, 2004).

Several studies in gastric epithelial cell lines have demonstrated that  $cag^+$  H. pylori strains are more potent in the stimulation of IL-8 and induce an enhanced inflammatory neutrophilic response in human tissue compared to those strains in which the island is not present (Crabtree *et al.*, 1994; Crabtree *et al.*, 1995; Crabtree & Farmery, 1995; Sharma *et al.*, 1995; Yamaoka *et al.*, 1996; Shimoyama *et al.*, 1998).  $cag^+$  strains have also been shown to activate mitogen activating protein kinase (MAPK) pathways (Keates *et al.*, 1999; Naumann *et al.*, 1999) activating the proto-oncogenes c-fos and c-jun via activation of the extracellular signal-related kinase (ERK) pathway which in turn activates the transcription factor AP-1 (Meyer-ter-Vehn *et al.*, 2000). The AP-1 family of transcription factors plays a pivotal role in cell proliferation and neoplastic transformation (Angel & Karin, 1991), therefore deregulation of AP-1 proteins or indeed the MAPK pathway may contribute to transformation of gastric epithelial cells.

Both  $cag^+$  and  $cag^-$  strains are able to transactivate the epidermal growth factor (EGF) receptor in gastric epithelial cells via activation of heparin binding-EGF (HB-EGF) which is required for Ras-mediated activation of ERK-1/2 and also for the induction of the early growth response gene *Egr-1* which is a regulator of cell growth, differentiation and survival (Keates *et al.*, 2001; Wallasch *et al.*,

2002; Keates *et al.*, 2005; Du *et al.*, 2007). Differences in gene expression induced by wild-type *H. pylori cag*<sup>+</sup> compared with wild-type *H. pylori cag*<sup>-</sup> showed many genes to be differentially expressed in the Kato III epithelial cell line with three genes, ADAM 10, amphiregulin and HPYR1 found to be upregulated *in vivo* in infected human gastric mucosa (Cox *et al.*, 2001).

*H. pylori* has been shown to up-regulate matrix metalloproteinase-7 (MMP-7) in epithelial cells both *in vivo* and *in vitro* in a Cag dependent manner. *H. pylori*  $cag^+$  strains enhance levels of MMP-7 within the inflamed gastric mucosa which is dependent on activation of ERK1/2 by specific components within the *cag* island (Crawford *et al.*, 2003). Bebb *et al* demonstrated that  $cagE^+$  *H. pylori* strains induced higher levels of MMP-7 expression than  $cagE^-$  strains (Bebb *et al.*, 2003).

#### 1.3.3.3 CagA

CagA is a 145kDa protein which requires the *cag*-PAI TFSS for it to be translocated into gastric epithelial cells leading to disruption of the epithelial apical-junctional complex and junction-mediated barrier function, processes previously shown to participate in the process of carcinogenesis (Amieva *et al.*, 2003). Infection of gastric epithelial cells (AGS) with *cagA*<sup>+</sup> *H. pylori* induces a 'hummingbird' phenotype characterised by cell spreading, elongated cell growth, lamellipodia and filopodia (Segal *et al.*, 1999). The inflammatory interleukins IL-1 $\beta$ , IL-6, IL-8 and tumour necrosis factor  $\alpha$  (TNF- $\alpha$ ) have all been detected in patients infected with *cagA*<sup>+</sup> strains of *H. pylori* (Yamaoka *et al.*, 1997).

Once inside the cell, CagA is subsequently phosphorylated by members of the Src family of kinases (Asahi *et al.*, 2000; Odenbreit *et al.*, 2000; Stein *et al.*, 2000) resulting in activation of C-terminal Src kinase, inhibiting the activity of Src leading to a decrease in the levels of phospho-CagA in a negative feedback loop (Selbach *et al.*, 2003; Tsutsumi *et al.*, 2003). Phospho-CagA interacts with several host signalling molecules, activating protein kinase cascades (Segal *et al.*, 1999). It activates both ERK which is a member of the MAPK family and a eukaryotic phosphatase (SHP-2) leading to morphological changes within the host cell due to rearrangement of the actin cytoskeleton (Higashi *et al.*, 2002; Higashi *et al.*, 2004).

The three prime repeat regions of *cagA* contain EPIYA motifs (consisting of glutamine-proline-isolucine-tyrosine-alanine), the number of which determines the level of CagA phosphorylation (Argent *et al.*, 2004). The EPIYA motif has a dual function in both membrane association and tyrosine phosphorylation which are both critically involved in the activity of CagA to deregulate intracellular signalling.

In gastric epithelial cells non-phosphorylated CagA intracellularly interacts with *c*-Met inducing a motogenic response (Churin *et al.*, 2003) and it also induces inactivation of Crc kinase, dephosphorylation of cortactin, and adapter molecule Grb-2 (Mimuro *et al.*, 2002; Selbach *et al.*, 2003). In AGS cells, *H. pylori* that co-express CagA with another virulence factor, HspB, (Dunn *et al.*, 1992; Macchia *et al.*, 1993) a protein involved in gastric carcinogenesis (Iaquinto *et al.*, 2000; De Luca *et al.*, 2003), are able to influence cell growth by inducing an increase in cyclin D3 and phosphorylated Retinoblastoma (Rb) gene product (De

Luca *et al.*, 2003; De Luca *et al.*, 2004). Interestingly, expression of both CagA and HspB are required to induce these effects in AGS cells, suggesting that cooperation among different *H. pylori* proteins is necessary to induce cell cycle alterations in infected cells.

### 1.3.3.4 VacA

An independent *H. pylori* locus associated with gastric adenocarcinoma development is *vacA*. This encodes a secreted protein that induces large cytoplasmic vacuole formation in eukaryotic cells (Cover & Blaser, 1992a; Cover & Blaser, 1992b; Phadnis *et al.*, 1994). The toxin inserts itself into the epithelial cell membrane allowing bicarbonate and organic anion release, facilitating the formation of transmembrane pores which permeabilise the gastric epithelium to urea (Papini *et al.*, 1998; Szabo *et al.*, 1999; Tombola *et al.*, 1999; Jungblut *et al.*, 2000).

All *H. pylori* strains possess the *vacA* gene, yet only approximately 50% of isolates express a toxin which has vacuolating activity (Cover *et al.*, 1994; Atherton *et al.*, 1995). This is due to two major sequence variations within *vacA* located to both the *vacA* secretion signal sequence which are known as allele types s1a, s1b, slc or s2 and to the midregion known as allele types m1 or m2 (Atherton *et al.*, 1995; van Doorn *et al.*, 1998; van Doorn *et al.*, 1999a). Enhanced gastric cell injury is associated with strains possessing the m1 allele (Ghiara *et al.*, 1995; Atherton *et al.*, 1997) and increased risk for distal gastric cancer compared to *vacA* m2 strains (Gerhard *et al.*, 1999; Miehlke *et al.*, 2000; Figueiredo *et al.*, 2002).

*vacA* induces apoptosis in gastric epithelial cells which is dependent upon strain variations in the structure of VacA (Kuck *et al.*, 2001; Cover *et al.*, 2003). It has also been shown that increased concentration of VacA are found in the gastric juice of patients with duodenal ulcers compared to those without, thus suggesting the quantity of VacA could also be important in determining clinical outcome of disease (Papini *et al.*, 1998). VacA binds to a unique receptor-type protein tyrosine phosphatase (PTP $\xi$ ) which is a member of a family of receptor like enzymes that are involved in regulation of proliferation, differentiation and adhesion (Fujikawa *et al.*, 2003).

Recent observations describe that VacA contributes to evasion of *H. pylori* to the adaptive immune response by interfering with T cell proliferation and activation via multiple mechanisms. VacA actively suppresses proliferation and activation of transformed Jurkat T cells *in vitro* via inhibition of nuclear factor of activated T cell (NFAT)-mediated IL-2 signalling (Gebert *et al.*, 2003). As well as this, VacA can also block activation of primary human T cells which is independent of NFAT (Sundrud *et al.*, 2004).

#### 1.3.3.5 Helicobacter felis

*H. pylori* can only colonize a limited number of hosts including non-human primates, germ-free or barrier raised piglets, germ-free dogs and recently, laboratory raised cats. Mice experimentally infected with *H. pylori* develop gastritis although this is less severe than that seen in human *H. pylori* infection (Ferrero *et al.*, 1994; van Doorn *et al.*, 1999b). However, a mouse-adapted strain of *H. pylori*, SS1, has proven useful in modelling gastric disease in mice (Lee *et al.*, 1997; Sutton *et al.*, 2000).

Infection of mice with the related gastric species *Helicobacter felis* (*H. felis*) results in severe gastritis (Sakagami *et al.*, 1996) and infection of C57BL/6 mice with *H. felis* mimics many of the pathogenic changes commonly found in humans infected with *H. pylori* (Lee *et al.*, 1990; Wang *et al.*, 1998). Although both *H. felis* and *H. pylori* are closely related (Dewhirst *et al.*, 1994), *H. felis* is deficient in production of the *H. pylori* CagA and the cytotoxin VacA (Xiang *et al.*, 1995).

## **1.3.4 Environmental factors**

Many environmental factors, particularly relating to dietary components are thought to affect the pathogenesis of gastric adenocarcinoma. The correlation between high salt diets, *H. pylori* infection and gastric cancer has been reported which appeared to be most prevalent in Japanese people (Go, 2002). These findings are consistent with a recent Japanese study on a large group of humans which also suggested that salt is an important environmental cofactor in patients infected with *H. pylori* and with atrophic gastritis (Shikata *et al.*, 2006). Salt is a dietary factor reported to promote *H. pylori* induced progression towards gastric atrophy and gastric cancer by increasing *H. pylori* colonisation and cellular proliferation in the antrum and corpus of C57BL/6 mice infected with the SS1 strain of *H. pylori* (Fox *et al.*, 1999).

*H. pylori* infected subjects on proton pump inhibitor (PPI) therapy to decrease acid secretion have been shown to have a modified gastric chemistry (Mowat *et al.*, 2000). At a neutral pH (Labenz *et al.*, 1996), ingested nitrite (derived from dietary nitrate) is not reduced to nitric oxide by ascorbic acid but accumulates in the stomach (Lundberg *et al.*, 2008). Proton pump inhibitor therapy also lowers the intragastric concentration of ascorbic acid, probably because of the relative

instability of the vitamin at a higher pH (Mowat & McColl, 2001). It is recognized that an elevated nitrite-to-ascorbic acid ratio predisposes to the formation of potentially carcinogenic N-nitroso compounds (NOCs) and H. *pylori* infection in animals has been shown to increase their susceptibility to NOC induced cancers (Sugiyama & Asaka, 1998).

## 1.3.5 Helicobacter and gastrin

Infection with *H. pylori* is accompanied by marked changes in gastric physiology and a role for gastrin in the progression of mucosal damage and development of cancer in response to this bacterium has emerged (reviewed Rozengurt & Walsh, 2001). Basal plasma gastrin concentrations are increased approximately two fold in *H. pylori* infected subjects and maybe elevated six fold in response to food (Dockray *et al.*, 2001). Hypergastrinaemia occurs early in the course of human *H. pylori* infection (Levi *et al.*, 1989) and the mechanism by which this occurs involves IL-1 $\beta$  and TNF $\alpha$  stimulated release of gastrin from antral G-cells (Weigert *et al.*, 1996).

Gastrin is negatively regulated by the release of somatostatin from D-cells which is inhibited by the pro-inflammatory cytokines TNF- $\alpha$  and interferon-gamma (IFN- $\gamma$ ; Beales *et al.*, 1997; Zavros *et al.*, 2003). *H. pylori* infection also reduces somatostatin with decreased levels reported in the antrum of infected patients (Kaneko *et al.*, 1992). Following *H. pylori* eradication in patients with duodenal ulcer disease there is an increase in the amount of somatostatin mRNA suggesting that the number of D-cells are reduced by infection (Moss *et al.*, 1992). The trophic actions of gastrin in the gastric mucosa and its interaction with *Helicobacter* have been further studied in mice over-expressing amidated gastrin (INS-GAS; discussed in section 1.3.7.1).

## 1.3.6 Host response to *H. pylori* infection

Several other factors are now thought to influence the susceptibility of infected *H. pylori* subjects to developing gastric carcinoma such as the nature of the immune response elicited towards *H. pylori*. This involves both humoral and cellular immune responses and has long been considered an important determinant of the pathology induced in the particular host (reviewed in Wilson & Crabtree, 2007). The gastric mucosa of infected patients has increased concentrations of pro-inflammatory cytokines such as IL-6, IL-8 and TNF- $\alpha$  (Crabtree *et al.*, 1991; Crabtree *et al.*, 1993; Peek *et al.*, 1995; Crabtree, 1996), which are likely to modulate exocrine and endocrine cell function, affecting local acid secretion, the colonisation pattern of *H. pylori* and thus the distribution of gastritis.

Pro-inflammatory cytokines are also responsible for leukocyte recruitment and are crucial for inflammation since IL-8 secretion is directly linked to severity of gastritis (Peek *et al.*, 1995). In addition to cytokine-mediated activation of leukocytes, *H. pylori* also directly activates monocytes and macrophages via toll-like receptor (TLR)-2 which are receptors involved in the innate immune response, resulting in the induction of lymphocyte T-helper (Th)-1 (Mandell *et al.*, 2004). T lymphocyte deficient mice were found to be largely protected from gastric mucosal changes due to *H. pylori* infection, demonstrating a direct involvement of T lymphocytes in gastric cancer progression (Eaton *et al.*, 2001). The pattern of cytokine expression in both human and the Mongolian gerbil

gastric mucosa infected with *H. pylori* is T-helper type 1 (Th1) polarised (Crabtree *et al.*, 2004; Yamaoka *et al.*, 2005) and probably accounts for the increased levels of plasma gastrin and reduced amounts of somatostatin in infected subjects (as discussed in section 1.3.5).

Host gene polymorphisms in the proinflammatory IL-1 gene cluster (IL-1B encoding IL-1 $\beta$  and IL-1RN encoding its receptor) increase the risk of gastric cancer and its precursors in the presence of *H. pylori* (el-Omar *et al.*, 2000). In defined patient cohorts, cytokine gene polymorphisms in the progression to gastric cancer have also been extensively studied and clearly play a role in the risk of *H. pylori* induced gastric adenocarcinoma. Potentially important polymorphisms have been found in IL-2, IL-6, IL-8, IL-10 and IL-12 (Perez-Perez *et al.*, 2005; Lobo *et al.*, 2005; Lu *et al.*, 2005b; Navaglia *et al.*, 2005; Ohyauchi *et al.*, 2005; Togawa *et al.*, 2005; Zabaleta *et al.*, 2006).

Several human cancers are initiated by chronic inflammation for example colon cancer by inflammatory bowel disease and oesophageal adenocarcinoma by gastroesophageal reflux disease (Houghton & Wang, 2005). A relationship between chronic inflammation and cancer was established as early as 1863 (Virchow, 1863). Chronic inflammation in the gastric mucosa as a result of *H. pylori* infection could promote gastric cancer development by promoting processes such as DNA damage, apoptosis and hyperproliferation of gastric epithelial cells. *H. pylori* infection increases the expression of cyclooxygenase (COX)-2 (Franco *et al.*, 1999; McCarthy *et al.*, 1999), which has been shown to promote carcinogenesis by inhibiting apoptosis (Kim *et al.*, 2000; Wambura *et* 

*al.*, 2002), increasing invasion, angiogenesis (Chang *et al.*, 2005; Wu *et al.*, 2005) and proliferation (Wambura *et al.*, 2002).

The involvement of oxidative stress in gastrointestinal carcinogenesis is also well established (Seril *et al.*, 2003). *H. pylori* infection leads to an increase in expression of an important polyamine catabolic enzyme, spermine oxidase, SMO(PAOh1) which oxidises spermine to produce the DNA damaging reactive oxygen species (ROS),  $H_2O_2$  (Xu *et al.*, 2004). Inflammation results in nitric oxide (NO) and ROS production by leukocytes, macrophages and gastric epithelial cells (Bagchi *et al.*, 1996; Mannick *et al.*, 1996; Wilson *et al.*, 1996), resulting in DNA damage and mutations (Jaiswal *et al.*, 2001). Apoptosis of epithelial cells is also promoted by chronic inflammation (Moss *et al.*, 1996; Piotrowski *et al.*, 1996). *H. pylori* infection can elicit a strong host response and an accumulation of mutations arising from hyperproliferation of epithelial cells and DNA damage caused by NO and ROS could lead to the formation of malignant transformations.

### 1.3.7 Animal models of *Helicobacter*-induced gastric carcinoma

Many *in vivo* studies have employed experimental animal models, especially infection of inbred C57BL/6 mice, hypergastrinaemic INS-GAS mice or Mongolian gerbils infected with *Helicobacter* (reviewed in Pritchard & Przemeck 2004; Peek & Crabtree, 2006) to provide an insight into the host, bacterial and environmental factors involved in gastric carcinogenesis.

Conventional human gastric carcinoma cells lines such as AGS, MKN28, MKN45 and KATO III have been useful for the *in vitro* investigation into the effects *H. pylori* has on cell signalling, gene expression, apoptosis and

proliferation (Chang *et al.*, 1993; Jones *et al.*, 1999; Peek *et al.*, 1999; Chiou *et al.*, 2001; Iwamoto *et al.*, 2005). However, limited information regarding the roles of these factors in gastric carcinogenesis can be obtained since they are all derived from established tumours. The development of mouse, and more recently human (Wroblewski *et al.*, 2003) primary epithelial cell cultures might prove to be more useful in providing an insight into the progression to gastric carcinoma.

#### 1.3.7.1 INS-GAS mice

INS-GAS mice express human gastrin under the control of the 0.4kb of the insulin promoter resulting in the production of amidated gastrin in pancreatic  $\beta$  cells and a 2 fold elevation (~150 pM) in plasma amidated gastrin levels at 1-3 months (Wang *et al.*, 1996). They show gastric mucosa hyperplasia which is associated with increased levels of TGF- $\alpha$  and HB-EGF. Initial hyperacidity is followed by hypoacidity, and at the age of 5 month onwards there is a steady decline in parietal cell number and invasive carcinomas are seen by the age of 20 months (Wang *et al.*, 2000).

INS-GAS mice are an attractive model of gastric cancer as they display many of the distinct histological and epidemiological steps as humans in the progression to intestinal-type adenocarcinoma. Both *H. felis* (Wang *et al.*, 2000; Fox *et al.*, 2003b) and *H. pylori* (Fox *et al.*, 2003a; Fox *et al.*, 2003b) have been shown to act synergistically with chronic hypergastrinaemia to accelerate the progression of gastric cancer with histopathological changes already present at 6-7 months in infected animals. Infection of INS-GAS mice induces gastric adenocarcinoma but is restricted to males which respond more aggressively and rapidly to infection, consistent with the greater incidence of gastric carcinoma in men (Fox

*et al.*, 2003b). Removal of *cagE* temporally slows down but does not abolish pathological progression (Fox *et al.*, 2003a), in agreement with the finding that CagE is important for the assembly of the TFSS that plays a role in the transfer of CagA from *Helicobacter* into the epithelial cell (Guillemin *et al.*, 2002).

Prior to the development of disease there is an increase in mucosal IL-1 $\beta$  levels. However, this is not gender specific (Fox *et al.*, 2003a). The only major drawback of this useful model is that hypergastinaemia preceded rather than results from *Helicobacter* infection and the site where gastric adenocarcinomas occurs is the only major difference between humans and the INS-GAS model of gastric cancer. Gastric adenocarcionmas tend to develop in the antrum or at the junction of the antrum and corpus (Correa, 1996; Peek & Blaser, 2002) whereas gastric cancer in INS-GAS mice develops more frequently in the corpus (Fox *et al.*, 2003b).

#### 1.3.7.2 Helicobacter vaccination

Even though current antimicrobial treatment regimens efficiently eradicate *H. pylori*, antibiotic resistance and reinfection remains a serious problem (Graham, 1998; Soto *et al.*, 2003; Koletzko *et al.*, 2006). Many strategies using animal models of infection have been attempted in order to try to develop either protective prophylactic or therapeutic vaccines against *H. pylori* which is an exciting prospect. Mice (Corthesy-Theulaz *et al.*, 1995) and ferrets (Cuenca *et al.*, 1996) in addition to beagle dogs (Rossi *et al.*, 2004) and Rhesus monkeys (Dubois *et al.*, 1998) have all shown that therapeutic immunisation reduces *H. pylori* colonisation and gastritis. A recently developed vaccine model in the

Mongolian gerbil showed no postimmunisation gastritis following *H. pylori* challenge in infected animals (Jeremy *et al.*, 2006). To date there have been very few clinical studies testing the immunogenicity and safety of potential *H. pylori* vaccines (Michetti *et al.*, 1999; Kotloff *et al.*, 2001; Banerjee *et al.*, 2002) and the ability to translate experimental model studies to human remains an elusive goal.

# 1.4 Epithelial-mesenchymal interactions and signalling

Two-way interactions between epithelial and underlying mesenchymal cells (such as myofibroblasts) are crucial for normal mucosal organisation and this occurs during development (Goke *et al.*, 1998), wound healing (McKaig *et al.*, 1999; Watanabe *et al.*, 2000) and progression to cancer. These interactions were first demonstrated in 1967 when the separation of embryonic epithelium of kidney, lung, salivary gland or pancreas from native mesenchymal cells resulted in the failure of these tissues to undergo branching morphogenesis but resumed to normal when these cell types were reintroduced (Grobstein, 1967).

The mechanism involved in epithelial-mesenchymal signalling often includes expression of specific growth factors by mesenchymal cells, which act in a paracrine fashion to stimulate epithelial cells. Intestinal mesenchymal cells do not express the hepatocyte growth factor (HGF)-receptor (c-Met) yet release HGF which acts via c-Met expressed on intestinal epithelial cell to stimulate proliferation (Goke *et al.*, 1998; Prat *et al.*, 1991; Fukamachi *et al.*, 1994). Keratinocyte growth factor (KGF) provides another example of this interaction as it originates solely from mesenchymal cells (Brauchle *et al.*, 1994; Boismenu & Havran, 1994) and acts exclusively on epithelial cells (Rubin *et al.*, 1995) where it plays a role in wound healing after injury (Werner *et al.*, 1992; Staiano-Coico *et al.*, 1993). Breast cancer epithelial cells show an up-regulation of PDGF yet do not expresses the PDGF-receptor. However, this is expressed by myofibroblasts (Ichiki *et al.*, 1995; Bostrom *et al.*, 1996) which show increased proliferation and migration when treated with PDGF (Sommer *et al.*, 2002; Tangkijvanich *et al.*, 2002).

Epithelial and mesenchymal cells are present in the specialised microenvironment of the gastric stem cell "niche" and are important for stem cell regulation and development via mesenchymal-epithelial cross-talk (Brittan & Wright, 2002; Brittan & Wright, 2004). Dysfunction of this system is likely to contribute to the process of epithelial tumour development arising on a background of chronic tissue injury or inflammation, an example of which is provided by the development of gastric cancer in patients infected with *H. pylori*.

The mechanisms of gastric cancer progression remain largely unknown but it does involve remodelling of the epithelium and alterations in cell type/ number as well as altered expression of growth factors, extra-cellular matrix (ECM) proteases and protease inhibitors (Dimaline & Varro, 2007). *H. pylori* increases the release of MMP-7 (a member of a family of proteolytic enzymes important in maintenance and remodelling of interactions between epithelial cells and basement membranes; McCawley, 2001) from the gastric epithelium (Wroblewski *et al.*, 2003). This stimulates gastric myofibroblasts to produce both insulin-like growth factor II (IGF-II) and IGF-binding protein (IGFBP)-5 (Hemers *et al.*, 2005; McCaig *et al.*, 2006). In turn, MMP-7 cleaves IGFBP-5 and the free bioavailable IGF-II not only acts on myofibroblasts to stimulate

migration, proliferation and adhesion of but also stimulates proliferation of gastric epithelial cells (McCaig *et al.*, 2006). These mechanisms might play an important role in altering the niche that favours premalignant changes in the gastric epithelium.

# 1.5 Urokinase plasminogen activator (uPA) system

The uPA system is an enzymatic system central to a spectrum of physiological processes with its primary function being the control of fibrinolysis. Binding of the protease uPA to its receptor, uPAR initiates a proteolytic cascade resulting in the conversion of the zymogen plasminogen into plasmin. Plasmin has a wide range of functions and can degrade fibrin, extracellular basement membrane components such as fibrinogen and fibronectin (Liotta *et al.*, 1981; Mochan & Keler 1984; Goldfarb *et al.*, 1986), whilst activating others such as MMPs (Keski-Oja *et al.*, 1992; Baramova *et al.*, 1997; Kazes *et al.*, 1998). The serpin inhibitor  $\alpha$ 2-antiplasmin is the primary inhibitor of plasmin (Wiman & Collen, 1978). The production of plasmin by the uPA system is tightly regulated by the activity of plasminogen activator inhibitors (PAIs) of which there are three: PAI-1, -2 and -3 (Figure 1.3). The activity of the uPA system also activates certain growth factors such as HGF (Naldini *et al.*, 1992; Mars *et al.*, 1993), TGF $\beta$  (Yee *et al.*, 1993) and IGF (Remacle-Bonnet *et al.*, 1997).

Tissue remodelling during normal and pathological processes such as embryogenesis (Werb *et al.*, 1999), wound healing (Lund *et al.*, 1999), angiogenesis (Pepper, 2001) and cancer invasion (Johnsen *et al.*, 1998) are regulated by the presence of ECM degrading proteases and their corresponding

inhibitors. As well as its role in fibrinolysis, numerous studies (often utilising uPA system transgenic mice) have highlighted an involvement of this system in most of these processes (Dano *et al.*, 1985; Liotta *et al.*, 1991; Andreasen *et al.*, 2000).

## 1.5.1 Urokinase plasminogen activator

The conversion of plasminogen to plasmin is activated by the highly specific protease uPA. The 2.5kb uPA mRNA (Verde *et al.*, 1984) gives rise to a 411 amino acid, 53-kDa catalytically inactive single-chain polypeptide known as prouPA or sc-uPA (Skriver et al., 1982; Wun *et al.*, 1982; Andreasen *et al.*, 1997). Pro-uPA is released from many cell types and lacks plasminogen-activating activity. It is cleaved at lysine 158-159 to yield its two-chain, active form, also known as high molecular weight uPA (HMW-uPA) or tc-UPA.

Plasmin is an important activator of uPA (Dano *et al.*, 1985; Andreasen *et al.*, 1997) and a positive feedback mechanism exists in which plasmin generates uPA activity which in turn leads to the formation of new active plasmin from plasminogen in a highly potent amplification mechanism (Petersen *et al.*, 1988). Several other proteases have been shown to catalyse activation of uPA including cathepsin B and cathepsin G (Kobayashi *et al.*, 1991), plasma kallikrein (Ichinose *et al.*, 1986), mast cell tryptase (Stack & Johnson, 1994) and nerve growth factor gamma (Wolf *et al.*, 1993). Certain proteins such as the angiogenic growth factors vascular endothelial growth factor (VEGF) can also activate uPA (Prager *et al.*, 2004).



#### Figure 1.3

Schematic outline of the urokinase plasminogen activator (uPA) system After binding to uPAR, uPA in its active form converts plasminogen to plasmin which is able to control fibrinolysis, breakdown the extracellular matrix (ECM) and activate growth factors and matrix metalloproteinases (MMPs), enhancing extracellular proteolysis. The plasminogen activator inhibitor PAI-1 (and PAI-2) control plasmin production by inhibiting uPA.  $\alpha$ 2-antiplasmin is also a specific inhibitor of plasmin.

High molecular weight-uPA is a two-chain protein (A and B chain), linked by a single sulphide bond. The tertiary structure of uPA is divided into three main domains. The N-terminal A-chain (light chain) growth factor domain (residues 1-49) contains the binding site for uPAR at residues 21-32 (Appella *et al.*, 1987; Saksela & Rifkin, 1988; Blasi & Carmeliet, 2002). This is followed by a kringle domain (residues 50-131) which together make up the amino-terminal fragment (ATF; Stoppelli *et al.*, 1985). The carboxyl-terminal region (B-chain/ heavy chain) contains the catalytic serine protease domain (residues 159-411).

Further cleavage of active uPA in the linker region between the kringle and serine protease domain (at lysine135-lysine136) splits uPA into two fragments:

the ATF fragment and B-chain (also designated low molecular weight uPA, LMW-uPA; Kasai *et al.*, 1985a; Kasai *et al.*, 1985b). LMW-uPA is soluble and maintains its proteolytic activity but does not bind to the uPAR receptor (Irigoyen *et al.*, 1999; Alfano *et al.*, 2005). Both pro-uPA and HMW-uPA bind to uPAR with high affinity (0.1-1nM) but the conversion of pro- to HMW-uPA occurs more efficiently when the zymogen is bound to uPAR (Irigoyen *et al.*, 1999). Binding of uPA to its cell surface receptor significantly increases the rate of its catalytic activity towards plasminogen at least 20-fold (Ellis *et al.*, 1989; Quax *et al.*, 1991).

The human gene coding for uPA is located on chromosome 10 and consists of 11 exons and 10 introns (Ibanez-Tallon et al., 1999). The 5' flanking sequence contains several elements that indicate a tight transcriptional regulation (Ibanez-Tallon et al., 1999; Ibanez-Tallon et al., 2002). Upstream of the TATA box lies a CG rich region containing several binding sites for the transcription factor Sp1. This has a prominent role in uPA gene expression in PC3 cells and is targeted by several signal transduction pathways (Benasciutti et al., 2004). Approximately 2 kb upstream of the transcription start site is a well-characterised enhancer containing an upstream Ets (PEA3) binding site and two AP-1 binding sites. There is a cooperativity mediator region containing the binding sites for upstream enhancer factor (UEF) transcription factors 1-4 (Ibanez-Tallon et al., 1999; Ibanez-Tallon et al., 2002). Two more elements have been identified in the regulatory region of the human uPA gene: an NK-kB binding site which mediates the transcriptional induction of gene expression by phorbol esters in the absence of the enhancer region (Hansen et al., 1992) and a cell type-specific silencer (Cannio et al., 1991).



#### Figure 1.4

#### Structure of uPA

Pro-uPA is secreted as a single-chain zymogen containing an amino terminal fragment (ATF) consisting of a growth factor domain, a kringle domain and also a carboxy-terminal fragment containing a serine protease domain. Pro-uPA is activated and cleaved by plasmin to give high molecular weight uPA (HMW-uPA) which can be further cleaved into low molecular weight uPA (LMW-uPA) and ATF.

Transcriptional activation of the uPA gene can be obtained through various different stimuli e.g. phorbol esters, growth factors and signal transduction pathways, which mainly target the enhancer. These include NF-κB (Guerrini *et al.*, 1996; Sliva *et al.*, 2002a; Sliva *et al.*, 2002b; Lee *et al.*, 2006a), phosphoinositol 3-kinase (PI3K)/Akt (Lee *et al.*, 2006a), jun N-terminal kinase (JNK; Miralles *et al.*, 1998; Parra *et al.*, 2000; Benasciutti *et al.*, 2004), MEK/ERK and p38-MAPK pathways (Irigoyen *et al.*, 1997; Shin *et al.*, 2002; Shin *et al.*, 2003; Lee *et al.*, 2006b).

Cell-surface bound and soluble uPA have been shown to promote cell proliferation (Rabbani *et al.*, 1990; He *et al.*, 1991; Rabbani *et al.*, 1992; Stepanova *et al.*, 1999), stimulate migration (Fibbi *et al.*, 1998; Odekon *et al.*,

1992; Resnati *et al.*, 1996) and modulate adhesion of several cells such as smooth muscle cells (Chang *et al.*, 1998; Sidenius *et al.*, 2002). The mitogenic activity of uPA has been shown in several cell-types and uPA stimulates growth in human epidermal cells, normal and malignant renal cells and in melanoma cells (Kirchheimer *et al.*, 1988; Kirchheimer *et al.*, 1989). The ATF of uPA has also been identified as a mitogenic factor isolated from osteoblast-like cells (Rabbani *et al.*, 1990).

## **1.5.2 Urokinase plasminogen activator receptor**

The specific binding of uPA to cell surfaces was first reported in 1985 and three years later Nielsen *et al.*, purified and characterised a membrane protein responsible for this binding and designated it uPAR (Nielsen *et al.*, 1988). Fully processed uPAR is heavily glycosylated (Behrendt *et al.*, 1990) and has a glycosyl-phosphatidyl-inositol moiety (GPI anchor) in its carboxyl-terminal region, integrating the receptor into the outer leaflet of the cell membrane (Ploug *et al.*, 1991a; Ploug *et al.*, 1993). Many cell types express uPAR including granulocytes (Nykjaer *et al.*, 1990), endothelial cells (Pepper *et al.*, 1993), B-lymphocyes (Plesner *et al.*, 1994), activated T-cells (Nykjaer *et al.*, 1994).

The human 1.4kb cDNA has been cloned and mapped to chromosome 19q13.2 (Roldan *et al.*, 1990; Webb *et al.*, 1994) and gives rise to a 270 amino acid polypeptide with a molecular mass of 50 kDa (Nielsen *et al.*, 1988; Ploug *et al.*, 1991b). The gene is composed of seven exons, seperated by six introns and occupies approximately 21.23kb. The 5' region of the gene contains a CpG-rich

island and several putative *cis*-regulatory elements. These include two AP-1, an SP2, two SP1 and a novel NF-κB sequence (Wang *et al.*, 1995).

Various factors such as phorbol 12-myristate 13-acetate (PMA) (Langer *et al.*, 1993), cytokines such as TNF- $\alpha$  (Sitrin *et al.*, 1994; Wang *et al.*, 1994), TGF- $\beta$ , IL-2,-3,-4,-7 (Nykjaer *et al.*, 1994), IFN- $\alpha$ , - $\gamma$  (Kirchheimer *et al.*, 1988), as well as growth factors including VEGF (Mandriota *et al.*, 1995) have all been shown to modulate uPAR expression in a number of different cell types.

The ternary structure of uPAR is composed of three cysteine rich extracellular domains known as domains 1, 2 and 3 (D1, D2 and D3), connected by short linker regions (Ploug & Ellis, 1994). The uPA binding site is near to the N-terminus located within D1 (residues 1-87), leaving the whole external surface available for other interactions (Behrendt *et al.*, 1991; Ploug *et al.*, 1994). High affinity binding of uPAR to pro-uPA, HMW-uPA and ATF occurs (Figure 1.4; Cubellis *et al.*, 1986). As well as this, uPAR also associates with the ECM protein vitronectin (VN), integrin family members (Wei *et al.*, 1996; Wei *et al.*, 2001), and with several other co-receptors, including the EGF-R (Liu *et al.*, 2002) and endocytosis receptors of the low density lipoprotein receptor (LDLR) family (Strickland *et al.*, 2002; discussed in sections 1.5.2.1 and 1.5.5.1).

#### 1.5.2.1 Cell signaling via uPAR

The binding of uPA to uPAR stimulates intracellular signaling (Kjoller, 2002). However, the mechanisms by which uPAR mediates signalling events remains to be fully elucidated as uPAR has no transmembrane domain. However, multiple transmembrane adapters and signaling receptors have been coimmunoprecipitated or co-localised with cell surface uPAR suggesting they act as signal transducers (Resnati *et al.*, 1996). These include integrins such as  $\beta$ 1 and  $\beta$ 3 (Xue *et al.*, 1997; Wei *et al.*, 2001),  $\beta$ 2 (Bohuslav *et al.*, 1995; May *et al.*, 1998) and  $\beta$ 5 (Carriero *et al.*, 1999) which can transduce signals into the cell through the src-kinases activated by caveolin (Stahl & Mueller, 1995; Wei *et al.*, 1996; Chapman *et al.*, 1999; Wei *et al.*, 1999), a protein associated with intracellular signaling involved in cell adhesion (Okamoto *et al.*, 1998).

FPR-like receptor-1/lipoxin A4 receptor (FPRL-1) is a G-protein coupled receptor reported to interact with uPAR and activate NF-κB, MEK/ERK and Rho pathways (Behrendt *et al.*, 2000; Resnati *et al.*, 2002). In certain cells, the protein gp130 seems to mediate uPAR responses leading to activation of the janus kinase/ signal transducers and activators of transcription (JAK-STAT) signaling pathways (Koshelnick *et al.*, 1997). Epidermal growth factor receptor coimmunoprecipitates with uPAR and can be constitutively activated by ligand bound uPAR, activating ERK to promote proliferation and migration (Liu *et al.*, 2002; Jo *et al.*, 2003; Jo *et al.*, 2005). Non-receptor tyrosine kinases such as hck, fyn, lyn, jak1 any tyk2 (Bohuslav *et al.*, 1995; Resnati *et al.*, 1996; Dumler *et al.*, 1998) also coimmunoprecipitate with uPAR and other pathways shown to be activated downstream of uPAR ligand binding include fos, jun, myc and rac (Rabbani *et al.*, 1997; Nguyen *et al.*, 1998; Kjoller & Hall, 2001).

## **1.5.3 Plasminogen activator inhibitors (PAIs)**

There are three specific serine protease inhibitors (serpins) of uPA known as plasminogen activator inhibitor (PAI) -1, -2 and -3 (Ginsburg *et al.*, 1986; Wilczynska *et al.*, 1995; Gettins, 2002). PAI-3 (also called SERPINA5, or protein C inhibitor, PCI) is a glycoprotein of 387 residues produced by several cell types and found in many tissues. PAI-3 is an inhibitor of activated protein C (APC) but also inhibits a wide range of other proteases including thrombin, kallikrein and prostate specific antigen (Espana *et al.*, 1989; Suzuki *et al.*, 1989; Espana *et al.*, 1991). Humans show a broad tissue expression pattern for PAI-3 (Laurell *et al.*, 1992; Radtke *et al.*, 1994; Elisen *et al.*, 1998) which functions primarily in the coagulation cascade and is involved in the process of reproduction demonstrated by male PAI-3<sup>-/-</sup> mice being infertile due to abnormal spermatogenesis (Uhrin *et al.*, 2000).

Although PAI-3 can inhibit uPA (Geiger *et al.*, 1989; Espana *et al.*, 1993), the major inhibitor of uPA is PAI-1 (SERPINE1) with PAI-2 (SERPINB2) being a less potent inhibitor. PAI-2 is a 47-kDa protein expressed in many cell types including gastric cancer cells (Ye *et al.*, 1989; Nakamura *et al.*, 1992), and can exist in two forms. Secreted PAI-2 modulates uPA activity but approximately 80% of PAI-2 is cytosolic acting within the cell to inhibit apoptosis (Kumar *et al.*, 1991; Dickinson *et al.*, 1995). Gene expression of PAI-2 is mediated by members of the CREB and AP-1 family of transcription factors (Schleuning *et al.*, 1987; Cousin *et al.*, 1991; Costa *et al.*, 2000). Within the PAI-2 promoter there are two putative AP-1 sites and a combined CRE/AP-1 site to which JunD has been shown to interact (Dear *et al.*, 1997; Costa *et al.*, 2000). Recently, PAI-2 expression has been shown to be regulated by gastrin in the AGS gastric cancer

cell line (Varro et al., 2002b) and to be increased by *H. pylori* (Varro et al., 2004).

# 1.5.4 Plasminogen activator inhibitor-1 (PAI-1)

PAI-1 was first discovered in rabbit endothelial cells in 1981 by Loskutoff *et al* (Loskutoff & Edgington, 1981). It is a single chain glycoprotein with a molecular mass of 50 kDA and is the principle physiologic regulator of plasminogen activation (Lawrence *et al.*, 1989; Fay *et al.*, 1992; Carmeliet *et al.*, 1993a; Carmeliet *et al.*, 1993b). The association of PAI-1 deficiency with clinical bleeding demonstrates that PAI-1 primarily functions to regulate haemostasis (Fay *et al.*, 1992).

The PAI-1 gene is an acute phase response gene approximately 12.2 kb in length, mapped to the long arm of chromosome 7 at 7q21.3-q22 and it is composed of 9 exons and 8 introns (Klinger *et al.*, 1987; Kruithof, 1988). Two distinct PAI-1 mRNAs (approximately 2.3 kb and 3.2 kb) are expressed by human cells, differing in the length of their 3' untranslated regions as a consequence of alternative polyadenylation. The 5'-flanking region of the PAI-1 gene has been extensively characterised and is considered to be the PAI-1 promoter containing the transcription initiation site, a TATA box, four Sp1 sites, an AP-1 and AP-2 binding site and several regulatory sequences.

PAI-1 can be rapidly induced suggesting it has a high biosynthetic rate and a variety of cultured cells synthesise PAI-1 due to the number of transcriptional elements within the PAI-1 promoter. These elements confer transcriptional responsiveness to a variety of mediators including growth factors, cytokines and

hormones (Keeton *et al.*, 1991; Fearns & Loskutoff, 1997; Stefansson *et al.*, 2003). Lipopolysaccharide (LPS) induces elevated PAI-1 which is mediated in part by the release of cytokines such as TNF $\alpha$  and IL-1 from inflammatory cells (Schleef *et al.*, 1988; van den Berg *et al.*, 1988; Sawdey & Loskutoff, 1991). Growth factors such as TGF $\beta$  also play a role in PAI-1 regulation (Sawdey & Loskutoff, 1991; Abe *et al.*, 1994; Dennler *et al.*, 1998) as does PMA through the induction of c-Jun/c-Fos. Glucocorticoids, glucose, very low density lipoprotein (VLDL), insulin and angiotensin as well as EGF and PDGF can induce PAI-1 expression (Loskutoff & Samad, 1998; Irigoyen *et al.*, 1999; Stefansson *et al.*, 2003; Hou *et al.*, 2004;).

Despite only 24% amino acid sequence identity between PAI-1 and PAI-2, both share a common tertiary structure consisting of three  $\beta$ -sheets, 8-9  $\alpha$ -helices and a reactive centre loop (RCL; Silverman *et al.*, 2001). The RCL is contained within a strained loop region at the carboxy terminus of the molecule and serves as the protease recognition site for uPA. The inhibition of uPA by PAI-1 occurs in a rapid, 1:1 stoichiometric manner leading to the formation of an SDS-stable complex in which PAI-1 adopts a more thermodynamically stable, relaxed conformation (Silverman *et al.*, 2001).

PAI-1 can exist in two different activity states, active and latent (Lawrence *et al.*, 1989; Loskutoff *et al.*, 1989), which is determined by conformational changes in the RCL (Levin *et al.*, 1987; Mottonen *et al.*, 1992). Active PAI-1 has the RCL exposed on the surface of the molecule which interacts with uPA. This unstable form is secreted by cells but spontaneously converts into the inactive (latent) form by insertion of the RCL (Levin *et al.*, 1987; Hekman *et al.*, 1988).

Human VN is a glycoprotein found in the circulation and in the ECM (Preissner, 1991) where it has a variety of functions dependent upon its ability to interact with humoral and cellular proteins. Physiologically, PAI-1 is present in plasma at 5-10 ng/ml (Declerck *et al.*, 1988a) and both here and in the ECM activated PAI-1 is in complex with VN (Declerck *et al.*, 1988b; Wiman *et al.*, 1988; Mimuro & Loskutoff., 1989). Binding to the somatomedin B (SMB) domain of VN stabalises PAI-1 in its active conformation (Lawrence *et al.*, 1994; Lawrence *et al.*, 1997), increasing its original half life of 90 minutes 2-10 fold (Hekman & Loskutoff, 1985; Declerck *et al.*, 1988b).

#### 1.5.4.1 PAI-1 in disease states

Due to the fundamental importance of regulation of the uPA system it was originally speculated that PAI-1 deficiency would be inconsistent with normal development. Although PAI-1 deficient humans (Fay *et al.*, 1997) and mice (Carmeliet *et al.*, 1993a; Carmeliet *et al.*, 1993b) show a mild hyperfibrinolytic state and bleeding tendency, they are fertile, develop normally and do not display any gross abnormalities. Many studies, often using PAI-1 transgenic animals have related alterations in PAI-1 expression to the pathogenesis of a variety of disorders, including vascular disease, fibrosis, obesity and the metabolic syndrome and cancer.

The expression of PAI-1 is induced during inflammation (Quax *et al.*, 1990) and is increased in tissue injury. It is an acute phase protein important in epithelial cell wound healing (Providence *et al.*, 2000; Providence & Higgins, 2004), epidermal wound repair and in renal injury repair. Mice deficient in PAI-1 have inhibited vascular wound healing after injury (Swaisgood *et al.*, 2000). Increased

PAI-1 expression is associated with fibrosis such as experimentally induced glomerulosclerosis, liver and pulmonary fibrosis and is also increased in human fibrotic diseases such as glomernulonephritis, diabetic nephropathy and bronchoalveolar lavage fluid from respiratory distress patients (Bertozzi *et al.*, 1990; Yamamoto *et al.*, 1996; Loskutoff *et al.*, 2000; Oda et *al.*, 2001). Mice deficient in PAI-1 show an attenuated fibrinogenic response and develop less severe fibrosis after challenge than wild-type or PAI-1 over-expressing mice (Eitzman *et al.*, 1996).

A role for PAI-1 in vascular/cardiovascular disease has been demonstrated in several experimental settings. Spontaneous thrombi develop in the extremities and coronary arteries of PAI-1 over-expressing mice (Eren *et al.*, 2002) and lack of PAI-1 has been shown to protect mice from venous thrombi induced by endotoxin. Individuals with increased PAI-1 expression have an increased risk for thrombosis including myocardial infraction (MI; Thogersen *et al.*, 1998). A circadian rhythm causes an increase concentration of PAI-1 in the morning (Angleton *et al.*, 1989), corresponding to the circadian peak of MI (Muller *et al.*, 1989). Excess PAI-1 is seen in youths surviving MI and also in recurrent MI patients (Hamsten *et al.*, 1985; Hamsten *et al.*, 1987). An increase in PAI-1 is seen in atherosclerotic plaques which may contribute to the pathogenesis of diseases of vessel walls (Schneiderman *et al.*, 1992; Lupu *et al.*, 1993).

There is a positive correlation between PAI-1 expression in human adipose tissue and body mass index with the concentration of PAI-1 often increased five-fold in obese mice and humans compared with lean controls. Adipose tissue has an increased capacity for PAI-1 production and it is found to be increased in adipose

tissue from obese ob/ob mice and in human visceral fat (Samad & Loskutoff, 1996; Samad *et al.*, 1996; Alessi *et al.*, 1997; Bastelica *et al.*, 2002). Weight loss dramatically reduces the concentration of PAI-1 and ob/ob mice crossed into a PAI-1 deficient background have reduced body weight compared to controls, therefore suggesting PAI-1 itself might influence the development of obesity (Samad & Loskutoff, 1997). Following on from this, hyperinsulinaemia patients also have increased PAI-1 expression with improvements of this condition being made by weight loss associated with a decreased PAI-1 concentration (Nordt *et al.*, 1995; Samad *et al.*, 1998; Carmassi *et al.*, 1999; Mavri *et al.*, 1999; Morange *et al.*, 2000; Juhan-Vague *et al.*, 2003).

Expression of PAI-1 is linked to tumour cell biology where it influences angiogenesis and is increased in many human cancers. A highly reproducible tumour model has been used to show that PAI-1 promotes tumor growth and invasion and angiogenesis which were inhibited using small molecule PAI-1 inhibitors (Bajou *et al.*, 1998; Bajou *et al.*, 2001; Brooks *et al.*, 2004). High PAI-1 expression levels correlate with a poor prognosis in many human cancers (discussed further in section 1.5.5.2; Foekens *et al.*, 2000; Dano *et al.*, 2005) and tumors in PAI-1 knock-out mice display lower proliferative and higher apoptotic indices compared to wild-type mice (Gutierrez *et al.*, 2000).

## 1.5.5 Pathophysiological functions of the uPA system

Although one of the principal identified functions of the uPA system is to regulate pericellular proteolysis and target the site of extracellular proteolysis at required sites on the cell surface (Blasi *et al.*, 1987; Blasi & Carmaliet, 2002), recent evidence has revealed that uPA binding to uPAR initiates intracellular

signaling events (discussed in section 1.5.2.1) affecting cell adhesion, migration, and proliferation in a variety of cell types, processes which are also influenced by PAI-1 (Chapman *et al.*, 1997; Ossowski & guirre-Ghiso, 2000; Preissner *et al.*, 2000; Blasi & Carmeliet, 2002; Kjoller, 2002).

## 1.5.5.1 Cell motility

Cell migration is crucial for normal and pathological processes including wound healing and cancer. Regulated changes in the affinity state of adhesion receptors are essential for optimal motility (Lauffenburger & Horwitz, 1996). It is now clear that this process is influenced by PAI-1 and other components of the uPA system (Deng *et al.*, 1996; Stefansson *et al.*, 1996; Kjoller *et al.*, 1997), resulting from direct interaction of system members with integrins and matrix proteins including VN.

As well as PAI-1, uPAR is also able to bind to the SMB domain on VN (Wei *et al.*, 1994; Okumura *et al.*, 2002) via an overlapping binding site to that of PAI-1 (Deng *et al.*, 1996) meaning both proteins compete for VN binding. The affinity of VN for PAI-1 is much higher than for uPAR (Deng *et al.*, 1996), and PAI-1 binding to VN blocks uPAR binding to VN (Deng *et al.*, 2001).

Integrins are a family of cell adhesion receptor heterodimers that interact with many ECM and cell-surface ligands, involving them in many biological and pathological situations (Hynes, 1992). Immediately adjacent to the SMB domain on VN is a distinct RGD site that binds  $\alpha_5\beta_3$ -integrins (Seiffert & Loskutoff, 1991). Binding of uPA to its receptor induces conformational changes in uPAR causing it to bind matrix-engaged integrins, with highest affinity for to the VN

receptors  $\alpha_3\beta_1$  and  $\alpha_5\beta_1$  (Chapman & Wei, 2001; Kugler *et al.*, 2003; Chaurasia *et al.*, 2006) which also induces activation of signalling molecules important for cell migration (Ossowski & guirre-Ghiso, 2000; Kjoller, 2002).

PAI-1 not only prevents cellular adhesion between uPAR and VN-rich ECM matrices (Deng *et al.*, 1996; Stefansson & Lawrence, 1996; Kjoller *et al.*, 1997), but also inhibits adhesive interactions between  $\alpha_5\beta_3$ -integrins and VN by steric hindrance (Okumura *et al.*, 2002; Stefansson *et al.*, 1996). The presence of PAI-1 can therefore inhibit VN-dependent cellular adhesion and migration independently of its inhibitory activity towards uPA and plasminogen activation. However, PAI-1 mediated cell attachment does have an absolute requirement for uPA and uPAR and is schematically outlined in Figure 1.5.

Besides the effect PAI-1 has on the adhesion of cells to VN, the above interaction may have other consequences as VN may function as a shuttle to transfer PAI-1 to uPAR bound uPA. When PAI-1 binds to uPA-uPAR, it undergoes a conformational change which destroys its affinity for VN and promotes its degradation as the PAI-1/uPA/uPAR/integrin complex is internalised from the cell surface (Cubellis *et al.*, 1990; Olson *et al.*, 1992; Czekay *et al.*, 2003) via the process of endocytosis. This is dependent on the  $\alpha_2$ -macroglobulin receptor/ low-density lipoprotein receptor related protein ( $\alpha_2$ M/LRP; Nykjaer *et al.*, 1997; Czekay *et al.*, 2003) or VLDL (Nykjaer *et al.*, 1992; Orth *et al.*, 1992; Conese *et al.*, 1995; Rodenburg *et al.*, 1998; Hussain *et al.*, 1999). Internalised uPAR is recycled to the cell surface (Nyjkaer *et al.*, 1997) and uPA and PAI-1 are targeted for lysosomal degradation (Figure 1.45). The

end result is the rapid disengagement of two adhesion receptors from their association with the ECM, and cell detachment.





Model depicting proposed events in PAI-1-induced cell detachment See section 1.5.6.1 text for details of interactions. (Diagram adapted from Czekay & Loskutoff, 2004). LRP, lipoprotein receptor-related protein.

PAI-1 stimulates cell migration of rat smooth muscle cells and human fibrosarcoma cells which is not dependent on the binding of PAI-1 to either VN or uPA, and is associated with LRP-induced activation of the Jak/Stat signaling pathway (Degryse *et al.*, 2004). A promigratory effect of PAI-1 is also seen in human breast cancer cells and melanoma cells (Stahl & Mueller., 1997; Chazaud *et al.*, 2002; Palmieri *et al.*, 2002).

## 1.5.5.2 Cancer

Many studies have focused attention on the involvement of the uPA system in cancer because it is heavily involved in matrix-proteolysis and uPA-initiated cell
signalling via uPAR activates many signal transduction pathways (as discussed in 1.5.2.1) effecting cell proliferation, migration and adhesion. For these reasons it is no surprise that uPA and uPAR have an adverse effect in cancer progression and are over-expressed in many epithelial tumours (Plesner *et al.*, 1994; Hudson & McReynolds, 1997; Morita *et al.*, 1998; De Petro *et al.*, 1998; Fisher *et al.*, 2000).

However, the finding that patients with increased PAI-1 concentrations also correlate with a poor prognosis for survival in a variety of metastatic human cancers including breast (Fisher *et al.*, 2000; Fox *et al.*, 2001; Duffy, 2002b), colorectal (Sakakibara *et al.*, 2005), lung (Pedersen *et al.*, 1994), renal (Hofmann *et al.*, 1996a; Hofmann *et al.*, 1996b; Ohba *et al.*, 2005), cervical (Kobayashi *et al.*, 1994; Hazelbag *et al.*, 2004), ovarian (Kuhn *et al.*, 1994; Chambers *et al.*, 1998) and endometrial cancer (Tecimer et *al.*, 2001) is suprising and unexpected given that PAI-1 is an inhibitor of extra-cellular proteolysis.

The expression of uPA, uPAR and PAI-1 has been shown to be increased in patients with gastric cancer (Herszenyi *et al.*, 1995; Ito *et al.*, 1996; Scicolone *et al.*, 2006). Using the technique of enzyme-linked immunosorbent assays (ELISA) both uPA and PAI-1 were found to be significantly increased in gastric cancerous tissue compared to tissue from control patients (Nekarda *et al.*, 1994). Immunohistochemical analysis was used to show expression of uPA, uPAR and PAI-1 to be significantly increased in primary gastric cancer patients and it has been suggested that these markers are used as new functional risk factors for gastric cancer prognosis (Heiss *et al.*, 1995; Kaneko *et al.*, 2003). Quantitative real-time PCR has been used to show that PAI-1 gene expression is substantially

increased in lymph node metastasis-positive gastric cancer patients (Sakakibara *et al.*, 2006) and ELISA has shown increased expression of uPA and uPAR in gastric cancer cell lines (MKN45 and KATO-III) after *H. pylori* infection (discussed further in section 3.4; Iwamoto *et al.*, 2005).

It has become evident that PAI-1 may influence the behavior of several cells through activities that are independent of its inhibitory effects on uPA and plasmin generation and the de-adhesive properties of PAI-1 (described in section 1.5.5.1) may begin to account for this observation. Angiogenesis is fundamental to tumour growth and PAI-1 may also influence this process (Stefansson *et al.*, 2003) depending on the concentration of the inhibitor and the specific experimental conditions. For example, one study observed no effect of PAI-1 in angiogenesis (Curino *et al.*, 2002), yet another showed impaired vessel formation in malignant keratinocytes transplanted into the skin of PAI-1<sup>-/-</sup> mice and increased tumour growth in mice that over-express PAI-1 (McMahon *et al.*, 2001). PAI-1 is expressed in endothelial cells of small vessels in human colon adenocarcinomas (Pyke *et al.*, 1991) and in proliferative vessels in high-grade gliomas and metastatic tumours (Kono *et al.*, 1994).

# 1.6 Overall aims and objectives

# 1.6.1 Objective 1: To determine the effect of *H. pylori* on PAI-1 expression in the gastric mucosa

Previous studies showed expression of a member of the uPA system, PAI-2 to be increased in *H. pylori* infection (Varro *et al.*, 2004). It is hypothesised that *H. pylori* infection increases expression of other key members of the uPA system including PAI-1, therefore the initial aims of this study were:

- 1. determine the effects of *H. pylori* infection on expression of PAI-1 within the gastric mucosa compared to uninfected controls;
- study the cellular localisation of PAI-1 in *H. pylori* positive and negative subjects;
- examine the relevance of studying PAI-1 expression using a mouse model of gastric cancer (INS-GAS).

# 1.6.2 Objective 2: To determine the effects of *H. pylori* on primary human gastric glands and the role of the uPA system in epithelial cell proliferation

Since the increased expression of PAI-1 by *H. pylori* was predominantly epithelial, and PAI-1 is increased in gastric cancer, primary human gastric epithelial gland cells were used to:

- 1. determine the effects of *H. pylori*, PAI-1 and uPA on gastric epithelial cell proliferation;
- 2. investigate the mechanism by which *H. pylori* can increase uPA system members using luciferase promotor-reporter constructs;
- study the endogenous expression of uPA system members and expression of their respective luciferase promotor-reporter constructs;

4. determine the effects of epithelial and sub-epithelial derived growth factors associated with *H. pylori* infection on PAI-1-luc expression.

# 1.6.3 Objective 3: To determine the contribution of PAI-1 to gastric mucosal protection

A possible role for PAI-1 in the long-term protection of the gastric mucosa was demonstrated by the fact it restrains *H. pylori* and uPA mediated gastric epithelial cell proliferation. Transgenic mice were developed and used to assess the role of PAI-1 in an acute model of gastric epithelial damage and specifically to:

- determine whether PAI-1 initially protects against, or aggravates gastric epithelial cell damage after challenge;
- investigate an independent role of PAI-1 in tissue damage in mice deficient in uPA signalling.

# Chapter 2

# **Materials and Methods**

## 2.1 Materials and Methods

### 2.1.1 Tissue culture, plasmids and transfection

Tissue culture reagents were purchased from Life Technologies (Paisley, Glasgow, UK) with the exception of HAMS/F12 medium, antibiotic-antimycotic solution and dithiothreitol (DTT) which were purchased from Sigma (Poole, Dorset, UK). Fetal bovine serum (FBS) was obtained from Perbio (Cheshire, UK). Routine tissue culture vessels were purchased from TPP (Trasadingen, Switzerland) and 24-well plates from Corning Incorporated (NY, USA). Promoter-reporter constructs consisting of 4.5kb of the human PAI-1 promoter plus exon 1 coupled to luciferase (PAI-1-luc), approximately 2.9kb of the human uPA promoter plus 120bp exon 1 coupled to luciferase (uPA-luc), and also 2kb plus 207bp exon 1 of the human uPAR promoter coupled to luciferase (uPARluc) all in PXP2 vector were made by Professor Rod Dimaline (The University of Liverpool, Liverpool, UK). The generation of PAI-2-luc has previously been described (Varro et al., 2002b) and contains 2.3kb of the human PAI-2 promoter together with 68bp of exon 1. The Renilla luciferase reporter phRL-SV40 was purchased from Promega (Southampton, UK) as were dual luciferase assay kits and TransFast reagent. CombiMag was obtained from Oz Biosciences (Marseille, France).

#### 2.1.2 Growth factors

Routine chemicals were obtained from BDH Laboratory Supplies (Lutterworth, Leicestershire, UK) or Sigma Chemicals. Heptadecapeptide amidated gastrin (G-17) was obtained from Peninsula (St. Helens, Merseyside, UK). HGF was bought from Genentech (CA, USA) and IGF-II from Calbiochem (Beeston, UK). Ethanol (99.7-100% v/v) was supplied by BDH Laboratory Supplies.

### 2.1.3 Bacteria

Bacteria were kindly routinely cultured by Debbie Sales (Medical Microbiology, The University of Liverpool) in a microaerophilic atmosphere at  $37^{\circ}$ C on fresh chocolatised Columbia blood agar. Wild-type *H. pylori* strain 60190 (American Type Culture Collection (ATCC) no.49503) was used unless otherwise stated. An alternative wild-type strain, 84-183 (ATCC no.53726) and isogenic *H. pylori* mutants that were *cagE<sup>-</sup>*, *cagA<sup>-</sup>* or *vacA<sup>-</sup>* were kind gifts from John Atherton (Nottingham University, Nottingham, UK) and information regarding these strains has previously been published (Argent *et al.*, 2008). Cells were infected at a multiplicity of infection (MOI) of 1:150 as this was the optimal concentration found to give the best stimulation of luciferase constructs described in section 2.1.1. In murine infection experiments *H. felis* was used (ATCC 49179) which had been minimally passaged.

#### 2.1.4 Antibodies and antisera

Antibodies used to identify differentiated cell types within gastric glands were obtained from a variety of sources and used at different concentrations (Table 2.1). Rabbit-anti-H<sup>+</sup>,K<sup>+</sup>-ATPase was obtained from Calbiochem and is raised against the  $\alpha$ -subunit, C-terminal region sequence: GVRCCPGSWWDQGLYY. Rabbit-anti-chromogranin A was purchased from Abcam (Cambridge, UK) and rabbit-anti-pepsinogen was a kind gift from Dr. Mike Samloff (Centre for Ulcer Research, CA, USA). Mouse-anti-human spasmolytic peptide (HSP/TFF-2) was

purchased from NovoCastra (Newcastle-upon-Tyne, UK). Mouse-anti-trefoil factor 1 (TFF-1 or pS2) was obtained from Dako (Cambridgeshire, UK), guineapig-anti-vimentin purchased from Fitzgerald (MA, USA) and rabbit-anti-VMAT-2 was raised in-house and has previously been described (Hussain *et al.*, 1999). Antibodies used to detect PAI-1, uPA and uPAR were purchased from either Santa Cruz Biotechnology (Wiltshire, UK) or American Diagnostica (CT, USA). Details of these antibodies and optimal concentrations are given in Table 2.2. Goat-anti-luciferase antibody was purchased from Rockland Immunochemicals (Gilbertsville, PA, USA) and used at a concentration of 1:200. BrdU was detected using fluorescin isothiocynate-conjugated fast immune anti-BrdU/DNase obtained from Becton Dickinson (Oxford, UK) and was used at an optimal concentration of 1:50. FITC and texas-red (TRed)-conjugated secondary antibodies were purchased from Jackson ImunoResearch Laboratories (Soham, UK) and were all used at an optimal concentration of 1:400.

| Cell type                   | Primary antibody                                      | Titre | Secondary<br>antibody | Titre |
|-----------------------------|---|-------|-----------------------|-------|
| Mucous neck                 | Anti-TFF2   | 1:50  | Anti-mouse IgM        | 1:400 |
| Surface mucous              | Anti-TFF1   | 1:100 | Anti-mouse IgG        | 1:400 |
| ECL                         | Anti-Chromogranin A                                   | 1:50  | Anti-Rabbit IgG       | 1:400 |
| ECL                         | Anti-VMAT-2   | 1:200 | Anti-Rabbit IgG       | 1:400 |
| Chief                       | Anti-pepsinogen                                       | 1:200 | Anti-Rabbit IgG       | 1:400 |
| Parietal                    | Parietal Anti- H <sup>+</sup> ,K <sup>+</sup> -ATPase |       | Anti-Rabbit IgG       | 1:400 |
| Myofibroblast Anti-Vimentin |   | 1:400 | Anti-G.pig IgG        | 1:400 |

Table 2.1 Antibodies used to identify main cell types within human primary gastric glands.

| Antibody | Origin                   | Species | Titre | Specificity  |
|----------|--------------------------|---------|-------|--|
| PAI-1    | Santa Cruz Biotechnology | Rabbit  | 1:200 | Raised against a recombinant protein corresponding to amino acids 24-158 of mature human PAI-1.  |
| PAI-1    | American Diagnostica     | Mouse   | 1:100 | Raised against purified active PAI-1 secreted by human<br>melanoma cell line, epitope specificity amino acid 110-145.                        |
| uPA      | Santa Cruz Biotechnology | Rabbit  | 1:200 | Raised against amino acids 136-275 of uPA.   |
| uPA      | American Diagnostica     | Mouse   | 1:200 | Directed against an epitope on the B-chain near the catalytic site.<br>Recognises the single pro-enzyme, HMW, LMW and receptor<br>bound uPA. |
| uPAR     | Santa Cruz Biotechnology | Goat    | 1:100 | Raised against a peptide mapping near the C terminus of uPAR.  |
| uPAR     | American Diagnostica     | Mouse   | 1:100 | Recognises domains 2 & 3 of the receptor. Binds with high affinity to both uPAR and uPAR/uPA complexes.                                      |

Table 2.2 Primary uPA system antibodies and dilutions used in immunocytochemistry to determine endogenous protein expression.

# 2.2 Animals

C57BL/6 were purchased from Charles River (MA, USA) at a specified age and sex as required. INS-GAS mice on a FVB/N background have previously been described (Wang *et al.*, 1996). PAI-1 KO and uPAR KO animals were initially purchased from JAX<sup>®</sup> MICE (Maine, USA) and are on a C57BL/6 background. Mice that are homozygous for targeted PAI-1 or uPAR mutation are viable, normal in size and do not display any gross physical or behavioural abnormalities. PAI-1 and uPA over-expressing mice are also on a C57BL/6 background and the generation of these animals is discussed in Chapter 5 of this thesis. All mice were housed in an appropriately controlled environment with a strict light to dark light cycle (12:12 hr) in polycarbonate-bottomed cages in which there was environment enrichment. Animals were not specific pathogen free and were fed on a commercial pellet diet and water was provided *ad libitum*. In experiments where fasting took place this was for no longer than 17 hr and all animals still had full access to water.

*Helicobacter felis* infections took place when animals were between 6-8 weeks old. They were oral dosed three times over a period of 1-2 weeks with 0.5ml (for a 25g mouse) of an *H. felis* suspension in TBS broth. The optical density of the suspension was above 1.5 and *H. felis* were not used if they had been passaged more than five times. The *H. felis* status of animals was determined by antral urease test (Prontodry, Medical Instruments Corporation, Solothrum, Switzerland) or by antral histology.

Animals were humanely killed by increasing  $CO_2$  concentrations followed by cervical dislocation according to schedule 1 methods as directed by the Home

Office. An abdominal incision was made down the linea alba and the stomach severed at both the oesophageal and pyloric sphincter, rapidly removed, and cut open along the greater curvature. Stomach contents were removed by careful washing in phosphate buffered saline (PBS) to allow the analysis of the gastric mucosa. For murine gastric gland isolation the stomach was ligated at both the oesophageal and pylori sphincters, removed and washed in Hanks Balanced Salts Solution (HBSS). The majority of the non-secretory epithelium was removed and the pyloric sphincter directed towards the newly made fundic opening, allowing complete inversion of the stomach. This was then ligated, washed in ice cold HBSS, injected with 0.5mg/ml collagenase A (Roche Molecular Biochemicals, East Sussex, UK) and placed on ice before culture (section 2.4.3).

# 2.3 Patients and ethical approval

Endoscopic pinch biopsies (maximum of 6) of the gastric corpus or antrum were oesophagogastroduodenoscopy from patients undergoing obtained by investigation for dyspepsia at the Royal Liverpool Gastroenterology Unit at Dr D.M Pritchard's clinic. Biopsies were placed in ice cold HBSS for primary culture (section 2.4.2). Blood samples were taken from patients to assay plasma gastrin and also for *H. pylori* serology. Both antral and corpus biopsies were fixed in formalin for histology at the Royal Liverpool Gastroenterology Unit and a rapid urease test carried out on antral biopsies of each patient (Prontodry, Medical Instruments Corporation). All patients gave informed consent and were included in the study if they were aged between 18 and 75 years of age, had no previous gastrointestinal (GI) surgery or GI malignancy, were not pregnant and did not have liver disease. Patients taking histamine  $(H_2)$  blockers and proton

pump inhibitors were included only if this medication had been stopped 7 days prior to endoscopy. All patients gave informed consent and local ethical committee approval was obtained from the Ethics Committees of Royal and Broadgreen Liverpool Hospitals NHS Trust. In addition, an experiment was performed on tissue obtained from patients with distal gastric cancer on a background of chronic *H. pylori* infection which was carried out in collaboration with Dr Suhail Ahmed, Aintree Hospital, Liverpool.

# 2.4 Tissue culture

## 2.4.1 Culture of AGS2 cells

AGS cells were obtained from the American Type Culture Collection (Manassas, VA) and have been stably transfected with the gastrin CCK2 receptor (AGS2) driven by the EF1- $\alpha$  promoter using TransFast reagent (Promega; Watson *et al.*, 2001). It has been shown that the receptor is functional (Varro *et al.*, 2002a). This cell line was cultured in Ham's F-12 medium supplemented with 10% FBS and 1% penicillin/ streptomycin in a humidified atmosphere of 5% CO<sub>2</sub> / 95% O<sub>2</sub>. Medium was replaced every two days. AGS2 cells were maintained under selection by supplementing the medium with puromycin (2µg/ml).

#### 2.4.2 Human primary gastric gland isolation and culture

Six endoscopic human pinch biopsies of gastric corpus were obtained with 2.4mm tip diameter, single-use, biopsy forceps (Diagmed healthcare, Thirsk, UK) from patients referred with dyspepsia. In all cases histology indicated superficial gastritis but not gastric atrophy or intestinal metaplasia and patients were excluded if there was a previous history of neoplastic disease, current

peptic ulcer or elevated plasma gastrin concentrations (>30pM). Results shown are from gastric glands isolated from *H. pylori* negative patients (determined as in section 2.3) unless otherwise stated.

Biopsies were placed in ice cold HBSS before being sliced into 2mm<sup>2</sup> segments using a sharp razor blade and washed in 3 changes of 37°C HBSS. Biopsies were incubated in 5ml 1mM dithiothreitol (DTT; Sigma) for 15 min and were continuously gassed with 5%  $CO_2$  / 95%  $O_2$  at 37°C whilst shaking at 100 rpm. Tissue was washed 3 times in HBSS (37°C), incubated in 5ml 0.5mg/ml collagenase A (Roche Molecular Biochemicals) for 30 min before being washed 3 times with HBSS (37°C). Tissue was subjected to a second 5ml collagenase A digestion (0.5mg/ml 37°C), triturated for 45 sec using a wide mouthed pipette and transferred to a clean universal. Larger fragments were allowed to settle under gravity for 45 sec, supernatant was removed, transferred to another clean universal, shaken vigorously and left to sediment on ice for 45 min. The supernatant was carefully removed, discarded and the remaining mixture containing isolated glands was equally divided between six wells of 24-well cell culture plate. Glands were routinely cultured in Dulbecco's Modified Eagles Medium (DMEM) supplemented with 10% FBS, 1% antibiotic-antimycotic solution (Sigma) and 1% L-glutamine at 37°C in a humidified atmosphere of 5% CO<sub>2</sub> / 95% O<sub>2</sub>. Glands were allowed to adhere overnight and medium was changed every 24 hr.

# 2.4.3 Murine gastric gland isolation and culture

Glands were isolated using a modified method that was first described in 1976 (Berglindh & Obrink, 1976). A very similar protocol to that used for the isolation

of human primary gastric glands (section 2.4.2) was used to obtain gastric glands from the inverted mouse stomachs (obtained as described in section 2.2). The protocols only differ when it comes to collagenase A digestion as mouse stomachs are incubated in 5ml, 0.32mg/ml collagenase A (Roche Molecular Biochemicals) and watched during the first 30 min incubation for rupturing of the inverted stomach. The development of an opalescent medium indicates that the stomach has been sufficiently digested to yield isolated glands. If this did not occur then stomachs were washed (3 times HBSS) and incubated for a further 30 min in 0.32mg/ml collagenase A, checking every 10 min for ruptures thus avoiding over-digestion. Once sufficiently digested, the tissue was triturated for 45 sec using a wide mouthed plastic pipette and then transferred to a clean universal. The remainder of the protocol is identical to that of human gland isolation (section 2.4.2). Isolated gastric glands were routinely cultured in DMEM supplemented with 10% FBS, 1% L-glutamine and 1% antibioticantimycotic solution at 37°C in a humidified atmosphere of 5% CO<sub>2</sub> / 95% O<sub>2</sub>. Glands were allowed to adhere overnight and medium was changed every 24 hr.

# 2.5 Transient transfection and luciferase assays

#### 2.5.1 AGS2 cell transfection

200,000 cells/well were plated in 6 well culture plates and allowed to adhere 24 hr prior to transfection. Cells were transfected using TransFast (Promega) with 3 µg plasmid DNA/well and 5ng/well of a constitutively active *Renilla* luciferase reporter construct, phRL-SV40 in 840µl SFM (Hams-F12) for 1 hr at 37°C. Full medium (FM; 3ml) was added and cells incubated for 20 hr. Cells were washed twice in SFM and then incubated with G17 as indicated. Luciferase activity was

measured by Dual-luciferase Reporter Assay System (Promega) using a Lumat LB 9057 luminometer according to manufactures instructions (Berthold, Redbourne, Herts, UK). Results are displayed as fold increase over unstimulated control where 1.0 signifies no change in luciferase activity.

## 2.5.2 Human primary gastric gland transfection and luciferase assay

Human gastric glands (prepared as in section 2.4.2) were transfected 24 hr after plating in 12 well plates using CombiMag (Oz Bioscience) and TransFast (Promega) according to manufacturers instructions.  $3\mu g$  of plasmid DNA/well (PAI-1-, uPAR- or uPA-luc) together with 5ng/well of a constitutively active *Renilla* luciferase reporter construct, phRL-SV40,  $18\mu$ l Combimag and  $18\mu$ l TransFast was added to  $410\mu$ l SFM, vortexed and incubated at room temperature for 15 min before being added to cells for 20 min under an applied magnetic field (in the form of a magnetic plate). Full medium (2.5ml) was added and cells incubated  $37^{\circ}$ C overnight. Cells were washed twice in PBS and SFM added before the addition of *H. pylori* at a specified multiplicity of infection (MOI) or any other compounds as indicated. Luciferase activity was measured by Dualluciferase Reporter Assay System (Promega) using a Lumat LB 9057 luminometer according to manufacturers instructions (Berthold). Results are displayed as fold increase over unstimulated control where 1.0 signifies no change in luciferase activity.

# 2.6 BrdU incorporation and anti-sense oligonucleotide treatment

Cells within human primary gastric glands in S phase of the cell cycle were labelled with 5'-bromo-2'-deoxyuridine (BrdU; Sigma) as outlined in Figure 2.1.

Human primary glands were prepared as in section 2.4.2, plated on glass coverslips and the following morning anti-sense oligonucleotides (ASO) added in FM. This was either control scrambled oligonucleotides (CSO) or PAI-1 antisense oligonucleotides (ASO; all at a final concentration of 2µM and details are given below; Biognostik, Germany). 8 hr later, cells were serum starved for 24 hr 1ml SFM and stimulated with either H. pylori MOI 1:150 or uPA (250 ng/ml; Chemicon), BrdU was added to cells at a final concentration of 3ng/ml, and a mouse monoclonal PAI-1 neutralising antibody (American Diagnostica) at a concentration of 10µg/ml. Throughout the experiment both CSO and ASO were replaced in the media when it was changed. Cells were fixed 18 hr later in 4% paraformaldehyde (PFA; Agar Scientific, Essex. UK) and immunocytochemistry carried out using a mouse-anti-BrdU antibody at a concentration of 1:50 (Becton Dickinson) and a guinea-pig-anti-vimentin antibody at a concentration of 1:400 (Fitzgerald) as described in section 2.7. An anti-vimentin antibody was used in order to exclude BrdU, vimentin positive myofibroblasts and 10 glands per patient per treatment were scored. BrdU labelled myofibroblasts were subtracted from the total number of BrdU labelled cells per gland to give percentage BrdU labelled epithelial cells.

> Human gastric glands prepared 18 hr J 3ml FM +control ASO / +PAI-1 ASO 8 hr J 1ml SFM +control ASO / +PAI-1 ASO 24 hr J 1ml SFM +control ASO / + PAI-1 ASO 24 hr J 1ml SFM +control ASO / + PAI-1 ASO + H. pylori or uPA +BrDU (+ PAI-1neutralising antibody) 18 hr J Fix cells 4% paraformaldehyde for immunocytochemistry

Figure 2.1 Method for BrdU incorporation into human gastric glands

Most anti-sense experiments produce sufficient inhibition grades without using transfection reagents and anti-sense oligonucleotides can be taken up by the cells simply by adding them to the cell culture media. The sequence of human PAI-1 anti-sense oligonucleotide (Biognostik) is based on Accession number X04429 and shows no relevant cross homologies with genes found in the GenBank database. Sequence (5'-3'): GAG AGA GGC ACC TCT TTT. Control scrambled oligonucleotides (CSO) were also supplied and are designed to avoid toxic motifs and to be nonhomologues to any known sequence, therefore little effect should be caused by controls. However, the sequence of CSO was not stated by the supplier at time of purchase.

# 2.7 Immunocytochemistry of adherent gastric glands

Biopsies from one individual patient were prepared (as in section 2.4.2) and cultured on glass coverslips. Isolated glands were treated with various compounds/ bacteria as indicated, media removed, washed 3 times in PBS and fixed in 1ml 4% w/v PFA for 30 min at room temperature. Fixed glands were washed 3 times in PBS and permeabilised for 30 min using PBS containing 0.1% triton X-100. Permeabilised glands were washed again 3 times in PBS and incubated with 5% w/v bovine serum albumin (BSA; Jackson ImmunoResearch Laboratories) for 30 min. Glands were washed 2 times in PBS and non-specific binding of the secondary antibody was blocked by incubating glands in 10% donkey serum (Jackson ImmunoResearch Laboratories) for 1 hr at room temperature. Serum was removed, glands washed 2 times in PBS and primary antibody (diluted in PBS) was added to glands overnight at 4°C in a humidified atmosphere. Details of all primary antibodies used are given in section 2.1.4.

The following day a series of 5 min PBS/sodium chloride (NaCl) washes were carried out: 0.14M NaCl, 0.5M NaCl, 0.14M NaCl. The appropriate secondary antibody (diluted in 10mM Hepes) was added to glands for 1 hr in the dark. Secondary antibodies were raised in donkey and were either TexasRed (TRed) or fluorescein isothiocyanate (FITC) conjugated (Jackson ImmunoResearch Laboratories) and used at a titre of 1:400. Secondary antibody was removed and glands washed (3 times 10 min in PBS) and mounted with Vectashield containing DAPI (Vector Laboratories, Peterborough, UK). Slides were examined using a Zeiss Axioplan-2 microscope (Zeiss Vision, Welwyn Garden City, UK) and images captured using JVC-3 charge-coupled device camera using 40X magnification (air immersion) and KS300 software combined with deconvolution software (Imaging Associates, Bicester, Oxfordshire, UK).

# 2.8 Immunohistochemistry of paraffin embedded tissue

Immunohistochemistry was carried out on formalin-fixed, paraffin-embedded tissue sections by Andy Dobson and analysed and evaluated by Dr Fiona Campbell (Consultant Pathologist, Department of Pathology, Royal Liverpool University Hospital). Briefly, sections were cut at  $4\mu$ m onto silane-coated glass microscope slides and dried overnight in an oven at  $45-50^{\circ}$ C. Sections were dewaxed through xylene and endogenous peroxidase activity blocked in methanol/H<sub>2</sub>O. Immunohistochemical staining was performed on an Autostainer (DakoCytomation, Cambridgeshire, UK). Sections were incubated with diluted primary antibody for 60 min. PAI-1 was detected using a rabbit-anti-PAI-1 antibody at a concentration of 1:40 (American Diagnostica). In addition, primary

antibody was omitted and used as a negative control. Detection was performed using the ChemMate<sup>TM</sup> EnVision<sup>TM</sup> HRP detection system (DakoCytomation) according to manufacturers recommendations.

# 2.9 Western blot analysis

Protein extract was prepared in lysis buffer (20mM Tris pH 7.8, 150mM NaCl, 2mM EDTA, 0.5% Nonidet P-40, 10mM NaF, 15mg/ml benzamide, 8.8mg/ml sodium orthovanadate, 0.5mM DTT, 10mg/ml phenylmethylsulfonyl fluoride) containing 10µl/ml protease inhibitor cocktail set III and 10µl/ml phosphatase inhibitor cocktail set III (Calbiochem). Equal amounts (40 µg) of protein were electrophoresed and resolved on an 8% w/v SDS-polyacrylamide gel and electrotransferred onto a nitrocellulose membrane (Amersham Pharmacia Biotech, Buckinghamshire, UK). After transfer, membranes were washed (3 times 10 min) in tris buffered saline (TBS) containing 0.1% Tween (TBST) then incubated in blocking buffer (TBST containing 5% Marvel) for 1 hr at room temperature. Membranes were incubated with primary antibody diluted in blocking buffer overnight at 4°C whilst shaking. The primary antibody used was mouse-anti-PAI-1 (1:100; American Diagnostica). Following this, membranes were washed in TBST (3 times 10 min) and incubated in blocking buffer containing horseradish peroxidase (HRP)-anti-mouse secondary antibody (1:5000; Sigma) for 1 hr at room temperature. Membranes were then washed (3 times 10 min) in TBST and protein was identified by incubating membranes in the chemiluminescent substrate SuperSignal<sup>®</sup> West Pico (Pierce, Illinois, USA) for 5 min. The signal was detected by exposure to HyperFilm (Amersham Biotech, Little Chalfont, UK), which was developed (Kodak D-19 developer)

and fixed (Kodak rapid fix). Samples were re-probed with goat-anti-β-actin (1:5000; Santa Cruz Biotechnology) to determine equal protein loading. This was identified using a HRP-anti-goat (1:5,000) secondary antibody.

# 2.10 Bacterial re-transformation with plasmid DNA

50 μl DH5α competent cells (Invitrogen, Paisley, UK) were incubated with 0.5μl plasmid DNA on ice for 30 min. Cells were heat shocked by subjection to  $42^{\circ}$ C, 50 sec and returned to ice for a further 2 min. 950μl SOC media (2% tryptone, 0.5% yeast extract, 10mM NaCl, 2.5mM KCL, autoclave, pH 7 then add 10mM MgCl<sub>2</sub>.6H<sub>2</sub>O, 20mM glcouse) was added and bacteria incubated at 37°C, 1.5 hr whilst shaking. Bacteria were streaked onto LB-agar plates (1.2% Agar added to Luria Bertani (LB) medium; 1% tryptone, 0.5% yeast extract, 1%NaCl, pH7 and autoclaved) containing ampicillin (100µg/ml).

# 2.11 Preparation of plasmid DNA

Single colonies were selected and cultured in LB medium containing 50µg/ml ampicillin at 37°C for 8 hr whilst shaking at 225rpm. The culture was transferred to 100ml LB medium and incubated overnight at 37°C whilst shaking at 225rpm. Plasmid DNA was recovered using QIAGEN Plasmid Maxi Kits (QIAGEN Ltd, West Sussex, UK) according to manufacturers protocol. Briefly, bacterial cells were pelleted by centrifugation (6000g, 15 min, 4°C), supernantant discarded and cells resuspended and lysed followed by neutralisation and precipitation. The precipitated material was centrifuged at 20000g, 30 min (4°C), the pellet discarded and the supernantant centrifuged again at 20000g, 30 min (4°C). The supernantant was applied to a resin and bound DNA was retained on the column.

This was eluted into a fresh tube and DNA was precipitated by adding 0.7 volumes room temperature isopropanol and pelleted by centrifugation at 15000g, 30 min, 4°C. The DNA pellet was washed with room temperature 70% ethanol and centrifuged at 15000g for 10 min. The supernantant was discarded and the pellet air-dried and redissolved in a suitable volume of Tris- EDTA (TE) buffer (Severn Biotech, Worcestershire, UK). Plasmid concentration was calculated using a Biowave II Optical Density reader (WPA, Cambridge, UK) by reading the absorbance at 260nm.

### 2.12 Real-time PCR

### 2.12.1 RNA extraction and cDNA preparation

Gastric corpus biopsies from male INS-GAS mice infected with *H. felis* as previously described (Wang *et al.*, 2000 and section 2.2) were obtained at 3, 6 and 9 months post-infection. Human antrum and corpus endoscopic pinch biopsies were taken from *H. pylori* infected and non-infected patients. Biopsy samples were stored in RNA-Later<sup>®</sup> (Ambicon, CA, USA) and tissue was later homogenised at 4°C using a Polytron homogeniser (Kinematica, Hessie, UK) in the presence of TRI<sup>®</sup> reagent (Sigma). A 20% volume of chloroform was added, mixed by inversion and centrifuged for 15 min at 14000rpm (4°C). Supernatant was transferred to a fresh tube, 50% volume of isopropanol added, mixed by inversion and left at room temperature for 15 min before centrifugation for 10 min at 14000rpm at 4°C. The supernantant was discarded and the pellet washed with 70% ethanol and air-dried.

RNA was resuspended in nuclease free water (MP Biomedicals, Cambridge, UK) or alternatively stored in 1ml 99.8% ethanol / 20µl 3M sodium acetate at -

80°C. RNA was recovered via centrifugation at 4°C for 15 min at 14000rpm, washed in 70% ethanol followed by ice-cold absolute ethanol. Supernantant was discarded and the pellet air-dried and resuspended in 50µl nuclease free water. RNA concentration was calculated using a Biowave II Optical Density reader (WPA) by reading the absorbance at 260nm. DNase treatment was performed on 30µg RNA by addition of RQ1-DNase (1µl per µg RNA; Promega), 10x buffer and DEPC-treated water were added according to manufacturers protocol. This was incubated at 37°C for 30 min and the volume made up to 200µl by the addition of DEPC-treated water. An equal volume of phenol/chloroform was added, the sample was mixed and centrifuged at 4°C for 5 min at 14000rpm. The supernantant was transferred to a fresh tube, 20µl of 3M sodium acetate added in addition to 1ml ice-cold 99.8% ethanol and samples were stored at -80°C. RNA was retrieved by 15 min centrifugation at 14000rpm (4°C), the pellet was washed in 70% ethanol followed by ice-cold absolute ethanol, air-dried and resuspended in 50µl nuclease-free water. RNA concentration was measured and 5µg reverse transcribed by the addition of 0.5µg of random primers (Promega) and incubation at 70°C for 5 min before slow cooling to below 40°C. Reverse transcription was carried out using AMV reverse transcriptase according to manufacturers instructions (Promega). The reaction mix was incubated at room temperature for 10 min followed by a 60 min incubation step at 42°C. Heating for 5 min at 95°C inactivated the enzyme and cDNA was stored at -20°C.

### 2.12.2 Quantitative real-time PCR

Real-time PCR was used to measure PAI-1 mRNA abundance using cDNA samples prepared from human and mouse gastric biopsies (prepared as in section

2.12.1) and standards (human or mouse cDNA). cDNA was added to a master mix containing QPCR Master Mix Plus (Eurogentec, Southampton, UK), forward and reverse primers, probes and ddH<sub>2</sub>O (all volumes as in Table 2.3). The sequences of human and mouse PAI-1 forward and reverse primers as well as probe sequences are shown in Table 2.4.

|                       | h-PAI-1 | m-PAI-1 | 18S        |  |
|-----------------------|---------|---------|------------|--|
| cDNA                  | 4µl     | 4µl     | 4µl        |  |
| Forward Primer (10µM) | 3µl     | 3µl     | <i>C</i> 1 |  |
| Reverse Primer (10µM) | 3µl     | 3µl     | ομι        |  |
| Probe (5nM)           | 2.5µl   | 3.5µl   | 2µl        |  |
| ddH <sub>2</sub> O    | 37.5µl  | 36.5µl  | 38µl       |  |
| Master Mix Plus       | 50µl    | 50µl    | 50µl       |  |

Table 2.3 Primer and probe volumes for real-time PCR of human and mouse cDNA samples.

Each cDNA sample was analysed in triplicate and 25µl of the above mix pipetted into each well of 96-well Q-PCR plate (Applied Biosystems, Warrington, UK) which was then centrifuged at 2000G for 2 min. An ABI Prism<sup>TM</sup>7700 Sequence Detector machine and Sequence Detector software (Applied Biosystems) was used to measure mRNA abundance. Taqman Technology was employed with the amplification cycles as follows: 50°C for 2 min, 95°C 10 min, 40 cycles of 95°C (15 sec) and 60°C (1 min). For each sample, the amount of 18S rRNA was assayed in a similar way using an 18S rRNA control kit (Eurogentec) and relative mRNA abundance was determined from the ratio mRNA/18SrRNA.

| Primer/Probe   | 5'-3'                        |
|----------------|------------------------------|
| hPAI-1 forward | GGCTGACTTCACGAGTCTTTCAG      |
| hPAI-1 reverse | CGTTCACCTCGATCTTCACTTTC      |
| hPAI-1 probe   | AAGAGCCTCTCCACGTCGCGCA       |
| mPAI-1 forward | CCCTGGCCGACTTCACAA           |
| mPAI-1 reverse | ACCTCGATCCTGACCTTTTGC        |
| mPAI-1 probe   | CAAGAGCAGCTCTCTGTAGCACAGGCAC |

Table 2.4 Human and mouse PAI-1 specific forward and reverse primer and probe sequences for real-time PCR.

# 2.13 Haemorrhagic lesion model

Male or female C57BL/6, PAI-1-KO, uPAR-KO, PAI-1/ uPA over-expressing mice (10-12 weeks) were fasted overnight (17 hr) before being gavarged with 100µl of either 50% or 99.8% (absolute) ethanol. Animals were sacrificed after 1 hr and stomachs removed by cutting at both the oesophageal and pyloric sphincter. The stomach was cut along the greater curvature and briefly washed in PBS. Gastric tissue was opened up and stretched out mucosa side up and pinned out on a piece of cork. The number and length of haemorrhagic lesions was recorded using callipers and a ruler. Stomachs were also graded as possessing either 'weak', 'moderate' or 'severe' lesions. All animal procedures had full ethical approval and were carried out according to Home Office regulations.

# 2.14 Statistical analysis

Results are expressed as mean  $\pm$  standard error about the mean (SEM), unless otherwise stated. One way analysis of variance (ANOVA) or Student's *t*-test was carried out to determine statistical significance of results and significance indicated (\*) at p<0.05. *n* = individual patients/ mouse or individual experiments.

# Chapter 3

# **Stimulation of PAI-1 expression**

# in the gastric epithelium and its

# significance for epithelial cell

function

# **3.1 Introduction**

Approximately 15% of all cancers develop on a background of chronic inflammation linked to infection. A classic example of this is the development of gastric cancer in patients infected with the gram negative, oncogenic bacterium *Helicobacter pylori* (*H. pylori*). Infection with *H. pylori* is associated with inflammation of the gastric mucosa, which in a small percentage of people triggers the progression to gastric cancer (reviewed in Bodger & Crabtree, 1998; Peek & Crabtree, 2006). Stomach cancer is the second most common cause of death due to malignancy worldwide and proceeds via a progression of distinct histological steps (discussed in section 1.3.2; Correa, 1982; Hohenberger & Gretschel, 2003; Correa, 2004).

The response of gastric epithelial cells to *H. pylori* infection that may influence preneoplastic changes, cancer initiation and progression remains largely unknown, but may include altered expression of growth factors, cytokines and extracellular proteases, as well as involving both host and bacterial factors (El-Omar *et al.*, 2000; Machado *et al.*, 2003; Peek & Crabtree, 2006). Damage to epithelial cells such as that caused by bacterial infection can elicit a strong host response, causing the release of paracrine factors, ECM proteins, proteases and protease inhibitors. This can trigger a chain of events to occur, leading to remodelling of the epithelium and the development of a hyperproliferative state, ultimately contributing to malignant transformations (Lechago & Correa, 1993).

Preliminary studies showed that expression of a member of the urokinase plasminogen activator (uPA) system, plasminogen activator inhibitor-2 (PAI-2), is increased in *H. pylori* infection (Varro *et al.*, 2004). Increased expression of uPA system members including uPA itself, its receptor (uPAR), and its main

extracellular inhibitor (PAI-1) have been shown to correlate with a poor prognosis in gastric cancer patients (discussed in section 1.5.5.2; Nekarda *et al.*, 1994; Heiss *et al.*, 1995; Ito *et al.*, 1996; Plebani *et al.*, 1997; Kaneko *et al.*, 2003; Scicolone *et al.*, 2006). However, the direct effect of *H. pylori* on PAI-1 expression, and the involvement of the uPA system with the preneoplastic stages involved in gastric cancer development have been poorly studied. It is hypothesised that *H. pylori* infection increases expression of other key members of the uPA system including PAI-1, which can influence gastric epithelium remodelling during infection.

Plasminogen activator inhibitor-1 is an important physiological regulator of fibrinolysis and ECM homeostasis by inhibiting the activities of the serine protease uPA, which acts to control pericellular proteolysis through the regulation of the conversion of plasminogen into the proteolytic enzyme plasmin (Irigoyen *et al.*, 1999). However, although one of the physiological roles of PAI-1 is to protect the ECM from degradation by proteases it is also associated with many pathological conditions making it a very interesting protein to study (discussed in section 1.5.4.1).

Increased PAI-1 expression is known to contribute to tissue fibrosis in several organs such as kidney (Ingelfinger, 2003), lung (Kaikita *et al.*, 2001) and heart (Takeshita *et al.*, 2004). Fibrosis within the stomach is often regarded as a premalignant state associated with *H. pylori* infection (Matsukura *et al.*, 1980). An increase in PAI-1 expression is not only associated with a poor prognosis in gastric cancer but in multiple human cancers (discussed in section 1.5.5.2). Tissue remodelling involves epithelial cell and ECM interactions which is heavily dependent upon the regulated release of ECM-degrading proteases.

Dysfunction in these interactions is thought, at least in part, to promote tumourigenesis. It is therefore essential to consider the fine balance between the protease uPA and its inhibitor PAI-1 as a disruption to this could have pathological consequences. The mechanism by which *H. pylori* promotes cancer development involves several different host responses (Naumann & Crabtree, 2004), and it is therefore interesting to investigate if expression of uPA system members is influenced by *H. pylori* infection and if this has any role to play in the multi-step process of remodelling of the gastric mucosa.

# 3.1.1 Aims

The aims of this chapter are to study the expression of PAI-1 in the gastric epithelium and determine the role it has on epithelial cell function. The specific objectives were to:

- determine the effects of *Helicobacter* infection on expression of PAI-1 mRNA- abundance within the stomach in both humans and mice;
- study the difference in cellular localisation of PAI-1 protein expression between *H. pylori* positive and negative subjects;
- investigate mRNA and protein expression of PAI-1 in gastric cancer patients;
- study endogenous PAI-1, uPA and uPAR expression in a human primary epithelial gastric gland cell culture;
- 5. examine the effects of *H. pylori*, PAI-1 and uPA on gastric epithelial cell proliferation.

# **3.2 Materials and Methods**

# 3.2.1 Patients

Endoscopic pinch biopsies of the gastric corpus and antrum were taken from human subjects as described in section 2.3. Samples from 10 *H. pylori* positive patients were compared to 15 *H. pylori* negative, control patients. Blood samples were taken to assay plasma gastrin (patients with levels greater than 30pM were excluded from this study) and also for *H. pylori* serology. In addition, a rapid urease test was carried out on antral biopsies of each patient. All patients gave informed consent and the study was approved by the Ethics Committee of Royal Liverpool and Broadgreen University Hospital NHS Trust. In addition, samples from 5 patients presenting with intestinal type, distal gastric cancer on a background of chronic *H. pylori* infection were also analysed. These samples were obtained from Dr Suhail Ahmed, Aintree Hospital, Liverpool, NHS trust.

## 3.2.2 Animals

Male INS-GAS mice on an FVB/N background were infected with *H. felis* as previously described (Wang *et al.*, 2000 and section 2.2) and infection status was determined by antral rapid urease test. PAI-1 mRNA abundance was measured from gastric corpus samples 3 and 6 months post-infection and compared to uninfected male INS-GAS mice. In a separate experiment, PAI-1 mRNA abundance was measured in uninfected INS-GAS mice and compared to uninfected FVB/N mice.

#### 3.2.3 PAI-1 protein expression in H. pylori infection and gastric cancer

PAI-1 protein expression and localisation was determined via immunohistochemistry using paraffin embedded stomach sections from a *H*.

pylori negative and positive patient as well as on gastric cancer tissue (as described in section 2.8). This technique was performed by the Pathology department at The Royal Liverpool University Hospital using a rabbit-anti-PAI-1 antibody at a concentration of 1:400 (American Diagnostica) and a mouse-anti- $\alpha$ -SMA antibody at 1:400 dilution (American Diagnostica). Negative controls were carried out with the omission of primary antibodies.

#### 3.2.4 Real-time PCR

Real-time PCR was performed on RNA extracted from biopsy samples taken from the corpus and antrum region of the stomach from either *H. pylori* negative or positive human subjects, *H. felis* positive or negative INS-GAS mice and *H. felis* negative FVB/N mice (as described in section 2.12.1). Real-time PCR was carried out as described in section 2.12.2 using either human or mouse PAI-1 primer and probe sets (volumes used and sequences given in Table 2.3 and Table 2.4 respectively). Relative mRNA abundance was determined from the ratio mRNA/18SrRNA.

## 3.2.5 Cellular localisation of endogenous PAI-1, uPA and uPAR

Human corpus gastric glands isolated from *H. pylori* negative patients as described in section 2.4.2 were cultured on glass cover slips in FM 48 hr and were stimulated with wild-type *H. pylori* M.O.I. 1:150 or left unstimulated in SFM 18 hr. The expression of uPA system members within gastric gland cells was assessed using dual immunofluorescence (as described in section 2.7). Endogenous PAI-1, uPA or uPAR protein was detected using antibodies listed in Table 2.2. Appropriate FITC conjugated secondary antibodies at a dilution of 1:400 (Jackson ImmunoResearch Labs) were used. Antibodies raised against cell

markers for the main terminally differentiated cell types of the gastric corpus were also used: parietal, chief, surface mucous, mucous neck and enterochromaffin-like cells with TRed labelled secondary antibodies (Jackson ImmunoResearch Labs) used as appropriate (details given in Table 2.1). A guinea-pig anti-vimentin antibody was also used to check for localisation of PAI-1, uPA and uPAR in myofibroblasts present in the preparation. Each cell specific antibody was used on at least 3 different patients before and after stimulation with *H. pylori* 18 hr, MOI 1:150 in SF media. Ten glands per antibody per patient were counted (minimum total of 30 glands) for the number of PAI-1, uPA and uPAR expressing cells per gland, the number of cells per gland expressing the cell specific marker, and the number of cells per gland expressing PAI-1, uPA or uPAR as well as the cell specific marker. The average number of cells per gland ranged from 60-250 cells.

#### 3.2.6 Analysis of proliferation

Incorporation of BrdU into human gastric glands was carried out as outlined in section 2.6 and Figure 2.1. Cells were treated with PAI-1 ASO (sequences given in section 2.6) as described and stimulated with wild-type *H. pylori* MOI 1:150 or uPA (250mg/ml). Immunocytochemistry was performed on PFA fixed glands as described in section 2.7. Cells were dual-stained with a guinea-pig-anti-vimentin antibody (1:400; Fitzgerald) and FITC-conjugated fast immune anti-BrdU/DNase antibody (1:50; Becton Dickinson). Ten glands per patient, per treatment were scored for BrdU positive cells and BrdU positive cells that were also vimentin positive. Vimentin positive myofibroblasts which were also labelled with BrdU were subtracted from the total number of BrdU labelled cells per gland to give the percentage of BrdU positive epithelial cells.

# **3.3 Results**

# 3.3.1 PAI-1 mRNA abundance is increased in *H. pylori* infection and in gastric cancer

Real-time PCR was used to determine if PAI-1 mRNA abundance is increased in the gastric corpus or antrum in *H. pylori* infection and in gastric cancer patients compared to *H. pylori* negative control subjects (Figure 3.1). A significant increase in PAI-1 mRNA abundance was seen in the corpus of infected subjects compared to non-infected subjects. This increase was enhanced in gastric cancer patients where PAI-1 mRNA abundance was further increased from than seen in *H. pylori* infection (*H. pylori* negative 100%  $\pm$  25, *H. pylori* positive 300%  $\pm$  90, gastric cancer 390%  $\pm$  111; p<0.05, ANOVA, n=5-15). Interestingly, real-time PCR was also performed on antrum samples from *H. pylori* positive (n=10) and negative patients (n=18) but no significant difference in PAI-1 mRNA abundance was seen in this region of the stomach. There was no significant difference in gastrin levels between *H. pylori* negative and positive patients used in this experiment and patients with gastrin levels higher than 30pM were excluded from the study.

## 3.3.2 H. pylori infection increases PAI-1 protein expression

To determine if PAI-1 protein expression and not just mRNA abundance is increased in *H. pylori* infection, immunohistochemical staining of human stomach sections was carried out on both a *H. pylori* negative and *H. pylori* positive patient using an anti-PAI-1 antibody (Figure 3.2). An increase in brown staining indicates an increase in protein expression and it is clear that there is a greater expression of PAI-1 protein in *H. pylori* infection compared to control, non-infected subject. The increase in expression is in the glandular region within

epithelial cells, including the large acid-secreting parietal cells determined by analysis of cell size and shape.



## Figure 3.1

# PAI-1 mRNA abundance is increased in *H. pylori* infection and gastric cancer.

Real time PCR normalised to 18s shows a significant increase in PAI-1 expression in the gastric corpus of *H. pylori* (H.p.) infected patients (n=10) and gastric cancer patients (n=5) compared to *H. pylori* negative subjects (n=15) (mean  $\pm$  S.E. \*p<0.05, ANOVA; Corpus real-time PCR was performed by John Sammut).



# Figure 3.2

# PAI-1 protein expression is increased in *H. pylori* infection.

Immunohistochemical staining of human stomach sections showed increased PAI-1 protein expression in *H. pylori* infected patients (B), compared to non-infected subjects (A). C, An enlarged area of B, showed increased PAI-1 staining is mainly localised to epithelial cells, possibly acid secreting parietal cells. (Carried out in collaboration with department of Pathology, The University of Liverpool).

# 3.3.3 PAI-1 protein is expressed in epithelial cells in gastric cancer

Immunohistochemical staining of human gastric cancer tissue showed PA I-1 protein to be expressed in epithelial cells (Figure 3.3).  $\alpha$ -SMA staining showed the presence of myofibroblasts surrounding the cancer cells and PAI-1 protein is expressed to a much lesser extent in this cell type.



# Figure 3.3

# Increased PAI-1 staining in gastric cancer is localised to epithelial cells.

A, Immunohistochemical localisation of PAI-1 in human gastric cancer tissue. B & C, Enlarged sections of A showing PAI-1 is mainly localised to epithelial cells and to a lesser extent in surrounding stromal cells. D,  $\alpha$ -SMA staining shows the myofibroblasts surrounding the cancer cells.

# 3.3.4 PAI-1 mRNA abundance is increased in a mouse model of epithelial remodelling

The finding that both PAI-1 mRNA and protein expression are up-regulated in humans infected with H. pylori was extended to a rodent model of transgenic (INS-GAS) mice infected with Helicobacter felis (H. felis). These animals have been documented to display a vast amount of tissue remodelling and show dramatic changes within the gastric architecture, eventually progressing to gastric atrophy and the development of gastric cancer approximately 6 months postinfection (discussed in section 1.2.5 and 1.3.7.1; Wang et al., 2000). Real-time PCR indicated that PAI-1 mRNA abundance was significantly increased in INS-GAS mice infected with H. felis compared to uninfected INS-GAS animals 6 months post infection (*H. felis* negative  $100\% \pm 18$ , *H.felis* positive  $199\% \pm 41$ ; p<0.05, t-test, n=6-7). There was no significant increase in PAI-1 mRNA abundance after 3 months of infection, and by 9 months post infection there was no further increase than that seen at the 6 month time-point (Figure 3.4-A). As INS-GAS mice are on FVB/N background, PAI-1 mRNA abundance was measured in uninfected mice of both groups of animals to be certain it was the H. felis infection and not that INS-GAS mice posses a transgene that was responsible for the increase in PAI-1 mRNA abundance. No significant difference between PAI-1 mRNA abundance was seen between uninfected INS-GAS and FBV/N mice (Figure 3.4-B). Also, although infected INS-GAS mice are known to display hypergastrinaemia there was no significant difference seen in plasma gastrin levels between H. felis infected and uninfected INS-GAS mice at the 6 month time-point (as published earlier; Steele et al., 2007). This data
indicates that in mouse, as in humans, *Helicobacter* infection stimulates PAI-1 expression.



### Figure 3.4

# PAI-1 mRNA abundance is increased in *H. felis* infected mice 6 months post infection.

A, Real time PCR normalised to 18s showed elevated PAI-1 expression in the gastric corpus of INS-GAS mice (n=6) after *H. felis* (*H.f.*<sup>+</sup>) infection 6 month post-infection compared to uninfected (*H.f.*<sup>-</sup>) mice (n=7). B, Real time PCR normalised to 18s showed no significant difference in PAI-1 expression between uninfected control FVB/N and INS-GAS mice (n=5) (mean  $\pm$  S.E, \*P<0.05, *t*-test).

# 3.3.5 Cultured gastric glands contain the same epithelial cell types as *in vivo* glands

As PAI-1 was found to be up-regulated by Helicobacter in both human and murine gastric corpus this was the region of the stomach chosen for further investigation. This allowed the expression and regulation of PAI-1 and other members of the uPA system to be studied, and to determine if this increase in expression had any significance upon epithelial cell function. In order to study this, primary cultured gastric glands from H. pylori negative patients (and positive patients only were indicated) were used which are an epithelial cell culture obtained from human corpus endoscopic pinch biopsies (prepared as in section 2.4.2). The gastric epithelium consists of a single layer of cells that invaginate forming highly ordered tubular structures, containing a variety of distinct cell types which are collectively known as gastric glands. The main differentiated cell types within the gastric corpus are parietal cells, chief cells, mucous neck cells, surface mucous neck cells and various endocrine cell types such as enterochromaffin-like (ECL) cells (discussed in section 1.2.2). Each cell type possesses a different cell-specific marker to which anti-bodies have been raised enabling identification of that cell population by immunocytochemistry (Table 2.1).

Primary gastric glands in culture consist of the same key epithelial cell types as *in vivo* gastric glands making them an excellent model to use. Figure 3.5 depicts a schematic structure of a corpus gastric gland showing the various different epithelial cell types and their approximate location within the gland. Also shown is how an important a sub-epithelial cell type, the myofibroblast is which lies underneath and in close proximity to the epithelium. The right hand side of the

90

figure shows a freshly isolated gastric gland, which after approximately 24 hr of culture adheres to the culture dish and spreads out to form an adherent primary gastric gland.



### Figure 3.5

# Schematic of the gastric corpus epithelium and the corresponding adherent cultured gland preparation.

A, Schematic representation showing the major differentiated cell types within a gastric gland of corpus origin (Adapted from Ito & Winchester, 1963). B, Freshly isolated gastric glands in suspension obtained from human corpus pinch biopsies. C, Adherent human primary gastric gland after 24 hr of culture.

# 3.3.6 Endogenous PAI-1 protein is expressed in multiple epithelial cell types within primary gastric glands

To determine the epithelial cell types which express PAI-1 protein, dual labelling immunocytochemistry was performed using an anti-PAI-1 antibody and various gastric cell specific markers (Table 2.1) on primary gastric glands which had been cultured for a total of 72 hr. PAI-1 was found to co-localised with TFF-1 ( $8.94 \pm 1.48\%$  of all PAI-1 positive cells), chromogranin A ( $51.83 \pm 2.49\%$  of all PAI-1 positive cells), pepsinogen ( $23.20 \pm 1.86\%$  of all PAI-1 positive cells) and H<sup>+</sup>, K<sup>+</sup>-ATPase ( $12.84 \pm 1.72\%$  of all PAI-1 positive cells; Table 3.1), indicating that PAI-1 is expressed in surface mucous, ECL, chief and parietal cells respectively (Figure 3.6). No significant difference in expression pattern was seen in the absence of *H. pylori* or when it was exogenously added to the gastric glands. PAI-1 was found not to co-localise with TFF-2 and is therefore not expressed in mucous neck cells.

# 3.3.7 Endogenous uPAR protein is expressed in multiple epithelial cell types within primary gastric glands

To determine the epithelial cell types which express uPAR protein dual labelling immunocytochemistry was performed using an anti-uPAR antibody and various gastric cell specific markers on primary gastric glands which had been cultured for a total of 72 hr. uPAR was found to co-localised with TFF-1 (11.91  $\pm$  1.88% of all uPAR positive cells), chromogranin A (17.15  $\pm$  1.66% of all uPAR positive cells), pepsinogen (28.30  $\pm$  2.49% of all uPAR positive cells) and H<sup>+</sup>, K<sup>+</sup>-ATPase (22.41  $\pm$  2.18% of all uPAR positive cells; Table 3.1), indicating that uPAR is endogenously expressed in surface mucous cells, ECL cells, chief cells

92

Sec.

and parietal cells respectively (Figure 3.7). No difference in expression pattern was seen in the presence or absence of wild-type *H. pylori* and uPAR was found not to co-localise with TFF-2, and is therefore not expressed in mucous neck cells.



### Figure 3.6

# PAI-1 endogenous protein expression in human primary cultured gastric glands.

A-C, PAI-1 protein expression (green) in TFF-1 (red) positive cells. D-F, PAI-1 protein expression (green) in chromogranin A (red) positive cells ECL cells. G-I, PAI-1 protein expression (green) in pepsinogen (red) positive cells chief cells. J-L, PAI-1 protein expression (green) in H<sup>+</sup>, K<sup>+</sup>-ATPase (red) positive parietal cells.



### Figure 3.7

### uPAR endogenous expression in human primary cultured gastric glands.

A-C, uPAR protein expression (green) in TFF-1 (red) positive cells. D-F, uPAR protein expression (green) in chromogranin A (red) positive cells ECL cells. G-I, uPAR protein expression (green) in pepsinogen (red) positive cells chief cells. J-L, uPAR protein expression (green) in H<sup>+</sup>, K<sup>+</sup>-ATPase (red) positive parietal cells. The cell type in which endogenous uPAR is expressed in was the same in the presence and absence of wild-type *H. pylori*.

#### 3.3.8 Both PAI-1 and uPAR are expressed in gastric myofibroblasts

Human primary gastric glands are primarily an epithelial cell culture but contain approximately a 3% population of gastric myofibroblasts. As both PAI-1 and uPAR are expressed within the epithelial cells, their expression within the present myofibroblasts was also tested. Dual-staining immunocytochemistry was performed on primary gastric glands which had been cultured for a total of 72 hr using an anti-PAI-1 or anti-uPAR antibody and a guinea-pig anti-vimentin antibody, which is a marker for myofibroblasts. PAI-1 was not found to be expressed in vimentin positive cells without stimulation with wild-type *H. pylori*. However, after stimulation with *H. pylori* some vimentin positive cells were found to express PAI-1 (Figure 3.8, Table 3.1). uPAR was found to be expressed in vimentin positive cells both before and after stimulation with wild-type *H. pylori* (Figure 3.8, Table 3.1).

#### 3.3.9 Endogenous uPA protein is expressed in gastric myofibroblasts

To determine the exact cell types which express uPA protein, dual labelling immunocytochemistry was performed using an anti-uPA antibody and various gastric cell specific markers (Table 2.1) on primary gastric glands which had been cultured for a total of 72 hr. uPA was found to co-localised with vimentin only (Table 3.1) indicating that uPA is endogenously expressed in gastric myofibroblasts (Figure 3.9). No difference in expression pattern was seen in the presence or absence of wild-type *H. pylori* and uPA was found not to co-localise with chromogaranin A,  $H^+$ ,  $K^+$ -ATPase, pepsinogen, TFF-1 or TFF-2.



## Figure 3.8

1

# PAI-1 and uPAR endogenous protein expression in human primary cultured gastric myofibroblasts.

A-C, PAI-1 protein is not expressed in vimentin positive myofibroblasts in the absence of *H. pylori*. D-F, PAI-1 is expressed in vimentin positive cells after wild-type *H. pylori* stimulation. G-I, uPAR protein expression (green) in vimentin positive myofibroblasts (red) without *H. pylori* stimulation but uPAR is also expressed in myofibroblasts after *H. pylori* stimulation.



## Figure 3.9

C

uPA endogenous protein expression in human primary cultured gastric glands.

uPA protein (green) is expressed in vimentin positive myofibroblasts (red) in the presence (A-C), and absence (D-F) of wild-type *H. pylori*.

|                          | TFF-1               | Chromogranin A      | Pepsinogen          | H <sup>+</sup> , K <sup>+</sup> -ATPase | Vimentin                       |
|--------------------------|---------------------|---------------------|---------------------|---|--------------------------------|
| % PAI-1 expressing cells | 8.94 <u>+</u> 1.45  | 51.83 <u>+</u> 2.49 | 23.20 <u>+</u> 1.86 | 12.84 <u>+</u> 1.72                     | Only after <i>H.</i><br>pylori |
| % uPA expressing cells   | <1                  | <1                  | <1                  | <1                                      | 100 <u>+</u> 0.00              |
| % uPAR expressing cells  | 11.91 <u>+</u> 1.88 | 17.15 <u>+</u> 1.66 | 28.30 <u>+</u> 2.49 | 22.41 <u>+</u> 2.18                     | 14.7 <u>+</u> 1.83             |

## Table 3.1

**Proportion of PAI-1, uPAR and uPA expressing cells in human cultured gastric gland cells after stimulation with wild-type** *H. pylori.* Proportion of PAI-1, uPA and uPAR expressing cells that also express either TFF-1, chromogranin A, pepsinogen,  $H^+$ ,  $K^+$ -ATPase or vimentin. Note <1% of PAI-1 and uPAR expressing cells also express TFF-2. *H. pylori* was added at an MOI 1:150 18 hr, SF media. Distribution of both proteins was found not to be significantly changed after *H. pylori* stimulation with the exception of PAI-1 only being expressed in vimentin positive cells after *H. pylori* stimulation (mean  $\pm$  S.E, n=3-5 patients without *H. pylori* and n=3 with *H. pylori*).

# 3.3.10 *H. pylori* stimulates epithelial cell proliferation which is increased by PAI-1 knock-down

The effect of *H. pylori* on epithelial cell proliferation was investigated using BrdU incorporation into human primary gastric glands. Exogenously added wild-type *H. pylori* M.O.I. 1:150, 18 hr was found to significantly increase the percentage of BrdU labelled cells  $(1.36 \pm 0.12 \text{ fold})$  in 3 separate experiments, each consisting of 4 individual patients (Figure 3.10). The role of PAI-1 in this process was determined by using PAI-1 anti-sense oligonucleotides (ASO).

After PAI-1 knock-down and stimulation with wild-type *H. pylori*, a further increase in the percentage of BrdU labelled gastric gland cells was seen compared to *H. pylori* alone (control ASO  $1.71 \pm 0.17\%$  BrdU labelled cells; control ASO & *H. pylori*  $2.74 \pm 0.36\%$  BrdU labelled cells; PAI-1 ASO & *H. pylori*  $3.44 \pm 0.20\%$  BrdU labelled cells. n=4 patients, ANOVA, \*p<0.05). There was an increase (29.1  $\pm$  3.6%) in BrdU labelled cells when cells were treated with PAI-1 ASO compared to control ASO. This data indicates that *H. pylori* stimulates gastric epithelial cells to proliferate and PAI-1 acts as positive regulator of proliferation. When PAI-1 is knocked-down the balance between protease and its inhibitor is disrupted and uPA can exert its effects on gastric epithelial cells causing an increase in proliferation.

100



### Figure 3.10

*H. pylori* stimulates proliferation of epithelial cells which is further increased by PAI-1 anti-sense oligonucleotide treatment.

BrdU incorporation into human gastric epithelial cells showed that wild-type *H. pylori* increases proliferation of these cells. Knocking-down PAI-1 with antisense oligonucleotides resulted in a further increase in proliferation compared to *H. pylori* alone (mean  $\pm$  S.E n=4 patients, ANOVA, \* p<0.05).

#### 3.3.11 Exogenous uPA stimulates proliferation of epithelial cells

As *H. pylori* was found to stimulate epithelial cell proliferation (Figure 3.10) and it is known to increase the expression of uPA within the stomach (Cédric Duval, PhD thesis, 2007) it was investigated whether exogenously added uPA could mimic the effect *H. pylori* has on proliferation of human primary cultured gastric glands. The role of PAI-1 in controlling uPA in this process was also studied by knocking-down PAI-1 using PAI-1 ASO. uPA alone was found to significantly increase the percentage of BrdU labelled cells within gastric glands which was further enhanced when the main secreted inhibitor of uPA, PAI-1 was knockeddown (Figure 3.11; control  $1.72 \pm 0.17\%$  BrdU labelled cells; uPA  $3.24 \pm 0.18\%$ BrdU labelled cells; uPA & PAI-1 ASO  $4.46 \pm 0.39\%$  BrdU labelled cells; p<0.05, ANOVA, n=5). These data indicate that uPA can act to increase proliferation of epithelial cells but is restrained to an extent by PAI-1.



## Figure 3.11

# uPA stimulates proliferation of epithelial cells which is further increased by PAI-1 anti-sense oligonucleotide treatment.

BrdU incorporation into human gastric epithelial cells showed that exogenous uPA increased proliferation. Knocking-down PAI-1 with anti-sense oligonucleotides (ASO) resulted in a further increase in proliferation. (mean  $\pm$  S.E n=5 patients, \*p<0.05, ANOVA).

#### 3.3.12 PAI-1 ASO reduce the percentage of PAI-1 expressing cells

To verify that PAI-1 ASO reduced the number of PAI-1 expressing cells within human primary gastric glands, immunocytochemistry using a goat anti-PAI-1 antibody was carried out (Figure 3.12). The average total number of cells per gastric gland expressing PAI-1 was quantified after treatment with either control ASO or PAI-1 ASO, before and after the addition of wild-type *H. pylori* M.O.I. 1:150 18 hr (Table 3.2). There was an approximate 60% decrease in the number of cells expressing PAI-1 in those glands treated with PAI-1 ASO compared to control ASO, both in the presence and absence of *H.pylori*. This confirms that PAI-1 ASO treatment is successful in knocking-down PAI-1 expression in human primary gastric glands.





#### Figure 3.12

# PAI-1 anti-sense oligonucleotides reduce the number of PAI-1 expressing cells in human gastric glands.

Immunocytochemistry to show PAI-1 expression in human gastric glands treated with control anti-sense oligonucleotides and wild-type *H. pylori* (A-C). Infected glands treated with PAI-1 anti-sense oligonucleotides and *H. pylori* showed a marked reduction in the number of PAI-1 expressing cells (D-F). Scale bar represents  $63 \mu m$ .

|                             | Control ASO        | PAI-1 ASO          | Control ASO<br>+ H. pylori | PAI-1 ASO<br>+ H. pylori |
|-----------------------------|--------------------|--------------------|----------------------------|--------------------------|
| % PAI-1<br>expressing cells | 8.71 <u>+</u> 0.19 | 3.66 <u>+</u> 0.22 | 10.04 <u>+</u> 0.24        | 3.81 <u>+</u> 0.21       |

### Table 3.2

# There is a decrease in PAI-1 expressing cells in human gastric glands treated with PAI-1 anti-sense oligonucleoides before and after the addition of *H. pylori*.

Average total number of cells per gastric gland expressing PAI-1 was quantified after treatment with control anti-sense oligonucleotides (ASO) and PAI-1 ASO before and after the addition of wild-type *H. pylori* M.O.I. 1:150. There was approximately a 60% decrease in the number of cells expressing PAI-1 in glands treated with PAI-1 ASO compared to control ASO in the presence and absence of *H. pylori* (n=30 glands, mean  $\pm$  S.E).

## **3.4 Discussion**

Results presented in this chapter have shown PAI-1 mRNA abundance to be increased by the oncogenic pathogen *H. pylori* in the gastric corpus, the region of the stomach where infection leads to glandular atrophy and is associated with elevated risk of gastric cancer (Mobley 1997; Correa, 2003). However, PAI-1 mRNA abundance was not increased in the antrum after *H. pylori* infection. An increase in PAI-1 protein expression was also seen in *H. pylori* infection which was predominately epithelial and localised in particular to the large acid secreting parietal cells. A critical stage in the development of intestinal metaplasia and gastric cancer is the induction of mature parietal cell loss (Canfield *et al.*, 1996; Correa, 2003). Therefore, it is an interesting finding that PAI-1 should be up-regulated in this cell type. However, it is not known at this stage if the increase in PAI-1 has any role to play in the loss of parietal cells.

Real-time PCR data showed *H. pylori* infection gives rise to a significant increase in uPA and uPAR mRNA abundance in the corpus region of the stomach (Cédric Duval, PhD Thesis, 2007). However, similar to PAI-1, there was no significant difference in mRNA abundance of these two genes in the antrum, and PAI-3 expression was unaltered in both regions with infection (John Sammaut, Msc Thesis, 2004). This highlights the fact that *H. pylori* up-regulates PAI-1 and other specific members of the uPA system.

It has previously been shown that PAI-2 is increased by *H. pylori* infection through the involvement of NF $\kappa$ B and RhoA (Varro *et al.*, 2004). However, the relevant signalling mechanisms by which *H. pylori* mediates PAI-1 expression remain to be elucidated. It is well recognised that *H. pylori* induces a variety of proteins, growth factors and cytokines (Cox *et al.*, 2001; Mills *et al.*, 2001), which could ultimately be candidates for influencing PAI-1 induction. Given the extensive range of transcriptional response elements contained within the human PAI-1 promoter, and number of mediators known to stimulate PAI-1 transcription (discussed in section 1.5.4) this is likely be a difficult task and is discussed further in Chapter 4 of this thesis.

Real-time PCR showed PAI-1 mRNA abundance to be significantly increased in gastric cancer patients on a background of chronic *H. pylori* infection. The extent of this finding is limited by the small patient size sampled during this experiment. However, it is in good agreement with previous data demonstrating uPA, uPAR and PAI-1 expression to be increased by *H. pylori* in human gastric carcinoma cell lines (Kitadai *et al.*, 2003; Kim *et al.*, 2005; Iwamoto *et al.*, 2005). It is also in agreement with work relating to the fact that increased PAI-1 expression correlates with a poor diagnosis in gastric cancer patients (Nekarda *et al.*, 1994; Heiss *et al.*, 1995; Ito *et al.*, 1996; Plebani *et al.*, 1997; Kaneko *et al.*, 2003; Scicolone *et al.*, 2006), and might serve as a new and promising parameter for both prediction and prognosis of gastric cancer (Sakakibara *et al.*, 2006).

The paradoxical fact that PAI-1 acts to control extracellular proteolysis, yet is associated with poor prognosis in a variety of human cancers could be explained by the many other functions of PAI-1 which are independent of uPA inhibition. These include its involvement in cell motility via promotion of cell detachment both by inactivating integrins and binding to the ECM molecule VN (Stefansson & Lawrence, 1996; Czekay *et al.*, 2003) and also by encouraging cell migration through an interaction with the LRP (discussed in section 1.5.5.1; Degryse *et al.*, 2004). In addition to these alternative functions, PAI-1 has also been documented to influence processes involved in tumour cell biology such as angiogenesis (discussed in section 1.5.5.2) and inhibition of apoptosis via the inactivation of caspase-3 (Chen *et al.*, 2004). Alternatively, as PAI-1 expression can be induced by uPA itself (Shetty *et al.*, 2003) high PAI-1 expression seen in human gastric cancer could simply reflect high expression of uPA.

In this study, PAI-1 expression was found to be localised within gastric cancer epithelial cells and to a much lesser extent in surrounding stromal cells. Previous immunohistochemical studies have identified strong PAI-1 expression in both tumour cells and fibroblasts in breast carcinomas (Dublin *et al.*, 2000). However, in colon cancer tissue the majority of PAI-1 expression was seen in myofibroblasts with no PAI-1 immunoreactivity detected in cancer cells (Illemann *et al.*, 2004) therefore showing varying expression patterns in different tissue types.

In an INS-GAS mouse model of gastric epithelial remodelling, PAI-1 mRNA abundance was found to be increased 6 months post *H. felis* infection. Infected INS-GAS mice show initial moderate hypergastrinaemia, followed later by parietal cell loss, development of gastric atrophy, achlorhydria and gradual progression to dysplasia (discussed in sections 1.2.5 and 1.3.7.1). Some of these observations are similar characteristics to that seen in *H. pylori* mediated human disease (Wang *et al.*, 2000), making it a good model to help determine the role of PAI-1 in the regulation of the gastric epithelium and its involvement in gastric remodelling and carcinogenesis.

There is a well documented role for the involvement of PAI-1 in the process of tissue repair (Providence *et al.*, 2000). However, it is not yet defined if the increase in PAI-1 expression seen in *Helicobacter* infection is initially part of a mucosal defensive mechanism, protecting the host from bacterial damage, or if elevated PAI-1 has a pathophysiological involvement, effecting epithelial cell function and contributing to the process of tissue remodelling.

As PAI-1 was found to be increased in epithelial cells within the gastric corpus in H. pylori infection, and also in gastric cancer, a primary culture consisting of all the gastric epithelium cell types as in vivo gastric glands was used to further investigate the effects of PAI-1 on gastric epithelial cell function. The endogenous expression of PAI-1, uPAR and uPA in primary cells showed cell restricted expression. However, both PAI-1 and uPAR are expressed in the same cell types but to a different extent; PAI-1 is primarily expressed in ECL cells whereas uPAR is principally expressed in chief and parietal cells. No previous reports have investigated the specific gastric cell types in which uPA system members are expressed, but the different expression patterns may reflect variations in spatial regulation and control of the uPA system. Myofibroblasts in several experimental models have been shown to strongly express PAI-1, yet interestingly it was only expressed in gastric myofibroblasts after stimulation with H. pylori. Gastric sub-epithelium cells in vivo would not normally be directly exposed to *H. pylori*. However, it is possible that in culture the presence of the bacteria may stimulate myofibroblasts to express PAI-1 or epithelial cells may secrete factors which act upon myofibroblasts promoting the expression of the protein.

Immunohistochemical staining of human stomach sections showed increased uPA expression in epithelial cells in *H. pylori* positive patients, with an abundant expression in sub-epithelial cells compared to H. pylori negative patients (Cédric Duval, PhD Thesis, 2007). However, in primary cultured human gastric glands, uPA was found to be expressed only in sub-epithelial myofibroblasts and not in epithelial cells, both before and after stimulation with exogenously added H. pylori. A possible reason for this difference in uPA expression is that cultured gastric glands were only acutely stimulated with bacteria and the patients from which these glands were derived from did not have chronic H. pylori corpus infection. However, the stomach sections from patients in which immunohistochemical data were obtained from did have chronic *H. pylori* corpus infection. Unfortunately, this is clinical information obtained retrospectively after experiments had been conducted. Ideally it would be useful to culture gastric glands from chronically H. pylori infected patients to determine if uPA can be detected in gastric epithelial cells, and if so in what specific cell types.

Infection with *H. pylori* is not only associated with the induction and increased expression of many genes, including those of the uPA system, but it also associated with increased gastric epithelial cell proliferation in both humans (Moss *et al.*, 2001) and animals (Israel & Peek, 2001). This is also seen in the primary gastric epithelial cell cultures used in this study. Exogenously added *H. pylori* increased the percentage of BrdU labelled cells which was mimicked when cells were stimulated with exogenous uPA. A further increase in proliferation in both cases was seen after the knock-down of the protease inhibitor PAI-1. These data implicate activation of uPA in the preneoplastic hyperproliferative condition associated with *H. pylori* infection which is at least

partly held in check by PAI-1. This also demonstrates the need for a tightly regulated, evenly balanced relationship between protease/protease inhibitor as a disruption to this can result in pathophysiological consequences, contributing to the metastatic cascade (Choong *et al.*, 2003).

However, it is also important to consider the balance of uPA and PAI-1 at the individual patient level. Activity assays for uPA (Cédric Duval, PhD Thesis, 2007) showed that approximately 70% of human patients responded with an increased uPA activity after *H. pylori* infection when PAI-1 was knocked-down. It is probable that these individuals originally had an elevated concentration of PAI-1, restraining the activity of uPA, displaying endogenous protection from the action of the protease. In patients whose uPA activity did not alter much after *H. pylori* infection and PAI-1 knock-down, probably already had increased concentrations of uPA activity due to a decreased concentration of PAI-1.

The data presented in the current chapter shows that *Helicobacter* infection increases gene expression of PAI-1. This increase might function as part of a protective host response helping to combat infection by contributing to tissue repair after bacterial damage. By inhibiting the protease activity of uPA and the production of plasmin, PAI-1 not only controls fibrinolysis thereby promoting the process of wound healing, but also acts to restrain the proliferative effects of uPA. An increase in PAI-1 could also function to inhibit cell invasion, again by limiting the production of plasmin. However, sustained levels of elevated PAI-1 in chronic *Helicobacter* infection could have a negative effect for the host, contributing to gastric epithelium remodelling, influence metastasis, a process to which there has been extensive experimental evidence supporting a major importance of the uPA system (reviewed Sidenius & Blasi., 2003).

# **3.5 Conclusions**

- PAI-1 mRNA abundance and protein expression are increased in *Helicobacter* infection and in gastric cancer.
- 2. The increased expression of PAI-1 by *H. pylori* is predominantly in epithelial cells, and seen especially in parietal cells.
- 3. In gastric cancer, PAI-1 expression is mostly epithelial.
- Endogenous PAI-1, uPAR and uPA proteins are expressed by sub-sets of the main differentiated gastric epithelial cells.
- 5. Gastric epithelial cell proliferation is increased by *H. pylori* and uPA, which is further increased in both cases after PAI-1 knock-down.

**Chapter 4** 

# Transcriptional regulation of uPA,

# uPAR and PAI-1 in human gastric

epithelial cells

# **4.1 Introduction**

The previous result chapter showed that infection with *Helicobacter* is associated with increased expression of PAI-1 in the gastric epithelium. As *Helicobacter* has also been shown to increase expression of other key uPA system members including PAI-2 (Varro *et al.*, 2004), uPA and uPAR (Cédric Duval PhD thesis, 2007) it is important to understand and investigate the cellular mechanisms by which these changes in expression are regulated.

The clinical outcome of *H. pylori* infection is dependent on several aspects including host (El-Omar *et al.*, 2000), environmental (Nomura & Stemmermann, 1989) and bacterial factors (Atherton, 1998). Many virulence determinants modulating its interactions with its host are produced by *H. pylori* with strains possessing the *cag* pathogenicity island (*cag*-PAI; encoding a type IV bacterial protein secretion system), and/ or an active form of VacA (a pore forming toxin) thought to be most virulent, and more strongly associated with disease (discussed in section 1.3.3; Blaser *et al.*, 1995; Kuipers *et al.*, 1995; Parsonnet *et al.*, 1997; Nomura *et al.*, 2002).

Previous studies have identified a role for *H. pylori* virulence factors in the upregulation of certain proteins in epithelial cells such as matrix metalloproteinase-7 (MMP-7; Bebb *et al.*, 2003; Crawford *et al.*, 2003). The possible involvement of *H. pylori* virulence factors in the induction of PAI-1, uPA and uPAR has been investigated in this results chapter using a number of isogenic mutant bacterial strains which have previously been described (Argent *et al.*, 2008).

As well as its classical function in controlling acid secretion (discussed in section 1.2.3), the gastric hormone gastrin is now well recognised for regulating

expression of several genes in the gastric epithelium (Dockray *et al.*, 2001). These genes include chromogranin A (Dimaline *et al.*, 1993), histidine decarboxylase (HDC; Dimaline & Sandvik, 1991), vesicular monoamine transporter type 2 (VMAT2; Dimaline & Struthers, 1996), Reg1 (Steele *et al.*, 2007) and PAI-2 (Varro *et al.*, 2002). Gastrin also stimulates gene and protein expression in human cultured gastric cancer cells of several genes such as hepatocyte growth factor (HGF; Konturek *et al.*, 2003). Gastrin, produced by G-cells exerts its effects by binding specifically to the CCK2R which is expressed on epithelial cells within the human gastric mucosa (Schmitz *et al.*, 2001). It is generally known that *H. pylori* infection results in a mild hypergastrinaemia state (Levi *et al.*, 1989; Mulholland *et al.*, 1993) and the effects of gastrin on PAI-1 gene expression have been investigated in the current chapter.

Gastric subepithelial cells such as myofibroblasts produce a number of key growth factors thought to act on epithelial cells and play a role in epithelialmesenchymal signalling (Powell *et al.*, 1999a). This process is important for normal mucosal organisation and occurs during development, wound healing and in progression to cancer (discussed in section 1.4). Infection with *H. pylori* has been shown to increase expression of a variety of growth factors in the human gastric mucosa including HGF (Kondo *et al.*, 1995; Taha *et al.*, 1996). The c-MET (mesenchymal epithelial transition factor) proto-oncogene encodes the MET protein, the high affinity receptor for HGF (Giordano *et al.*, 1989; Bottaro *et al.*, 1991; Naldini *et al.*, 1991). The MET receptor is expressed solely on epithelial cells (Prat *et al.*, 1991; Sonnenberg *et al.*, 1993; Fukamochi *et al.*, 1994) and is activated in a paracrine manner by HGF, produced and released from underlying mesenchymal cells. Several reports have implicated HGF in the regulation of components of the uPA system and it has been demonstrated to mediate PAI-1 production in liver cells (Wojta *et al.*, 1994; Imagawa *et al.*, 2006), human keratinocytes (Wojta *et al.*, 1999), and has also been shown to increase expression of uPA and uPAR in kidney epithelial cells (Pepper *et al.*, 1992). The activation of HGF has been shown to require uPA (Naldini *et al.*, 1992), and uPA production has been found to be stimulated by HGF in bovine capillary endothelial cells (Grant *et al.*, 1993) and in human stomach cancer cell lines (Lee *et al.*, 2006c). Strikingly, hepatocyte cells transduced with activated MET show a prominent up-regulation of PAI-1 (Boccaccio *et al.*, 2005), providing a strong association between the two. Over-expression of *c-MET* has been found in many human cancers, including stomach cancer (Wu *et al.*, 1998; Huang *et al.*, 2001).

Work from our laboratory has shown *H. pylori* infection to increase the expression of MMP-7 in gastric epithelial cells (Wroblewski *et al.*, 2003) which stimulates myofibroblasts resulting in increased bio-availability of IGF-II (discussed in section 1.4; Hermes *et al.*, 2005; McCaig *et al.*, 2006). Acting as an autocrine growth factor, IGF-II influences migration, proliferation and adhesion of myofibroblasts, and proliferation of gastric epithelial cells (McCaig *et al.*, 2006; Varro *et al.*, 2007). As well as this, IGF-II is known to be increased in gastric cancer (Ajisaka *et al.*, 2001). The effects of gastrin, HGF and IGF-II on PAI-1 expression within the gastric epithelium were investigated as these are three epithelial or mesenchymal derived factors known to be important for epithelial cell function, and for gastric epithelial-mesenchymal interactions (Figure 4.1).



### Figure 4.1

Simplified model of several epithelial and mesenchymal factors (and their respective receptors) which are known to play a role in epithelial-mesenchymal interactions within the gastric mucosa.

## 4.1.1 Aims

The aims of this chapter were to study the regulation of PAI-1 and other uPA system members in the gastric epithelium using luciferase promoter reporter consructs. The specific objectives were to:

- 1. study the cellular targeting and expression of uPA-, uPAR- and PAI-1luciferase promoter-reporter constructs in primary gastric gland cells;
- determine the effects of *H. pylori* on uPA-, uPAR- and PAI-1- luc expression;
- 3. investigate the involvement of *H. pylori* virulence factors on uPA-, uPAR- and PAI-1- luc expression;
- 4. examine whether gastric mucosa growth factors such as gastrin, IGF-II and HGF regulate PAI-1-luc expression within the gastric epithelium.

## 4.2 Materials and methods

### 4.2.1 Transient transfection and luciferase assays

AGS2 cells and primary gastric glands (prepared as in section 2.4.1 and 2.4.2) were cultured for 24 hr before being transfected (as described in section 2.5) with  $3\mu g$  of DNA per well of uPAR-, uPA- or PAI-1-luc, together with 5ng per well of a *Renilla* luciferase reporter construct, phRL-SV40 (Promega) in SFM 20 min. Details of all plasmids used are given in section 2.1.1. Cells were stimulated with *H. pylori* at an M.O.I. of 1:150 (details of bacterial strains used are in section 2.1.3), 1nM heptadecapeptide amidated gastrin (G17), 40ng/ml HGF or 100ng/ml IGF-II in 1ml SFM. Luciferase activity was measured by dual luciferase assay (Promega) in a Lumat LB9507 luminometer (Berthold). Results are displayed as fold increase over unstimulated control, where 1.0 signifies no change in luciferase activity.

# 4.2.2 Cellular targeting of uPAR-, uPA- and PAI-1-luc in primary human adherent glands

To study the cellular targeting of the luciferase vectors, PAI-1-, uPA- and uPARluc, immunocytochemistry was carried out (as described in section 2.7) on transfected human gastric glands (as described in section 2.5.2). Cells were dual immunolabelled with a goat-anti-luciferase antibody (Rockland Immunochemicals), together with one of the cell specific markers for differentiated gastric gland cells (Table 2.1). The appropriate FITC or TRed conjugated secondary antibodies were used (Jackson Immunoresearch). Ten fields of cells from at least 3 patients were counted and results expressed as a percentage of total cell number as reported earlier (Wroblewski *et al.*, 2003).

## 4.3 Results

# 4.3.1 Gastrin stimulates PAI-1, uPA- and uPAR-luc expression in AGS2 cells

The functionality of 4.5kb of the human PAI-1 promoter, 1.94kb of the human uPA promoter and 2.3kb of the human uPAR promoter all coupled to a luciferase reporter (PAI-1-, uPA- and uPAR-luc) were tested by transiently transfecting these constructs into the AGS2 cell line. These were newly made promoter-reporter constructs so their activity in a cell line was verified before their use in experiments utilising valuable human primary cells. As AGS2 cells are stably transfected with the CCK2R, gastrin was the first choice of stimuli since it has previously been shown to stimulate PAI-2 expression (Varro *et al.*, 2002b). Stimulation with 1nM G17, 8 hr resulted in an increase in luciferase activity with all constructs compared to unstimulated controls (Figure 4.2; uPAR-luc 2.53  $\pm$  0.19 fold increase; uPA-luc 2.40  $\pm$  0.09 fold increase; PAI-1-luc 1.78  $\pm$  0.06 fold increase. Mean  $\pm$  S.E, n=9). This showed all three constructs to be functionally correct and transcriptionally active, enabling their use in further experiments.

# 4.3.2 Transient transfection of human gastric gland cells with PAI-1-, uPAor uPAR-luc

To allow the transcriptional regulation of PAI-1, uPA and uPAR by both *H. pylori* and specific growth factors to be studied in primary gastric epithelial cells, a methodology was developed to transfect these cells with luciferase promoter-reporter constructs. The functionality of PAI-1-, uPA- and uPAR-luc had previously been tested in AGS2 cells (Figure 4.2), but never before in primary human gastric glands in culture. Transfection was carried out as described in

section 2.5.2 using Magnetofection and cells were stimulated with exogenously added *H. pylori* or growth factors as indicated. The technique of Magnetofection using CombiMag reagent is able to transfect every cell type within the gland as an adherent cell culture, is highly efficient and ideal for transfection of primary cells. Further details of this methodology found can be at Briefly, www.ozbiosciences.com/magentofection.html. Magnetofection associates nucleic acids/ plasmids with magnetic nanoparticles coated with cationic molecules. The resulting molecular complexes are targeted to and endocyted by cells using an applied magnetic field in the form of a magnetic plate. Magnetic nanoparticles are completely biodegradable and are not toxic to the cells. Nucleic acids/ plasmids are released into the cytoplasm by destabilisation of the endosome by cationic lipids (TransFast) coated on the particles which release plasmids by a flip-flop mechanism. Appropriate concentrations of reagents used are given in section 2.5.2. Figure 4.3 outlines the protocol used to identify the specific cell types within the gastric glands that express the luciferase-reporter constructs or alternatively, luciferase activity measured after *H. pylori* or growth factor stimulation.



## Figure 4.2

### Gastrin stimulates expression of uPAR-, uPA- and PAI-1-luc.

Luciferase expression in AGS2 cells transfected with uPAR-, uPA or PAI-1-luc promoter-reporter constructs. Expression of all three constructs was significantly increased in response to 1 nM G17, 8 hr. Results are expressed as fold increase of luciferase expression after stimulation over unstimulated controls. (Mean  $\pm$  S.E. n=9, \*p<0.05, *t*-test).

### 1. Human gastric biopsy



### Figure 4.3

# Schematic protocol for transfection of human gastric glands with uPAR, uPA- or PAI-1-luc.

Adherent gastric glands can be transfected with uPA system member luciferase promoter-reporter constructs using Magnetofection (see section 2.5.2 and 4.3.2 for details), allowing the transcriptional regulation of these genes by *H. pylori* or growth factors to be studied in luciferase assays. The specific cells types in which the constructs are expressed can be identified via immunocytochemistry using an anti-luciferase antibody and in addition various cell specific markers.

**4.3.3 PAI-1-luc is targeted to cell types which also express endogenous PAI-1** To elucidate the cell types which 4.5kb of the human PAI-1 promoter coupled to a luciferase reporter (PAI-1-luc) are targeted to, and expressed by, human primary gastric glands were transiently transfected and dual labelling immunofluorescence carried out using an anti-luciferase antibody and various gastric cell specific markers. PAI-1-luc was targeted to, and expressed by, parietal cells, ECL cells and myofibroblasts as luciferase co-localised with  $H^+$ ,  $K^+$ -ATPase, VMAT2 and vimentin (Figure 4.4 and Table 4.1). PAI-1-luc was highly expressed in parietal cells, ECL cells and to a lesser extent in myofibroblasts. The expression pattern of PAI-1-luc was not significantly different when cells were stimulated with wild-type *H. pylori* (Table 4.1). As PAI-1-luc is expressed in the same cell population as the endogenous protein (shown in section 3.3.6) it is a good reporter construct to use to study the regulation of PAI-1 expression.

#### 4.3.4 uPA-luc is targeted to cell types which also express endogenous uPA

To confirm the cell types which 2.9kb of the human uPA promoter coupled to a luciferase reporter (uPA-luc) is targeted to and expressed by, similar experiments to that described above for PAI-1-luc were carried out. uPA-luc was found to directly target to, and be expressed in, parietal cells, ECL cells, chief cells and myofibroblasts as luciferase co-localised with H<sup>+</sup>, K<sup>+</sup>-ATPase, VMAT2, pepsinogen and vimentin (Figure 4.5 and Table 4.1). The expression pattern of uPA-luc was not significantly different when cells were stimulated with wild-type *H. pylori* (Table 4.1).

# 4.3.5 uPAR-luc is expressed by the same cell types which express endogenous uPAR

Similar experiments to that described above for uPA-luc and PAI-1-luc were carried out to determine if 2kb of the human uPAR promoter coupled to a luciferase reporter (uPAR-luc) exhibits cell restricted expression within human primary gastric glands. uPAR-luc was found to have direct targeted expression to parietal cells, ECL cells, chief cells, surface mucous cells and myofibroblasts as luciferase co-localised with H<sup>+</sup>, K<sup>+</sup>-ATPase, VMAT2, pepsinogen, TFF-1 and vimentin (Figure 4.6 and Table 4.1). The expression pattern of uPAR-luc was not significantly different when cells were stimulated with *H. pylori* (Table 4.1). Endogenous uPAR was also found to be expressed in all five of the cell types in which uPAR-luc expressed in (shown in section 3.3.7), therefore making it a very suitable expression vector to use in further studies of uPAR.

#### 4.3.6 PAI-2-luc is expressed by a variety of gastric epithelial cells

Endogenous PAI-2 has previously been shown to be expressed by a variety of gastric epithelial cells (Varro *et al.*, 2004) and the 2.3kb of the human PAI-2 promoter coupled to luciferase reporter (PAI-2-luc) has been extensively used in AGS2 cell line work. In the context of the other uPA system members expression patterns, it was also determined if PAI-2-luc exhibited cell restricted expression. Primary human gastric glands transiently transfected with PAI-2-luc showed that this reporter construct was expressed in parietal cells, ECL cells, surface mucous cells, neck cells and myofibroblasts. Luciferase was found to co-localised with H<sup>+</sup>, K<sup>+</sup>-ATPase, VMAT2, TFF-1, TFF-2 and vimentin (Figure 4.7). PAI-2-luc expression correlates well with endogenous PAI-2 protein expression which was
shown to be expressed in mucous cells, chief cells, parietal cells and ECL cells within primary human gastric glands (Varro *et al.*, 2004).



4.5kb of the human PAI-1 promoter targets expression of PAI-1-luc to specific cell types in human primary gastric glands.

A-C, PAI-1-luc (green) expression in H<sup>+</sup>, K<sup>+</sup>-ATPase (red) positive parietal cell. D-F, PAI-1-luc (green) expression in VMAT-2 (red) positive ECL cell. G-I, PAI-1-luc (green) expression in vimentin (red) positive myofibroblast.



## 2.9kb of the human uPA promoter targets expression of uPA-luc to specific cell types in human primary gastric glands.

A-C, uPA-luc (green) expression in  $H^+$ ,  $K^+$ -ATPase (red) positive parietal cell. D-F, uPA-luc (green) expression in VMAT-2 (red) positive ECL cell. G-I, uPAluc (green) expression in pepsinogen (red) positive chief cell. J-L, uPA-luc (green) expression in vimentin (red) positive myofibroblast.



# 2kb of the human uPAR promoter targets expression of uPAR-luc to specific cell types in human primary gastric glands.

A-C, uPAR-luc (green) expression in H<sup>+</sup>, K<sup>+</sup>-ATPase (red) positive parietal cell. D-F, uPAR-luc (green) expression in VMAT-2 (red) positive ECL cell. G-I, uPAR-luc (green) expression in pepsinogen (red) positive chief cell. J-L, uPARluc (green) expression in TFF-1 (red) positive cell. M-O, uPA-luc (green) expression in vimentin (red) positive myofibroblast.

|                              |      | TFF-1           | VMAT-2              | Pepsinogen         | H <sup>+</sup> , K <sup>+</sup> -ATPase | Vimentin           |
|------------------------------|------|-----------------|---------------------|--------------------|---|--------------------|
| % PAI-1-luc expressing cells | -H.p | <1              | 48.83 <u>+</u> 8.50 | <1                 | 68.06 <u>+</u> 7.58                     | 6.91 <u>+</u> 2.54 |
| % uPA-luc expressing cells   | -H.p | <1              | $36.55 \pm 3.61$    | $22.00 \pm 3.93$   | 39.94 <u>+</u> 3.55                     | $13.01 \pm 3.92$   |
| % uPAR-luc expressing cells  | -H.p | $7.00 \pm 3.59$ | 47.67 <u>+</u> 5.05 | 3.43 <u>+</u> 2.33 | 34.04 <u>+</u> 1.91                     | 7.7 <u>+</u> 2.66  |

#### Table 4.1

#### Proportion of PAI-1-, uPA- and uPAR-luc expressing cells in human primary cultured gastric glands.

Proportion of PAI-1-, uPA- and uPAR-luc expressing cells that also express either TFF-1, VMAT-2, pepsinogen, H<sup>+</sup>, K<sup>+</sup>-ATPase or vimentin. Note <1% of PAI-1-luc, uPA-luc or uPAR-luc expressing cells express TFF-2. Table shows quantification data in control, *H. pylori* negative glands. Stimulation with *H. pylori* MOI 1:150 18 hr gave no significant difference in distribution patterns. (mean  $\pm$  S.E, n=3-5 patients).



## 2.3kb of the human PAI-2 promoter targets expression of PAI-2-luc to specific cell types in human primary gastric glands.

A-C, PAI-2-luc (green) expression in H<sup>+</sup>, K<sup>+</sup>-ATPase (red) positive parietal cell. D-F, PAI-2-luc (green) expression in VMAT-2 (red) positive ECL cell. G-I, PAI-2-luc (green) expression in TFF-1 (red) positive cell. J-L, PAI-2-luc (green) expression in TFF-2 (red) positive cell. M-O, PAI-2-luc (green) expression in vimentin (red) positive myofibroblast.

# 4.3.7 Induction of uPAR-, uPA- and PAI-1-luc by *H. pylori* in gastric epithelial cells

The protocol outlined in Figure 4.3 was used to investigate if *H. pylori* directly stimulates expression of uPAR-, uPA- and PAI-1-luc in primary gastric epithelial cells. Results presented in Chapter 3 showed increased PAI-1 mRNA abundance and protein expression in gastric epithelial cells in response to *H. pylori* infection in human patients with the same being true for both uPA and uPAR (Cédric Duval, PhD thesis 2007). It is therefore interesting to study the regulation of these three proteins by *H. pylori* in primary epithelial cells in culture using promoter-reporter constructs thus determining if this *in vitro* model can be used to explain the cellular *in vivo* regulation mechanisms. Wild-type *H. pylroi* (strain 60190) significantly increased the expression of uPAR-, uPA- and PAI-1-luc in human gastric gland cells compared to non-stimulated controls (Figure 4.8; uPAR-luc  $1.8 \pm 0.2$  fold increase; uPA-luc  $2.1 \pm 0.5$  fold increase; PAI-1-luc 3.0  $\pm 0.3$  fold increase. Mean  $\pm$  S.E, n=6 patients).



## H. pylori stimulates uPAR-, uPA- and PAI-1-luc in human gastric gland cells.

*H. pylori* M.O.I. 1:150 18 hr stimulated the expression of 2kb of the human uPAR, 2.9kb of the human uPA and 4.5kb of the human PAI-1 promoter in gastric epithelial cells. Results are expressed as fold increase in luciferase expression after stimulation compared to non-stimulated controls. (mean  $\pm$  S.E, n=6 patients. \*p<0.05, *t*-test).

#### 4.3.8 Mechanism of H. pylori induction of uPAR-, uPA- and PAI-1-luc

To examine the mechanisms by which *H. pylori* up-regulates uPAR-, uPA- and PAI-1-luc, and to define the precise role of bacterial virulence factors in induction of these three genes, transiently transfected human gastric gland cells were stimulated with a series of isogenic *H. pylori* mutants. A similar response for the increase in uPAR-, and PAI-1-luc to that seen with the wild-type 60190 strain (Figure 4.8) was obtained when cells were stimulated with another wild-type *H. pylori* strain (84-183), and mutant strains that were *cagE* null, *cagA* null and *vacA* null (Figure 4.9 A and C). In contrast, uPA-luc expression was stimulated by the alternative wild-type strain (84-183) but expression was abolished by a *cagE* null *H. pylori* mutant (Figure 4.9 B).

#### 4.3.9 Regulation of PAI-1-luc by growth factors

Previous studies have indicated that *H. pylori* induces epithelial cell release of various chemokine, cytokine and paracrine mediators. The possibility that *H. pylori* stimulates PAI-1-luc via paracrine mechanisms involving several growth factors was considered. Gastrin is known to stimulate PAI-2 expression in AGS2 cells (Varro *et al.*, 2002b) and real-time PCR data has previously shown PAI-1 mRNA abundance to be increased in patients with elevated gastrin concentrations ( $\geq$ 30 pM; John Sammut, Msc Thesis, 2004). The possibility of PAI-1 being a target of gastrin was investigated. As gastrin is both produced by, and known to act on epithelial cells the concept of PAI-1 expression being increased by growth factors such as HGF and IGF-II secreted from mesenchymal cells (such as myofibroblasts), but known to act on epithelial cells was explored (Sonnenberg *et al.*, 1993; McCaig *et al.*, 2006). PAI-1-luc expression was found

to be significantly increased by all three growth factors (Figure 4.10; IGF-II 2.02  $\pm$  0.20 fold; HGF 2.33  $\pm$  0.57 fold; G17 2.38  $\pm$  0.74 fold; Mean  $\pm$  S.E, n=3-9 patients).



uPA but not uPAR or PAI-1 luciferase expression is abolished by a CagE null *H. pylori* mutant in human primary gastric epithelial cells. Luciferase expression in human primary gastric glands transfected with either uPAR-luc (A), uPA-luc (B), or PAI-1-luc (C) promoter-reporter constructs. Expression of all three constructs was significantly increased in response to wild-type (wt) *H. pylori* infection (M.O.I. 1:150). Deletion of *cagE*, *cagA* and *vacA* had no effect on PAI-1-luc or uPAR-luc expression. However, uPA-luc expression was abolished by a *cagE* null *H. pylori* mutant. Results are shown as fold increase in luciferase expression compared to non-stimulated controls. (Mean  $\pm$  S.E, n=5 patients, \* p<0.05, ANOVA).



#### IGF-II, HGF and gastrin increase PAI-1-luc expression

IGF-II (100ng/ml), HGF (40ng/ml) and gastrin (1nM) 18 hr stimulated 4.5kb of the human PAI-1 promoter in gastric epithelial cells. Results are expressed as fold increase in luciferase expression after stimulation over non-stimulated controls. (Mean + S.E, n=3-9 patients, \*p<0.05, ANOVA).

#### **4.4 Discussion**

Results presented in this chapter show that *H. pylori* can stimulate expression of luciferase promoter reporter constructs containing either 4.5kb of the human PAI-1 promoter, 2.9kb of the human uPA promoter or 2kb of the human uPAR promoter in human primary gastric epithelial cells. This is in excellent agreement with RT-PCR and immunohistochemical data displayed in Chapter 3 of this thesis (Figures 3.1 and 3.2), showing increased PAI-1 mRNA and endogenous protein expression in *H. pylori* infection. Endogenous uPA and uPAR protein and mRNA abundance have also been shown to be increased by *H. pylori* in infected human subjects (Cedric Duval, PhD thesis, 2007).

The promoter sequences of PAI-1-, PAI-2-, uPA- and uPAR-luciferase constructs were found to selectively target expression to specific cell types within human primary gastric gland cells. It is important to note that these luciferase vectors are generally expressed in cell types normally associated with expression of the corresponding endogenous protein.

Recent investigations into mechanisms underlying *H. pylori* induced gastric cancer have highlighted that disease involves well choreographed interactions between host and pathogen involving differences between bacterial strains, variations in individual hosts or environmental influences (discussed in section 1.3; Blaser & Berg, 2001). Patients infected with *H. pylori cag+* strains are at a higher risk of developing distal gastric cancer (Crabtree *et al.*, 1993, Kuipers *et al.*, 1995, Blaser *et al.*, 1995; McColl *et al.*, 1997), and genes contained within the *cag-*PAI are required for

137

induction of several epithelial cell responses relevant to pathogenesis (Tummuru *et al* 1995; Sharma *et al.*, 1998; Naumann *et al.*, 1999). Stimulation of PAI-1-, uPA- and uPAR-luc by *H. pylori* occurs via distinct mechanisms as expression of all three were unaltered by isogenic mutants lacking CagA and VacA. However, uPA expression was virtually abolished by a CagE null mutant. This demonstrates a requirement for an intact and fully functional *cag*-PAI island and a role for CagE in the up-regulation of uPA by *H. pylori*.

Interestingly, INS-GAS mice infected with *H. pylori* with inactivated CagE show delayed progression to neoplasia compared to those animals infected with wild-type bacteria strains (Fox *et al.*, 2003a). The release of certain chemokines such as IL-8 from gastric epithelial cells has been found to require *cagE* (Keates *et al.*, 1999) and this gene is also needed for activation of several transcription factors in gastric cancer cell lines, including NF- $\kappa$ B and AP1 which are known to play a role in transcriptional regulation of uPA (Novak *et al.*, 1991; Kawasaki *et al.*, 1998; Mitsuno *et al.*, 2001; Guillemin *et al.*, 2002).

As well as *cag*-PAI and VacA there are at least 32 outer membrane proteins involved in bacterial adherence that have been identified on the cell surface of *H. pylori* which could also act as virulence factors (discussed in section 1.3.3.1). Possible important proteins include the blood group antigenbinding adhesin (BabA), the outer membrane inflammatory protein (OipA) and sialic acid-binding adhesin (SabA; Lu *et al.*, 2005a). *H. pylori* strains containing a functional copy of OipA are linked with more severe gastric

138

inflammation, increased levels of IL-8 and higher bacterial colonisation (Peek, 2002). It could be proposed that *H. pylori* virulence factors other than *cagE*, *cagA* or *vacA* are responsible for the increase in expression of PAI-1 and uPAR such as plasminogen binding proteins.

There is accumulating evidence that the transformation of the 'normal' *H. pylori* infected gastric mucosal cells into malignant cells and the development of gastric cancer requires the action of mucosal promoting factors such as gastrin, EGF, HGF, IL-8 and TGF- $\alpha$  (Crabtree *et al.*, 1993; Boyer *et al.*, 2000) and epithelial-mesenchymal interactions (discussed in section 1.4). The increase in PAI-1 expression after stimulation with gastrin and growth factors such as HGF and IGF-II demonstrates that both epithelial- and mesenchymal-derived factors are important for the regulation of this protease inhibitor.

The specific signalling pathways involved in the induction of PAI-1 gene expression in gastric epithelial cells have not been investigated but are likely to be complex and further work is needed to elucidate mediators which are responsible for *H. pylori*, HGF, IGF-II and gastrin induction of PAI-1. The human PAI-1 promoter has been extensively characterised and is known to contain consensus sequences for several known regulatory elements (discussed in section 1.5.4; Kunz *et al.*, 1995; Grubic *et al.*, 1996; Dennler *et al.*, 1998) and is induced by a number of controls (Emeis *et al.*, 1986; Alessi *et al.*, 1988; Hopkins *et al.*, 1991; Sperti *et al.*, 1992; Nordt *et al.*, 1995; Le Magueresse- Battistoni *et al.*, 1997; Pinsky *et al.*, 1998; Brown *et al.*, 2000).

There is a body of evidence linking PAI-1 expression to the processes of wound healing and epithelium repair (Providence et al., 2000; Xia et al., 2003; Providence et al., 2004). Enhanced PAI-1 expression in the gastric epithelium by acute infection with H. pylori (and growth factors associated with infection) suggests this protease inhibitor could initially act to protect gastric epithelial cells by decreasing fibrinolysis in areas of damaged mucosa. Also, Chapter 3 of this thesis highlights a role for PAI-1 in the control of epithelial cell proliferation by inhibiting the proliferative effects of uPA (also shown to increase the proliferation, migration and adhesion of gastric myofibroblats which were further increased by PAI-1 knock-down; Cédric Duval, PhD thesis, 2007). This signifies that increased PAI-1 not only acts to restrain uPA within the gastric epithelium, therefore controlling cellular proliferation, but it could also participate in epithelial-mesenchymal interactions helping to protect the gastric mucosa from processes which are thought to play an important role in altering the 'niche' that favours premalignant changes.

In premalignant states such as long-term infection with *H. pylori*, associated with chronic atrophic gastritis, increased PAI-1 may act in an oncogenic manner, accelerating epithelial remodelling contributing to the progression of gastric cancer. Correlating with clinical data of high PAI-1 expression in patients suffering from a variety of tumours (Andreason *et al.*, 2000; Foekens *et al.*, 2000; Rakic *et al.*, 2003), PAI-1 can modulate cellular migration and have a promigratory effect (Deng *et al.*, 1996; Stefansson *et al.*, 1996; Kjoller *et al.* 1997).

### **4.5 Conclusions**

- PAI-1-, uPA- and uPAR-luc exhibit cell restricted expression and are expressed in similar gastric gland cells as their endogenous proteins.
- 2. *H. pylori* stimulates expression of PAI-1-, uPA- and uPAR-luc in human primary gastric gland cells.
- uPA, but not uPAR or PAI-1 expression is abolished by CagE null *H. pylori* mutant.
- 4. Both epithelial and myofibroblast derived growth factors such as IGF-II, HGF and G17 stimulate PAI-1-luc expression in gastric epithelial cells.

Chapter 5

## **Contribution of PAI-1 to the**

## response of epithelial cell damage

## using uPA system transgenic mice

### **5.1 Introduction**

Haemostasis is a complex interplay between platelet aggregation, coagulation and fibrinolysis and is a vital defence system that rapidly stops haemorrhage after injury. The uPA system is fundamental to this process as it is responsible for the activation of plasminogen to plasmin (discussed in section 1.5), the protease that degrades fibrin, the main component of clots. The importance of PAI-1 in the stabilisation of the primary haemostatic plug formed after injury is not fully established, but it is known that PAI-1 is the primary inhibitor of plasminogen activators, and the release of PAI-1 by aggregating platelets (Erickson *et al.*, 1984; Kruithof *et al.*, 1986), may provide a means to prevent premature fibrinolysis, therefore contributing to the stabilisation of the haemostatic plug in the early stages of haemostasis.

Gastric ulcer is a multifaceted disease with a complex pluricasual etiology that is not fully understood. To illustrate the complexity of gastric ulcer pathogenesis several experimental models of gastric mucosal damage exist which mimic ulcer disease and have been well characterised. These include an ethanol-induced acute gastric mucosal injury model (Mozsik & Javor, 1988), and a stress-induced mucosal injury model (Cho & Ogle, 1992). Examination of the former indicates that gastric damage is produced as a consequence of many interacting factors and is representative of human gastric ulcer disease (Silen, 1988). A hypothesis used to account for gastric mucosal injury or resistance to injury is that of an interaction and balance between aggressive and defensive factors.

The gastric epithelium is exposed to various physical and chemical factors such as infection with *H. pylori*, which is regarded as an aggressive factor and is

associated with epithelial cell damage, inflammation of the gastric mucosa (gastritis) and duodenal ulcerogenesis (discussed in section 1.3.2; Tytgat & Rauws, 1990). Defensive mechanisms within the gastric mucosa are represented by classic factors such as mucus (Wallace & Whittle, 1986) and bicarbonate (Kivilaakso, 1981) secretions, prostaglandins (Konturek & Pawlik, 1986) and NO (discussed in section 1.2.6; MacNaughton *et al.*, 1989).

Gastric mucosal damage can be produced by direct application of ethanol where injury is limited to the gastric epithelium and involves mostly the inter-foveolar epithelium and gastric pits. However, deeper lesions involve intramucosal haemorrhage and vascular engorgement (Guth *et al.*, 1984). Due to damage, microvessels leak inflammatory mediators and vasoconstriction of submucosal arteries results in ischemia which eventually enhances the formation of more severe necrotic mucosal injury.

Ethanol-induced gastric mucosal lesions can be produced reliably and simply by intragastric administration of concentrated (50 - absolute) ethanol. Depending on the amount of ethanol given, 1-2 hr after administration between 10 and 40% of the glandular stomach of both rats and mice have been shown to display epithelial damage, characterised by mucosal edema, subepithelial haemorrhages, cellular exfoliation and inflammatory cell infiltration (Guslandi, 1987). Injury is attributed to congestion of the mucosa caused by constrictions of collecting venules (Ohno *et al.*, 1995; Ohno *et al.*, 1999; Saeki *et al.*, 2004). Time course studies have revealed that most gastric damage observed after ethanol administration occurs within 1-3 min of its instillation into the gut (Szabo, 1987).

Several studies have highlighted a tight correlation between the expression of uPA system components and wound re-epithelisation, with the majority of studies indicating PAI-1 as the major player (Pawar *et al*; 1995; Li *et al.*, 2000; Providence *et al.*, 2000; Providence *et al.*, 2002; Providence *et al.*, 2004). PAI-1 is an acute phase protein, can be induced during inflammation (Quax *et al.*, 1990; Swaisgood *et al.*, 2000; Luyendyk *et al.*, 2004; Lagoa *et al.*, 2005) and is increased in response to injury (Varela *et al.*, 2002). As well as its role in control of fibrinolysis, PAI-1 is also involved in the process of tissue repair in several different experimental settings (Providence *et al.*, 2000; Providence *&* Higgins 2004; Maquerlot *et al.*, 2006). The expression of PAI-1 in the gastric epithelium during mucosal injury has yet to be fully investigated although high levels of PAI-1 are found in duodenal ulcers (Herszenyi *et al.*, 1997).

#### 5.1.1 Aims

The aim of this chapter was to investigate the involvement of PAI-1 in short term gastric mucosal protection using an acute *in vivo* model of gastric mucosal injury. The specific objectives were to:

- develop transgenic mice with targeted over-expression of PAI-1 in the parietal cells of the stomach;
- study the role of PAI-1 in gastric mucosal protection by examining the degree and severity of epithelial cell damage after challenge in wild-type, PAI-1 KO and PAI-1 over-expressing mice;
- 3. demonstrate a possible uPA independent role of PAI-1 in gastric mucosal protection, using mice in which the uPAR receptor is deficient.

145

#### 5.2 Materials and methods

#### 5.2.1 Induction of haemorrhagic lesions within the gastric mucosa

All animals were housed as described in section 2.2. Haemorrhagic lesions were induced within the gastric mucosa of male or female C57BL/6, PAI-1-KO, uPAR-KO, PAI-1 over-expressing and uPA over-expressing mice (10-12 weeks) using an acute ethanol model of epithelial damage (as described in section 2.13). Both the number and length of haemorrhagic lesions per animal was recorded using callipers and a small ruler. Individual mouse stomachs were also graded as possessing overall either 'weak', 'moderate' or 'severe' lesions and photographed using a Canon Digital IXUS iZoom camera with 5.0 mega pixels and a 2.4 x zoom lens. Lesion length per animal was determined by totalling both the number and length of each individual lesion. Average lesion length was calculated per group of animals.

#### 5.2.2 Development and production of PAI-1 over-expressing mice

#### 5.2.2.1 Expression of PAI-1

AGS2 cells were transfected with either pcDNA3 or pcDNA3-PAI-1 plasmids (as described in section 2.5.2) and left to express for 48 hr. Protein was extracted and western blot analysis carried out (as described in section 2.9). A mouse-anti-PAI-1 antibody (American Diagnostica) was used at a concentration of 1:100 and blots were re-probed with a goat-anti- $\beta$ -actin antibody used at a concentration of 1:5000 (Santa Cruz Biotechnology).

#### 5.2.2.2 Transfection of murine gastric gland cells

Murine gastric gland cells (isolated as in section 2.4.3) were transfected ( $3\mu g$  total DNA) with an expression plasmid in which murine the PAI-1 cDNA sequence had been coupled to the  $\beta$ -subunit of the H<sup>+</sup>, K<sup>+</sup>-ATPase promoter (HKB-PAI-1; as described in section 5.5.4 and plasmid map shown in Figure 5.6-A) using the same protocol as for human gastric gland transfection (as described in section 2.5.2). Cells were fixed 48 hr after transfection in 4% PFA. Dual-immunolabelling was carried out (as described in section 2.7) using a rabbit-anti-PAI-1 antibody at a concentration of 1:200 (Santa Cruz Biotechnology) and a goat-anti-ezrin (C-15) antibody at a concentration of 1:100 (Santa Cruz Biotechnology). Control, un-transfected murine gastric gland cells were also dual-immunolabelled with the above antibodies. More than 20 control and transfected glands were scored for the percentage of cells expressing, PAI-1, ezrin, or both. Results are displayed as percentage relative increase compared to control glands (taken to be 100%).

#### 5.3 Results

## 5.3.1 PAI-1 KO mice develop more severe haemorrhagic lesions than wildtype or uPAR KO mice

To investigate the contribution of PAI-1 in the process of epithelial cell damage, an experimental *in vivo* model of gastric mucosal injury was carried out. Ethanolinduced gastric mucosal lesions effect epithelial cells within the stomach mucosa and can be produced reliably by intragastric administration of varying ethanol concentrations (50% and absolute ethanol). This model has previously been shown to induce gastric ulcers or what are termed 'haemorrhagic lesions' within the stomach of both rats and mice (Szabo *et al.*, 1987; Mozsik *et al.*, 1988; Pan *et al.*, 2005).

Commercially available PAI-1 KO mice, uPAR KO mice and wild-type C57BL/6 mice all showed the presence of haemorrhagic lesions within the gastric mucosa after challenge with 50% or absolute ethanol, although to varying extents (Figure 5.1). Untreated control animals showed the appearance of normal, unaltered tissue. However, 50% ethanol induced several small, pinpoint like haemorrhagic lesions termed petichae in C57BL/6 mice and slightly longer, thicker and more abundant lesions were seen in uPAR KO mice. The length, thickness and number of haemorrhagic lesions increased further in PAI-1 KO mice.

Wild-type mice developed more severe lesions when challenged with absolute ethanol compared to 50% ethanol, as did both uPAR KO and PAI-1 KO mice. However, the severity of the lesions formed in PAI-1 KO mice after absolute ethanol treatment were more severe than those seen in uPAR KO or C57BL/6

148

mice. These lesions were very thick, long and aggressive often covering the majority of the surface area of the mucosa.



#### Figure 5.1

## PAI-1 KO mice develop more severe haemorrhagic lesions than wild-type or uPAR KO mice after ethanol administration.

Representative photographs of haemorrhagic lesions formed inside the stomachs of C57BL/6, uPAR KO and PAI-1 KO mice after 1 hr treatment with 50% or absolute ethanol. A-C, C57BL/6 untreated control, 50% ethanol and absolute ethanol. D-F, uPAR untreated control, 50% ethanol and absolute ethanol. G-I, PAI-1 KO untreated control, 50% ethanol and absolute ethanol. G-I,

## 5.3.2 PAI-1 KO mice display a more severe phenotype when challenged with ethanol that uPAR KO or wild-type mice

To quantify the extent of epithelial cell damage caused after ethanol administration, the gastric mucosa of each animal was graded as having the formation of 'weak', 'moderate' or 'severe' haemorrhagic lesions, depending on the number, length, aggressiveness and the area of the mucosa covered by lesions. The percentage of animals in each group with weak, moderate or severe lesions was recorded (Figure 5.2).

After challenge with 50% ethanol, 100% of C57BL/6 mice were graded as having weak lesions and no animals in this group displayed moderate or severe lesions. Weak lesions were developed by 72% of uPAR KO mice and the remaining 28% developed moderate lesions. Only 17% of PAI-1 KO animals challenged with 50% ethanol displayed weak lesions with a massive 83% graded as possessing moderate haemorrhagic lesions.

With absolute ethanol treatment fewer C57BL/6 mice developed weak lesions compared to with 50% ethanol, as more animals developed moderate lesions. However, only 10% of wild-type mice developed severe lesions after absolute ethanol treatment compared to 25% of uPAR KO mice and 55% of PAI-1 KO mice. This data shows that PAI-1 KO mice develop a more severe phenotype and display a greater extent of tissue damage after challenge than uPAR KO or wild-type mice.



## PAI-1 KO mice have a more severe phenotype when challenged with ethanol than uPAR KO or wild type mice.

Percentage of wild-type (C57BL/6), uPAR KO and PAI-1 KO mice graded as developing weak, moderate or severe haemorrhagic lesions after treatment with either 50% or absolute ethanol. There are no severe lesions after 50% ethanol insult in any strain of mouse, and no moderate lesions in C57BL/6 mice.

# 5.3.3 PAI-1 KO mice develop more severe haemorrhagic lesions with an overall greater lesion length than wild-type or uPAR KO mice

As another measure of quantification of tissue damage after ethanol challenge, both the number and length of lesions formed within the gastric mucosa of each animal was measured as accurately as possible and each animal given a lesion score. This corresponded to the total length of each individual lesion. The average lesion score for each group of animals (C57BL/6, uPAR KO and PAI-1 KO) was recorded after 50% and absolute ethanol treatment. When challenged with 50% ethanol (Figure 5.3), PAI-KO mice had a significantly greater average lesion length score  $(34.25 \pm 4.74 \text{mm})$  compared to uPAR KO  $(13.86 \pm 2.63 \text{mm})$ or C57BL/6 mice (7.64 + 0.91 mm). All three groups of animals displayed a greater lesion score with absolute ethanol than with 50% ethanol treatment, but the lesion score for PA I-1 KO mice remained signific antly higher (51.91  $\pm$ 5.11mm) than either uPAR KO ( $39.38 \pm 5.37$ mm) or wild-type mice ( $33.42 \pm$ 2.90mm). Although there was a slight difference in lesion length between uPAR KO and C57BL/6 mice with both concentrations of ethanol (significant only with 50% ethanol), it was not as substantial as the difference in lesion length between PAI-1 KO and wild-type mice. This data highlights that knocking out PAI-1 gives rise to more severe gastric mucosal injury after ethanol challenge.



# PAI-1 KO mice have a greater average lesion length after challenge with 50% or absolute ethanol compared to uPAR KO or wild-type C57BL/6 mice.

The lengths and number of lesions per animal was scored and summed up to give a lesion score per animal. Average lesion scores per group of animals was then calculated after challenge with 50% and absolute ethanol for C57BL/6, uPAR KO and PAI-KO mice (Mean  $\pm$  S.E, n=7-12 animals, \* p>0.05, ANOVA).

#### 5.3.4 Generation of PAI-1 over-expressing transgenic mice

To investigate the contribution of elevated PAI-1 to the process of gastric mucosal protection after damage, and to compare this to results obtained from PAI-1 KO mice, over-expressing PAI-1 transgenic mice were made in collaboration with Dr Nikolina Vlatkovic, The University of Liverpool. In addition, uPA over-expressing mice were also developed simultaneously using the same strategy. It is important to note that these animals have over-expression of the given gene directly targeted to the stomach. More specifically, over-expression was targeted to a particular epithelial cell type within the stomach, the acid secreting parietal cell.

Parietal cells are one of the main differentiated cell type within the gastric corpus and in mice they undergo bilateral migration to be distributed throughout the gland (Karam, 1993e). The H<sup>+</sup>, K<sup>+</sup>-ATPase is the primary gastric proton pump and is situated on the apical membrane of a stimulated parietal (Forte & Yao, 1996). It is the product of 2 genes, *Atp4a* encodes the  $\alpha$ -subunit and *Atp4b* encodes the  $\beta$ -subunit (Nguyen *et al.*, 2004). The murine  $\beta$ -subunit of the H<sup>+</sup>, K<sup>+</sup>-ATPase promoter sequence was used to drive expression of either PAI-1 or uPA to parietal cells. The  $\beta$ -subunit of the H<sup>+</sup>, K<sup>+</sup>-ATPase promoter has previously been used to successfully drive over-expression of other genes in the stomach of transgenic mice (Lorenz & Gordon, 1993), and has been used to direct expression of an attenuated diphtheria toxin or the herpes simplex virus 1 thymidine kinase to parietal cells (Canfield *et al* 1996; Li *et al.*, 1996). Parietal cells were chosen as a direct target as these are a major gastric epithelial cell type and are also the cell type in which up-regulation of uPA system members (including PAI-1) by *H. pylori* occurred (as shown in Figure 3.2). There were several stages involved in the making of both transgenic animals. Due to the precision needed to specifically over-express the genes in the parietal cells of the stomach, as well as the cost involved in the project, it was important to ensure the correct expression constructs were used and that these were functionally sound. The molecular biology and production of constructs used in the making of these animals was performed by Professor Rod Dimaline, The University of Liverpool. The three different stages of plasmid generation are briefly outlined in Figure 5.4.

Initially, the  $\beta$ -subunit of the mouse H<sup>+</sup>, K<sup>+</sup>-ATPase promoter sequence was inserted into the pEGFP vector and was used to drive the expression of green fluorescent protein (GFP; Figure 5.4-A). This was transfected into primary human gastric glands to ensure the promoter sequence was correct and that it directed GFP expression into parietal cells (Dr Islay Steele, personal communication).

Next, the cDNA sequence of murine PAI-1 was tested by inserting it into a pcDNA3 vector (pcDNA3-PAI-1) which uses a strong viral CMV promoter to drive gene expression (Figure 5.4-B). This expression plasmid or pcDNA3 as an empty vector were transfected into AGS2 cells. After 48 hr cell protein was extracted and western blot carried out. The directional orientation and sequence of PAI-1 cDNA was found to be correct as there was increased PAI-1 protein expression in cells transfected with pcDNA3-PAI-1 compared to when the empty pcDNA3 vector was used alone (Figure 5.5).

The final stage of plasmid production involved the removal of GFP from the original pEGFP vector, and the insertion of the murine PAI-1 cDNA sequence

155

(Figure 5.4-C). This resulted in an expression plasmids were PAI-1 was coupled to the  $\beta$ -subunit of the H<sup>+</sup>, K<sup>+</sup>-ATPase promoter (HKB-PAI-1). As a final precaution to verify plasmid expression in mouse parietal cells, HKB-PAI-1 was transfected into primary mouse gastric glands which were dual-immunolabelled with an anti-PAI-1 antibody and Ezrin (a marker of parietal cells). Gastric glands transfected with HKB-PAI-1 had a higher number of cells expressing PAI-1 and the percentage of parietal cells expressing PAI-1 was approximately 26% greater compared to control, un-transfected glands (Figure 5.7). HKB-PAI-1 transfected glands also displayed a stronger PAI-1 staining compared to control glands.

The detailed maps of HKB-PAI-1 and HKB-uPA are shown in Figure 5.6. The PAI-1 transgene was cut out from HKB-PAI-1 using the restriction enzymes *Hind III* and *Afl II* to yield a fragment of approximately 3.3kb containing a polyA sequence. The uPA transgene was yielded from HKB-uPA in a triple digest using the restriction enzymes *Afe I*, *Af II* and *Cla I* to give a fragment of approximately 3.3kb also containing a polyA sequence. Both transgenes were purified by agarose gel electrophoresis and were injected into pseudopregnant female C57BL/6 mice by pronuclear injection.

The overview described here for the generation of PAI-1/uPA over-expressing mice only outlines the stages prior to the actually establishment and breeding of true homozygous (+/+) PAI-1 and uPA over-expressing mice. Many stages of genotyping from DNA obtained from mouse tail biopsies were required and was performed by Cédric Duval, The University of Liverpool.

156



Schematic diagram of the stages and various expression plasmids involved in the generation of PAI-1 over-expressing mice.



#### Increased PAI-1 expression in epithelial cells transfected with pcDNA3-PAI-1 expression plasmid.

Western blot showing the expression of PAI-1 (top panel) in AGS2 cells 48 hr after being transfected with either pcDNA3 empty vector or pcDNA3 vector in which the murine PAI-1 cDNA sequence had been inserted (pcDNA3-PAI-1).  $\beta$ -actin (bottom panel) shows protein loading controls.



## Map of expression plasmids used to over-express PAI-1 or uPA in murine gastric parietal cells.

A. HKB-PAI-1; Expression plasmid used to generate PAI-1 over-expressing transgenic mice. The promoter sequence of the β-subunit of the H<sup>+</sup>, K<sup>+</sup>-ATPase proton pump was used to drive PAI-1 gene expression in parietal cells. B. HKB-uPA; Expression plasmid used to drive uPA gene expression in murine parietal cells using the promoter sequence of the β-subunit of the H<sup>+</sup>, K<sup>+</sup>-ATPase proton pump.



## A higher percentage of parietal cells express PAI-1 in murine gastric glands transfected with HKB-PAI-1.

Dual-immunolabelled murine gastric gland cells transfected with HKB-PAI-1 showed that a higher percentage of parietal cells express PAI-1 compared to untransfected control glands (Mean  $\pm$  S.E, n=20 glands. \* p>0.05 ANOVA).
# 5.3.5 PAI-1 over-expressing mice develop less severe haemorrhagic lesions after ethanol challenge than PAI-1 KO mice

An acute model of gastric mucosal injury using ethanol administration was carried out on PAI-1 over-expressing mice (generated as described in section 5.5.4). This experiment had previously been carried out on PAI-1 KO mice which showed a more severe phenotype after challenge than wild-type mice (Figure 5.1). A comparison between PAI-1 KO and PAI-1 over-expressing mice was made to allow a thorough investigation into the involvement of PAI-1 in the process of tissue damage and gastric mucosal protection.

The over-expressing PAI-1 mice used in this experiment were from one founder and had 2-4 copies of the PAI-1 transgene at one integration site. After absolute ethanol administration, PAI-1 over-expressing mice had a smaller average lesion length score ( $20.25 \pm 5.02$ mm) compared to C57BL/6 mice ( $30.17 \pm 1.90$ mm; Figure 5.8). The average lesion score obtained for PAI-1 KO mice (Figure 5.3), was significantly greater ( $51.91 \pm 5.11$ mm) than that obtained for PAI-1 overexpressing mice.



### Figure 5.8

## PAI-1 over-expressing mice show a less severe phenotype when challenged with ethanol compared to wild-type mice.

The lengths and number of lesions per animal was scored and totalled to give a lesion score per animal. Average lesion scores per group of animals (C57BL/6 and PAI-1 over-expressing mice) were then calculated after challenge with absolute ethanol. (Mean  $\pm$  S.E, n=6-12 animals, \* p>0.05, ANOVA).

## 5.3.6 uPA over-expressing mice develop a more severe phenotype when challenged with ethanol compared to wild-type mice

An acute model of gastric epithelial damage was carried out on uPA overexpressing mice (generated using the same strategy as described for PAI-1 overexpressing mice in section 5.5.4) to establish if uPA has any significant role to play in the process of gastric mucosal protection after tissue injury. The overexpressing animals were from one founder and had 10-20 copies of the uPA transgene at one integration site. Both the number and length of lesions formed within the gastric mucosa of each animal was measured as accurately as possible after 50% ethanol administration and each animal given a lesion score. This corresponded to the total length of each individual lesion. The average lesion score for both group of animals, C57BL/6 and uPA over-expressing was calculated. After challenge, uPA over-expressing mice had a significantly greater average lesion length score compared to C57BL/6 mice (Figure 5.9).



### Figure 5.9

## uPA over-expressing mice have a greater average lesion length after challenge with ethanol compared to wild-type mice.

The length and number of lesions per mouse was scored and summed up to give a lesion score per animal. Average lesion scores per group of mice (C57BL/6 and uPA over-expressing mice) were then calculated after challenge with 50% ethanol (Mean  $\pm$  S.E, n=5 animals, \* p>0.05, ANOVA).

## 5.4 Discussion

The results presented in this Chapter show that members of the uPA system have a possible involvement in the process of gastric mucosal protection after acute epithelial cell damage. In particular, it has been demonstrated through the use of transgenic animals that PAI-1 has a protective role to play in the gastric epithelium during acute tissue injury, as knocking out this gene gives rise to a more severe phenotype compared to wild-type or uPAR KO mice after ethanol administration. Not only is it evident macroscopically that the gastric mucosa of PAI-1 KO mice is dramatically different after challenge with 50% and absolute ethanol (Figure 5.1) compared to uPAR KO and wild-type mice, but careful analysis of the formation of haemorrhagic lesions confirmed this result (Figure 5.2 and Figure 5.3).

However, over-expression of PAI-1 seems to provide protection to the gastric mucosa after the epithelium is damaged, as the extent of tissue damage caused after ethanol administration resulted in the development of a less severe phenotype compared to PAI-1 KO mice. Interestingly, the effect of ethanol administration seen in PAI-1 KO mice was reversed in PAI-1 over-expressing mice which developed a phenotype less severe than wild-type mice (Figure 5.8).

In addition, it is likely that the observed effects of ethanol administration in these transgenic mice are dependent on the actions of PAI-1 in the control of uPA. Comparison of haemorrhagic lesions formed between uPAR KO and wild-type mice revealed very little difference with absolute ethanol. When uPAR is deleted, the action of uPA in fibrinolysis is inhibited as it is unable to bind to its receptor and ultimately plasmin is unavailable to degrade fibrin. However, when

PAI-1 is deleted, it can no longer inhibit uPA, and as a result a significant effect on the degree of severity of lesions formed in PAI-1 KO mice is seen. This could possibly be a consequence of plasmin over-production and degradation of fibrin within the fibrin plug. The importance of PAI-1 in the stabilisation of the haemostatic plug has previously been demonstrated in a 36 year old patient who showed a lifelong bleeding disorder which was associated with a deficiency of PAI-1 and the inability to form fibrin clots (Dieval *et al.*, 1991).

This is also reflected in uPA over-expressing mice which show more severe haemorrhagic lesions than wild-type mice, again possibly caused by the inability of these animals to form a fibrin plug at the site of injury. Over-expression of PAI-1 reversed these effects as uPA is now inhibited, fibrin is not degraded thus haemorrhagic lesion development is reduced. It can be concluded that in part, the protective effects of PAI-1 in an ethanol induced model of acute gastric epithelial injury are dependent on not only a vital balance between protease and protease inhibitor, but the presence of uPAR and ultimately the down stream effects of uPA binding to its receptor are necessary.

In addition to the pivotal role of PAI-1 in haemostasis, the possibility of well established agents which are involved in gastric mucosal protection (such as those discussed in section 1.2.6) mediating PAI-1 production could also help explain the protective effect of PAI-1 in the gastric epithelial after damage. However, this has proved difficult to evaluate due to a current lack of literature. For example, the role of the potent vasodilator NO on PAI-1 production is still under debate as some authors have found that NO suppresses PAI-1 expression and synthesis in vascular smooth muscle cells (Wolfsgruber *et al.*, 2003), and

reduces PAI-1 release from platelets (Numano *et al.*, 1995). Instead, other researchers found a dual effect of NO on PAI-1 expression in endothelial cells (Swiatkowska *et al.*, 2000). In particular, NO endogenously generated by NOS can potentiate PAI-1 expression, while NO exogenously released from NO donors inhibits its generation. However, no *in vivo* data are available to confirm these findings, leaving the question of whether NO up- or down-regulates PAI-1 synthesis unconfirmed.

Cyclooxygenase is a rate-limiting enzyme for PG synthesis which plays a crucial role in protecting the gastric mucosa against injury. In the gastric mucosa, COX-1 is constitutively expressed in normal tissues, whereas there is a low but significant expression of COX-2 (Jackson *et al.*, 2000) which is induced under pathogenic conditions, such as inflammation (Vane *et al.*, 1994). The induction of PAI-2 expression in gastric epithelial cells by *H. pylori* has been shown to be mediated by COX-2 activation (Varro *et al.*, 2004). Anti-inflammatory drugs such as NSAIDs are widely used to inhibit the action of the COX pathway (Vane *et al.*, 1998). Cyclooxygenase-2 could be a possible mediator of PAI-1 expression within the gastric epithelium since six different NSAIDs have been shown to down-regulate the expression of COX enzymes and also PAI-1 (Yang *et al.*, 2008).

The ethanol induced haemorrhagic lesion model used was chosen as it acutely induces epithelial damage. However, the gastric mucosa is frequently exposed to a wide range of other aggressive insults, and numerous stimuli such as oxidative stress, mechanical trauma and inflammation which all induce injury. Superficial damage heals rapidly via the migration of epithelial cells recruited from the

adjacent mucosa onto the denuded basement membrane through a process known as restitution (Lacy & Ito, 1984; Silen, 1985; Lacy *et al.*, 1993). Gastric myofibroblasts can also migrate through the basement membrane to aid this process (Wu *et al.*, 1999). However, deeper damage such as the formation of erosions and ulcers within the gastric mucosa (which are often a result of chronic *H. pylori* infection) requires a complex sequence of events for re-epithelialisation to occur (Miyake *et al.*, 1980).

Restoration of epithelial integrity and remodelling of the ECM after injury is an essential feature of wound repair and for it to be successful the net result of the expression of proteases and inhibitors must allow for the formation and deposition of new ECM proteins, while at the same time permitting cell migration and remodelling (Wysocki *et al.*, 1999). There is a well documented role for PAI-1 in influencing cell migration (discussed in section 1.5.5.1) and in the process of tissue repair in several different experimental settings. PAI-1 has been found to be an important protein in epithelial alveolar cell wound healing (Maquerlot *et al.*, 2006), is required for epidermal wound repair (Providence & Higgins, 2004), and also participates in renal epithelial injury repair (Providence *et al.*, 2000). PAI-1 may also act by influencing restoration of epithelial integrity in the gastric epithelium during injury and repair.

### **5.5 Conclusions**

 PAI-1 KO mice develop a more severe phenotype after an acute model of gastric epithelial mucosal injury induced by ethanol administration compared to uPAR KO or wild-type C57BL/6 mice.

- PAI-1 over-expressing mice develop a less severe phenotype than PAI-1 KO mice and wild-type mice after an acute model of gastric epithelial mucosal injury induced by ethanol administration.
- 3. uPA over-expressing mice develop a more severe phenotype after induction of gastric epithelial cell damage than wild-type mice.

**Chapter 6** 

**Conclusions and Future Work** 

## 6.1 Major Findings

The main findings of this thesis are: PAI-1 mRNA abundance and protein expression are increased in *Helicobacter* infection and gastric cancer where expression is predominantly epithelial; Both *H. pylori* and uPA increase gastric epithelial cell proliferation which is further increased after PAI-1 knock-down; Expression of endogenous PAI-1, uPAR and uPA proteins show cell restricted expression in human primary gastric glands as do their respective luciferase reporter constructs; *H. pylori* stimulates expression of PAI-1-, uPAR- and uPA-luc with the later requiring the CagE virulence factor; PAI-1 KO mice and uPA over-expressing mice develop a more severe phenotype after acute gastric mucosal damage compared to wild-type mice.

Previous data has shown *H. pylori* infection of the human gastric corpus to be associated with increased expression of uPA and its receptor, uPAR (Cédric Duval, PhD Thesis, 2007). In addition, the present data show that *H. pylori* infection of the human gastric corpus mucosa is also associated with increased expression of the main extracellular inhibitor of uPA, PAI-1. Expression of PAI-1 was also found to be increased in gastric cancer patients, and in a mouse model leading to gastric cancer. The expression of PAI-1 in *H. pylori* infected patients appears to be predominantly epithelial. Within the epithelium, both parietal and ECL cells have the capacity to increase uPA, uPAR and PAI-1 expression in response to *H. pylori*, while pepsinogen containing cells can also express uPA and surface epithelial cells may express uPAR. Moreover, the CagE virulence factor is required for uPA responses to *H. pylori* but not PAI-1 or uPAR, therefore it seems probable that separate mechanisms are associated with increased expression of uPA, uPAR and PAI-1 during infection. Both *H. pylori* 

and uPA stimulate epithelial cell proliferation which is at least partly held in check by PAI-1. To elucidate the role of the uPA system in the initial response of gastric epithelial cells after insult, an acute model of ethanol induced epithelial injury was used in uPA system transgenic mice. Expression of PAI-1 was found to protect the gastric epithelium from the damaging effects of ethanol, while uPA expression gave rise to the development of a more severe phenotype compared to wild-type mice.

The data implicate increased expression of PAI-1 during both acute and chronic *H. pylori* infection to be part of a defensive host response; it is vital for the restraint of uPA in allowing the initial formation of a haemostatic fibrin plug, thus helping to protect and repair the gastric epithelium after damage. In addition to its role during initial infection, PAI-1 also acts to control the activation and proliferative effects of uPA on gastric epithelial cells which could be associated with the preneoplastic hyperproliferative condition seen during *H. pylori* infection. However, in chronic *H. pylori* infection, associated with gastric cancer development, a sustained increase in PAI-1 expression could have a pathophysiological role explaining its correlation with a poor prognosis. The control of uPA system activity may therefore present a new target in arresting *H. pylori* stimulated progression to cancer.

## 6.2 Conceptual advances and future work

Previous work has established that uPA, uPAR and PAI-1 are increased in gastric cancer and in chronic atrophic gastritis (Nekarda *et al.*, 1994; Ito *et al.*, 1996; Plebani *et al.*, 1997; Kawasaki *et al.*, 1998; Kaneko *et al.*, 2003), and that increased expression in cancer is associated with an adverse outcome. Expression

of uPA, uPAR and PAI-1 occurs in both stromal and tumour cells, and it is generally thought that the expression of these proteins is associated with progression of the disease via increased cell invasion, metastasis, and angiogenesis (Ito et al., 1996; Johnsen et al., 1998; Peper, 2001). In the case of uPA, binding to uPAR generates a focal point of proteolytic activity that facilitates cell invasion (discussed in section 1.5.2.1; Heiss et al., 1995). In addition to inhibition of uPA, through accelerated internalisation of the uPA/uPAR complex, PAI-1 also exhibits uPA-independent actions. Specifically, by binding to VN it can increase cell dissociation from the ECM and also accelerate invasion (discussed in section 1.5.5.1; Deng et al., 1996). It has been suggested that the latter effects account for the association of increased PAI-1 in gastric and other human cancers with an adverse outcome. This is often referred to as "paradoxical" given that uPA is also associated with disease progression (discussed in section 1.5.5.2). In spite of the interest in uPA and PAI-1 in gastric cancer, little attention has been given to how this system might be implicated in the response of the gastric epithelium to insult and in preneoplastic conditions.

It has previously been shown that PAI-2 is increased in gastric epithelial cells in response to *H. pylori* via activation of NF $\kappa$ B (Varro *et al.*, 2004). A pool of PAI-2 is retained within cells and is associated with inhibition of apoptosis (Kumar *et al.*, 1991), while some is secreted and has been shown to be involved in inhibition of invasion of a cancer cell line (Andreason *et al.*, 1997). The expression of uPA and uPAR have also been shown to be increased in *H. pylori* infection (Cédric Duval, PhD Thesis, 2007) and the present data now make it clear that acute infection with *H. pylori* also induces another main member of the uPA system, the main extracellular inhibitor of uPA, PAI-1. No evidence is

available that a third inhibitor, PAI-3, is changed with *H. pylori* infection and it was not studied further.

Previous work using microarrays to identify H. pylori regulated genes, demonstrated increased uPA in gastric cancer cell lines (Kitadai et al., 2003), and there has been work using gastric cancer cell lines that indicates H. pylori stimulation of uPAR expression (Kim et al., 2005; Kim et al., 2007). Some work has also been carried out in cell lines regarding the effect of H. pylori on PAI-1 expression (Iwamoto et al., 2005). However, given the nature of cancer cell lines, it remained uncertain how these observations related to expression of members of the uPA system in primary gastric epithelial cells. Data presented here show that with acute infection there is the capacity to increase uPA, uPAR and PAI-1 expression in two cell types associated with the gastric corpus namely the acidsecreting parietal cell and the histamine releasing enterochromaffin-like cell; the restriction of both cell types to the gastric corpus is consistent with the observation that there was no difference in expression of PAI-1 (or uPA/uPAR; Cédric Duval, PhD Thesis, 2007) in the antral region of the stomach (where these cells types are absent). In contrast, PAI-2 is prominently expressed in mucin and pepsinogen-expressing cells and is only occasionally found in parietal or ECL cells (Varro et al., 2004); it is also increased in the antrum in H. pylori positive patients (personal communication, Andrea Varro; unpublished observation).

*H. pylori* increases plasma gastrin concentrations, and parietal and ECL cells express the CCK2R at which gastrin acts (discussed in section 1.2.3 and 1.3.5; Levi *et al.*, 1989; Dockray *et al.*, 2001). However, increased abundance of PAI-1 transcripts were found in *H. pylori* infected patients with plasma gastrin concentrations in the normal range and *H. pylori* stimulation of promoter-

reporter luciferase constructs occurred in primary epithelial cells in vitro (in the absence of gastrin), therefore the effects of H. pylori are not secondary to increased gastrin. However, gastrin was found to be able to stimulate expression of PAI-l-luc in primary epithelial cells in vitro, and uPA-, uPAR-, and PAI-1-luc in a gastric cancer epithelial cell line. While stimulation of uPAR and PAI-1 was independent of the pathogenicity island, a *cagE* isogenic mutant was defective in uPA induction. Disruption of cagE means that H. pylori cannot produce a functional type IV secretory apparatus. This has two effects as it blocks Nod1-NF $\kappa$ B signalling (discussed in section 1.3.3.2) and also blocks delivery of CagA into epithelial cells, which ultimately has consequences on cell signalling events (discussed in section 1.3.3.3). As disruption of cagA had no effect on uPA induction, it is likely that the lack of induction of uPA by the *cagE* mutant is due to lack of stimulation of Nod1-NFkB. There is an already described role of  $NF\kappa B$  in the regulation of uPA transcription and it is known that the human uPA promotor contains an NF $\kappa$ B binding site (discussed in section 1.5.1; Novak *et al.*, 1991; Hansen et al., 1992; Mitsuno et al., 2001; Guillemin et al., 2002). There are many different ways that bacteria are known to interact with the uPA system (Bergmann & Hammerschmidt, 2007). In the case of H. pylori, two plasminogen binding proteins have been identified that are proposed to coat the organism and with the subsequent activation of bound plasminogen to plasmin these proteins are thought to increase virulence (Jonsson et al., 2004).

Both *H. pylori* and exogenous uPA stimulate proliferation of primary gastric epithelial cells. These data link *H. pylori* induction of the uPA system to one of the key characteristics of preneoplastic, *H. pylori*-infected epithelium, namely increased cell proliferation which was enhanced further by blocking PAI-1

expression, compatible with a dynamic equilibrium between enzyme and its inhibitor. It is unknown at this stage if the proliferative response of gastric epithelial cells to *H. pylori* is mediated as a direct consequence of increased uPA expression, or indeed how the response of gastric epithelial cells to uPA is mediated. The involvement of uPA binding to uPAR in this process also warrants further investigation and all three questions can be addressed by carrying out proliferation assays on primary gastric epithelial cells treated with various inhibitors, anti-sense oligonuclecotides and neutralistation antibodies.

Previous work has shown that *H. pylori* stimulates proliferation of epithelial cells and that this includes transactivation of the EGF-R and stimulation of HB-EGF cleavage (Keates *et al.*, 2001; Wallasch *et al.*, 2002; Ashktorab *et al.*, 2007). The main ligands of EGF-R produced in the gastric epithelium, are HB-EGF, amphiregulin and TGF- $\alpha$ . It is known that HB-EGF is cleaved from its cell bound precursor via proteases that include TACE (ADAM-17; Sunnarborg *et al.*, 2002). It has also been shown in prostatic DU-145 cells that uPA stimulated invasion is mediated by HB-EGF in an autocrine mechanism (Caceres *et al.*, 2005). Epidermal growth factor receptor is known to coimmunopreciptate with uPAR and can be constitutively activated by ligand bound uPAR, activating ERK to promote proliferation and migration of a variety of cell types (Liu *et al.*, 2002; Jo *et al.*, 2003; Jo *et al.*, 2005). Whether the effects of uPA on gastric epithelial cell proliferation involves EGF-R, or if any of the ligands for EGF-R are required for this response remains to be determined but is an interesting avenue to explore.

The primary function of uPA is the cleavage of plasminogen to plasmin and hence control of fibrin degradation. It is possible that with acute *H. pylori* 

infection, the induction of PAI-1 is part of a defense mechanism initiated to control the protease activity of increased uPA, therefore preventing premature fibrin lysis in order for a stable haemostatic fibrin plug to be formed. Fibrinolysis is a reparative process that occurs in response to a haemostatic plug (Loskutoff *et al.*, 1993). The involvement of PAI-1 in this process is well established (Dieval *et al.*, 1991), and studies have also shown that PAI-1 can bind to the fibrin clot either directly or via VN (Murayama *et al.*, 1987; Wagner *et al.*, 1989). However, the fact that PAI-1 protects the gastric epithelium from the formation of haemorrhagic lesions after ethanol administration is novel, and is therefore difficult to discuss in the context of other literature.

The fibrin clot formed during haemostasis must be remodelled and removed during tissue repair to restore normal structure and function. The fibrinolytic system is the principle effecter of clot removal as it controls the enzymatic degradation of fibrin and is therefore important for tissue repair. It has been shown here that PAI-1 expression is associated with gastric mucosal protection; however, its role in gastric epithelium repair after acute injury has yet to be determined. To explore this, a cryoinjury ulcer model (Ashurst et al., 2008) could be used comparing the repair process of the gastric mucosa between wild-type, PAI-1 KO and PAI-1 over-expressing mice. It has been previously shown that wild-type, C57BL/6 mice show a progressive repair of the ulcer crater at 6 and 9 days postinjury, and by 13 days the mucosal ulceration is barely discernible macroscopically (Ashurst et al., 2008). As PAI-1 has an already well documented role in tissue repair in several experimental settings (discussed in section 5.4; Providence et al., 2000; Providence & Higgins, 2004; Maquerlot et al., 2006), it could be hypothesised that the repair process would be retarded in PAI-1 KO mice, but accelerated in PAI-1 over-expressing mice.

So far it has been highlighted that PAI-1 expression is not only pivotal in the control of fibrinolysis during initial gastric mucosal injury but increased expression is required to restrain the proliferative effects of H. pylori and uPA during infection. However, aside from the uPA-independent, 'paradoxical' actions of PAI-1 (described above and discussed in section 1.5.5.1) which accounts for the pathophysiological association of increased PAI-1 in gastric and other human cancers with an adverse outcome, sustained, increased antifibrinolytic concentrations of PAI-1 could contribute to fibrosis, a process intimately tied to chronic infection. Analysis of PAI-1 KO mice have offered an insight into the relative contributions of PAI-1 in fibrosis; in a mouse model of cholestatic liver injury, PAI-1 KO mice are robustly protected against hepatic fibrosis (Bergheim et al., 2006) and mice genetically deficient in PAI-1 are protected from developing fibrosis after lung injury (Eitzman et al., 1996; Hattori et al., 2000; Chuang-Tsai et al., 2003) whilst fibrosis is worsened in mice that have a constitutively active PAI-1 transgene (Eitzman et al., 1996). PAI-1 KO mice also have attenuated interstitial fibrosis after unilateral ureteral obstruction compared to wild-type mice (Oda et al., 2001).

The role of PAI-1 in gastric remodelling and cancer progression can be further studied in *H. felis* or *H. pylori* infected PAI-1 KO and PAI-1 over-expressing mice and comparing the extent of tissue remodelling to that observed in wild-type, C57BL/6 mice. *H. felis* infection in the C57BL/6 mouse model reproducibly results in the classic sequence of histological changes seen in human infection (discussed in section 1.3.2) with adenocarcinoma occurring in 100% of mice by 15 months of infection (Wang *et al.*, 1998; Houghton *et al.*, 2004; Rogers *et al.*, 2004; Cai *et al.*, 2005). As there is an association between high PAI-1 expression and a poor outcome for gastric cancer patients, it could be

suggested that PAI-1 KO mice display a less severe phenotype after *H. felis* infection with PAI-1 over-expressing mice developing a more aggressive phenotype.

The present data suggest a role for PAI-1 in gastric mucosal protection during acute injury which could account for the initial increase in expression during acute *H. pylori* infection. However, in prolonged infection increased PAI-1 expression acts to restrain the proliferative effects of increased uPA expression, helping to inhibit hyperproliferation of gastric epithelial cells. During chronic infection, associated with gastric cancer development, it is suggested that a sustained increase in PAI-1 expression has a pathophysiological role by contributing to fibrosis and by acting independently of uPA to influence cell migration and invasion. In addition to this it can be suggested that increased proliferation favours the acquisition of mutations leading to cancer. The present data therefore raise the possibility that intervention of the uPA system might be therapeutically useful in those patients at risk of progression from gastric atrophy to cancer.



**Figure 6.1** (See opposite page for legend).

## Chapter 7

## References

Chapter 7 - References

#### References

Abe M, Harpel JG, Metz CN, Nunes I, Loskutoff DJ and Rifkin DB. An assay for transforming growth factor-beta using cells transfected with a plasminogen activator inhibitor-1 promoter-luciferase construct. *Anal Biochem* 216: 276-284, 1994.

Ajisaka H, Fushida S, Yonemura Y and Miwa K. Expression of insulin-like growth factor-2, c-MET, matrix metalloproteinase-7 and MUC-1 in primary lesions and lymph node metastatic lesions of gastric cancer. *Hepatogastroenterology* 48: 1788-1792, 2001.

Akopyants NS, Clifton SW, Kersulyte D, Crabtree JE, Youree BE, Reece CA, Bukanov NO, Drazek ES, Roe BA and Berg DE. Analyses of the cag pathogenicity island of Helicobacter pylori. *Mol Microbiol* 28: 37-53, 1998.

Alessi MC, Juhan-Vague I, Kooistra T, Declerck PJ and Collen D. Insulin stimulates the synthesis of plasminogen activator inhibitor 1 by the human hepatocellular cell line Hep G2. *Thromb* Haemost 60: 491-494, 1988.

Alessi MC, Peiretti F, Morange P, Henry M, Nalbone G and Juhan-Vague I. Production of plasminogen activator inhibitor 1 by human adipose tissue: possible link between visceral fat accumulation and vascular disease. *Diabetes* 46: 860-867, 1997.

Alfano D, Franco P, Vocca I, Gambi N, Pisa V, Mancini A, Caputi M, Carriero MV, Iaccarino I and Stoppelli MP. The urokinase plasminogen activator and its receptor: role in cell growth and apoptosis. *Thromb Haemost* 93: 205-211, 2005.

Allen A. The relation of structure to function in gastric mucus. Mod Probl Paediatr 19: 22-33, 1976.

Allen A, Hutton DA, Pearson JP and Sellers LA. Mucus glycoprotein structure, gel formation and gastrointestinal mucus function. *Ciba Found Symp* 109: 137-156, 1984.

Alm RA, Ling LS, Moir DT, King BL, Brown ED, Doig PC, Smith DR, Noonan B, Guild BC, deJonge BL, Carmel G, Tummino PJ, Caruso A, Uria-Nickelsen M, Mills DM, Ives C, Gibson R, Merberg D, Mills SD, Jiang Q, Taylor DE, Vovis GF and Trust TJ. Genomic-sequence comparison of two unrelated isolates of the human gastric pathogen Helicobacter pylori. *Nature* 397: 176-180, 1999.

Amieva MR, Vogelmann R, Covacci A, Tompkins LS, Nelson WJ and Falkow S. Disruption of the epithelial apical-junctional complex by Helicobacter pylori CagA. *Science* 300: 1430-1434, 2003.

Andreasen PA, Kjoller L, Christensen L and Duffy MJ. The urokinase-type plasminogen activator system in cancer metastasis: a review. Int J Cancer 72: 1-22, 1997.

Andreasen PA, Egelund R and Petersen HH. The plasminogen activation system in tumor growth, invasion, and metastasis. *Cell Mol Life Sci* 57: 25-40, 2000.

Angel P and Karin M. The role of Jun, Fos and the AP-1 complex in cell-proliferation and transformation. *Biochim Biophys Acta* 1072: 129-157, 1991.

Angleton P, Chandler WL and Schmer G. Diurnal variation of tissue-type plasminogen activator and its rapid inhibitor (PAI-1). *Circulation* 79: 101-106, 1989.

Anichini E, Fibbi G, Pucci M, Caldini R, Chevanne M and Del RM. Production of second messengers following chemotactic and mitogenic urokinase-receptor interaction in human fibroblasts and mouse fibroblasts transfected with human urokinase receptor. *Exp Cell Res* 213: 438-448, 1994.

Appella E, Robinson EA, Ullrich SJ, Stoppelli MP, Corti A, Cassani G and Blasi F. The receptor-binding sequence of urokinase. A biological function for the growth-factor module of proteases. *J Biol Chem* 262: 4437-4440, 1987.

Argent RH, Kidd M, Owen RJ, Thomas RJ, Limb MC and Atherton JC. Determinants and consequences of different levels of CagA phosphorylation for clinical isolates of Helicobacter pylori. *Gastroenterology* 127: 514-523, 2004.

Argent RH, Thomas RJ, Letley DP, Rittig MG, Hardie KR and Atherton JC. Functional association between the Helicobacter pylori virulence factors VacA and CagA. *J Med Microbiol* 57: 145-150, 2008.

Asahi M, Azuma T, Ito S, Ito Y, Suto H, Nagai Y, Tsubokawa M, Tohyama Y, Maeda S, Omata M, Suzuki T and Sasakawa C. Helicobacter pylori CagA protein can be tyrosine phosphorylated in gastric epithelial cells. *J Exp Med* 191: 593-602, 2000.

Ashktorab H, Daremipouran M, Wilson M, Siddiqi S, Lee EL, Rakhshani N, Malekzadeh R, Johnson AC, Hewitt SM and Smoot DT. Transactivation of the EGFR by AP-1 is induced by Helicobacter pylori in gastric cancer. *Am J Gastroenterol* 102: 2135-2146, 2007.

Ashurst HL, Varro A and Dimaline R. Regulation of mammalian gastrin/CCK receptor (CCK2R) expression in vitro and in vivo. *Exp Physiol* 93: 223-236, 2008.

Atherton JC, Cao P, Peek RM, Jr., Tummuru MK, Blaser MJ and Cover TL. Mosaicism in vacuolating cytotoxin alleles of Helicobacter pylori. Association of specific vacA types with cytotoxin production and peptic ulceration. *J Biol Chem* 270: 17771-17777, 1995.

Atherton JC, Peek RM, Jr., Tham KT, Cover TL and Blaser MJ. Clinical and pathological importance of heterogeneity in vacA, the vacuolating cytotoxin gene of Helicobacter pylori. *Gastroenterology* 112: 92-99, 1997.

Atherton JC. H. pylori virulence factors. Br Med Bull 54: 105-120, 1998.

Axon A and Forman D. Helicobacter gastroduodenitis: a serious infectious disease. *BMJ* 314: 1430-1431, 1997.

Bagchi D, Bhattacharya G and Stohs SJ. Production of reactive oxygen species by gastric cells in association with Helicobacter pylori. *Free Radic Res* 24: 439-450, 1996.

Bajou K, Noel A, Gerard RD, Masson V, Brunner N, Holst-Hansen C, Skobe M, Fusenig NE, Carmeliet P, Collen D and Foidart JM. Absence of host plasminogen activator inhibitor 1 prevents cancer invasion and vascularization. *Nat Med* 4: 923-928, 1998.

Bajou K, Masson V, Gerard RD, Schmitt PM, Albert V, Praus M, Lund LR, Frandsen TL, Brunner N, Dano K, Fusenig NE, Weidle U, Carmeliet G, Loskutoff D, Collen D, Carmeliet P, Foidart JM and Noel A. The plasminogen activator inhibitor PAI-1 controls in vivo tumor vascularization by interaction with proteases, not vitronectin. Implications for antiangiogenic strategies. J Cell Biol 152: 777-784, 2001.

Banerjee S, Medina-Fatimi A, Nichols R, Tendler D, Michetti M, Simon J, Kelly CP, Monath TP and Michetti P. Safety and efficacy of low dose Escherichia coli enterotoxin adjuvant for urease based oral immunisation against Helicobacter pylori in healthy volunteers. *Gut* 51: 634-640, 2002.

Baramova EN, Bajou K, Remacle A, L'Hoir C, Krell HW, Weidle UH, Noel A and Foidart JM. Involvement of PA/plasmin system in the processing of pro-MMP-9 and in the second step of pro-MMP-2 activation. *FEBS Lett* 405: 157-162, 1997.

**Bastelica D, Mavri A, Verdierl M, Berthet B, Juhan-Vague I and Alessi MC**. Relationships between fibrinolytic and inflammatory parameters in human adipose tissue: strong contribution of TNFalpha receptors to PAI-1 levels. *Thromb Haemost* 88: 481-487, 2002.

**Beales I, Calam J, Post L, Srinivasan S, Yamada T and DelValle J**. Effect of transforming growth factor alpha and interleukin 8 on somatostatin release from canine fundic D cells. *Gastroenterology* 112: 136-143, 1997.

Bebb JR, Letley DP, Thomas RJ, Aviles F, Collins HM, Watson SA, Hand NM, Zaitoun A and Atherton JC. Helicobacter pylori upregulates matrilysin (MMP-7) in epithelial cells in vivo and in vitro in a Cag dependent manner. *Gut* 52: 1408-1413, 2003.

Behrendt N, Ronne E, Ploug M, Petri T, Lober D, Nielsen LS, Schleuning WD, Blasi F, Appella E and Dano K. The human receptor for urokinase plasminogen activator. NH2-terminal amino acid sequence and glycosylation variants. *J Biol Chem* 265: 6453-6460, 1990.

Behrendt N, Ploug M, Patthy L, Houen G, Blasi F and Dano K. The ligand-binding domain of the cell surface receptor for urokinase-type plasminogen activator. *J Biol Chem* 266: 7842-7847, 1991.

Behrendt N, Jensen ON, Engelholm LH, Mortz E, Mann M and Dano K. A urokinase receptorassociated protein with specific collagen binding properties. *J Biol Chem* 275: 1993-2002, 2000.

Benasciutti E, Pages G, Kenzior O, Folk W, Blasi F and Crippa MP. MAPK and JNK transduction pathways can phosphorylate Sp1 to activate the uPA minimal promoter element and endogenous gene transcription. *Blood* 104: 256-262, 2004.

Bergheim I, Guo L, Davis MA, Lambert JC, Beier JI, Duveau I, Luyendyk JP, Roth RA and Arteel GE. Metformin prevents alcohol-induced liver injury in the mouse: Critical role of plasminogen activator inhibitor-1. *Gastroenterology* 130: 2099-2112, 2006.

Berglindh T and Obrink KJ. A method for preparing isolated glands from the rabbit gastric mucosa. Acta Physiol Scand 96: 150-159, 1976.

Bergmann S and Hammerschmidt S. Fibrinolysis and host response in bacterial infections. *Thromb* Haemost 98: 512-520, 2007.

Bertozzi P, Astedt B, Zenzius L, Lynch K, LeMaire F, Zapol W and Chapman HA, Jr. Depressed bronchoalveolar urokinase activity in patients with adult respiratory distress syndrome. *N* Engl J Med 322: 890-897, 1990.

Blaser MJ, Perez-Perez GI, Kleanthous H, Cover TL, Peek RM, Chyou PH, Stemmermann GN and Nomura A. Infection with Helicobacter pylori strains possessing cagA is associated with an increased risk of developing adenocarcinoma of the stomach. *Cancer Res* 55: 2111-2115, 1995.

Blaser MJ and Berg DE. Helicobacter pylori genetic diversity and risk of human disease. J Clin Invest 107: 767-773, 2001.

Blasi F, Vassalli JD and Dano K. Urokinase-type plasminogen activator: proenzyme, receptor, and inhibitors. *J Cell Biol* 104: 801-804, 1987.

Blasi F and Carmeliet P. uPAR: a versatile signalling orchestrator. *Nat Rev Mol Cell Biol* 3: 932-943, 2002.

Boccaccio C, Sabatino G, Medico E, Girolami F, Follenzi A, Reato G, Sottile A, Naldini L and Comoglio PM. The MET oncogene drives a genetic programme linking cancer to haemostasis. *Nature* 434: 396-400, 2005.

**Bodger K and Crabtree JE**. Helicobacter pylori and gastric inflammation. *Br Med Bull* 54: 139-150, 1998.

Bohuslav J, Horejsi V, Hansmann C, Stockl J, Weidle UH, Majdic O, Bartke I, Knapp W and Stockinger H. Urokinase plasminogen activator receptor, beta 2-integrins, and Src-kinases within a single receptor complex of human monocytes. *J Exp Med* 181: 1381-1390, 1995.

**Boismenu R and Havran WL**. Modulation of epithelial cell growth by intraepithelial gamma delta T cells. *Science* 266: 1253-1255, 1994.

Bordi C, D'Adda T, Azzoni C and Ferraro G. Classification of gastric endocrine cells at the light and electron microscopical levels. *Microsc Res Tech* 48: 258-271, 2000a.

Bordi C, Azzoni C, Ferraro G, Corleto VD, Gibril F, Delle FG, Lubensky IA, Venzon DJ and Jensen RT. Sampling strategies for analysis of enterochromaffin-like cell changes in Zollinger-Ellison syndrome. *Am J Clin Pathol* 114: 419-425, 2000b.

Bostrom H, Willetts K, Pekny M, Leveen P, Lindahl P, Hedstrand H, Pekna M, Hellstrom M, Gebre-Medhin S, Schalling M, Nilsson M, Kurland S, Tornell J, Heath JK and Betsholtz C. PDGF-A signaling is a critical event in lung alveolar myofibroblast development and alveogenesis. *Cell* 85: 863-873, 1996.

Bottaro DP, Rubin JS, Faletto DL, Chan AM, Kmiecik TE, Vande Woude GF and Aaronson SA. Identification of the hepatocyte growth factor receptor as the c-met proto-oncogene product. *Science* 251: 802-804, 1991.

Boyer B, Valles AM and Edme N. Induction and regulation of epithelial-mesenchymal transitions. *Biochem Pharmacol* 60: 1091-1099, 2000.

**Brand SJ and Stone D**. Reciprocal regulation of antral gastrin and somatostatin gene expression by omeprazole-induced achlorhydria. *J Clin Invest* 82: 1059-1066, 1988.

Brauchle M, Angermeyer K, Hubner G and Werner S. Large induction of keratinocyte growth factor expression by serum growth factors and pro-inflammatory cytokines in cultured fibroblasts. *Oncogene* 9: 3199-3204, 1994.

Brittan M and Wright NA. Gastrointestinal stem cells. J Pathol 197: 492-509, 2002.

Brittan M and Wright NA. The gastrointestinal stem cell. Cell Prolif 37: 35-53, 2004.

Brooks TD, Wang SW, Brunner N and Charlton PA. XR5967, a novel modulator of plasminogen activator inhibitor-1 activity, suppresses tumor cell invasion and angiogenesis in vitro. *Anticancer* Drugs 15: 37-44, 2004.

Brown JF, Keates AC, Hanson PJ and Whittle BJ. Nitric oxide generators and cGMP stimulate mucus secretion by rat gastric mucosal cells. *Am J Physiol* 265: G418-G422, 1993.

Brown NJ, Nakamura S, Ma L, Nakamura I, Donnert E, Freeman M, Vaughan DE and Fogo AB. Aldosterone modulates plasminogen activator inhibitor-1 and glomerulosclerosis in vivo. *Kidney Int* 58: 1219-1227, 2000.

Brzozowski T, Konturek PC, Konturek SJ, Brzozowska I and Pawlik T. Role of prostaglandins in gastroprotection and gastric adaptation. *J Physiol Pharmacol* 56 Suppl 5: 33-55, 2005.

Caceres M, Tobar N, Guerrero J, Smith PC and Martinez J. c-jun-NH2JNK mediates invasive potential and EGFR activation by regulating the expression of HB-EGF in a urokinase-stimulated pathway. *J Cell Biochem* 103: 986-993, 2008

Cai X, Carlson J, Stoicov C, Li H, Wang TC and Houghton J. Helicobacter felis eradication restores normal architecture and inhibits gastric cancer progression in C57BL/6 mice. *Gastroenterology* 128: 1937-1952, 2005.

Calam J, Gibbons A, Healey ZV, Bliss P and Arebi N. How does Helicobacter pylori cause mucosal damage? Its effect on acid and gastrin physiology. *Gastroenterology* 113: S43-S49, 1997.

Caletti G, Bolondi L and Labo G. Anatomical aspects in ultrasonic endoscopy for the stomach. Scand J Gastroenterol Suppl 94: 34-42, 1984.

**Canfield V, West AB, Goldenring JR and Levenson R**. Genetic ablation of parietal cells in transgenic mice: a new model for analyzing cell lineage relationships in the gastric mucosa. *Proc Natl Acad Sci US A* 93: 2431-2435, 1996.

Cannio R, Rennie PS and Blasi F. A cell-type specific and enhancer-dependent silencer in the regulation of the expression of the human urokinase plasminogen activator gene. *Nucleic Acids Res* 19: 2303-2308, 1991.

Carmassi F, Morale M, Ferrini L, Dell'Omo G, Ferdeghini M, Pedrinelli R and De NF. Local insulin infusion stimulates expression of plasminogen activator inhibitor-1 and tissue-type plasminogen activator in normal subjects. *Am J Med* 107: 344-350, 1999.

Carmeliet P, Stassen JM, Schoonjans L, Ream B, van den Oord JJ, De MM, Mulligan RC and Collen D. Plasminogen activator inhibitor-1 gene-deficient mice. II. Effects on hemostasis, thrombosis, and thrombolysis. *J Clin Invest* 92: 2756-2760, 1993a.

Carmeliet P, Kieckens L, Schoonjans L, Ream B, van NA, Prendergast G, Cole M, Bronson R, Collen D and Mulligan RC. Plasminogen activator inhibitor-1 gene-deficient mice. I. Generation by homologous recombination and characterization. *J Clin Invest* 92: 2746-2755, 1993b.

Carriero MV, Del VS, Capozzoli M, Franco P, Fontana L, Zannetti A, Botti G, D'Aiuto G, Salvatore M and Stoppelli MP. Urokinase receptor interacts with alpha(v)beta5 vitronectin receptor, promoting urokinase-dependent cell migration in breast cancer. *Cancer Res* 59: 5307-5314, 1999.

Caterina MJ, Schumacher MA, Tominaga M, Rosen TA, Levine JD and Julius D. The capsaicin receptor: a heat-activated ion channel in the pain pathway. *Nature* 389: 816-824, 1997.

Censini S, Lange C, Xiang Z, Crabtree JE, Ghiara P, Borodovsky M, Rappuoli R and Covacci A. cag, a pathogenicity island of Helicobacter pylori, encodes type I-specific and disease-associated virulence factors. *Proc Natl Acad Sci U S A* 93: 14648-14653, 1996.

Chambers SK, Ivins CM and Carcangiu ML. Plasminogen activator inhibitor-1 is an independent poor prognostic factor for survival in advanced stage epithelial ovarian cancer patients. *Int J Cancer* 79: 449-454, 1998.

Chang AW, Kuo A, Barnathan ES and Okada SS. Urokinase receptor-dependent upregulation of smooth muscle cell adhesion to vitronectin by urokinase. *Arterioscler Thromb Vasc Biol* 18: 1855-1860, 1998.

Chang K, Fujiwara Y, Wyle F and Tarnawski A. Helicobacter pylori toxin inhibits growth and proliferation of cultured gastric cells-Kato III. *J Physiol Pharmacol* 44: 17-22, 1993.

Chang YJ, Wu MS, Lin JT and Chen CC. Helicobacter pylori-Induced invasion and angiogenesis of gastric cells is mediated by cyclooxygenase-2 induction through TLR2/TLR9 and promoter regulation. *J Immunol* 175: 8242-8252, 2005.

Chapman HA. Plasminogen activators, integrins, and the coordinated regulation of cell adhesion and migration. *Curr Opin Cell Biol* 9: 714-724, 1997.

Chapman HA, Wei Y, Simon DI and Waltz DA. Role of urokinase receptor and caveolin in regulation of integrin signaling. *Thromb Haemost* 82: 291-297, 1999.

Chapman HA and Wei Y. Protease crosstalk with integrins: the urokinase receptor paradigm. *Thromb Haemost* 86: 124-129, 2001.

Chaurasia P, guirre-Ghiso JA, Liang OD, Gardsvoll H, Ploug M and Ossowski L. A region in urokinase plasminogen receptor domain III controlling a functional association with alpha5beta1 integrin and tumor growth. *J Biol Chem* 281: 14852-14863, 2006.

Chazaud B, Ricoux R, Christov C, Plonquet A, Gherardi RK and Barlovatz-Meimon G. Promigratory effect of plasminogen activator inhibitor-1 on invasive breast cancer cell populations. *Am J Pathol* 160: 237-246, 2002.

Chen D, Zhao CM, Al-Haider W, Hakanson R, Rehfeld JF and Kopin AS. Differentiation of gastric ECL cells is altered in CCK(2) receptor-deficient mice. *Gastroenterology* 123: 577-585, 2002.

Chen J and Yan H. [Effect of hypoxia on vasodilation induced by substance P (SP) and calcitonin gene-related peptide (CGRP)]. *Zhongguo Yi Xue Ke Xue Yuan Xue Bao* 14: 114-117, 1992.

Chen Y, Kelm RJ, Jr., Budd RC, Sobel BE and Schneider DJ. Inhibition of apoptosis and caspase-3 in vascular smooth muscle cells by plasminogen activator inhibitor type-1. *J Cell Biochem* 92: 178-188, 2004.

Chiou CC, Chan CC, Sheu DL, Chen KT, Li YS and Chan EC. Helicobacter pylori infection induced alteration of gene expression in human gastric cells. *Gut* 48: 598-604, 2001.

Cho CH and Ogle CW. The pharmacological differences and similarities between stress- and ethanol-induced gastric mucosal damage. *Life Sci* 51: 1833-1842, 1992.

Choong PF and Nadesapillai AP. Urokinase plasminogen activator system: a multifunctional role in tumor progression and metastasis. *Clin Orthop Relat Res* S46-S58, 2003.

Chuang-Tsai S, Sisson TH, Hattori N, Tsai CG, Subbotina NM, Hanson KE and Simon RH. Reduction in fibrotic tissue formation in mice genetically deficient in plasminogen activator inhibitor-1. Am J Pathol 163: 445-452, 2003.

Churin Y, Al-Ghoul L, Kepp O, Meyer TF, Birchmeier W and Naumann M. Helicobacter pylori CagA protein targets the c-Met receptor and enhances the motogenic response. *J Cell Biol* 161: 249-255, 2003.

Clyne M, Dillon P, Daly S, O'Kennedy R, May FE, Westley BR and Drumm B. Helicobacter pylori interacts with the human single-domain trefoil protein TFF1. *Proc Natl Acad Sci U S A* 101: 7409-7414, 2004.

Conese M, Nykjaer A, Petersen CM, Cremona O, Pardi R, Andreasen PA, Gliemann J, Christensen EI and Blasi F. alpha-2 Macroglobulin receptor/Ldl receptor-related protein(Lrp)-dependent internalization of the urokinase receptor. *J Cell Biol* 131: 1609-1622, 1995.

Correa P, Haenszel W and Tannenbaum S. Epidemiology of gastric carcinoma: review and future prospects. *Natl Cancer Inst Monogr* 62: 129-134, 1982.

Correa P. Diet modification and gastric cancer prevention. J Natl Cancer Inst Monogr 75-78, 1992.

**Correa P.** Helicobacter pylori and gastric cancer: state of the art. *Cancer Epidemiol Biomarkers Prev* 5: 477-481, 1996.

**Correa P.** Helicobacter pylori as a pathogen and carcinogen. *J Physiol Pharmacol* 48 Suppl 4: 19-24, 1997.

Correa P. Helicobacter pylori infection and gastric cancer. Cancer Epidemiol Biomarkers Prev 12: 238s-241s, 2003.

Correa P. The biological model of gastric carcinogenesis. IARC Sci Publ 301-310, 2004.

Correa P and Houghton J. Carcinogenesis of Helicobacter pylori. *Gastroenterology* 133: 659-672, 2007.

Corthesy-Theulaz I, Porta N, Glauser M, Saraga E, Vaney AC, Haas R, Kraehenbuhl JP, Blum AL and Michetti P. Oral immunization with Helicobacter pylori urease B subunit as a treatment against Helicobacter infection in mice. *Gastroenterology* 109: 115-121, 1995.

**Costa M, Shen Y and Medcalf RL**. Overexpression of a dominant negative CREB protein in HT-1080 cells selectively disrupts plasminogen activator inhibitor type 2 but not tissue-type plasminogen activator gene expression. *FEBS Lett* 482: 75-80, 2000.

Cousin E, Medcalf RL, Bergonzelli GE and Kruithof EK. Regulatory elements involved in constitutive and phorbol ester-inducible expression of the plasminogen activator inhibitor type 2 gene promoter. *Nucleic Acids Res* 19: 3881-3886, 1991.

Covacci A, Censini S, Bugnoli M, Petracca R, Burroni D, Macchia G, Massone A, Papini E, Xiang Z, Figura N and . Molecular characterization of the 128-kDa immunodominant antigen of Helicobacter pylori associated with cytotoxicity and duodenal ulcer. *Proc Natl Acad Sci U S A* 90: 5791-5795, 1993.

Cover TL and Blaser MJ. Purification and characterization of the vacuolating toxin from Helicobacter pylori. *J Biol Chem* 267: 10570-10575, 1992.

Cover TL, Tummuru MK, Cao P, Thompson SA and Blaser MJ. Divergence of genetic sequences for the vacuolating cytotoxin among Helicobacter pylori strains. *J Biol Chem* 269: 10566-10573, 1994.

Cover TL, Glupczynski Y, Lage AP, Burette A, Tummuru MK, Perez-Perez GI and Blaser MJ. Serologic detection of infection with cagA+ Helicobacter pylori strains. *J Clin Microbiol* 33: 1496-1500, 1995.

Cover TL and Blaser MJ. Helicobacter pylori factors associated with disease. *Gastroenterology* 117: 257-261, 1999.

Cover TL, Krishna US, Israel DA and Peek RM, Jr. Induction of gastric epithelial cell apoptosis by Helicobacter pylori vacuolating cytotoxin. *Cancer Res* 63: 951-957, 2003.

Cox JM, Clayton CL, Tomita T, Wallace DM, Robinson PA and Crabtree JE. cDNA array analysis of cag pathogenicity island-associated Helicobacter pylori epithelial cell response genes. *Infect Immun* 69: 6970-6980, 2001.

Crabtree JE, Taylor JD, Wyatt JI, Heatley RV, Shallcross TM, Tompkins DS and Rathbone BJ. Mucosal IgA recognition of Helicobacter pylori 120 kDa protein, peptic ulceration, and gastric pathology. *Lancet* 338: 332-335, 1991.

Crabtree JE, Peichl P, Wyatt JI, Stachl U and Lindley IJ. Gastric interleukin-8 and IgA IL-8 autoantibodies in Helicobacter pylori infection. *Scand J Immunol* 37: 65-70, 1993.

Crabtree JE, Wyatt JI, Trejdosiewicz LK, Peichl P, Nichols PH, Ramsay N, Primrose JN and Lindley IJ. Interleukin-8 expression in Helicobacter pylori infected, normal, and neoplastic gastroduodenal mucosa. J Clin Pathol 47: 61-66, 1994.

Crabtree JE, Xiang Z, Lindley IJ, Tompkins DS, Rappuoli R and Covacci A. Induction of interleukin-8 secretion from gastric epithelial cells by a cagA negative isogenic mutant of Helicobacter pylori. *J Clin Pathol* 48: 967-969, 1995a.

Crabtree JE and Farmery SM. Helicobacter pylori and gastric mucosal cytokines: evidence that CagA-positive strains are more virulent. *Lab Invest* 73: 742-745, 1995b.

Crabtree JE. Gastric mucosal inflammatory responses to Helicobacter pylori. *Aliment Pharmacol Ther* 10 Suppl 1: 29-37, 1996.

Crabtree JE, Kersulyte D, Li SD, Lindley IJ and Berg DE. Modulation of Helicobacter pylori induced interleukin-8 synthesis in gastric epithelial cells mediated by cag PAI encoded VirD4 homologue. *J Clin Pathol* 52: 653-657, 1999.

Crabtree JE, Court M, Aboshkiwa MA, Jeremy AH, Dixon MF and Robinson PA. Gastric mucosal cytokine and epithelial cell responses to Helicobacter pylori infection in Mongolian gerbils. *J Pathol* 202: 197-207, 2004.

Crampton JR. Gastroduodenal mucus and bicarbonate: the defensive zone. *Q J Med* 67: 269-272, 1988.

Crawford HC, Krishna US, Israel DA, Matrisian LM, Washington MK and Peek RM, Jr. Helicobacter pylori strain-selective induction of matrix metalloproteinase-7 in vitro and within gastric mucosa. *Gastroenterology* 125: 1125-1136, 2003.

Cubellis MV, Nolli ML, Cassani G and Blasi F. Binding of single-chain prourokinase to the urokinase receptor of human U937 cells. *J Biol Chem* 261: 15819-15822, 1986.

Cubellis MV, Wun TC and Blasi F. Receptor-mediated internalization and degradation of urokinase is caused by its specific inhibitor PAI-1. *EMBO J* 9: 1079-1085, 1990.

Cuenca R, Blanchard TG, Czinn SJ, Nedrud JG, Monath TP, Lee CK and Redline RW. Therapeutic immunization against Helicobacter mustelae in naturally infected ferrets. *Gastroenterology* 110: 1770-1775, 1996.

Curino A, Mitola DJ, Aaronson H, McMahon GA, Raja K, Keegan AD, Lawrence DA and Bugge TH. Plasminogen promotes sarcoma growth and suppresses the accumulation of tumor-infiltrating macrophages. *Oncogene* 21: 8830-8842, 2002.

Czekay RP, Aertgeerts K, Curriden SA and Loskutoff DJ. Plasminogen activator inhibitor-1 detaches cells from extracellular matrices by inactivating integrins. *J Cell Biol* 160: 781-791, 2003.

Czekay RP and Loskutoff DJ. Unexpected role of plasminogen activator inhibitor 1 in cell adhesion and detachment. *Exp Biol Med (Maywood)* 229: 1090-1096, 2004.

**Danesh J.** Helicobacter pylori infection and gastric cancer: systematic review of the epidemiological studies. *Aliment Pharmacol Ther* 13: 851-856, 1999.

Dano K, Andreasen PA, Grondahl-Hansen J, Kristensen P, Nielsen LS and Skriver L. Plasminogen activators, tissue degradation, and cancer. Adv Cancer Res 44: 139-266, 1985.

Dano K, Behrendt N, Hoyer-Hansen G, Johnsen M, Lund LR, Ploug M and Romer J. Plasminogen activation and cancer. *Thromb Haemost* 93: 676-681, 2005.

De LA, Baldi A, Russo P, Todisco A, Altucci L, Giardullo N, Pasquale L, Iaquinto S, D'Onofrio V, Parodi MC, Paggi MG and Iaquinto G. Coexpression of Helicobacter pylori's proteins CagA and HspB induces cell proliferation in AGS gastric epithelial cells, independently from the bacterial infection. *Cancer Res* 63: 6350-6356, 2003.

**De LA, De FM, Iaquinto S and Iaquinto G**. Effects of Helicobacter pylori infection on cell cycle progression and the expression of cell cycle regulatory proteins. *J Cell Physiol* 200: 334-342, 2004.

De PG, Tavian D, Copeta A, Portolani N, Giulini SM and Barlati S. Expression of urokinase-type plasminogen activator (u-PA), u-PA receptor, and tissue-type PA messenger RNAs in human hepatocellular carcinoma. *Cancer Res* 58: 2234-2239, 1998.

**Dear AE, Costa M and Medcalf RL**. Urokinase-mediated transactivation of the plasminogen activator inhibitor type 2 (PAI-2) gene promoter in HT-1080 cells utilises AP-1 binding sites and potentiates phorbol ester-mediated induction of endogenous PAI-2 mRNA. *FEBS Lett* 402: 265-272, 1997.

Declerck PJ, Alessi MC, Verstreken M, Kruithof EK, Juhan-Vague I and Collen D. Measurement of plasminogen activator inhibitor 1 in biologic fluids with a murine monoclonal antibody-based enzyme-linked immunosorbent assay. *Blood* 71: 220-225, 1988a.

Declerck PJ, De MM, Alessi MC, Baudner S, Paques EP, Preissner KT, Muller-Berghaus G and Collen D. Purification and characterization of a plasminogen activator inhibitor 1 binding protein from human plasma. Identification as a multimeric form of S protein (vitronectin). J Biol Chem 263: 15454-15461, 1988b.

**Degryse B, Neels JG, Czekay RP, Aertgeerts K, Kamikubo Y and Loskutoff DJ**. The low density lipoprotein receptor-related protein is a motogenic receptor for plasminogen activator inhibitor-1. *J Biol Chem* 279: 22595-22604, 2004.

**Deng G, Curriden SA, Wang S, Rosenberg S and Loskutoff DJ**. Is plasminogen activator inhibitor-1 the molecular switch that governs urokinase receptor-mediated cell adhesion and release? *J Cell Biol* 134: 1563-1571, 1996.

**Deng G, Curriden SA, Hu G, Czekay RP and Loskutoff DJ**. Plasminogen activator inhibitor-1 regulates cell adhesion by binding to the somatomedin B domain of vitronectin. *J Cell Physiol* 189: 23-33, 2001.

**Dennler S, Itoh S, Vivien D, ten DP, Huet S and Gauthier JM**. Direct binding of Smad3 and Smad4 to critical TGF beta-inducible elements in the promoter of human plasminogen activator inhibitor-type 1 gene. *EMBO J* 17: 3091-3100, 1998.

**Dewhirst FE, Seymour C, Fraser GJ, Paster BJ and Fox JG**. Phylogeny of Helicobacter isolates from bird and swine feces and description of Helicobacter pametensis sp. nov. *Int J Syst Bacteriol* 44: 553-560, 1994.

**Dickinson JL, Bates EJ, Ferrante A and Antalis TM**. Plasminogen activator inhibitor type 2 inhibits tumor necrosis factor alpha-induced apoptosis. Evidence for an alternate biological function. *J Biol Chem* 270: 27894-27904, 1995.

**Dieval J, Nguyen G, Gross S, Delobel J and Kruithof EK**. A lifelong bleeding disorder associated with a deficiency of plasminogen activator inhibitor type 1. *Blood* 77: 528-532, 1991.

Dimaline R, Evans D, Varro A and Dockray GJ. Reversal by omeprazole of the depression of gastrin cell function by fasting in the rat. *J Physiol* 433: 483-493, 1991a.

**Dimaline R and Sandvik AK**. Histidine decarboxylase gene expression in rat fundus is regulated by gastrin. *FEBS Lett* 281: 20-22, 1991b.

Dimaline R, Evans D, Forster ER, Sandvik AK and Dockray GJ. Control of gastric corpus chromogranin A messenger RNA abundance in the rat. *Am J Physiol* 264: G583-G588, 1993.

**Dimaline R and Struthers J.** Expression and regulation of a vesicular monoamine transporter in rat stomach: a putative histamine transporter. *J Physiol* 490 (Pt 1): 249-256, 1996.

**Dimaline R and Varro A**. Attack and defence in the gastric epithelium - a delicate balance. *Exp Physiol* 92: 591-601, 2007.

**Dixon MF, Genta RM, Yardley JH and Correa P.** Classification and grading of gastritis. The updated Sydney System. International Workshop on the Histopathology of Gastritis, Houston 1994. *Am J Surg Pathol* 20: 1161-1181, 1996.

Dockray GJ. Topical review. Gastrin and gastric epithelial physiology. J Physiol 518 (Pt 2): 315-324, 1999.

Dockray GJ, Varro A, Dimaline R and Wang T. The gastrins: their production and biological activities. *Annu Rev Physiol* 63: 119-139, 2001.

**Domotor A, Peidl Z, Vincze A, Hunyady B, Szolcsanyi J, Kereskay L, Szekeres G and Mozsik G.** Immunohistochemical distribution of vanilloid receptor, calcitonin-gene related peptide and substance P in gastrointestinal mucosa of patients with different gastrointestinal disorders. *Inflammopharmacology* 13: 161-177, 2005.

**Du Y, Danjo K, Robinson PA and Crabtree J.** In-cell western analysis of helicobacter pyloriinduced phosphorylation of extracellular-signal related kinase via the transactivation of the epidermal growth factor receptor. *Microbes Infect* 9: 838-46, 2007.

**Dublin E, Hanby A, Patel NK, Liebman R and Barnes D**. Immunohistochemical expression of uPA, uPAR, and PAI-1 in breast carcinoma. Fibroblastic expression has strong associations with tumor pathology. *Am J Pathol* 157: 1219-1227, 2000.

Dubois A, Lee CK, Fiala N, Kleanthous H, Mehlman PT and Monath T. Immunization against natural Helicobacter pylori infection in nonhuman primates. *Infect Immun* 66: 4340-4346, 1998.

**Duffy MJ**. Urokinase-type plasminogen activator: a potent marker of metastatic potential in human cancers. *Biochem Soc Trans* 30: 207-210, 2002a.

**Duffy MJ**. Urokinase plasminogen activator and its inhibitor, PAI-1, as prognostic markers in breast cancer: from pilot to level 1 evidence studies. *Clin Chem* 48: 1194-1197, 2002b.

**Dumler I, Weis A, Mayboroda OA, Maasch C, Jerke U, Haller H and Gulba DC**. The Jak/Stat pathway and urokinase receptor signaling in human aortic vascular smooth muscle cells. *J Biol Chem* 273: 315-321, 1998.

**Dunn BE, Roop RM, Sung CC, Sharma SA, Perez-Perez GI and Blaser MJ**. Identification and purification of a cpn60 heat shock protein homolog from Helicobacter pylori. *Infect Immun* 60: 1946-1951, 1992.

Eaton KA, Mefford M and Thevenot T. The role of T cell subsets and cytokines in the pathogenesis of Helicobacter pylori gastritis in mice. *J Immunol* 166: 7456-7461, 2001.

Eberhart CE and DuBois RN. Eicosanoids and the gastrointestinal tract. *Gastroenterology* 109: 285-301, 1995.

Eitzman DT, McCoy RD, Zheng X, Fay WP, Shen T, Ginsburg D and Simon RH. Bleomycininduced pulmonary fibrosis in transgenic mice that either lack or overexpress the murine plasminogen activator inhibitor-1 gene. *J Clin Invest* 97: 232-237, 1996.

El-Omar EM, Carrington M, Chow WH, McColl KE, Bream JH, Young HA, Herrera J, Lissowska J, Yuan CC, Rothman N, Lanyon G, Martin M, Fraumeni JF, Jr. and Rabkin CS. Interleukin-1 polymorphisms associated with increased risk of gastric cancer. *Nature* 404: 398-402, 2000.

Elisen MG, van Kooij RJ, Nolte MA, Marquart JA, Lock TM, Bouma BN and Meijers JC. Protein C inhibitor may modulate human sperm-oocyte interactions. *Biol Reprod* 58: 670-677, 1998.

Ellis V, Scully MF and Kakkar VV. Plasminogen activation initiated by single-chain urokinasetype plasminogen activator. Potentiation by U937 monocytes. *J Biol Chem* 264: 2185-2188, 1989. **Emeis JJ and Kooistra T**. Interleukin 1 and lipopolysaccharide induce an inhibitor of tissue-type plasminogen activator in vivo and in cultured endothelial cells. *J Exp Med* 163: 1260-1266, 1986.

Eren M, Painter CA, Atkinson JB, Declerck PJ and Vaughan DE. Age-dependent spontaneous coronary arterial thrombosis in transgenic mice that express a stable form of human plasminogen activator inhibitor-1. *Circulation* 106: 491-496, 2002.

Erickson LA, Ginsberg MH and Loskutoff DJ. Detection and partial characterization of an inhibitor of plasminogen activator in human platelets. *J Clin Invest* 74: 1465-1472, 1984.

Ernst PB and Gold BD. The disease spectrum of Helicobacter pylori: the immunopathogenesis of gastroduodenal ulcer and gastric cancer. Annu Rev Microbiol 54: 615-640, 2000.

Espana F, Berrettini M and Griffin JH. Purification and characterization of plasma protein C inhibitor. *Thromb Res* 55: 369-384, 1989.

Espana F, Estelles A, Griffin JH and Aznar J. Interaction of plasma kallikrein with protein C inhibitor in purified mixtures and in plasma. *Thromb Haemost* 65: 46-51, 1991.

Espana F, Estelles A, Fernandez PJ, Gilabert J, Sanchez-Cuenca J and Griffin JH. Evidence for the regulation of urokinase and tissue type plasminogen activators by the serpin, protein C inhibitor, in semen and blood plasma. *Thromb Haemost* 70: 989-994, 1993.

Everhart JE. Recent developments in the epidemiology of Helicobacter pylori. *Gastroenterol Clin* North Am 29: 559-578, 2000.

Fay WP, Shapiro AD, Shih JL, Schleef RR and Ginsburg D. Brief report: complete deficiency of plasminogen-activator inhibitor type 1 due to a frame-shift mutation. *N Engl J Med* 327: 1729-1733, 1992.

Fay WP, Parker AC, Condrey LR and Shapiro AD. Human plasminogen activator inhibitor-1 (PAI-1) deficiency: characterization of a large kindred with a null mutation in the PAI-1 gene. *Blood* 90: 204-208, 1997.

Fearns C and Loskutoff DJ. Induction of plasminogen activator inhibitor 1 gene expression in murine liver by lipopolysaccharide. Cellular localization and role of endogenous tumor necrosis factor-alpha. *Am J Pathol* 150: 579-590, 1997.

Ferrero RL, Thiberge JM, Huerre M and Labigne A. Recombinant antigens prepared from the urease subunits of Helicobacter spp.: evidence of protection in a mouse model of gastric infection. *Infect Immun* 62: 4981-4989, 1994.

Festuccia C, Guerra F, D'Ascenzo S, Giunciuglio D, Albini A and Bologna M. In vitro regulation of pericellular proteolysis in prostatic tumor cells treated with bombesin. *Int J Cancer* 75: 418-431, 1998.

Fibbi G, Pucci M, Serni U, Cerinic MM and Del RM. Antisense targeting of the urokinase receptor blocks urokinase-dependent proliferation, chemoinvasion, and chemotaxis of human synovial cells and chondrocytes in vitro. *Proc Assoc Am Physicians* 110: 340-350, 1998.

Figueiredo C, Machado JC, Pharoah P, Seruca R, Sousa S, Carvalho R, Capelinha AF, Quint W, Caldas C, van Doorn LJ, Carneiro F and Sobrinho-Simoes M. Helicobacter pylori and interleukin 1 genotyping: an opportunity to identify high-risk individuals for gastric carcinoma. *J Natl Cancer Inst* 94: 1680-1687, 2002.

Fisher JL, Field CL, Zhou H, Harris TL, Henderson MA and Choong PF. Urokinase plasminogen activator system gene expression is increased in human breast carcinoma and its bone metastases--a comparison of normal breast tissue, non-invasive and invasive carcinoma and osseous metastases. *Breast Cancer Res Treat* 61: 1-12, 2000.

**Flemstrom G.** Active alkalinization by amphibian gastric fundic mucosa in vitro. *Am J Physiol* 233: E1-12, 1977.

Foekens JA, Peters HA, Look MP, Portengen H, Schmitt M, Kramer MD, Brunner N, Janicke F, Meijer-van Gelder ME, Henzen-Logmans SC, van Putten WL and Klijn JG. The urokinase system of plasminogen activation and prognosis in 2780 breast cancer patients. *Cancer Res* 60: 636-643, 2000.

Forman D, Newell DG, Fullerton F, Yarnell JW, Stacey AR, Wald N and Sitas F. Association between infection with Helicobacter pylori and risk of gastric cancer: evidence from a prospective investigation. *BMJ* 302: 1302-1305, 1991.

Forte JG and Yao X. The membrane-recruitment-and-recycling hypothesis of gastric HCl secretion. *Trends Cell Biol* 6: 45-48, 1996.

Fox JG, Dangler CA, Taylor NS, King A, Koh TJ and Wang TC. High-salt diet induces gastric epithelial hyperplasia and parietal cell loss, and enhances Helicobacter pylori colonization in C57BL/6 mice. *Cancer Res* 59: 4823-4828, 1999.

Fox JG, Wang TC, Rogers AB, Poutahidis T, Ge Z, Taylor N, Dangler CA, Israel DA, Krishna U, Gaus K and Peek RM, Jr. Host and microbial constituents influence Helicobacter pylori-induced cancer in a murine model of hypergastrinemia. *Gastroenterology* 124: 1879-1890, 2003a.

Fox JG, Rogers AB, Ihrig M, Taylor NS, Whary MT, Dockray G, Varro A and Wang TC. Helicobacter pylori-associated gastric cancer in INS-GAS mice is gender specific. *Cancer Res* 63: 942-950, 2003b.

Fox SB, Taylor M, Grondahl-Hansen J, Kakolyris S, Gatter KC and Harris AL. Plasminogen activator inhibitor-1 as a measure of vascular remodelling in breast cancer. *J Pathol* 195: 236-243, 2001.
Franco L, Talamini G, Carra G and Doria D. Expression of COX-1, COX-2, and inducible nitric oxide synthase protein in human gastric antrum with Helicobacter pylori infection. *Prostaglandins Other Lipid Mediat* 58: 9-17, 1999.

Friedenstein AJ, Chailakhyan RK, Latsinik NV, Panasyuk AF and Keiliss-Borok IV. Stromal cells responsible for transferring the microenvironment of the hemopoietic tissues. Cloning in vitro and retransplantation in vivo. *Transplantation* 17: 331-340, 1974.

Friis-Hansen L, Sundler F, Li Y, Gillespie PJ, Saunders TL, Greenson JK, Owyang C, Rehfeld JF and Samuelson LC. Impaired gastric acid secretion in gastrin-deficient mice. *Am J Physiol* 274: G561-G568, 1998.

Friis-Hansen L. Achlorhydria is associated with gastric microbial overgrowth and development of cancer: lessons learned from the gastrin knockout mouse. *Scand J Clin Lab Invest* 66: 607-621, 2006a.

Friis-Hansen L, Rieneck K, Nilsson HO, Wadstrom T and Rehfeld JF. Gastric inflammation, metaplasia, and tumor development in gastrin-deficient mice. *Gastroenterology* 131: 246-258, 2006b.

Fujikawa A, Shirasaka D, Yamamoto S, Ota H, Yahiro K, Fukada M, Shintani T, Wada A, Aoyama N, Hirayama T, Fukamachi H and Noda M. Mice deficient in protein tyrosine phosphatase receptor type Z are resistant to gastric ulcer induction by VacA of Helicobacter pylori. *Nat Genet* 33: 375-381, 2003.

Fukamachi H, Ichinose M, Tsukada S, Kakei N, Suzuki T, Miki K, Kurokawa K and Masui T. Hepatocyte growth factor region specifically stimulates gastro-intestinal epithelial growth in primary culture. *Biochem Biophys Res Commun* 205: 1445-1451, 1994.

Gabbiani G, Ryan GB and Majne G. Presence of modified fibroblasts in granulation tissue and their possible role in wound contraction. *Experientia* 27: 549-550, 1971.

Garner A, Flemstrom G and Heylings JR. Effects of antiinflammatory agents and prostaglandins on acid and bicarbonate secretions in the amphibian-isolated gastric mucosa. *Gastroenterology* 77: 451-457, 1979.

Gebert B, Fischer W, Weiss E, Hoffmann R and Haas R. Helicobacter pylori vacuolating cytotoxin inhibits T lymphocyte activation. *Science* 301: 1099-1102, 2003.

Geiger M, Huber K, Wojta J, Stingl L, Espana F, Griffin JH and Binder BR. Complex formation between urokinase and plasma protein C inhibitor in vitro and in vivo. *Blood* 74: 722-728, 1989.

Gerhard M, Lehn N, Neumayer N, Boren T, Rad R, Schepp W, Miehlke S, Classen M and Prinz C. Clinical relevance of the Helicobacter pylori gene for blood-group antigen-binding adhesin. *Proc Natl Acad Sci USA* 96: 12778-12783, 1999.

Gettins PG. Serpin structure, mechanism, and function. Chem Rev 102: 4751-4804, 2002.

Ghiara P, Marchetti M, Blaser MJ, Tummuru MK, Cover TL, Segal ED, Tompkins LS and Rappuoli R. Role of the Helicobacter pylori virulence factors vacuolating cytotoxin, CagA, and urease in a mouse model of disease. *Infect Immun* 63: 4154-4160, 1995.

Ginsburg D, Zeheb R, Yang AY, Rafferty UM, Andreasen PA, Nielsen L, Dano K, Lebo RV and Gelehrter TD. cDNA cloning of human plasminogen activator-inhibitor from endothelial cells. *J Clin Invest* 78: 1673-1680, 1986.

Giordano S, Di Renzo MF, Narsimhan RP, Cooper CS, Rosa C and Comoglio PM. Biosynthesis of the protein encoded by the c-met proto-oncogene. *Oncogene* 4: 1383-1388, 1989.

Glocker E, Lange C, Covacci A, Bereswill S, Kist M and Pahl HL. Proteins encoded by the cag pathogenicity island of Helicobacter pylori are required for NF-kappaB activation. *Infect Immun* 66: 2346-2348, 1998.

Go MF. Review article: natural history and epidemiology of Helicobacter pylori infection. *Aliment Pharmacol Ther* 16 Suppl 1: 3-15, 2002.

Gobert AP, McGee DJ, Akhtar M, Mendz GL, Newton JC, Cheng Y, Mobley HL and Wilson KT. Helicobacter pylori arginase inhibits nitric oxide production by eukaryotic cells: a strategy for bacterial survival. *Proc Natl Acad Sci USA* 98: 13844-13849, 2001.

Goke M, Kanai M and Podolsky DK. Intestinal fibroblasts regulate intestinal epithelial cell proliferation via hepatocyte growth factor. *Am J Physiol* 274: G809-G818, 1998.

Goldfarb RH, Murano G, Brundage R, Siegal GP, Terranova V, Garbisa S and Liotta LA. Degradation of glycoprotein and collagenous components of the basement membrane: studies with urokinase-type plasminogen activator, alpha-thrombin, and plasmin. *Semin Thromb Hemost* 12: 335-336, 1986.

**Graham DY**. Antibiotic resistance in Helicobacter pylori: implications for therapy. *Gastroenterology* 115: 1272-1277, 1998.

Grant DS, Kleinman HK, Goldberg ID, Bhargava MM, Nickoloff BJ, Kinsella JL, Polverini P and Rosen EM. Scatter factor induces blood vessel formation in vivo. *Proc Natl Acad Sci US A* 90: 1937-1941, 1993.

Green T and Dockray GJ. Characterization of the peptidergic afferent innervation of the stomach in the rat, mouse and guinea-pig. *Neuroscience* 25: 181-193, 1988.

**GREGORY RA and TRACY HJ**. THE CONSTITUTION AND PROPERTIES OF TWO GASTRINS EXTRACTED FROM HOG ANTRAL MUCOSA. *Gut* 5: 103-114, 1964.

Grobstein C. Mechanisms of organogenetic tissue interaction. *Natl Cancer Inst Monogr* 26: 279-299, 1967.

Grubic N, Stegnar M, Peternel P, Kaider A and Binder BR. A novel G/A and the 4G/5G polymorphism within the promoter of the plasminogen activator inhibitor-1 gene in patients with deep vein thrombosis. *Thromb Res* 84: 431-443, 1996.

Guerrini L, Casalino L, Corti A and Blasi F. NF-kappa B-mediated regulation of urokinase gene expression by PMA and TNF-alpha in human A549 cells. *FEBS Lett* 393: 69-73, 1996.

Guillemin K, Salama NR, Tompkins LS and Falkow S. Cag pathogenicity island-specific responses of gastric epithelial cells to Helicobacter pylori infection. *Proc Natl Acad Sci U S A* 99: 15136-15141, 2002.

Guslandi M. Effects of ethanol on the gastric mucosa. Dig Dis 5: 21-32, 1987.

Guth PH, Paulsen G and Nagata H. Histologic and microcirculatory changes in alcohol-induced gastric lesions in the rat: effect of prostaglandin cytoprotection. *Gastroenterology* 87: 1083-1090, 1984.

Gutierrez LS, Schulman A, Brito-Robinson T, Noria F, Ploplis VA and Castellino FJ. Tumor development is retarded in mice lacking the gene for urokinase-type plasminogen activator or its inhibitor, plasminogen activator inhibitor-1. *Cancer Res* 60: 5839-5847, 2000.

Hall PA. What are stem cells and how are they controlled? J Pathol 158: 275-277, 1989.

Hamsten A, Wiman B, de FU and Blomback M. Increased plasma levels of a rapid inhibitor of tissue plasminogen activator in young survivors of myocardial infarction. *N Engl J Med* 313: 1557-1563, 1985.

Hamsten A, de FU, Walldius G, Dahlen G, Szamosi A, Landou C, Blomback M and Wiman B. Plasminogen activator inhibitor in plasma: risk factor for recurrent myocardial infarction. *Lancet* 2: 3-9, 1987.

Hansen SK, Nerlov C, Zabel U, Verde P, Johnsen M, Baeuerle PA and Blasi F. A novel complex between the p65 subunit of NF-kappa B and c-Rel binds to a DNA element involved in the phorbol ester induction of the human urokinase gene. *EMBO J* 11: 205-213, 1992.

Hansson LE, Engstrand L, Nyren O, Evans DJ, Jr., Lindgren A, Bergstrom R, Andersson B, Athlin L, Bendtsen O and Tracz P. Helicobacter pylori infection: independent risk indicator of gastric adenocarcinoma. *Gastroenterology* 105: 1098-1103, 1993.

Hansson LE, Nyren O, Hsing AW, Bergstrom R, Josefsson S, Chow WH, Fraumeni JF, Jr. and Adami HO. The risk of stomach cancer in patients with gastric or duodenal ulcer disease. *N Engl J Med* 335: 242-249, 1996.

Harris PR, Weber HC, Wilcox CM, Jensen RT and Smith PD. Cytokine gene profile in gastric mucosa in Helicobacter pylori infection and Zollinger-Ellison syndrome. *Am J Gastroenterol* 97: 312-318, 2002.

Hattori N, Degen JL, Sisson TH, Liu H, Moore BB, Pandrangi RG, Simon RH and Drew AF. Bleomycin-induced pulmonary fibrosis in fibrinogen-null mice. *J Clin Invest* 106: 1341-1350, 2000.

Hattori Y, Ohno T, Ae T, Saeki T, Arai K, Mizuguchi S, Saigenji K and Majima M. Gastric mucosal protection against ethanol by EP2 and EP4 signaling through the inhibition of leukotriene C4 production. *Am J Physiol Gastrointest Liver Physiol* 294: G80-G87, 2008.

Hazelbag S, Kenter GG, Gorter A and Fleuren GJ. Prognostic relevance of TGF-beta1 and PAI-1 in cervical cancer. *Int J Cancer* 112: 1020-1028, 2004.

He CJ, Rebibou JM, Peraldi MN, Meulders Q and Rondeau E. Growth factor-like effect of urokinase type plasminogen activator in human renal cells. *Biochem Biophys Res Commun* 176: 1408-1416, 1991.

Heiss MM, Babic R, Allgayer H, Gruetzner KU, Jauch KW, Loehrs U and Schildberg FW. Tumor-associated proteolysis and prognosis: new functional risk factors in gastric cancer defined by the urokinase-type plasminogen activator system. *J Clin Oncol* 13: 2084-2093, 1995.

Hekman CM and Loskutoff DJ. Endothelial cells produce a latent inhibitor of plasminogen activators that can be activated by denaturants. *J Biol Chem* 260: 11581-11587, 1985.

Hekman CM and Loskutoff DJ. Bovine plasminogen activator inhibitor 1: specificity determinations and comparison of the active, latent, and guanidine-activated forms. *Biochemistry* 27: 2911-2918, 1988.

Hemers E, Duval C, McCaig C, Handley M, Dockray GJ and Varro A. Insulin-like growth factor binding protein-5 is a target of matrix metalloproteinase-7: implications for epithelial-mesenchymal signaling. *Cancer Res* 65: 7363-7369, 2005.

Henning N, Heinkel K and Gloor M. [Gastric secretion and gastritis. Secretion studies and suctionbioptic examinations]. *Munch Med Wochenschr* 108: 1977-1982, 1966.

Herszenyi L, Plebani M, Cardin R, Carraro P, De PM, Roveroni G, Naccarato R and Farinati F. The role of urokinase-type plasminogen activator and its inhibitor PAI-1 in gastric cancer. *Acta Physiol Hung* 83: 213-221, 1995.

Herszenyi L, Plebani M, Carraro P, De PM, Cardin R, Di MF, Kusstatscher S, Naccarato R and Farinati F. Impaired fibrinolysis and increased protease levels in gastric and duodenal mucosa of patients with active duodenal ulcer. *Am J Gastroenterol* 92: 843-847, 1997.

Hessey SJ, Spencer J, Wyatt JI, Sobala G, Rathbone BJ, Axon AT and Dixon MF. Bacterial adhesion and disease activity in Helicobacter associated chronic gastritis. *Gut* 31: 134-138, 1990.

Higashi H, Tsutsumi R, Muto S, Sugiyama T, Azuma T, Asaka M and Hatakeyama M. SHP-2 tyrosine phosphatase as an intracellular target of Helicobacter pylori CagA protein. *Science* 295: 683-686, 2002.

Higashi H, Nakaya A, Tsutsumi R, Yokoyama K, Fujii Y, Ishikawa S, Higuchi M, Takahashi A, Kurashima Y, Teishikata Y, Tanaka S, Azuma T and Hatakeyama M. Helicobacter pylori CagA induces Ras-independent morphogenetic response through SHP-2 recruitment and activation. *J Biol Chem* 279: 17205-17216, 2004.

Hofmann R, Lehmer A, Buresch M, Hartung R and Ulm K. Clinical relevance of urokinase plasminogen activator, its receptor, and its inhibitor in patients with renal cell carcinoma. *Cancer* 78: 487-492, 1996.

Hofmann R, Lehmer A, Hartung R, Robrecht C, Buresch M and Grothe F. Prognostic value of urokinase plasminogen activator and plasminogen activator inhibitor-1 in renal cell cancer. *J Urol* 155: 858-862, 1996.

Hohenberger P and Gretschel S. Gastric cancer. Lancet 362: 305-315, 2003.

Hold GL, Rabkin CS, Chow WH, Smith MG, Gammon MD, Risch HA, Vaughan TL, McColl KE, Lissowska J, Zatonski W, Schoenberg JB, Blot WJ, Mowat NA, Fraumeni JF, Jr. and El-Omar EM. A functional polymorphism of toll-like receptor 4 gene increases risk of gastric carcinoma and its precursors. *Gastroenterology* 132: 905-912, 2007.

HOLLANDER F. The two-component mucous barrier; its activity in protecting the gastroduodenal mucosa against peptic ulceration. AMA Arch Intern Med 93: 107-120, 1954.

Holzer-Petsche U. Modulation of gastric contractions in response to tachykinins and bethanechol by extrinsic nerves. *Br J Pharmacol* 103: 1958-1962, 1991.

Holzer P and Sametz W. Gastric mucosal protection against ulcerogenic factors in the rat mediated by capsaicin-sensitive afferent neurons. *Gastroenterology* 91: 975-981, 1986.

Holzer P. Local effector functions of capsaicin-sensitive sensory nerve endings: involvement of tachykinins, calcitonin gene-related peptide and other neuropeptides. *Neuroscience* 24: 739-768, 1988.

Holzer P. Nerves and gastric mucosal protection. Dig Dis Sci 35: 1048-1051, 1990a.

Holzer P, Peskar BM, Peskar BA and Amann R. Release of calcitonin gene-related peptide induced by capsaicin in the vascularly perfused rat stomach. *Neurosci Lett* 108: 195-200, 1990b.

Holzer P, Lippe II, Raybould HE, Pabst MA, Livingston EH, Amann R, Peskar BM, Peskar BA, Tache Y and Guth PH. Role of peptidergic sensory neurons in gastric mucosal blood flow and protection. *Ann N Y Acad Sci* 632: 272-282, 1991.

Holzer P, Schicho R, Holzer-Petsche U and Lippe IT. The gut as a neurological organ. *Wien Klin Wochenschr* 113: 647-660, 2001.

Hopkins WE, Westerhausen DR, Jr., Sobel BE and Billadello JJ. Transcriptional regulation of plasminogen activator inhibitor type-1 mRNA in Hep G2 cells by epidermal growth factor. *Nucleic Acids Res* 19: 163-168, 1991.

Hou B, Eren M, Painter CA, Covington JW, Dixon JD, Schoenhard JA and Vaughan DE. Tumor necrosis factor alpha activates the human plasminogen activator inhibitor-1 gene through a distal nuclear factor kappaB site. *J Biol Chem* 279: 18127-18136, 2004.

Hou W and Schubert ML. Gastric secretion. Curr Opin Gastroenterol 22: 593-598, 2006.

Houghton J, Fox JG and Wang TC. Gastric cancer: laboratory bench to clinic. J Gastroenterol Hepatol 17: 495-502, 2002.

Houghton J, Stoicov C, Nomura S, Rogers AB, Carlson J, Li H, Cai X, Fox JG, Goldenring JR and Wang TC. Gastric cancer originating from bone marrow-derived cells. *Science* 306: 1568-1571, 2004.

Houghton J and Wang TC. Helicobacter pylori and gastric cancer: a new paradigm for inflammation-associated epithelial cancers. *Gastroenterology* 128: 1567-1578, 2005.

Huang TJ, Wang JY, Lin SR, Lian ST and Hsieh JS. Overexpression of the c-met protooncogene in human gastric carcinoma--correlation to clinical features. *Acta Oncol* 40: 638-643, 2001.

Hudson MA and McReynolds LM. Urokinase and the urokinase receptor: association with in vitro invasiveness of human bladder cancer cell lines. J Natl Cancer Inst 89: 709-717, 1997.

Hussain I, Bate GW, Henry J, Djali P, Dimaline R, Dockray GJ and Varro A. Modulation of gastrin processing by vesicular monoamine transporter type 1 (VMAT1) in rat gastrin cells. *J Physiol* 517 (Pt 2): 495-505, 1999a.

Hussain MM, Strickland DK and Bakillah A. The mammalian low-density lipoprotein receptor family. Annu Rev Nutr 19: 141-172, 1999b.

Hynes RO. Integrins: versatility, modulation, and signaling in cell adhesion. Cell 69: 11-25, 1992.

Iaquinto G, Todisco A, Giardullo N, D'Onofrio V, Pasquale L, De LA, Andriulli A, Perri F, Rega C, De CG, Landi M, Taccone W, Leandro G and Figura N. Antibody response to Helicobacter pylori CagA and heat-shock proteins in determining the risk of gastric cancer development. *Dig Liver Dis* 32: 378-383, 2000.

**Ibanez-Tallon I, Caretti G, Blasi F and Crippa MP**. In vivo analysis of the state of the human uPA enhancer following stimulation by TPA. *Oncogene* 18: 2836-2845, 1999.

**Ibanez-Tallon I, Ferrai C, Longobardi E, Facetti I, Blasi F and Crippa MP**. Binding of Sp1 to the proximal promoter links constitutive expression of the human uPA gene and invasive potential of PC3 cells. *Blood* 100: 3325-3332, 2002.

Ichiki Y, Smith E, LeRoy EC and Trojanowska M. Different effects of basic fibroblast growth factor and transforming growth factor-beta on the two platelet-derived growth factor receptors' expression in scleroderma and healthy human dermal fibroblasts. *J Invest Dermatol* 104: 124-127, 1995.

Ichinose A, Fujikawa K and Suyama T. The activation of pro-urokinase by plasma kallikrein and its inactivation by thrombin. *J Biol Chem* 261: 3486-3489, 1986.

Illemann M, Hansen U, Nielsen HJ, Andreasen PA, Hoyer-Hansen G, Lund LR, Dano K and Nielsen BS. Leading-edge myofibroblasts in human colon cancer express plasminogen activator inhibitor-1. *Am J Clin Pathol* 122: 256-265, 2004.

Ilver D, Arnqvist A, Ogren J, Frick IM, Kersulyte D, Incecik ET, Berg DE, Covacci A, Engstrand L and Boren T. Helicobacter pylori adhesin binding fucosylated histo-blood group antigens revealed by retagging. *Science* 279: 373-377, 1998.

Imagawa S, Fujii S, Dong J, Furumoto T, Kaneko T, Zaman T, Satoh Y, Tsutsui H and Sobel BE. Hepatocyte growth factor regulates E box-dependent plasminogen activator inhibitor type 1 gene expression in HepG2 liver cells. *Arterioscler Thromb Vasc Biol* 26: 2407-2413, 2006.

Ingelfinger JR. Forestalling fibrosis. N Engl J Med 349: 2265-2266, 2003.

International Agency for Research on Cancer. Infection with *Helicobacter pylori*. In IARC monographs on the evaluation of the carcinogenic risks to humans. Schistosomes, liver flukes and *Helicobacter pylori*; Lyon, France: World Health Organistaion. 61: 177-241, 1994.

Irigoyen JP, Besser D and Nagamine Y. Cytoskeleton reorganization induces the urokinase-type plasminogen activator gene via the Ras/extracellular signal-regulated kinase (ERK) signaling pathway. *J Biol Chem* 272: 1904-1909, 1997.

Irigoyen JP, Munoz-Canoves P, Montero L, Koziczak M and Nagamine Y. The plasminogen activator system: biology and regulation. *Cell Mol Life Sci* 56: 104-132, 1999.

**Isaacson PG and Du MQ**. Gastrointestinal lymphoma: where morphology meets molecular biology. *J Pathol* 205: 255-274, 2005.

Israel DA and Peek RM. pathogenesis of Helicobacter pylori-induced gastric inflammation. *Aliment Pharmacol Ther* 15: 1271-1290, 2001.

Ito H, Yonemura Y, Fujita H, Tsuchihara K, Kawamura T, Nojima N, Fujimura T, Nose H, Endo Y and Sasaki T. Prognostic relevance of urokinase-type plasminogen activator (uPA) and plasminogen activator inhibitors PAI-1 and PAI-2 in gastric cancer. *Virchows Arch* 427: 487-496, 1996.

Ito M. [Quantitative morphological studies of the human gastric parietal cell--a lectin histochemical study]. *Nippon Shokakibyo Gakkai Zasshi* 84: 863-867, 1987.

Ito S and WINCHESTER RJ. The fine structure of the gastric mucosa in the bat. J Cell Biol 16: 541-577, 1963.

Iwamoto J, Mizokami Y, Takahashi K, Nakajima K, Ohtsubo T, Miura S, Narasaka T, Takeyama H, Omata T, Shimokobe K, Ito M, Takehara H and Matsuoka T. Expressions of urokinase-type plasminogen activator, its receptor and plasminogen activator inhibitor-1 in gastric cancer cells and effects of Helicobacter pylori. *Scand J Gastroenterol* 40: 783-793, 2005.

Jackson LM, Wu KC, Mahida YR, Jenkins D and Hawkey CJ. Cyclooxygenase (COX) 1 and 2 in normal, inflamed, and ulcerated human gastric mucosa. *Gut* 47: 762-770, 2000.

Jaiswal M, LaRusso NF and Gores GJ. Nitric oxide in gastrointestinal epithelial cell carcinogenesis: linking inflammation to oncogenesis. *Am J Physiol Gastrointest Liver Physiol* 281: G626-G634, 2001.

Jeremy AH, Du Y, Dixon MF, Robinson PA and Crabtree JE. Protection against Helicobacter pylori infection in the Mongolian gerbil after prophylactic vaccination. *Microbes Infect* 8: 340-346, 2006.

Jo M, Thomas KS, O'Donnell DM and Gonias SL. Epidermal growth factor receptor-dependent and -independent cell-signaling pathways originating from the urokinase receptor. *J Biol Chem* 278: 1642-1646, 2003.

Jo M, Thomas KS, Marozkina N, Amin TJ, Silva CM, Parsons SJ and Gonias SL. Dynamic assembly of the urokinase-type plasminogen activator signaling receptor complex determines the mitogenic activity of urokinase-type plasminogen activator. *J Biol Chem* 280: 17449-17457, 2005.

Johnsen M, Lund LR, Romer J, Almholt K and Dano K. Cancer invasion and tissue remodeling: common themes in proteolytic matrix degradation. *Curr Opin Cell Biol* 10: 667-671, 1998.

Jones NL, Day AS, Jennings HA and Sherman PM. Helicobacter pylori induces gastric epithelial cell apoptosis in association with increased Fas receptor expression. *Infect Immun* 67: 4237-4242, 1999.

Jonsson K, Guo BP, Monstein HJ, Mekalanos JJ and Kronvall G. Molecular cloning and characterization of two Helicobacter pylori genes coding for plasminogen-binding proteins. *Proc Natl Acad Sci US A* 101: 1852-1857, 2004.

Juhan-Vague I, Alessi MC, Mavri A and Morange PE. Plasminogen activator inhibitor-1, inflammation, obesity, insulin resistance and vascular risk. *J Thromb Haemost* 1: 1575-1579, 2003.

Jungblut PR, Bumann D, Haas G, Zimny-Arndt U, Holland P, Lamer S, Siejak F, Aebischer A and Meyer TF. Comparative proteome analysis of Helicobacter pylori. *Mol Microbiol* 36: 710-725, 2000.

Kaikita K, Fogo AB, Ma L, Schoenhard JA, Brown NJ and Vaughan DE. Plasminogen activator inhibitor-1 deficiency prevents hypertension and vascular fibrosis in response to long-term nitric oxide synthase inhibition. *Circulation* 104: 839-844, 2001.

Kaneko H, Nakada K, Mitsuma T, Uchida K, Furusawa A, Maeda Y and Morise K. Helicobacter pylori infection induces a decrease in immunoreactive-somatostatin concentrations of human stomach. *Dig Dis Sci* 37: 409-416, 1992.

Kaneko T, Konno H, Baba M, Tanaka T and Nakamura S. Urokinase-type plasminogen activator expression correlates with tumor angiogenesis and poor outcome in gastric cancer. *Cancer Sci* 94: 43-49, 2003.

Kang JY, Teng CH, Wee A and Chen FC. Effect of capsaicin and chilli on ethanol induced gastric mucosal injury in the rat. *Gut* 36: 664-669, 1995.

Karam SM and Leblond CP. Dynamics of epithelial cells in the corpus of the mouse stomach. I. Identification of proliferative cell types and pinpointing of the stem cell. *Anat Rec* 236: 259-279, 1993a.

Karam SM and Leblond CP. Dynamics of epithelial cells in the corpus of the mouse stomach. II. Outward migration of pit cells. *Anat Rec* 236: 280-296, 1993b.

Karam SM and Leblond CP. Dynamics of epithelial cells in the corpus of the mouse stomach. III. Inward migration of neck cells followed by progressive transformation into zymogenic cells. *Anat Rec* 236: 297-313, 1993c.

Karam SM and Leblond CP. Dynamics of epithelial cells in the corpus of the mouse stomach. V. Behavior of entero-endocrine and caveolated cells: general conclusions on cell kinetics in the oxyntic epithelium. *Anat Rec* 236: 333-340, 1993d.

Karam SM. Dynamics of epithelial cells in the corpus of the mouse stomach. IV. Bidirectional migration of parietal cells ending in their gradual degeneration and loss. *Anat Rec* 236: 314-332, 1993e.

Karam SM, Straiton T, Hassan WM and Leblond CP. Defining epithelial cell progenitors in the human oxyntic mucosa. *Stem Cells* 21: 322-336, 2003.

Kargman S, Charleson S, Cartwright M, Frank J, Riendeau D, Mancini J, Evans J and O'Neill G. Characterization of Prostaglandin G/H Synthase 1 and 2 in rat, dog, monkey, and human gastrointestinal tracts. *Gastroenterology* 111: 445-454, 1996.

Karnes WE, Jr., Samloff IM, Siurala M, Kekki M, Sipponen P, Kim SW and Walsh JH. Positive serum antibody and negative tissue staining for Helicobacter pylori in subjects with atrophic body gastritis. *Gastroenterology* 101: 167-174, 1991.

Kasai S, Arimura H, Nishida M and Suyama T. Proteolytic cleavage of single-chain pro-urokinase induces conformational change which follows activation of the zymogen and reduction of its high affinity for fibrin. *J Biol Chem* 260: 12377-12381, 1985a.

Kasai S, Arimura H, Nishida M and Suyama T. Primary structure of single-chain pro-urokinase. J Biol Chem 260: 12382-12389, 1985b.

Kawasaki K, Hayashi Y, Wang Y, Suzuki S, Morita Y, Nakamura T, Narita K, Doe W, Itoh H and Kuroda Y. Expression of urokinase-type plasminogen activator receptor and plasminogen activator inhibitor-1 in gastric cancer. *J Gastroenterol Hepatol* 13: 936-944, 1998.

Kazes I, Delarue F, Hagege J, Bouzhir-Sima L, Rondeau E, Sraer JD and Nguyen G. Soluble latent membrane-type 1 matrix metalloprotease secreted by human mesangial cells is activated by urokinase. *Kidney Int* 54: 1976-1984, 1998.

Keates S, Keates AC, Warny M, Peek RM, Jr., Murray PG and Kelly CP. Differential activation of mitogen-activated protein kinases in AGS gastric epithelial cells by cag+ and cag- Helicobacter pylori. *J Immunol* 163: 5552-5559, 1999.

Keates S, Sougioultzis S, Keates AC, Zhao D, Peek RM, Jr., Shaw LM and Kelly CP. cag+ Helicobacter pylori induce transactivation of the epidermal growth factor receptor in AGS gastric epithelial cells. *J Biol Chem* 276: 48127-48134, 2001.

Keates S, Keates AC, Nath S, Peek RM, Jr. and Kelly CP. Transactivation of the epidermal growth factor receptor by cag+ Helicobacter pylori induces upregulation of the early growth response gene Egr-1 in gastric epithelial cells. *Gut* 54: 1363-1369, 2005.

Keeton MR, Curriden SA, van Zonneveld AJ and Loskutoff DJ. Identification of regulatory sequences in the type 1 plasminogen activator inhibitor gene responsive to transforming growth factor beta. *J Biol Chem* 266: 23048-23052, 1991.

Keski-Oja J, Lohi J, Tuuttila A, Tryggvason K and Vartio T. Proteolytic processing of the 72,000-Da type IV collagenase by urokinase plasminogen activator. *Exp Cell Res* 202: 471-476, 1992.

Khan ZE, Wang TC, Cui G, Chi AL and Dimaline R. Transcriptional regulation of the human trefoil factor, TFF1, by gastrin. *Gastroenterology* 125: 510-521, 2003.

Kim JM, Kim JS, Jung HC, Song IS and Kim CY. Apoptosis of human gastric epithelial cells via caspase-3 activation in response to Helicobacter pylori infection: possible involvement of neutrophils through tumor necrosis factor alpha and soluble Fas ligands. *Scand J Gastroenterol* 35: 40-48, 2000.

Kim MH, Yoo HS, Chang HJ, Hong MH, Kim HD, Chung IJ, Shin BA, Cho MJ, Ahn BW and Jung YD. Urokinase plasminogen activator receptor is upregulated by Helicobacter pylori in human gastric cancer AGS cells via ERK, JNK, and AP-1. *Biochem Biophys Res Commun* 333: 874-880, 2005.

Kim MH, Yoo HS, Kim MY, Jang HJ, Baek MK, Kim HR, Kim KK, Shin BA, Ahn BW and Jung YD. Helicobacter pylori stimulates urokinase plasminogen activator receptor expression and cell invasiveness through reactive oxygen species and NF-kappaB signaling in human gastric carcinoma cells. *Int J Mol Med* 19: 689-697, 2007.

Kirchheimer JC, Wojta J, Christ G, Hienert G and Binder BR. Mitogenic effect of urokinase on malignant and unaffected adjacent human renal cells. *Carcinogenesis* 9: 2121-2123, 1988.

Kirchheimer JC, Wojta J, Christ G and Binder BR. Functional inhibition of endogenously produced urokinase decreases cell proliferation in a human melanoma cell line. *Proc Natl Acad Sci U S A* 86: 5424-5428, 1989.

Kirton CM, Wang T and Dockray GJ. Regulation of parietal cell migration by gastrin in the mouse. Am J Physiol Gastrointest Liver Physiol 283: G787-G793, 2002.

Kitadai Y, Sasaki A, Ito M, Tanaka S, Oue N, Yasui W, Aihara M, Imagawa K, Haruma K and Chayama K. Helicobacter pylori infection influences expression of genes related to angiogenesis and invasion in human gastric carcinoma cells. *Biochem Biophys Res Commun* 311: 809-814, 2003.

Kitano M, Norlen P, Ding XQ, Nakamura S and Hakanson R. Long-lasting cholecystokinin(2) receptor blockade after a single subcutaneous injection of YF476 or YM022. *Br J Pharmacol* 130: 699-705, 2000.

Kivilaakso E. High plasma HCO3-protects gastric mucosa against acute ulceration in the rat. Gastroenterology 81: 921-927, 1981.

Kjoller L, Kanse SM, Kirkegaard T, Rodenburg KW, Ronne E, Goodman SL, Preissner KT, Ossowski L and Andreasen PA. Plasminogen activator inhibitor-1 represses integrin- and vitronectin-mediated cell migration independently of its function as an inhibitor of plasminogen activation. *Exp Cell Res* 232: 420-429, 1997.

**Kjoller L and Hall A**. Rac mediates cytoskeletal rearrangements and increased cell motility induced by urokinase-type plasminogen activator receptor binding to vitronectin. *J Cell Biol* 152: 1145-1157, 2001.

**Kjoller L**. The urokinase plasminogen activator receptor in the regulation of the actin cytoskeleton and cell motility. *Biol Chem* 383: 5-19, 2002.

Klinger KW, Winqvist R, Riccio A, Andreasen PA, Sartorio R, Nielsen LS, Stuart N, Stanislovitis P, Watkins P, Douglas R and . Plasminogen activator inhibitor type 1 gene is located at region q21.3-q22 of chromosome 7 and genetically linked with cystic fibrosis. *Proc Natl Acad Sci US A* 84: 8548-8552, 1987.

Kobayashi H, Schmitt M, Goretzki L, Chucholowski N, Calvete J, Kramer M, Gunzler WA, Janicke F and Graeff H. Cathepsin B efficiently activates the soluble and the tumor cell receptorbound form of the proenzyme urokinase-type plasminogen activator (Pro-uPA). *J Biol Chem* 266: 5147-5152, 1991.

Kobayashi H, Fujishiro S and Terao T. Impact of urokinase-type plasminogen activator and its inhibitor type 1 on prognosis in cervical cancer of the uterus. *Cancer Res* 54: 6539-6548, 1994.

Koh TJ, Goldenring JR, Ito S, Mashimo H, Kopin AS, Varro A, Dockray GJ and Wang TC. Gastrin deficiency results in altered gastric differentiation and decreased colonic proliferation in mice. *Gastroenterology* 113: 1015-1025, 1997.

Koletzko S, Richy F, Bontems P, Crone J, Kalach N, Monteiro ML, Gottrand F, Celinska-Cedro D, Roma-Giannikou E, Orderda G, Kolacek S, Urruzuno P, Martinez-Gomez MJ, Casswall T, Ashorn M, Bodanszky H and Megraud F. Prospective multicentre study on antibiotic resistance of Helicobacter pylori strains obtained from children living in Europe. *Gut* 55: 1711-1716, 2006.

Kondo S, Shinomura Y, Kanayama S, Higashimoto Y, Kiyohara T, Yasunaga Y, Kitamura S, Ueyama H, Imamura I, Fukui H and . Helicobacter pylori increases gene expression of hepatocyte growth factor in human gastric mucosa. *Biochem Biophys Res Commun* 210: 960-965, 1995.

Kono S, Rao JS, Bruner JM and Sawaya R. Immunohistochemical localization of plasminogen activator inhibitor type 1 in human brain tumors. *J Neuropathol Exp Neurol* 53: 256-262, 1994.

Konturek PC, Kania J, Kukharsky V, Ocker S, Hahn EG and Konturek SJ. Influence of gastrin on the expression of cyclooxygenase-2, hepatocyte growth factor and apoptosis-related proteins in gastric epithelial cells. *J Physiol Pharmacol* 54: 17-32, 2003.

Konturek SJ, Radecki T, Brzozowski T, Piastucki I, Zmuda A and mbinska-Kiec A. Aspirininduced gastric ulcers in cats. Prevention by prostacyclin. *Dig Dis Sci* 26: 1003-1012, 1981a.

Konturek SJ, Piastucki I, Brzozowski T, Radecki T, mbinska-Kiec A, Zmuda A and Gryglewski R. Role of prostaglandins in the formation of aspirin-induced gastric ulcers. *Gastroenterology* 80: 4-9, 1981b.

Konturek SJ and Pawlik W. Physiology and pharmacology of prostaglandins. *Dig Dis Sci* 31: 6S-19S, 1986.

Koshelnick Y, Ehart M, Hufnagl P, Heinrich PC and Binder BR. Urokinase receptor is associated with the components of the JAK1/STAT1 signaling pathway and leads to activation of this pathway upon receptor clustering in the human kidney epithelial tumor cell line TCL-598. *J Biol Chem* 272: 28563-28567, 1997.

Kotloff KL, Sztein MB, Wasserman SS, Losonsky GA, DiLorenzo SC and Walker RI. Safety and immunogenicity of oral inactivated whole-cell Helicobacter pylori vaccine with adjuvant among volunteers with or without subclinical infection. *Infect Immun* 69: 3581-3590, 2001.

Kruithof EK, Tran-Thang C and Bachmann F. Studies on the release of a plasminogen activator inhibitor by human platelets. *Thromb Haemost* 55: 201-205, 1986.

Kruithof EK. Plasminogen activator inhibitors--a review. Enzyme 40: 113-121, 1988.

Kuck D, Kolmerer B, Iking-Konert C, Krammer PH, Stremmel W and Rudi J. Vacuolating cytotoxin of Helicobacter pylori induces apoptosis in the human gastric epithelial cell line AGS. *Infect Immun* 69: 5080-5087, 2001.

Kugler MC, Wei Y and Chapman HA. Urokinase receptor and integrin interactions. Curr Pharm Des 9: 1565-1574, 2003.

Kuhn W, Pache L, Schmalfeldt B, Dettmar P, Schmitt M, Janicke F and Graeff H. Urokinase (uPA) and PAI-1 predict survival in advanced ovarian cancer patients (FIGO III) after radical surgery and platinum-based chemotherapy. *Gynecol Oncol* 55: 401-409, 1994.

Kuhn W, Schmalfeldt B, Reuning U, Pache L, Berger U, Ulm K, Harbeck N, Spathe K, Dettmar P, Hofler H, Janicke F, Schmitt M and Graeff H. Prognostic significance of urokinase (uPA) and its inhibitor PAI-1 for survival in advanced ovarian carcinoma stage FIGO IIIc. Br J Cancer 79: 1746-1751, 1999.

Kuipers EJ, Uyterlinde AM, Pena AS, Roosendaal R, Pals G, Nelis GF, Festen HP and Meuwissen SG. Long-term sequelae of Helicobacter pylori gastritis. *Lancet* 345: 1525-1528, 1995.

Kumar S and Baglioni C. Protection from tumor necrosis factor-mediated cytolysis by overexpression of plasminogen activator inhibitor type-2. *J Biol Chem* 266: 20960-20964, 1991.

Kunz C, Pebler S, Otte J and von der AD. Differential regulation of plasminogen activator and inhibitor gene transcription by the tumor suppressor p53. *Nucleic Acids Res* 23: 3710-3717, 1995.

Labenz J and Stolte M. [Long-term inhibition of gastric acid secretion. Evaluation and healing of Helicobacter pylori infection?]. Leber Magen Darm 26: 241-242, 1996.

Lacy ER and Ito S. Rapid epithelial restitution of the rat gastric mucosa after ethanol injury. Lab Invest 51: 573-583, 1984.

Lacy ER, Morris GP and Cohen MM. Rapid repair of the surface epithelium in human gastric mucosa after acute superficial injury. *J Clin Gastroenterol* 17 Suppl 1: S125-S135, 1993.

Lagoa CE, Vodovotz Y, Stolz DB, Lhuillier F, McCloskey C, Gallo D, Yang R, Ustinova E, Fink MP, Billiar TR and Mars WM. The role of hepatic type 1 plasminogen activator inhibitor (PAI-1) during murine hemorrhagic shock. *Hepatology* 42: 390-399, 2005.

Langer DJ, Kuo A, Kariko K, Ahuja M, Klugherz BD, Ivanics KM, Hoxie JA, Williams WV, Liang BT, Cines DB and . Regulation of the endothelial cell urokinase-type plasminogen activator receptor. Evidence for cyclic AMP-dependent and protein kinase C-dependent pathways. *Circ Res* 72: 330-340, 1993.

Langhans N, Rindi G, Chiu M, Rehfeld JF, Ardman B, Beinborn M and Kopin AS. Abnormal gastric histology and decreased acid production in cholecystokinin-B/gastrin receptor-deficient mice. *Gastroenterology* 112: 280-286, 1997.

Lauffenburger DA and Horwitz AF. Cell migration: a physically integrated molecular process. *Cell* 84: 359-369, 1996.

Laurell M, Christensson A, Abrahamsson PA, Stenflo J and Lilja H. Protein C inhibitor in human body fluids. Seminal plasma is rich in inhibitor antigen deriving from cells throughout the male reproductive system. *J Clin Invest* 89: 1094-1101, 1992.

Lawrence D, Strandberg L, Grundstrom T and Ny T. Purification of active human plasminogen activator inhibitor 1 from Escherichia coli. Comparison with natural and recombinant forms purified from eucaryotic cells. *Eur J Biochem* 186: 523-533, 1989.

Lawrence DA, Berkenpas MB, Palaniappan S and Ginsburg D. Localization of vitronectin binding domain in plasminogen activator inhibitor-1. *J Biol Chem* 269: 15223-15228, 1994.

Lawrence DA, Palaniappan S, Stefansson S, Olson ST, Francis-Chmura AM, Shore JD and Ginsburg D. Characterization of the binding of different conformational forms of plasminogen activator inhibitor-1 to vitronectin. Implications for the regulation of pericellular proteolysis. *J Biol Chem* 272: 7676-7680, 1997.

Le Magueresse-Battistoni B, Pernod G, Kolodie L, Morera AM and Benahmed M. Tumor necrosis factor-alpha regulates plasminogen activator inhibitor-1 in rat testicular peritubular cells. Endocrinology 138: 1097-1105, 1997.

Lechago J and Correa P. Prolonged achlorhydria and gastric neoplasia: is there a causal relationship? *Gastroenterology* 104: 1554-1557, 1993.

Lee A, Fox JG, Otto G and Murphy J. A small animal model of human Helicobacter pylori active chronic gastritis. *Gastroenterology* 99: 1315-1323, 1990.

Lee A, O'Rourke J, De Ungria MC, Robertson B, Daskalopoulos G and Dixon MF. A standardized mouse model of Helicobacter pylori infection: introducing the Sydney strain. *Gastroenterology* 112: 1386-1397, 1997.

Lee J, Duk J, I, Gyo PC, Han JW and Young LH. Autotaxin stimulates urokinase-type plasminogen activator expression through phosphoinositide 3-kinase-Akt-nuclear [corrected] factor kappa B signaling cascade in human melanoma cells. *Melanoma Res* 16: 445-452, 2006a.

Lee KH, Choi EY, Kim MK, Hyun MS, Jang BI, Kim TN, Kim SW, Song SK, Kim JH and Kim JR. Regulation of hepatocyte growth factor-mediated urokinase plasminogen activator secretion by MEK/ERK activation in human stomach cancer cell lines. *Exp Mol Med* 38: 27-35, 2006b.

Lee KH, Choi EY, Hyun MS, Jang BI, Kim TN, Kim SW, Song SK, Kim JH and Kim JR. Hepatocyte growth factor/c-met signaling in regulating urokinase plasminogen activator in human stomach cancer: A potential therapeutic target for human stomach cancer. *Korean J Intern Med* 21: 20-27, 2006c.

Levi S, Beardshall K, Swift I, Foulkes W, Playford R, Ghosh P and Calam J. Antral Helicobacter pylori, hypergastrinaemia, and duodenal ulcers: effect of eradicating the organism. *BMJ* 299: 1504-1505, 1989.

Levin EG and Santell L. Conversion of the active to latent plasminogen activator inhibitor from human endothelial cells. *Blood* 70: 1090-1098, 1987.

Li F, Goncalves J, Faughnan K, Steiner MG, Pagan-Charry I, Esposito D, Chin B, Providence KM, Higgins PJ and Staiano-Coico L. Targeted inhibition of wound-induced PAI-1 expression alters migration and differentiation in human epidermal keratinocytes. *Exp Cell Res* 258: 245-253, 2000.

Li Q, Karam SM and Gordon JI. Diphtheria toxin-mediated ablation of parietal cells in the stomach of transgenic mice. *J Biol Chem* 271: 3671-3676, 1996.

Limongi P, Resnati M, Hernandez-Marrero L, Cremona O, Blasi F and Fazioli F. Biosynthesis and apical localization of the urokinase receptor in polarized MDCK epithelial cells. *FEBS Lett* 369: 207-211, 1995.

Linz B and Schuster SC. Genomic diversity in Helicobacter and related organisms. *Res Microbiol* 158: 737-744, 2007.

Liotta LA, Goldfarb RH, Brundage R, Siegal GP, Terranova V and Garbisa S. Effect of plasminogen activator (urokinase), plasmin, and thrombin on glycoprotein and collagenous components of basement membrane. *Cancer Res* 41: 4629-4636, 1981.

Liotta LA, Stetler-Stevenson WG and Steeg PS. Cancer invasion and metastasis: positive and negative regulatory elements. *Cancer Invest* 9: 543-551, 1991.

Liu D, Aguirre GJ, Estrada Y and Ossowski L. EGFR is a transducer of the urokinase receptor initiated signal that is required for in vivo growth of a human carcinoma. *Cancer Cell* 1: 445-457, 2002.

Lobo GL, Zambaldi TM, de Labio RW, Silva LC, de Arruda Cardoso SM and Marques Payao SL. Interleukin-6 polymorphism and Helicobacter pylori infection in Brazilian adult patients with chronic gastritis. *Clin Exp Med* 5: 112-116, 2005.

Lopez-Belmonte J, Whittle BJ and Moncada S. The actions of nitric oxide donors in the prevention or induction of injury to the rat gastric mucosa. *Br J Pharmacol* 108: 73-78, 1993.

Lorenz RG and Gordon JI. Use of transgenic mice to study regulation of gene expression in the parietal cell lineage of gastric units. *J Biol Chem* 268: 26559-70, 1993.

Loskutoff DJ and Edgington TS. An inhibitor of plasminogen activator in rabbit endothelial cells. J Biol Chem 256: 4142-4145, 1981.

Loskutoff DJ, Sawdey M and Mimuro J. Type 1 plasminogen activator inhibitor. Prog Hemost Thromb 9: 87-115, 1989.

Loskutoff DJ, Sawdey M, Keeton M and Schneiderman J. Regulation of PAI-1 gene expression in vivo. *Thromb Haemost* 70: 135-137, 1993.

Loskutoff DJ and Samad F. The adipocyte and hemostatic balance in obesity: studies of PAI-1. Arterioscler Thromb Vasc Biol 18: 1-6, 1998.

Loskutoff DJ, Fujisawa K and Samad F. The fat mouse. A powerful genetic model to study hemostatic gene expression in obesity/NIDDM. *Ann N Y Acad Sci* 902: 272-281, 2000.

Lu H, Yamaoka Y and Graham DY. Helicobacter pylori virulence factors: facts and fantasies. Curr Opin Gastroenterol 21: 653-659, 2005a.

Lu W, Pan K, Zhang L, Lin D, Miao X and You W. Genetic polymorphisms of interleukin (IL)-1B, IL-1RN, IL-8, IL-10 and tumor necrosis factor {alpha} and risk of gastric cancer in a Chinese population. *Carcinogenesis* 26: 631-636, 2005b.

Lund LR, Romer J, Bugge TH, Nielsen BS, Frandsen TL, Degen JL, Stephens RW and Dano K. Functional overlap between two classes of matrix-degrading proteases in wound healing. *EMBO J* 18: 4645-4656, 1999.

Lundberg JO, Weitzberg E and Gladwin MT. The nitrate-nitrite-nitric oxide pathway in physiology and therapeutics. *Nat Rev Drug Discov* 7: 156-167, 2008.

Lupu F, Bergonzelli GE, Heim DA, Cousin E, Genton CY, Bachmann F and Kruithof EK. Localization and production of plasminogen activator inhibitor-1 in human healthy and atherosclerotic arteries. *Arterioscler Thromb* 13: 1090-1100, 1993.

Luria EA, Samoylina NL, Gerasimov YV and Chertkov JL. Proliferation of hemopoietic stem cells in culture of embryonal liver of mice. *J Cell Physiol* 78: 461-464, 1971.

Luyendyk JP, Mattes WB, Burgoon LD, Zacharewski TR, Maddox JF, Cosma GN, Ganey PE and Roth RA. Gene expression analysis points to hemostasis in livers of rats cotreated with lipopolysaccharide and ranitidine. *Toxicol Sci* 80: 203-213, 2004.

Macchia G, Massone A, Burroni D, Covacci A, Censini S and Rappuoli R. The Hsp60 protein of Helicobacter pylori: structure and immune response in patients with gastroduodenal diseases. *Mol Microbiol* 9: 645-652, 1993.

Machado JC, Figueiredo C, Canedo P, Pharoah P, Carvalho R, Nabais S, Castro AC, Campos ML, van Doorn LJ, Caldas C, Seruca R, Carneiro F and Sobrinho-Simoes M. A proinflammatory genetic profile increases the risk for chronic atrophic gastritis and gastric carcinoma. *Gastroenterology* 125: 364-371, 2003.

MacNaughton WK, Cirino G and Wallace JL. Endothelium-derived relaxing factor (nitric oxide) has protective actions in the stomach. *Life Sci* 45: 1869-1876, 1989.

Mahdavi J, Sonden B, Hurtig M, Olfat FO, Forsberg L, Roche N, Angstrom J, Larsson T, Teneberg S, Karlsson KA, Altraja S, Wadstrom T, Kersulyte D, Berg DE, Dubois A, Petersson C, Magnusson KE, Norberg T, Lindh F, Lundskog BB, Arnqvist A, Hammarstrom L and Boren T. Helicobacter pylori SabA adhesin in persistent infection and chronic inflammation. *Science* 297: 573-578, 2002.

Mandell L, Moran AP, Cocchiarella A, Houghton J, Taylor N, Fox JG, Wang TC and Kurt-Jones EA. Intact gram-negative Helicobacter pylori, Helicobacter felis, and Helicobacter hepaticus bacteria activate innate immunity via toll-like receptor 2 but not toll-like receptor 4. *Infect Immun* 72: 6446-6454, 2004.

Mandriota SJ, Seghezzi G, Vassalli JD, Ferrara N, Wasi S, Mazzieri R, Mignatti P and Pepper MS. Vascular endothelial growth factor increases urokinase receptor expression in vascular endothelial cells. *J Biol Chem* 270: 9709-9716, 1995.

Mannick EE, Bravo LE, Zarama G, Realpe JL, Zhang XJ, Ruiz B, Fontham ET, Mera R, Miller MJ and Correa P. Inducible nitric oxide synthase, nitrotyrosine, and apoptosis in Helicobacter pylori gastritis: effect of antibiotics and antioxidants. *Cancer Res* 56: 3238-3243, 1996.

Maquerlot F, Galiacy S, Malo M, Guignabert C, Lawrence DA, d'Ortho MP and Barlovatz-Meimon G. Dual role for plasminogen activator inhibitor type 1 as soluble and as matricellular regulator of epithelial alveolar cell wound healing. *Am J Pathol* 169: 1624-1632, 2006. Mars WM, Zarnegar R and Michalopoulos GK. Activation of hepatocyte growth factor by the plasminogen activators uPA and tPA. *Am J Pathol* 143: 949-958, 1993.

Marshall BJ and Warren JR. Unidentified curved bacilli in the stomach of patients with gastritis and peptic ulceration. *Lancet* 1: 1311-1315, 1984.

Matsukura N, Suzuki K, Kawachi T, Aoyagi M, Sugimura T, Kitaoka H, Numajiri H, Shirota A, Itabashi M and Hirota T. Distribution of marker enzymes and mucin in intestinal metaplasia in human stomach and relation to complete and incomplete types of intestinal metaplasia to minute gastric carcinomas. J Natl Cancer Inst 65: 231-240, 1980.

Mavri A, Stegnar M, Krebs M, Sentocnik JT, Geiger M and Binder BR. Impact of adipose tissue on plasma plasminogen activator inhibitor-1 in dieting obese women. *Arterioscler Thromb Vasc Biol* 19: 1582-1587, 1999.

May AE, Kanse SM, Lund LR, Gisler RH, Imhof BA and Preissner KT. Urokinase receptor (CD87) regulates leukocyte recruitment via beta 2 integrins in vivo. *J Exp Med* 188: 1029-1037, 1998.

McCaig C, Duval C, Hemers E, Steele I, Pritchard DM, Przemeck S, Dimaline R, Ahmed S, Bodger K, Kerrigan DD, Wang TC, Dockray GJ and Varro A. The role of matrix metalloproteinase-7 in redefining the gastric microenvironment in response to Helicobacter pylori. *Gastroenterology* 130: 1754-1763, 2006.

McCarthy CJ, Crofford LJ, Greenson J and Scheiman JM. Cyclooxygenase-2 expression in gastric antral mucosa before and after eradication of Helicobacter pylori infection. Am J Gastroenterol 94: 1218-1223, 1999.

McColl KE, el-Omar E, Gillen D and Banerjee S. The role of Helicobacter pylori in the pathophysiology of duodenal ulcer disease and gastric cancer. *Semin Gastrointest Dis* 8: 142-155, 1997.

McColl KE, el-Omar E and Gillen D. Helicobacter pylori gastritis and gastric physiology. *Gastroenterol Clin North Am* 29: 687-703, viii, 2000.

McColl KE, Gillen D and el-Omar E. The role of gastrin in ulcer pathogenesis. *Baillieres Best Pract Res Clin Gastroenterol* 14: 13-26, 2000.

McGuigan JE. Gastric mucosal intracellular localization of gastrin by immunofluorescence. *Gastroenterology* 55: 315-327, 1968.

McKaig BC, Makh SS, Hawkey CJ, Podolsky DK and Mahida YR. Normal human colonic subepithelial myofibroblasts enhance epithelial migration (restitution) via TGF-beta3. *Am J Physiol* 276: G1087-G1093, 1999.

McMahon GA, Petitclerc E, Stefansson S, Smith E, Wong MK, Westrick RJ, Ginsburg D, Brooks PC and Lawrence DA. Plasminogen activator inhibitor-1 regulates tumor growth and angiogenesis. *J Biol Chem* 276: 33964-33968, 2001.

Meyer-ter-Vehn T, Covacci A, Kist M and Pahl HL. Helicobacter pylori activates mitogenactivated protein kinase cascades and induces expression of the proto-oncogenes c-fos and c-jun. J Biol Chem 275: 16064-16072, 2000.

Michetti P, Kreiss C, Kotloff KL, Porta N, Blanco JL, Bachmann D, Herranz M, Saldinger PF, Corthesy-Theulaz I, Losonsky G, Nichols R, Simon J, Stolte M, Ackerman S, Monath TP and Blum AL. Oral immunization with urease and Escherichia coli heat-labile enterotoxin is safe and immunogenic in Helicobacter pylori-infected adults. *Gastroenterology* 116: 804-812, 1999.

Miehlke S, Kirsch C, gha-Amiri K, Gunther T, Lehn N, Malfertheiner P, Stolte M, Ehninger G and Bayerdorffer E. The Helicobacter pylori vacA s1, m1 genotype and cagA is associated with gastric carcinoma in Germany. *Int J Cancer* 87: 322-327, 2000.

Mills JC, Syder AJ, Hong CV, Guruge JL, Raaii F and Gordon JI. A molecular profile of the mouse gastric parietal cell with and without exposure to Helicobacter pylori. *Proc Natl Acad Sci US A* 98: 13687-13692, 2001.

Mimuro H, Suzuki T, Tanaka J, Asahi M, Haas R and Sasakawa C. Grb2 is a key mediator of helicobacter pylori CagA protein activities. *Mol Cell* 10: 745-755, 2002.

**Mimuro J and Loskutoff DJ**. Purification of a protein from bovine plasma that binds to type 1 plasminogen activator inhibitor and prevents its interaction with extracellular matrix. Evidence that the protein is vitronectin. *J Biol Chem* 264: 936-939, 1989.

Miralles F, Parra M, Caelles C, Nagamine Y, Felez J and Munoz-Canoves P. UV irradiation induces the murine urokinase-type plasminogen activator gene via the c-Jun N-terminal kinase signaling pathway: requirement of an AP1 enhancer element. *Mol Cell Biol* 18: 4537-4547, 1998.

Mitsuno Y, Yoshida H, Maeda S, Ogura K, Hirata Y, Kawabe T, Shiratori Y and Omata M. Helicobacter pylori induced transactivation of SRE and AP-1 through the ERK signalling pathway in gastric cancer cells. *Gut* 49: 18-22, 2001.

Miyake T, Suzaki T and Oishi M. Correlation of gastric ulcer healing features by endoscopy, stereoscopic microscopy, and histology, and a reclassification of the epithelial regenerative process. *Dig Dis Sci* 25: 8-14, 1980.

Mobley HL, Hu LT and Foxal PA. Helicobacter pylori urease: properties and role in pathogenesis. Scand J Gastroenterol Suppl 187: 39-46, 1991.

**Mobley HL**. Helicobacter pylori factors associated with disease development. *Gastroenterology* 113: S21-S28, 1997.

Mochan E and Keler T. Plasmin degradation of cartilage proteoglycan. *Biochim Biophys Acta* 800: 312-315, 1984.

Modlin IM, Kidd M, Marks IN and Tang LH. The pivotal role of John S. Edkins in the discovery of gastrin. *World J Surg* 21: 226-234, 1997.

Morange PE, Aubert J, Peiretti F, Lijnen HR, Vague P, Verdier M, Negrel R, Juhan-Vague I and Alessi MC. Glucocorticoids and insulin promote plasminogen activator inhibitor 1 production by human adipose tissue. *Diabetes* 48: 890-895, 1999.

Morita S, Sato A, Hayakawa H, Ihara H, Urano T, Takada Y and Takada A. Cancer cells overexpress mRNA of urokinase-type plasminogen activator, its receptor and inhibitors in human non-small-cell lung cancer tissue: analysis by Northern blotting and in situ hybridization. *Int J Cancer* 78: 286-292, 1998.

Moss SF, Legon S, Bishop AE, Polak JM and Calam J. Effect of Helicobacter pylori on gastric somatostatin in duodenal ulcer disease. *Lancet* 340: 930-932, 1992.

Moss SF, Calam J, Agarwal B, Wang S and Holt PR. Induction of gastric epithelial apoptosis by Helicobacter pylori. *Gut* 38: 498-501, 1996.

Moss SF, Sordillo EM, Abdalla AM, Makarov V, Hanzely Z, Perez-Perez GI, Blaser MJ and Holt PR. Increased gastric epithelial cell apoptosis associated with colonization with cagA + Helicobacter pylori strains. *Cancer Res* 61: 1406-1411, 2001.

Mottonen J, Strand A, Symersky J, Sweet RM, Danley DE, Geoghegan KF, Gerard RD and Goldsmith EJ. Structural basis of latency in plasminogen activator inhibitor-1. *Nature* 355: 270-273, 1992.

Mowat C, Williams C, Gillen D, Hossack M, Gilmour D, Carswell A, Wirz A, Preston T and McColl KE. Omeprazole, Helicobacter pylori status, and alterations in the intragastric milieu facilitating bacterial N-nitrosation. *Gastroenterology* 119: 339-347, 2000.

Mowat C and McColl KE. Alterations in intragastric nitrite and vitamin C levels during acid inhibitory therapy. *Best Pract Res Clin Gastroenterol* 15: 523-537, 2001.

Mozsik G and Javor T. A biochemical and pharmacological approach to the genesis of ulcer disease. I. A model study of ethanol-induced injury to gastric mucosa in rats. *Dig Dis Sci* 33: 92-105, 1988.

Mozsik G, Szolcsanyi J and Racz I. Gastroprotection induced by capsaicin in healthy human subjects. *World J Gastroenterol* 11: 5180-5184, 2005.

Mulholland G, Ardill JE, Fillmore D, Chittajallu RS, Fullarton GM and McColl KE. Helicobacter pylori related hypergastrinaemia is the result of a selective increase in gastrin 17. *Gut* 34: 757-761, 1993.

Muller JE, Tofler GH and Stone PH. Circadian variation and triggers of onset of acute cardiovascular disease. *Circulation* 79: 733-743, 1989.

Murayama H and Bang NU. Incorporation of plasminogen activator inhibitor into fibrin, an alternative regulatory pathway for fibrinolysis. *Thromb Haemost* 58: 1643, 1987 (abstr).

Nagata A, Ito M, Iwata N, Kuno J, Takano H, Minowa O, Chihara K, Matsui T and Noda T. G protein-coupled cholecystokinin-B/gastrin receptors are responsible for physiological cell growth of the stomach mucosa in vivo. *Proc Natl Acad Sci US A* 93: 11825-11830, 1996.

Nakamura M, Konno H, Tanaka T, Maruo Y, Nishino N, Aoki K, Baba S, Sakaguchi S, Takada Y and Takada A. Possible role of plasminogen activator inhibitor 2 in the prevention of the metastasis of gastric cancer tissues. *Thromb Res* 65: 709-719, 1992.

Naldini L, Vigna E, Narsimhan RP, Gaudino G, Zarnegar R, Michalopoulos GK and Comoglio PM. Hepatocyte growth factor (HGF) stimulates the tyrosine kinase activity of the receptor encoded by the proto-oncogene c-MET. *Oncogene* 6: 501-504, 1991.

Naldini L, Tamagnone L, Vigna E, Sachs M, Hartmann G, Birchmeier W, Daikuhara Y, Tsubouchi H, Blasi F and Comoglio PM. Extracellular proteolytic cleavage by urokinase is required for activation of hepatocyte growth factor/scatter factor. *EMBO J* 11: 4825-4833, 1992.

Naumann M, Wessler S, Bartsch C, Wieland B, Covacci A, Haas R and Meyer TF. Activation of activator protein 1 and stress response kinases in epithelial cells colonized by Helicobacter pylori encoding the cag pathogenicity island. *J Biol Chem* 274: 31655-31662, 1999.

Naumann M and Crabtree JE. Helicobacter pylori-induced epithelial cell signalling in gastric carcinogenesis. *Trends Microbiol* 12: 29-36, 2004.

Navaglia F, Basso D, Zambon CF, Ponzano E, Caenazzo L, Gallo N, Falda A, Belluco C, Fogar P, Greco E, Di MF, Rugge M and Plebani M. Interleukin 12 gene polymorphisms enhance gastric cancer risk in H pylori infected individuals. *J Med Genet* 42: 503-510, 2005.

Nekarda H, Schmitt M, Ulm K, Wenninger A, Vogelsang H, Becker K, Roder JD, Fink U and Siewert JR. Prognostic impact of urokinase-type plasminogen activator and its inhibitor PAI-1 in completely resected gastric cancer. *Cancer Res* 54: 2900-2907, 1994.

Nguyen DH, Hussaini IM and Gonias SL. Binding of urokinase-type plasminogen activator to its receptor in MCF-7 cells activates extracellular signal-regulated kinase 1 and 2 which is required for increased cellular motility. *J Biol Chem* 273: 8502-8507, 1998.

Nguyen NV, Gleeson PA, Courtois-Coutry N, Caplan MJ and Van D, I. Gastric parietal cell acid secretion in mice can be regulated independently of H/K ATPase endocytosis. *Gastroenterology* 127: 145-154, 2004.

Nielsen LS, Kellerman GM, Behrendt N, Picone R, Dano K and Blasi F. A 55,000-60,000 Mr receptor protein for urokinase-type plasminogen activator. Identification in human tumor cell lines and partial purification. *J Biol Chem* 263: 2358-2363, 1988.

Nomura A and Stemmermann GN. Helicobacter pylori and gastric cancer. *J Gastroenterol Hepatol* 8: 294-303, 1993.

Nomura AM, Perez-Perez GI, Lee J, Stemmermann G and Blaser MJ. Relation between Helicobacter pylori cagA status and risk of peptic ulcer disease. Am J Epidemiol 155: 1054-1059, 2002.

Nordt TK, Sawa H, Fujii S and Sobel BE. Induction of plasminogen activator inhibitor type-1 (PAI-1) by proinsulin and insulin in vivo. *Circulation* 91: 764-770, 1995.

Novak U, Cocks BG and Hamilton JA. A labile repressor acts through the NFkB-like binding sites of the human urokinase gene. *Nucleic Acids Res* 19: 3389-3393, 1991.

Nozawa Y, Nishihara K, Peek RM, Nakano M, Uji T, Ajioka H, Matsuura N and Miyake H. Identification of a signaling cascade for interleukin-8 production by Helicobacter pylori in human gastric epithelial cells. *Biochem Pharmacol* 64: 21-30, 2002.

Numano F, Kishi Y, Ashikaga T, Hata A, Makita T and Watanabe R. What effect does controlling platelets have on atherosclerosis? Ann N Y Acad Sci 748: 383-392, 1995.

Nykjaer A, Petersen CM, Christensen EI, Davidsen O and Gliemann J. Urokinase receptors in human monocytes. *Biochim Biophys Acta* 1052: 399-407, 1990.

Nykjaer A, Petersen CM, Moller B, Jensen PH, Moestrup SK, Holtet TL, Etzerodt M, Thogersen HC, Munch M, Andreasen PA and . Purified alpha 2-macroglobulin receptor/LDL receptor-related protein binds urokinase.plasminogen activator inhibitor type-1 complex. Evidence that the alpha 2-macroglobulin receptor mediates cellular degradation of urokinase receptor-bound complexes. *J Biol Chem* 267: 14543-14546, 1992.

Nykjaer A, Moller B, Todd RF, III, Christensen T, Andreasen PA, Gliemann J and Petersen CM. Urokinase receptor. An activation antigen in human T lymphocytes. *J Immunol* 152: 505-516, 1994.

Nykjaer A, Conese M, Christensen EI, Olson D, Cremona O, Gliemann J and Blasi F. Recycling of the urokinase receptor upon internalization of the uPA:serpin complexes. *EMBO J* 16: 2610-2620, 1997.

Oda T, Jung YO, Kim HS, Cai X, Lopez-Guisa JM, Ikeda Y and Eddy AA. PAI-1 deficiency attenuates the fibrogenic response to ureteral obstruction. *Kidney Int* 60: 587-596, 2001.

**Odekon LE, Sato Y and Rifkin DB**. Urokinase-type plasminogen activator mediates basic fibroblast growth factor-induced bovine endothelial cell migration independent of its proteolytic activity. *J Cell Physiol* 150: 258-263, 1992.

Odenbreit S, Puls J, Sedlmaier B, Gerland E, Fischer W and Haas R. Translocation of Helicobacter pylori CagA into gastric epithelial cells by type IV secretion. *Science* 287: 1497-1500, 2000.

Ogawa M, LaRue AC and Drake CJ. Hematopoietic origin of fibroblasts/myofibroblasts: Its pathophysiologic implications. *Blood* 108: 2893-2896, 2006.

**Ohba K, Miyata Y, Kanda S, Koga S, Hayashi T and Kanetake H**. Expression of urokinase-type plasminogen activator, urokinase-type plasminogen activator receptor and plasminogen activator inhibitors in patients with renal cell carcinoma: correlation with tumor associated macrophage and prognosis. *J Urol* 174: 461-465, 2005.

Ohno T, Katori M, Nishiyama K and Saigenji K. Direct observation of microcirculation of the basal region of rat gastric mucosa. *J Gastroenterol* 30: 557-564, 1995.

Ohno T, Katori M, Majima M, Saeki T, Boku K, Nishiyama K, Hayashi H and Saigenji K. Dilatation and constriction of rat gastric mucosal microvessels through prostaglandin EP2 and EP3 receptors. *Aliment Pharmacol Ther* 13: 1243-1250, 1999.

Ohyauchi M, Imatani A, Yonechi M, Asano N, Miura A, Iijima K, Koike T, Sekine H, Ohara S and Shimosegawa T. The polymorphism interleukin 8 -251 A/T influences the susceptibility of Helicobacter pylori related gastric diseases in the Japanese population. *Gut* 54: 330-335, 2005.

Okamoto T, Schlegel A, Scherer PE and Lisanti MP. Caveolins, a family of scaffolding proteins for organizing "preassembled signaling complexes" at the plasma membrane. *J Biol Chem* 273: 5419-5422, 1998.

**Okuda K.** Discovery of vitamin B12 in the liver and its absorption factor in the stomach: a historical review. *J Gastroenterol Hepatol* 14: 301-308, 1999.

Okumura Y, Kamikubo Y, Curriden SA, Wang J, Kiwada T, Futaki S, Kitagawa K and Loskutoff DJ. Kinetic analysis of the interaction between vitronectin and the urokinase receptor. J Biol Chem 277: 9395-9404, 2002.

Olson D, Pollanen J, Hoyer-Hansen G, Ronne E, Sakaguchi K, Wun TC, Appella E, Dano K and Blasi F. Internalization of the urokinase-plasminogen activator inhibitor type-1 complex is mediated by the urokinase receptor. *J Biol Chem* 267: 9129-9133, 1992.

Orth K, Madison EL, Gething MJ, Sambrook JF and Herz J. Complexes of tissue-type plasminogen activator and its serpin inhibitor plasminogen-activator inhibitor type 1 are internalized by means of the low density lipoprotein receptor-related protein/alpha 2-macroglobulin receptor. *Proc* Natl Acad Sci US A 89: 7422-7426, 1992.

Ossowski L and guirre-Ghiso JA. Urokinase receptor and integrin partnership: coordination of signaling for cell adhesion, migration and growth. *Curr Opin Cell Biol* 12: 613-620, 2000.

Pagliocca A, Khan ZE, Hemers E, Pennington SR, Dimaline R, Varro A and Dockray GJ. Gastrin regulates parietal cell maturation: identification of ezrin as a novel target. *Gastroenterology* 124: A-22, 2003.

Palmer JB, Cuss FM, Mulderry PK, Ghatei MA, Springall DR, Cadieux A, Bloom SR, Polak JM and Barnes PJ. Calcitonin gene-related peptide is localised to human airway nerves and potently constricts human airway smooth muscle. *Br J Pharmacol* 91: 95-101, 1987.

Palmieri D, Lee JW, Juliano RL and Church FC. Plasminogen activator inhibitor-1 and -3 increase cell adhesion and motility of MDA-MB-435 breast cancer cells. *J Biol Chem* 277: 40950-40957, 2002.

Pan LR, Tang Q, Fu Q, Hu BR, Xiang JZ and Qian JQ. Roles of nitric oxide in protective effect of berberine in ethanol-induced gastric ulcer mice. *Acta Pharmacol Sin* 26: 1334-1338, 2005.

Papini E, Satin B, de BM, Molinari M, Arico B, Galli C, Telford JR, Rappuoli R and Montecucco C. Action site and cellular effects of cytotoxin VacA produced by Helicobacter pylori. *Folia Microbiol (Praha)* 43: 279-284, 1998.

**Parra M, Lluis F, Miralles F, Caelles C and Munoz-Canoves P**. The cJun N-terminal kinase (JNK) signaling pathway mediates induction of urokinase-type plasminogen activator (uPA) by the alkylating agent MNNG. *Blood* 96: 1415-1424, 2000.

Parsonnet J, Vandersteen D, Goates J, Sibley RK, Pritikin J and Chang Y. Helicobacter pylori infection in intestinal- and diffuse-type gastric adenocarcinomas. *J Natl Cancer Inst* 83: 640-643, 1991.

**Parsonnet J, Friedman GD, Orentreich N and Vogelman H**. Risk for gastric cancer in people with CagA positive or CagA negative Helicobacter pylori infection. *Gut* 40: 297-301, 1997.

**Pasare C and Medzhitov R**. Toll-like receptors: linking innate and adaptive immunity. *Adv Exp Med Biol* 560: 11-18, 2005.

Pawar S, Kartha S and Toback FG. Differential gene expression in migrating renal epithelial cells after wounding. *J Cell Physiol* 165: 556-565, 1995.

Pedersen H, Brunner N, Francis D, Osterlind K, Ronne E, Hansen HH, Dano K and Grondahl-Hansen J. Prognostic impact of urokinase, urokinase receptor, and type 1 plasminogen activator inhibitor in squamous and large cell lung cancer tissue. *Cancer Res* 54: 4671-4675, 1994a.

Pedersen H, Grondahl-Hansen J, Francis D, Osterlind K, Hansen HH, Dano K and Brunner N. Urokinase and plasminogen activator inhibitor type 1 in pulmonary adenocarcinoma. *Cancer Res* 54: 120-123, 1994b.

Peek RM, Jr., Miller GG, Tham KT, Perez-Perez GI, Zhao X, Atherton JC and Blaser MJ. Heightened inflammatory response and cytokine expression in vivo to cagA+ Helicobacter pylori strains. *Lab Invest* 73: 760-770, 1995.

Peek RM, Jr., Wirth HP, Moss SF, Yang M, Abdalla AM, Tham KT, Zhang T, Tang LH, Modlin IM and Blaser MJ. Helicobacter pylori alters gastric epithelial cell cycle events and gastrin secretion in Mongolian gerbils. *Gastroenterology* 118: 48-59, 2000.

Peek RM, Jr. and Blaser MJ. Helicobacter pylori and gastrointestinal tract adenocarcinomas. Nat Rev Cancer 2: 28-37, 2002.

**Peek RM, Jr.** Helicobacter pylori strain-specific modulation of gastric mucosal cellular turnover: implications for carcinogenesis. *J Gastroenterol* 37 Suppl 13: 10-16, 2002.

**Peek RM, Jr.** Events at the host-microbial interface of the gastrointestinal tract IV. The pathogenesis of Helicobacter pylori persistence. *Am J Physiol Gastrointest Liver Physiol* 289: G8-12, 2005.

Peek RM, Jr. and Crabtree JE. Helicobacter infection and gastric neoplasia. *J Pathol* 208: 233-248, 2006.

**Pepper MS, Matsumoto K, Nakamura T, Orci L and Montesano R**. Hepatocyte growth factor increases urokinase-type plasminogen activator (u-PA) and u-PA receptor expression in Madin-Darby canine kidney epithelial cells. *J Biol Chem* 267: 20493-20496, 1992.

Pepper MS, Sappino AP, Stocklin R, Montesano R, Orci L and Vassalli JD. Upregulation of urokinase receptor expression on migrating endothelial cells. *J Cell Biol* 122: 673-684, 1993.

**Pepper MS.** Role of the matrix metalloproteinase and plasminogen activator-plasmin systems in angiogenesis. *Arterioscler Thromb Vasc Biol* 21: 1104-1117, 2001.

Perez-Perez GI, Garza-Gonzalez E, Portal C and Olivares AZ. Role of cytokine polymorphisms in the risk of distal gastric cancer development. *Cancer Epidemiol Biomarkers Prev* 14: 1869-1873, 2005.

**Petersen LC, Lund LR, Nielsen LS, Dano K and Skriver L**. One-chain urokinase-type plasminogen activator from human sarcoma cells is a proenzyme with little or no intrinsic activity. *J Biol Chem* 263: 11189-11195, 1988.

## Chapter 7 - References

Petersson C, Forsberg M, Aspholm M, Olfat FO, Forslund T, Boren T and Magnusson KE. Helicobacter pylori SabA adhesin evokes a strong inflammatory response in human neutrophils which is down-regulated by the neutrophil-activating protein. *Med Microbiol Immunol* 195: 195-206, 2006.

Phadnis SH, Ilver D, Janzon L, Normark S and Westblom TU. Pathological significance and molecular characterization of the vacuolating toxin gene of Helicobacter pylori. *Infect Immun* 62: 1557-1565, 1994.

Pinsky DJ, Liao H, Lawson CA, Yan SF, Chen J, Carmeliet P, Loskutoff DJ and Stern DM. Coordinated induction of plasminogen activator inhibitor-1 (PAI-1) and inhibition of plasminogen activator gene expression by hypoxia promotes pulmonary vascular fibrin deposition. *J Clin Invest* 102: 919-928, 1998.

Piotrowski J, Skrodzka D, Slomiany A and Slomiany BL. Helicobacter pylori lipopolysaccharide induces gastric epithelial cells apoptosis. *Biochem Mol Biol Int* 40: 597-602, 1996.

Pique JM, Whittle BJ and Esplugues JV. The vasodilator role of endogenous nitric oxide in the rat gastric microcirculation. *Eur J Pharmacol* 174: 293-296, 1989.

Pisani P, Parkin DM, Bray F and Ferlay J. Erratum: Estimates of the worldwide mortality from 25 cancers in 1990. Int. J. Cancer, 83, 18-29 (1999). Int J Cancer 83: 870-873, 1999.

Plebani M, Herszenyi L, Carraro P, De PM, Roveroni G, Cardin R, Tulassay Z, Naccarato R and Farinati F. Urokinase-type plasminogen activator receptor in gastric cancer: tissue expression and prognostic role. *Clin Exp Metastasis* 15: 418-425, 1997.

Plesner T, Ralfkiaer E, Wittrup M, Johnsen H, Pyke C, Pedersen TL, Hansen NE and Dano K. Expression of the receptor for urokinase-type plasminogen activator in normal and neoplastic blood cells and hematopoietic tissue. *Am J Clin Pathol* 102: 835-841, 1994.

Ploug M, Ronne E, Behrendt N, Jensen AL, Blasi F and Dano K. Cellular receptor for urokinase plasminogen activator. Carboxyl-terminal processing and membrane anchoring by glycosyl-phosphatidylinositol. *J Biol Chem* 266: 1926-1933, 1991a.

Ploug M, Behrendt N, Lober D and Dano K. Protein structure and membrane anchorage of the cellular receptor for urokinase-type plasminogen activator. *Semin Thromb Hemost* 17: 183-193, 1991b.

Ploug M, Kjalke M, Ronne E, Weidle U, Hoyer-Hansen G and Dano K. Localization of the disulfide bonds in the NH2-terminal domain of the cellular receptor for human urokinase-type plasminogen activator. A domain structure belonging to a novel superfamily of glycolipid-anchored membrane proteins. *J Biol Chem* 268: 17539-17546, 1993.

**Ploug M and Ellis V**. Structure-function relationships in the receptor for urokinase-type plasminogen activator. Comparison to other members of the Ly-6 family and snake venom alpha-neurotoxins. *FEBS Lett* 349: 163-168, 1994.

Powell DW, Mifflin RC, Valentich JD, Crowe SE, Saada JI and West AB. Myofibroblasts. II. Intestinal subepithelial myofibroblasts. Am J Physiol 277: C183-C201, 1999a.

**Powell DW, Mifflin RC, Valentich JD, Crowe SE, Saada JI and West AB**. Myofibroblasts. I. Paracrine cells important in health and disease. *Am J Physiol* 277: C1-C9, 1999b.

Prager GW, Breuss JM, Steurer S, Olcaydu D, Mihaly J, Brunner PM, Stockinger H and Binder BR. Vascular endothelial growth factor receptor-2-induced initial endothelial cell migration depends on the presence of the urokinase receptor. *Circ Res* 94: 1562-1570, 2004.

Prat M, Narsimhan RP, Crepaldi T, Nicotra MR, Natali PG and Comoglio PM. The receptor encoded by the human c-MET oncogene is expressed in hepatocytes, epithelial cells and solid tumors. *Int J Cancer* 49: 323-328, 1991.

Preissner KT. Structure and biological role of vitronectin. Annu Rev Cell Biol 7: 275-310, 1991.

Preissner KT, Kanse SM and May AE. Urokinase receptor: a molecular organizer in cellular communication. *Curr Opin Cell Biol* 12: 621-628, 2000.

Prinz C, Schoniger M, Rad R, Becker I, Keiditsch E, Wagenpfeil S, Classen M, Rosch T, Schepp W and Gerhard M. Key importance of the Helicobacter pylori adherence factor blood group antigen binding adhesin during chronic gastric inflammation. *Cancer Res* 61: 1903-1909, 2001.

Pritchard DM and Przemeck SM. Review article: How useful are the rodent animal models of gastric adenocarcinoma? *Aliment Pharmacol Ther* 19: 841-859, 2004.

**Providence KM, Kutz SM, Staiano-Coico L and Higgins PJ**. PAI-1 gene expression is regionally induced in wounded epithelial cell monolayers and required for injury repair. *J Cell Physiol* 182: 269-280, 2000.

**Providence KM, White LA, Tang J, Gonclaves J, Staiano-Coico L and Higgins PJ**. Epithelial monolayer wounding stimulates binding of USF-1 to an E-box motif in the plasminogen activator inhibitor type 1 gene. *J Cell Sci* 115: 3767-3777, 2002.

**Providence KM and Higgins PJ**. PAI-1 expression is required for epithelial cell migration in two distinct phases of in vitro wound repair. *J Cell Physiol* 200: 297-308, 2004.

Pyke C, Kristensen P, Ralfkiaer E, Eriksen J and Dano K. The plasminogen activation system in human colon cancer: messenger RNA for the inhibitor PAI-1 is located in endothelial cells in the tumor stroma. *Cancer Res* 51: 4067-4071, 1991.

Quax PH, van den Hoogen CM, Verheijen JH, Padro T, Zeheb R, Gelehrter TD, van Berkel TJ, Kuiper J and Emeis JJ. Endotoxin induction of plasminogen activator and plasminogen activator inhibitor type 1 mRNA in rat tissues in vivo. *J Biol Chem* 265: 15560-15563, 1990.

Quax PH, Pedersen N, Masucci MT, Weening-Verhoeff EJ, Dano K, Verheijen JH and Blasi F. Complementation between urokinase-producing and receptor-producing cells in extracellular matrix degradation. *Cell Regul* 2: 793-803, 1991.

Rabbani SA, Desjardins J, Bell AW, Banville D, Mazar A, Henkin J and Goltzman D. An amino-terminal fragment of urokinase isolated from a prostate cancer cell line (PC-3) is mitogenic for osteoblast-like cells. *Biochem Biophys Res Commun* 173: 1058-1064, 1990.

Rabbani SA, Mazar AP, Bernier SM, Haq M, Bolivar I, Henkin J and Goltzman D. Structural requirements for the growth factor activity of the amino-terminal domain of urokinase. *J Biol Chem* 267: 14151-14156, 1992.

**Rabbani SA, Gladu J, Mazar AP, Henkin J and Goltzman D**. Induction in human osteoblastic cells (SaOS2) of the early response genes fos, jun, and myc by the amino terminal fragment (ATF) of urokinase. *J Cell Physiol* 172: 137-145, 1997.

Radtke KP, Fernandez JA, Greengard JS, Tang WW, Wilson CB, Loskutoff DJ, Scharrer I and Griffin JH. Protein C inhibitor is expressed in tubular cells of human kidney. *J Clin Invest* 94: 2117-2124, 1994.

Rakic JM, Maillard C, Jost M, Bajou K, Masson V, Devy L, Lambert V, Foidart JM and Noel A. Role of plasminogen activator-plasmin system in tumor angiogenesis. *Cell Mol Life Sci* 60: 463-473, 2003.

**Remacle-Bonnet MM, Garrouste FL and Pommier GJ.** Surface-bound plasmin induces selective proteolysis of insulin-like-growth-factor (IGF)-binding protein-4 (IGFBP-4) and promotes autocrine IGF-II bio-availability in human colon-carcinoma cells. *Int J Cancer* 72: 835-843, 1997.

**Resnati M, Guttinger M, Valcamonica S, Sidenius N, Blasi F and Fazioli F**. Proteolytic cleavage of the urokinase receptor substitutes for the agonist-induced chemotactic effect. *EMBO J* 15: 1572-1582, 1996.

**Resnati M, Pallavicini I, Wang JM, Oppenheim J, Serhan CN, Romano M and Blasi F**. The fibrinolytic receptor for urokinase activates the G protein-coupled chemotactic receptor FPRL1/LXA4R. *Proc Natl Acad Sci US A* 99: 1359-1364, 2002.

Reubi JC, Waser B, Laderach U, Stettler C, Friess H, Halter F and Schmassmann A. Localization of cholecystokinin A and cholecystokinin B-gastrin receptors in the human stomach. *Gastroenterology* 112: 1197-1205, 1997.

Rindi G, Necchi V, Savio A, Torsello A, Zoli M, Locatelli V, Raimondo F, Cocchi D and Solcia E. Characterisation of gastric ghrelin cells in man and other mammals: studies in adult and fetal tissues. *Histochem Cell Biol* 117: 511-519, 2002.

Robert A. Cytoprotection by prostaglandins. Gastroenterology 77: 761-767, 1979.

**Rodenburg KW, Kjoller L, Petersen HH and Andreasen PA**. Binding of urokinase-type plasminogen activator-plasminogen activator inhibitor-1 complex to the endocytosis receptors alpha2-macroglobulin receptor/low-density lipoprotein receptor-related protein and very-low-density lipoprotein receptor involves basic residues in the inhibitor. *Biochem J* 329 (Pt 1): 55-63, 1998.

Rogers AB and Fox JG. Inflammation and Cancer. I. Rodent models of infectious gastrointestinal and liver cancer. Am J Physiol Gastrointest Liver Physiol 286: G361-G366, 2004.

**Roldan AL, Cubellis MV, Masucci MT, Behrendt N, Lund LR, Dano K, Appella E and Blasi F**. Cloning and expression of the receptor for human urokinase plasminogen activator, a central molecule in cell surface, plasmin dependent proteolysis. *EMBO J* 9: 467-474, 1990.

Rossi G, Ruggiero P, Peppoloni S, Pancotto L, Fortuna D, Lauretti L, Volpini G, Mancianti S, Corazza M, Taccini E, Di PF, Rappuoli R and Del GG. Therapeutic vaccination against Helicobacter pylori in the beagle dog experimental model: safety, immunogenicity, and efficacy. *Infect Immun* 72: 3252-3259, 2004.

Rozengurt E and Walsh JH. Gastrin, CCK, signaling, and cancer. Annu Rev Physiol 63: 49-76, 2001.

Rubin JS, Bottaro DP, Chedid M, Miki T, Ron D, Cunha GR and Finch PW. Keratinocyte growth factor as a cytokine that mediates mesenchymal-epithelial interaction. *EXS* 74: 191-214, 1995.

Rugge M, Correa P, Dixon MF, Hattori T, Leandro G, Lewin K, Riddell RH, Sipponen P and Watanabe H. Gastric dysplasia: the Padova international classification. *Am J Surg Pathol* 24: 167-176, 2000.

Sachs G, Zeng N and Prinz C. Physiology of isolated gastric endocrine cells. *Annu Rev Physiol* 59: 243-256, 1997.

Saeki T, Ohno T, Kamata K, Arai K, Mizuguchi S, Katori M, Saigenji K and Majima M. Mild irritant prevents ethanol-induced gastric mucosal microcirculatory disturbances through actions of calcitonin gene-related peptide and PGI2 in rats. *Am J Physiol Gastrointest Liver Physiol* 286: G68-G75, 2004.

Sakagami T, Dixon M, O'Rourke J, Howlett R, Alderuccio F, Vella J, Shimoyama T and Lee A. Atrophic gastric changes in both Helicobacter felis and Helicobacter pylori infected mice are host dependent and separate from antral gastritis. *Gut* 39: 639-648, 1996.

Sakakibara T, Hibi K, Koike M, Fujiwara M, Kodera Y, Ito K and Nakao A. Plasminogen activator inhibitor-1 as a potential marker for the malignancy of colorectal cancer. *Br J Cancer* 93: 799-803, 2005.

Sakakibara T, Hibi K, Koike M, Fujiwara M, Kodera Y, Ito K and Nakao A. Plasminogen activator inhibitor-1 as a potential marker for the malignancy of gastric cancer. *Cancer Sci* 97: 395-399, 2006.

Saksela O and Rifkin DB. Cell-associated plasminogen activation: regulation and physiological functions. *Annu Rev Cell Biol* 4: 93-126, 1988.

Samad F and Loskutoff DJ. Tissue distribution and regulation of plasminogen activator inhibitor-1 in obese mice. *Mol Med* 2: 568-582, 1996.

Samad F, Yamamoto K and Loskutoff DJ. Distribution and regulation of plasminogen activator inhibitor-1 in murine adipose tissue in vivo. Induction by tumor necrosis factor-alpha and lipopolysaccharide. *J Clin Invest* 97: 37-46, 1996.

Samad F and Loskutoff DJ. The fat mouse: a powerful genetic model to study elevated plasminogen activator inhibitor 1 in obesity/NIDDM. *Thromb Haemost* 78: 652-655, 1997.

Samad F, Pandey M and Loskutoff DJ. Tissue factor gene expression in the adipose tissues of obese mice. *Proc Natl Acad Sci U S A* 95: 7591-7596, 1998.

Samad F, Pandey M, Bell PA and Loskutoff DJ. Insulin continues to induce plasminogen activator inhibitor 1 gene expression in insulin-resistant mice and adipocytes. *Mol Med* 6: 680-692, 2000.

Sands BE and Podolsky DK. The trefoil peptide family. Annu Rev Physiol 58: 253-273, 1996.

Sandvik AK, Waldum HL, Kleveland PM and Schulze SB. Gastrin produces an immediate and dose-dependent histamine release preceding acid secretion in the totally isolated, vascularly perfused rat stomach. *Scand J Gastroenterol* 22: 803-808, 1987.

Sawdey MS and Loskutoff DJ. Regulation of murine type 1 plasminogen activator inhibitor gene expression in vivo. Tissue specificity and induction by lipopolysaccharide, tumor necrosis factoralpha, and transforming growth factor-beta. *J Clin Invest* 88: 1346-1353, 1991.

Schicho R, Florian W, Liebmann I, Holzer P and Lippe IT. Increased expression of TRPV1 receptor in dorsal root ganglia by acid insult of the rat gastric mucosa. *Eur J Neurosci* 19: 1811-1818, 2004.

Schleef RR, Bevilacqua MP, Sawdey M, Gimbrone MA, Jr. and Loskutoff DJ. Cytokine activation of vascular endothelium. Effects on tissue-type plasminogen activator and type 1 plasminogen activator inhibitor. *J Biol Chem* 263: 5797-5803, 1988.

Schleuning WD, Medcalf RL, Hession C, Rothenbuhler R, Shaw A and Kruithof EK. Plasminogen activator inhibitor 2: regulation of gene transcription during phorbol ester-mediated differentiation of U-937 human histiocytic lymphoma cells. *Mol Cell Biol* 7: 4564-4567, 1987.

Schmitz F, Goke MN, Otte JM, Schrader H, Reimann B, Kruse ML, Siegel EG, Peters J, Herzig KH, Folsch UR and Schmidt WE. Cellular expression of CCK-A and CCK-B/gastrin receptors in human gastric mucosa. *Regul Pept* 102: 101-110, 2001.

Schneiderman J, Sawdey MS, Keeton MR, Bordin GM, Bernstein EF, Dilley RB and Loskutoff DJ. Increased type 1 plasminogen activator inhibitor gene expression in atherosclerotic human arteries. *Proc Natl Acad Sci U S A* 89: 6998-7002, 1992.

Scicolone G, Sanchez V, Vauthay L, Fuentes F, Scicolone A, Scicolone L, Rapacioli M and Flores V. Tissue-type plasminogen activator activity in morphologically normal tissues adjacent to gastrointestinal carcinomas is associated with the degree of tumor progression. *J Cancer Res Clin Oncol* 132: 309-319, 2006.

Segal ED, Cha J, Lo J, Falkow S and Tompkins LS. Altered states: involvement of phosphorylated CagA in the induction of host cellular growth changes by Helicobacter pylori. *Proc Natl Acad Sci U S A* 96: 14559-14564, 1999.

Seiffert D and Loskutoff DJ. Evidence that type 1 plasminogen activator inhibitor binds to the somatomedin B domain of vitronectin. *J Biol Chem* 266: 2824-2830, 1991.

Selbach M, Moese S, Meyer TF and Backert S. Functional analysis of the Helicobacter pylori cag pathogenicity island reveals both VirD4-CagA-dependent and VirD4-CagA-independent mechanisms. *Infect Immun* 70: 665-671, 2002.

Selbach M, Moese S, Hurwitz R, Hauck CR, Meyer TF and Backert S. The Helicobacter pylori CagA protein induces cortactin dephosphorylation and actin rearrangement by c-Src inactivation. *EMBO J* 22: 515-528, 2003.

Seril DN, Liao J, Yang GY and Yang CS. Oxidative stress and ulcerative colitis-associated carcinogenesis: studies in humans and animal models. *Carcinogenesis* 24: 353-362, 2003.

Sharma SA, Tummuru MK, Miller GG and Blaser MJ. Interleukin-8 response of gastric epithelial cell lines to Helicobacter pylori stimulation in vitro. *Infect Immun* 63: 1681-1687, 1995.

Sharma SA, Tummuru MK, Blaser MJ and Kerr LD. Activation of IL-8 gene expression by Helicobacter pylori is regulated by transcription factor nuclear factor-kappa B in gastric epithelial cells. *J Immunol* 160: 2401-2407, 1998.

Shetty S, Bdeir K, Cines DB and Idell S. Induction of plasminogen activator inhibitor-1 by urokinase in lung epithelial cells. *J Biol Chem* 278: 18124-18131, 2003.

Shikata K, Kiyohara Y and Iida M. [Effect of dietary factors on gastric cancer]. Nippon Shokakibyo Gakkai Zasshi 103: 501-507, 2006.

Shimoyama T, Fukuda S, Tanaka M, Mikami T, Munakata A and Crabtree JE. CagA seropositivity associated with development of gastric cancer in a Japanese population. *J Clin Pathol* 51: 225-228, 1998.

Shin BA, Yoo HG, Kim HS, Kim MH, Hwang YS, Chay KO, Lee KY, Ahn BW and Jung YD. P38 MAPK pathway is involved in the urokinase plasminogen activator expression in human gastric SNU-638 cells. *Oncol Rep* 10: 1467-1471, 2003.

Shin EY, Ma EK, Kim CK, Kwak SJ and Kim EG. Src/ERK but not phospholipase D is involved in keratinocyte growth factor-stimulated secretion of matrix metalloprotease-9 and urokinase-type plasminogen activator in SNU-16 human stomach cancer cell. J Cancer Res Clin Oncol 128: 596-602, 2002.

Sidenius N, Andolfo A, Fesce R and Blasi F. Urokinase regulates vitronectin binding by controlling urokinase receptor oligomerization. *J Biol Chem* 277: 27982-27990, 2002.

Sidenius N and Blasi F. The urokinase plasminogen activator system in cancer: recent advances and implication for prognosis and therapy. *Cancer Metastasis Rev* 22: 205-222, 2003.

Silen W and Ito S. Mechanisms for rapid re-epithelialization of the gastric mucosal surface. Annu Rev Physiol 47: 217-229, 1985.

Silen W. What is cytoprotection of the gastric mucosa? Gastroenterology 94: 232-235, 1988.

Silverman GA, Bird PI, Carrell RW, Church FC, Coughlin PB, Gettins PG, Irving JA, Lomas DA, Luke CJ, Moyer RW, Pemberton PA, Remold-O'Donnell E, Salvesen GS, Travis J and Whisstock JC. The serpins are an expanding superfamily of structurally similar but functionally diverse proteins. Evolution, mechanism of inhibition, novel functions, and a revised nomenclature. J Biol Chem 276: 33293-33296, 2001.

Sipponen P, Kosunen TU, Valle J, Riihela M and Seppala K. Helicobacter pylori infection and chronic gastritis in gastric cancer. *J Clin Pathol* 45: 319-323, 1992.

Sipponen P and Marshall BJ. Gastritis and gastric cancer. Western countries. *Gastroenterol Clin* North Am 29: 579-5vi, 2000.

Sitrin RG, Todd RF, III, Mizukami IF, Gross TJ, Shollenberger SB and Gyetko MR. Cytokinespecific regulation of urokinase receptor (CD87) expression by U937 mononuclear phagocytes. *Blood* 84: 1268-1275, 1994.

Skriver L, Nielsen LS, Stephens R and Dano K. Plasminogen activator released as inactive proenzyme from murine cells transformed by sarcoma virus. *Eur J Biochem* 124: 409-414, 1982.

Sliva D, English D, Lyons D and Lloyd FP, Jr. Protein kinase C induces motility of breast cancers by upregulating secretion of urokinase-type plasminogen activator through activation of AP-1 and NF-kappaB. *Biochem Biophys Res Commun* 290: 552-557, 2002a.

Sliva D, Rizzo MT and English D. Phosphatidylinositol 3-kinase and NF-kappaB regulate motility of invasive MDA-MB-231 human breast cancer cells by the secretion of urokinase-type plasminogen activator. *J Biol Chem* 277: 3150-3157, 2002b.

Sommer M, Eismann U, Gerth J and Stein G. Interleukin 4 co-stimulates the PDGF-BB- and bFGF-mediated proliferation of mesangial cells and myofibroblasts. *Nephron* 92: 868-880, 2002.

Sonnenberg E, Meyer D, Weidner KM and Birchmeier C. Scatter factor/hepatocyte growth factor and its receptor, the c-met tyrosine kinase, can mediate a signal exchange between mesenchyme and epithelia during mouse development. *J Cell Biol* 123: 223-235, 1993.

Soto G, Bautista CT, Roth DE, Gilman RH, Velapatino B, Ogura M, Dailide G, Razuri M, Meza R, Katz U, Monath TP, Berg DE and Taylor DN. Helicobacter pylori reinfection is common in Peruvian adults after antibiotic eradication therapy. *J Infect Dis* 188: 1263-1275, 2003.

Sperti G, van LR, Maseri A and Kluft C. Platelet-derived growth factor increases plasminogen activator inhibitor-1 activity and mRNA in rat cultured vascular smooth muscle. *Ann N Y Acad Sci* 667: 178-180, 1992.

Stack MS and Johnson DA. Human mast cell tryptase activates single-chain urinary-type plasminogen activator (pro-urokinase). *J Biol Chem* 269: 9416-9419, 1994.

Stahl A and Mueller BM. The urokinase-type plasminogen activator receptor, a GPI-linked protein, is localized in caveolae. *J Cell Biol* 129: 335-344, 1995.

Stahl A and Mueller BM. Melanoma cell migration on vitronectin: regulation by components of the plasminogen activation system. *Int J Cancer* 71: 116-122, 1997.

Staiano-Coico L, Krueger JG, Rubin JS, D'limi S, Vallat VP, Valentino L, Fahey T, III, Hawes A, Kingston G, Madden MR and . Human keratinocyte growth factor effects in a porcine model of epidermal wound healing. *J Exp Med* 178: 865-878, 1993.

Steele IA, Dimaline R, Pritchard DM, Peek RM, Jr., Wang TC, Dockray GJ and Varro A. Helicobacter and gastrin stimulate Reg1 expression in gastric epithelial cells through distinct promoter elements. *Am J Physiol Gastrointest Liver Physiol* 293: G347-G354, 2007.

**Stefansson S and Lawrence DA**. The serpin PAI-1 inhibits cell migration by blocking integrin alpha V beta 3 binding to vitronectin. *Nature* 383: 441-443, 1996.

Stefansson S, McMahon GA, Petitclerc E and Lawrence DA. Plasminogen activator inhibitor-1 in tumor growth, angiogenesis and vascular remodeling. *Curr Pharm Des* 9: 1545-1564, 2003.

Stein M, Rappuoli R and Covacci A. Tyrosine phosphorylation of the Helicobacter pylori CagA antigen after cag-driven host cell translocation. *Proc Natl Acad Sci US A* 97: 1263-1268, 2000.

Stepanova V, Mukhina S, Kohler E, Resink TJ, Erne P and Tkachuk VA. Urokinase plasminogen activator induces human smooth muscle cell migration and proliferation via distinct receptor-dependent and proteolysis-dependent mechanisms. *Mol Cell Biochem* 195: 199-206, 1999.

Stone MA. Non-invasive testing for Helicobacter pylori. Postgrad Med J 75: 74-77, 1999.

Stoppelli MP, Corti A, Soffientini A, Cassani G, Blasi F and Assoian RK. Differentiationenhanced binding of the amino-terminal fragment of human urokinase plasminogen activator to a specific receptor on U937 monocytes. *Proc Natl Acad Sci US A* 82: 4939-4943, 1985.

Strickland DK, Gonias SL and Argraves WS. Diverse roles for the LDL receptor family. *Trends Endocrinol Metab* 13: 66-74, 2002.

Stroff T, Plate S, Ebrahim JS, Ehrlich KH, Respondek M and Peskar BM. Tachykinin-induced increase in gastric mucosal resistance: role of primary afferent neurons, CGRP, and NO. Am J Physiol 271: G1017-G1027, 1996.

Su B, Ceponis PJ, Lebel S, Huynh H and Sherman PM. Helicobacter pylori activates Toll-like receptor 4 expression in gastrointestinal epithelial cells. *Infect Immun* 71: 3496-3502, 2003.

Sugiyama T and Asaka M. [Pathophysiological mechanism of Helicobacter infections--host factors]. Nippon Naika Gakkai Zasshi 87: 813-818, 1998.

Sundrud MS, Torres VJ, Unutmaz D and Cover TL. Inhibition of primary human T cell proliferation by Helicobacter pylori vacuolating toxin (VacA) is independent of VacA effects on IL-2 secretion. *Proc Natl Acad Sci USA* 101: 7727-7732, 2004.

Sunnarborg SW, Hinkle CL, Stevenson M, Russell WE, Raska CS, Peschon JJ, Castner BJ, Gerhart MJ, Paxton RJ, Black RA and Lee DC. Tumor necrosis factor-alpha converting enzyme (TACE) regulates epidermal growth factor receptor ligand availability. *J Biol Chem* 277: 12838-12845, 2002.

Sutton P, Wilson J and Lee A. Further development of the Helicobacter pylori mouse vaccination model. *Vaccine* 18: 2677-2685, 2000.

Suzuki K, Deyashiki Y, Nishioka J and Toma K. Protein C inhibitor: structure and function. *Thromb Haemost* 61: 337-342, 1989.

Swaisgood CM, French EL, Noga C, Simon RH and Ploplis VA. The development of bleomycininduced pulmonary fibrosis in mice deficient for components of the fibrinolytic system. *Am J Pathol* 157: 177-187, 2000. Swiatkowska M, Cierniewska-Cieslak A, Pawlowska Z and Cierniewski CS. Dual regulatory effects of nitric oxide on plasminogen activator inhibitor type 1 expression in endothelial cells. *Eur J Biochem* 267: 1001-1007, 2000.

Szabo I, Brutsche S, Tombola F, Moschioni M, Satin B, Telford JL, Rappuoli R, Montecucco C, Papini E and Zoratti M. Formation of anion-selective channels in the cell plasma membrane by the toxin VacA of Helicobacter pylori is required for its biological activity. *EMBO J* 18: 5517-5527, 1999.

Szabo S. Mechanisms of mucosal injury in the stomach and duodenum: time-sequence analysis of morphologic, functional, biochemical and histochemical studies. *Scand J Gastroenterol Suppl* 127: 21-28, 1987.

Taha AS, Curry GW, Morton R, Park RH and Beattie AD. Gastric mucosal hepatocyte growth factor in Helicobacter pylori gastritis and peptic ulcer disease. *Am J Gastroenterol* 91: 1407-1409, 1996.

Takahashi S, Shigeta J, Inoue H, Tanabe T and Okabe S. Localization of cyclooxygenase-2 and regulation of its mRNA expression in gastric ulcers in rats. *Am J Physiol* 275: G1137-G1145, 1998.

Takeshita K, Hayashi M, Iino S, Kondo T, Inden Y, Iwase M, Kojima T, Hirai M, Ito M, Loskutoff DJ, Saito H, Murohara T and Yamamoto K. Increased expression of plasminogen activator inhibitor-1 in cardiomyocytes contributes to cardiac fibrosis after myocardial infarction. *Am J Pathol* 164: 449-456, 2004.

Tangkijvanich P, Melton AC, Chitapanarux T, Han J and Yee HF. Platelet-derived growth factor-BB and lysophosphatidic acid distinctly regulate hepatic myofibroblast migration through focal adhesion kinase. *Exp Cell Res* 281: 140-147, 2002.

Tecimer C, Doering DL, Goldsmith LJ, Meyer JS, Abdulhay G and Wittliff JL. Clinical relevance of urokinase-type plasminogen activator, its receptor, and its inhibitor type 1 in endometrial cancer. *Gynecol Oncol* 80: 48-55, 2001.

**Tepperman BL and Whittle BJ.** Endogenous nitric oxide and sensory neuropeptides interact in the modulation of the rat gastric microcirculation. *Br J Pharmacol* 105: 171-175, 1992.

Terano A, Ota S, Mach T, Hiraishi H, Stachura J, Tarnawski A and Ivey KJ. Prostaglandin protects against taurocholate-induced damage to rat gastric mucosal cell culture. *Gastroenterology* 92: 669-677, 1987.

Thogersen AM, Jansson JH, Boman K, Nilsson TK, Weinehall L, Huhtasaari F and Hallmans G. High plasminogen activator inhibitor and tissue plasminogen activator levels in plasma precede a first acute myocardial infarction in both men and women: evidence for the fibrinolytic system as an independent primary risk factor. *Circulation* 98: 2241-2247, 1998.

Tindberg Y, Bengtsson C, Granath F, Blennow M, Nyren O and Granstrom M. Helicobacter pylori infection in Swedish school children: lack of evidence of child-to-child transmission outside the family. *Gastroenterology* 121: 310-316, 2001.

**Togawa S, Joh T, Itoh M, Katsuda N, Ito H, Matsuo K, Tajima K and Hamajima N**. Interleukin-2 gene polymorphisms associated with increased risk of gastric atrophy from Helicobacter pylori infection. *Helicobacter* 10: 172-178, 2005.

Tomb JF, White O, Kerlavage AR, Clayton RA, Sutton GG, Fleischmann RD, Ketchum KA, Klenk HP, Gill S, Dougherty BA, Nelson K, Quackenbush J, Zhou L, Kirkness EF, Peterson S, Loftus B, Richardson D, Dodson R, Khalak HG, Glodek A, McKenney K, Fitzegerald LM, Lee N, Adams MD, Hickey EK, Berg DE, Gocayne JD, Utterback TR, Peterson JD, Kelley JM, Cotton MD, Weidman JM, Fujii C, Bowman C, Watthey L, Wallin E, Hayes WS, Borodovsky M, Karp PD, Smith HO, Fraser CM and Venter JC. The complete genome sequence of the gastric pathogen Helicobacter pylori. *Nature* 388: 539-547, 1997.

Tombola F, Carlesso C, Szabo I, de BM, Reyrat JM, Telford JL, Rappuoli R, Montecucco C, Papini E and Zoratti M. Helicobacter pylori vacuolating toxin forms anion-selective channels in planar lipid bilayers: possible implications for the mechanism of cellular vacuolation. *Biophys J* 76: 1401-1409, 1999.

**Tsutsumi R, Higashi H, Higuchi M, Okada M and Hatakeyama M**. Attenuation of Helicobacter pylori CagA x SHP-2 signaling by interaction between CagA and C-terminal Src kinase. *J Biol Chem* 278: 3664-3670, 2003.

Tu S, Chi AL, Lim S, Cui G, Dubeykovskaya Z, Ai W, Fleming JV, Takaishi S and Wang TC. Gastrin regulates the TFF2 promoter through gastrin-responsive cis-acting elements and multiple signaling pathways. *Am J Physiol Gastrointest Liver Physiol* 292: G1726-G1737, 2007.

Tummuru MK, Sharma SA and Blaser MJ. Helicobacter pylori picB, a homologue of the Bordetella pertussis toxin secretion protein, is required for induction of IL-8 in gastric epithelial cells. *Mol Microbiol* 18: 867-876, 1995.

Tytgat GN and Rauws EA. Campylobacter pylori and its role in peptic ulcer disease. *Gastroenterol Clin North Am* 19: 183-196, 1990.

Uemura N, Okamoto S, Yamamoto S, Matsumura N, Yamaguchi S, Yamakido M, Taniyama K, Sasaki N and Schlemper RJ. Helicobacter pylori infection and the development of gastric cancer. *N* Engl J Med 345: 784-789, 2001.

Uhrin P, Dewerchin M, Hilpert M, Chrenek P, Schofer C, Zechmeister-Machhart M, Kronke G, Vales A, Carmeliet P, Binder BR and Geiger M. Disruption of the protein C inhibitor gene results in impaired spermatogenesis and male infertility. *J Clin Invest* 106: 1531-1539, 2000.

Unemo M, spholm-Hurtig M, Ilver D, Bergstrom J, Boren T, Danielsson D and Teneberg S. The sialic acid binding SabA adhesin of Helicobacter pylori is essential for nonopsonic activation of human neutrophils. *J Biol Chem* 280: 15390-15397, 2005.
van den Berg EA, Sprengers ED, Jaye M, Burgess W, Maciag T and van H, V. Regulation of plasminogen activator inhibitor-1 mRNA in human endothelial cells. *Thromb Haemost* 60: 63-67, 1988.

van Doorn LJ, Figueiredo C, Sanna R, Pena S, Midolo P, Ng EK, Atherton JC, Blaser MJ and Quint WG. Expanding allelic diversity of Helicobacter pylori vacA. *J Clin Microbiol* 36: 2597-2603, 1998.

van Doorn NE, Namavar F, van Doorn LJ, Durrani Z, Kuipers EJ and Vandenbroucke-Grauls CM. Analysis of vacA, cagA, and IS605 genotypes and those determined by PCR amplification of DNA between repetitive sequences of Helicobacter pylori strains isolated from patients with nonulcer dyspepsia or mucosa-associated lymphoid tissue lymphoma. *J Clin Microbiol* 37: 2348-2349, 1999a.

van Doorn NE, Namavar F, Sparrius M, Stoof J, van Rees EP, van Doorn LJ and Vandenbroucke-Grauls CM. Helicobacter pylori-associated gastritis in mice is host and strain specific. *Infect Immun* 67: 3040-3046, 1999b.

Vane JR, Mitchell JA, Appleton I, Tomlinson A, Bishop-Bailey D, Croxtall J and Willoughby DA. Inducible isoforms of cyclooxygenase and nitric-oxide synthase in inflammation. *Proc Natl Acad Sci US A* 91: 2046-2050, 1994.

Vane JR and Botting RM. Mechanism of action of nonsteroidal anti-inflammatory drugs. *Am J Med* 104: 2S-8S, 1998.

Varela JC, Goldstein MH, Baker HV and Schultz GS. Microarray analysis of gene expression patterns during healing of rat corneas after excimer laser photorefractive keratectomy. *Invest Ophthalmol Vis Sci* 43: 1772-1782, 2002.

Varro A, Noble PJ, Wroblewski LE, Bishop L and Dockray GJ. Gastrin-cholecystokinin(B) receptor expression in AGS cells is associated with direct inhibition and indirect stimulation of cell proliferation via paracrine activation of the epidermal growth factor receptor. *Gut* 5 0: 827-833, 2002a.

Varro A, Hemers E, Archer D, Pagliocca A, Haigh C, Ahmed S, Dimaline R and Dockray GJ. Identification of plasminogen activator inhibitor-2 as a gastrin-regulated gene: Role of Rho GTPase and menin. *Gastroenterology* 123: 271-280, 2002b.

Varro A, Noble PJ, Pritchard DM, Kennedy S, Hart CA, Dimaline R and Dockray GJ. Helicobacter pylori induces plasminogen activator inhibitor 2 in gastric epithelial cells through nuclear factor-kappaB and RhoA: implications for invasion and apoptosis. *Cancer Res* 64: 1695-1702, 2004.

Varro A, Kenny S, Hemers E, McCaig C, Przemeck S, Wang TC, Bodger K and Pritchard DM. Increased gastric expression of MMP-7 in hypergastrinemia and significance for epithelialmesenchymal signaling. *Am J Physiol Gastrointest Liver Physiol* 292: G1133-G1140, 2007. Verde P, Stoppelli MP, Galeffi P, Di NP and Blasi F. Identification and primary sequence of an unspliced human urokinase poly(A)+ RNA. *Proc Natl Acad Sci U S A* 81: 4727-4731, 1984.

Viala J, Chaput C, Boneca IG, Cardona A, Girardin SE, Moran AP, Athman R, Memet S, Huerre MR, Coyle AJ, DiStefano PS, Sansonetti PJ, Labigne A, Bertin J, Philpott DJ and Ferrero RL. Nod1 responds to peptidoglycan delivered by the Helicobacter pylori cag pathogenicity island. *Nat Immunol* 5: 1166-1174, 2004.

**Virchow R.** Cellular pathology as based upon physiological and pathological histology. (English translation of  $2^{nd}$  German edition (1859). New York: Dover, 1971) 1863.

Wagner OF, de VC, Hohmann C, Veerman H and Pannekoek H. Interaction between plasminogen activator inhibitor type 1 (PAI-1) bound to fibrin and either tissue-type plasminogen activator (t-PA) or urokinase-type plasminogen activator (u-PA). Binding of t-PA/PAI-1 complexes to fibrin mediated by both the finger and the kringle-2 domain of t-PA. J Clin Invest 84: 647-655, 1989.

**Wallace JL and Whittle BJ.** Role of mucus in the repair of gastric epithelial damage in the rat. Inhibition of epithelial recovery by mucolytic agents. *Gastroenterology* 91: 603-611, 1986.

Wallace JL. COX-2: a pivotal enzyme in mucosal protection and resolution of inflammation. *ScientificWorldJournal* 6: 577-588, 2006.

Wallasch C, Crabtree JE, Bevec D, Robinson PA, Wagner H and Ullrich A. Helicobacter pyloristimulated EGF receptor transactivation requires metalloprotease cleavage of HB-EGF. *Biochem Biophys Res Commun* 295: 695-701, 2002.

Walsh JH. Gastrin. In gut peptides: Biochemistry and Physiology. Ed. JH Walsh, GF Dockray pp. 75-121. New York: Raven, 1994.

Wambura C, Aoyama N, Shirasaka D, Sakai T, Ikemura T, Sakashita M, Maekawa S, Kuroda K, Inoue T, Ebara S, Miyamoto M and Kasuga M. Effect of Helicobacter pylori-induced cyclooxygenase-2 on gastric epithelial cell kinetics: implication for gastric carcinogenesis. *Helicobacter* 7: 129-138, 2002.

Wang TC, Koh TJ, Varro A, Cahill RJ, Dangler CA, Fox JG and Dockray GJ. Processing and proliferative effects of human progastrin in transgenic mice. *J Clin Invest* 98: 1918-1929, 1996.

Wang TC, Goldenring JR, Dangler C, Ito S, Mueller A, Jeon WK, Koh TJ and Fox JG. Mice lacking secretory phospholipase A2 show altered apoptosis and differentiation with Helicobacter felis infection. *Gastroenterology* 114: 675-689, 1998.

Wang TC, Dangler CA, Chen D, Goldenring JR, Koh T, Raychowdhury R, Coffey RJ, Ito S, Varro A, Dockray GJ and Fox JG. Synergistic interaction between hypergastrinemia and Helicobacter infection in a mouse model of gastric cancer. *Gastroenterology* 118: 36-47, 2000.

Wang Y, Jones CJ, Dang J, Liang X, Olsen JE and Doe WF. Human urokinase receptor expression is inhibited by amiloride and induced by tumor necrosis factor and phorbol ester in colon cancer cells. *FEBS Lett* 353: 138-142, 1994.

Wang Y, Dang J, Johnson LK, Selhamer JJ and Doe WF. Structure of the human urokinase receptor gene and its similarity to CD59 and the Ly-6 family. *Eur J Biochem* 227: 116-122, 1995.

Watanabe S, Hirose M, Wang XE, Kobayashi O, Nagahara A, Murai T, Iwazaki R, Miwa H, Miyazaki A and Sato N. Epithelial-mesenchymal interaction in gastric mucosal restoration. J Gastroenterol 35 Suppl 12: 65-68, 2000.

Watson F, Kiernan RS, Deavall DG, Varro A and Dimaline R. Transcriptional activation of the rat vesicular monoamine transporter 2 promoter in gastric epithelial cells: regulation by gastrin. *J Biol Chem* 276: 7661-7671, 2001.

Webb G, Baker MS, Nicholl J, Wang Y, Woodrow G, Kruithof E and Doe WF. Chromosomal localization of the human urokinase plasminogen activator receptor and plasminogen activator inhibitor type-2 genes: implications in colorectal cancer. *J Gastroenterol Hepatol* 9: 340-343, 1994.

Webb PM, Crabtree JE and Forman D. Gastric cancer, cytotoxin-associated gene A-positive Helicobacter pylori, and serum pepsinogens: an international study. The Eurogst Study Group. *Gastroenterology* 116: 269-276, 1999.

Wei Y, Waltz DA, Rao N, Drummond RJ, Rosenberg S and Chapman HA. Identification of the urokinase receptor as an adhesion receptor for vitronectin. *J Biol Chem* 269: 32380-32388, 1994.

Wei Y, Lukashev M, Simon DI, Bodary SC, Rosenberg S, Doyle MV and Chapman HA. Regulation of integrin function by the urokinase receptor. *Science* 273: 1551-1555, 1996.

Wei Y, Yang X, Liu Q, Wilkins JA and Chapman HA. A role for caveolin and the urokinase receptor in integrin-mediated adhesion and signaling. *J Cell Biol* 144: 1285-1294, 1999.

Wei Y, Eble JA, Wang Z, Kreidberg JA and Chapman HA. Urokinase receptors promote betal integrin function through interactions with integrin alpha3beta1. *Mol Biol Cell* 12: 2975-2986, 2001.

Weigert N, Schaffer K, Schusdziarra V, Classen M and Schepp W. Gastrin secretion from primary cultures of rabbit antral G cells: stimulation by inflammatory cytokines. *Gastroenterology* 110: 147-154, 1996.

Werb Z and Chin JR. Extracellular matrix remodeling during morphogenesis. Ann N Y Acad Sci 857: 110-118, 1998.

Werner S, Peters KG, Longaker MT, Fuller-Pace F, Banda MJ and Williams LT. Large induction of keratinocyte growth factor expression in the dermis during wound healing. *Proc Natl Acad Sci US A* 89: 6896-6900, 1992.

Wilczynska M, Fa M, Ohlsson PI and Ny T. The inhibition mechanism of serpins. Evidence that the mobile reactive center loop is cleaved in the native protease-inhibitor complex. *J Biol Chem* 270: 29652-29655, 1995.

Wilson KT, Ramanujam KS, Mobley HL, Musselman RF, James SP and Meltzer SJ. Helicobacter pylori stimulates inducible nitric oxide synthase expression and activity in a murine macrophage cell line. *Gastroenterology* 111: 1524-1533, 1996.

Wilson KT and Crabtree JE. Immunology of Helicobacter pylori: insights into the failure of the immune response and perspectives on vaccine studies. *Gastroenterology* 133: 288-308, 2007.

Wimalawansa SJ. Effects of in vivo stimulation on molecular forms of circulatory calcitonin and calcitonin gene-related peptide in man. *Mol Cell Endocrinol* 71: 13-19, 1990.

Wiman B and Collen D. On the kinetics of the reaction between human antiplasmin and plasmin. *Eur J Biochem* 84: 573-578, 1978.

Wiman B, Almquist A, Sigurdardottir O and Lindahl T. Plasminogen activator inhibitor 1 (PAI) is bound to vitronectin in plasma. *FEBS Lett* 242: 125-128, 1988.

Wojta J, Nakamura T, Fabry A, Hufnagl P, Beckmann R, McGrath K and Binder BR. Hepatocyte growth factor stimulates expression of plasminogen activator inhibitor type 1 and tissue factor in HepG2 cells. *Blood* 84: 151-157, 1994.

Wojta J, Kaun C, Breuss JM, Koshelnick Y, Beckmann R, Hattey E, Mildner M, Weninger W, Nakamura T, Tschachler E and Binder BR. Hepatocyte growth factor increases expression of vascular endothelial growth factor and plasminogen activator inhibitor-1 in human keratinocytes and the vascular endothelial growth factor receptor flk-1 in human endothelial cells. *Lab Invest* 79: 427-438, 1999.

Wolf BB, Vasudevan J, Henkin J and Gonias SL. Nerve growth factor-gamma activates soluble and receptor-bound single chain urokinase-type plasminogen activator. *J Biol Chem* 268: 16327-16331, 1993.

Wolfsgruber W, Feil S, Brummer S, Kuppinger O, Hofmann F and Feil R. A proatherogenic role for cGMP-dependent protein kinase in vascular smooth muscle cells. *Proc Natl Acad Sci U S A* 100: 13519-13524, 2003.

Wroblewski LE, Pritchard DM, Carter S and Varro A. Gastrin-stimulated gastric epithelial cell invasion: the role and mechanism of increased matrix metalloproteinase 9 expression. *Biochem J* 365: 873-879, 2002.

Wroblewski LE, Noble PJ, Pagliocca A, Pritchard DM, Hart CA, Campbell F, Dodson AR, Dockray GJ and Varro A. Stimulation of MMP-7 (matrilysin) by Helicobacter pylori in human gastric epithelial cells: role in epithelial cell migration. *J Cell Sci* 116: 3017-3026, 2003.

Wu AH, Crabtree JE, Bernstein L, Hawtin P, Cockburn M, Tseng CC and Forman D. Role of Helicobacter pylori CagA+ strains and risk of adenocarcinoma of the stomach and esophagus. Int J Cancer 103: 815-821, 2003.

Wu CW, Li AF, Chi CW, Chung WW, Liu TY, Lui WY and P'eng FK. Hepatocyte growth factor and Met/HGF receptors in patients with gastric adenocarcinoma. *Oncol Rep* 5: 817-822, 1998.

Wu CY, Wang CJ, Tseng CC, Chen HP, Wu MS, Lin JT, Inoue H and Chen GH. Helicobacter pylori promote gastric cancer cells invasion through a NF-kappaB and COX-2-mediated pathway. *World J Gastroenterol* 11: 3197-3203, 2005.

Wu KC, Jackson LM, Galvin AM, Gray T, Hawkey CJ and Mahida YR. Phenotypic and functional characterisation of myofibroblasts, macrophages, and lymphocytes migrating out of the human gastric lamina propria following the loss of epithelial cells. *Gut* 44: 323-330, 1999.

Wun TC, Ossowski L and Reich E. A proenzyme form of human urokinase. J Biol Chem 257: 7262-7268, 1982.

Wysocki AB, Kusakabe AO, Chang S and Tuan TL. Temporal expression of urokinase plasminogen activator, plasminogen activator inhibitor and gelatinase-B in chronic wound fluid switches from a chronic to acute wound profile with progression to healing. *Wound Repair Regen* 7: 154-165, 1999.

Xia W, de BC, Murrell GA and Wang Y. Expression of urokinase-type plasminogen activator and its receptor is up-regulated during tendon healing. *J Orthop Res* 21: 819-825, 2003.

Xiang Z, Censini S, Bayeli PF, Telford JL, Figura N, Rappuoli R and Covacci A. Analysis of expression of CagA and VacA virulence factors in 43 strains of Helicobacter pylori reveals that clinical isolates can be divided into two major types and that CagA is not necessary for expression of the vacuolating cytotoxin. *Infect Immun* 63: 94-98, 1995.

Xu H, Chaturvedi R, Cheng Y, Bussiere FI, Asim M, Yao MD, Potosky D, Meltzer SJ, Rhee JG, Kim SS, Moss SF, Hacker A, Wang Y, Casero RA, Jr. and Wilson KT. Spermine oxidation induced by Helicobacter pylori results in apoptosis and DNA damage: implications for gastric carcinogenesis. *Cancer Res* 64: 8521-8525, 2004.

Xue W, Mizukami I, Todd RF, III and Petty HR. Urokinase-type plasminogen activator receptors associate with beta1 and beta3 integrins of fibrosarcoma cells: dependence on extracellular matrix components. *Cancer Res* 57: 1682-1689, 1997.

Yamamoto T, Noble NA, Cohen AH, Nast CC, Hishida A, Gold LI and Border WA. Expression of transforming growth factor-beta isoforms in human glomerular diseases. *Kidney Int* 49: 461-469, 1996.

Yamaoka Y, Kita M, Kodama T, Sawai N and Imanishi J. Helicobacter pylori cagA gene and expression of cytokine messenger RNA in gastric mucosa. *Gastroenterology* 110: 1744-1752, 1996.

Yamaoka Y, Kita M, Kodama T, Sawai N, Kashima K and Imanishi J. Induction of various cytokines and development of severe mucosal inflammation by cagA gene positive Helicobacter pylori strains. *Gut* 41: 442-451, 1997.

Yamaoka Y, Kodama T, Kita M, Imanishi J, Kashima K and Graham DY. Relationship of vacA genotypes of Helicobacter pylori to cagA status, cytotoxin production, and clinical outcome. *Helicobacter* 3: 241-253, 1998.

Yamaoka Y, Kikuchi S, El-Zimaity HM, Gutierrez O, Osato MS and Graham DY. Importance of Helicobacter pylori oipA in clinical presentation, gastric inflammation, and mucosal interleukin 8 production. *Gastroenterology* 123: 414-424, 2002.

Yamaoka Y, Yamauchi K, Ota H, Sugiyama A, Ishizone S, Graham DY, Maruta F, Murakami M and Katsuyama T. Natural history of gastric mucosal cytokine expression in Helicobacter pylori gastritis in Mongolian gerbils. *Infect Immun* 73: 2205-2212, 2005.

Yang SF, Hsieh YS, Lue KH, Chu SC, Chang IC and Lu KH. Effects of nonsteroidal antiinflammatory drugs on the expression of urokinase plasminogen activator and inhibitor and gelatinases in the early osteoarthritic knee of humans. *Clin Biochem* 41: 109-116, 2008.

Yao X and Forte JG. Cell biology of acid secretion by the parietal cell. Annu Rev Physiol 65: 103-131, 2003.

Ye RD, Ahern SM, Le Beau MM, Lebo RV and Sadler JE. Structure of the gene for human plasminogen activator inhibitor-2. The nearest mammalian homologue of chicken ovalbumin. *J Biol Chem* 264: 5495-5502, 1989.

Yee JA, Yan L, Dominguez JC, Allan EH and Martin TJ. Plasminogen-dependent activation of latent transforming growth factor beta (TGF beta) by growing cultures of osteoblast-like cells. *J Cell Physiol* 157: 528-534, 1993.

Yeoh KG, Ho KY, Guan R and Kang JY. How does chili cause upper gastrointestinal symptoms? A correlation study with esophageal mucosal sensitivity and esophageal motility. *J Clin Gastroenterol* 21: 87-90, 1995.

Yonei Y, Holzer P and Guth PH. Laparotomy-induced gastric protection against ethanol injury is mediated by capsaicin-sensitive sensory neurons. *Gastroenterology* 99: 3-9, 1990.

Zabaleta J, Camargo MC, Piazuelo MB, Fontham E, Schneider BG, Sicinschi LA, Ferrante W, Balart L, Correa P and Ochoa AC. Association of interleukin-1beta gene polymorphisms with precancerous gastric lesions in African Americans and Caucasians. *Am J Gastroenterol* 101: 163-171, 2006.

Zavros Y, Rieder G, Ferguson A, Samuelson LC and Merchant JL. Genetic or chemical hypochlorhydria is associated with inflammation that modulates parietal and G-cell populations in mice. *Gastroenterology* 122: 119-133, 2002.

### Chapter 7 - References

Zavros Y, Rathinavelu S, Kao JY, Todisco A, Del VJ, Weinstock JV, Low MJ and Merchant JL. Treatment of Helicobacter gastritis with IL-4 requires somatostatin. *Proc Natl Acad Sci U S A* 100: 12944-12949, 2003.

## **Publications**

# Increased gastric expression of MMP-7 in hypergastrinemia and significance for epithelial-mesenchymal signaling

Andrea Varro,<sup>1</sup> Susan Kenny,<sup>1</sup> Elaine Hemers,<sup>1</sup> Catherine McCaig,<sup>1</sup> Sabine Przemeck,<sup>2</sup> Timothy. C. Wang,<sup>3</sup> Keith Bodger,<sup>2</sup> and D. Mark Pritchard<sup>2</sup>

<sup>1</sup>Physiological Laboratory, School of Biomedical Sciences, and <sup>2</sup>Division of Gastroenterology, School of Clinical Sciences, University of Liverpool, Liverpool, United Kingdom; and <sup>3</sup>Department of Medicine, Columbia University, Columbia, New York

Submitted 10 November 2006; accepted in final form 3 January 2007

Varro A, Kenny S, Hemers E, McCaig C, Przemeck S, Wang TC, Bodger K, Pritchard DM. Increased gastric expression of MMP-7 in hypergastrinemia and significance for epithelial-mesenchymal signaling. Am J Physiol Gastrointest Liver Physiol 292: G1133-G1140, 2007. First published January 11, 2007; doi:10.1152/ajpgi.00526.2006.-Chronic hypergastrinemia is associated with enterochromaffin-like (ECL) cell hyperplasia, which may progress to gastric carcinoid tumors. The latter consists of epithelial cells and stroma, and both compartments usually regress after normalization of hypergastrinemia. We previously showed that matrix metalloproteinase (MMP)-7 in gastric epithelial cells was upregulated by Heliobacter pylori and described MMP-7-dependent reciprocal signaling between the epithelium and a key stromal cell type, the myofibroblast. Here, we describe the regulation of gastric MMP-7 by gastrin and the potential significance for recruiting and maintaining myofibroblast populations. Biopsies of the gastric corpus and ECL cell carcinoid tumors were obtained from hypergastrinemic patients. Western blot analysis, ELISA, immunohistochemistry, and promoter-luciferase (luc) reporter assays were used to study MMP-7 expression. Gastric myofibroblasts were identified by  $\alpha$ -smooth muscle actin ( $\alpha$ -SMA) expression, and the effects of MMP-7 on myofibroblast proliferation were investigated. In hypergastrinemic patients, there was an increased abundance of MMP-7 and  $\alpha$ -SMA in gastric corpus biopsies and ECL cell carcinoid tumors. In the latter, MMP-7 was localized to ECL cells but not stromal cells, which were nevertheless well represented. Gastrin stimulated MMP-7-luc expression in both AGS-GR and primary human gastric epithelial cells. Conditioned medium from gastrin-treated human gastric glands stimulated myofibroblast proliferation, which was inhibited by neutralizing antibodies to MMP-7. MMP-7 increased the proliferation of myofibroblasts via the MAPK and phosphatidylinositol 3-kinase (PI3K) pathways. In conclusion, stimulation of gastric MMP-7 by elevated plasma gastrin may activate epithelial-mesenchymal signaling pathways regulating myofibroblast function via MAPK and PI3K pathways and contribute to stromal deposition in ECL cell carcinoid tumors.

matrix metalloproteinase-7; pernicious anemia; multiple endocrine neoplasia type 1; enterochromaffin-like cell carcinoid tumors

MUÇOSÁL ORGANIZATION throughout the gastrointestinal tract depends on interactions between the epithelium and subepithelial cells including fibroblasts and myofibroblasts as well as blood vessels, immune cells, and neurons. Recent work has suggested an unexpected diversity in the relevant signaling systems (13). In particular, in addition to growth factors and cytokines, it is now recognized that proteolytic enzymes may also play a role in these signaling processes (13, 20). Matrix

Address for reprint requests and other correspondence: A. Varro, Physiological Laboratory, Univ. of Liverpool, Crown St., Liverpool L69 3BX, UK (e-mail: avarro@liverpool.ac.uk). metalloproteinases (MMPs) are involved in remodeling of the extracellular matrix and the liberation of growth factors and are frequently increased in inflammatory conditions, injury, and cancer (10, 24). Most MMPs are produced in subepithelial cells, but an exception is MMP-7 (also known as matrilysin or PUMP), which is predominantly expressed in epithelial cells both of the gut and other organs including airways, mammary glands, and the urogenital tract (9, 19, 25). Recent work has shown that in the stomach, there is increased MMP-7 with *Heliobacter* pylori infection (2, 4, 33). This is associated with increased myofibroblast abundance, involving both cell proliferation and migration via increased IGF-II bioavailability through cleavage of its binding protein, IGF binding protein (IGFBP)-5, which is released from myofibroblasts (13, 20).

It is well established that a prolonged elevation of plasma gastrin is found in achlorhydria, e.g., pernicious anemia (PA), as well as in gastrinoma, including on a background of multiple endocrine neoplasia (MEN) type-1, and in some patients on prolonged proton pump inhibitors (5, 7, 17, 30). Elevated plasma gastrin is associated with hyperplasia of histamine-secreting enterochromaffin-like (ECL) cells in the gastric corpus (3, 17, 18). In some hypergastrinemic patients, there is a progression via dysplastic ECL cells to ECL cell carcinoid tumors (11, 23); thus, the latter are reported to occur in patients with hypergastrinemia on a background of either gastric inflammation, as occurs in PA, or MEN-1 but are generally uncommon in patients with sporadic gastrinoma (5, 17). In the case of patients with hypergastrinemia of antral origin, resection of the antrum to lower plasma gastrin has been reported by several groups to lead to regression of ECL cell carcinoid tumors (12, 14, 15). Interestingly, both ECL cells and the stroma regress, suggesting that gastrin acts via ECL cells to trigger a stromal reaction. The present study was based on the specific hypothesis that gastrin stimulates MMP-7 expression, which, in turn, stimulates proliferation of a key stromal cell type, the myofibroblast. We therefore asked whether 1) there is increased expression of MMP-7 in gastric biopsies of hypergastrinemic patients and in a transgenic mouse model overexpressing human gastrin, 2) there is also increased MMP-7 in ECL cell carcinoid tumors, 3) gastrin stimulates expression of a MMP-7 promoter/luciferase reporter construct in cultured cells in vitro, 4) increased MMP-7 is accompanied by increased myofibroblast abundance, and 5) MMP-7 stimulates myofibroblast proliferation through mechanisms involving MAPK and phosphatidylinositol 3-kinase (PI3K) pathways.

0193-1857/07 \$8.00 Copyright © 2007 the American Physiological Society

The costs of publication of this article were defrayed in part by the payment of page charges. The article must therefore be hereby marked "*advertisement*" in accordance with 18 U.S.C. Section 1734 solely to indicate this fact.

#### MATERIALS AND METHODS

*Cells, plasmids, and drugs.* The gastric cancer cell line AGS-G<sub>R</sub>, permanently transfected with the CCK-2 receptor, was routinely cultured in Ham's-F-12 medium as previously described (27). Human gastric adherent glands and myofibroblasts were prepared as previously described (20, 33). A vector consisting of 2.3 kb of the human MMP-7 promoter coupled to firefly luciferase (i.e., MMP-7-luc) was kindly donated by Lynn Matrisian (Vanderbilt University, Nashville, TN). Human recombinant (r)MMP-7, MMP-7 fluorogenic substrate, AG-1478, AG-825, Ro-340432, and mouse purified EGF were obtained from Calbiochem (Nottingham, UK). Heptadecapeptide amidated gastrin (G17) was purchased from Peninsula (Merseyside, UK). All other chemicals were obtained from Sigma (Dorset, UK).

Patients. Six endoscopic pinch biopsies of the gastric corpus were obtained from 12 patients with PA and 7 patients with MEN-1 with hypergastrinemia. Five patients with MEN-1 and six patients with PA had macroscopic ECL cell carcinoid tumors, as confirmed by histology. Dyspeptic patients (n = 41) with normal endoscopy, normal plasma gastrin concentrations (<30 pM), negative H. pylori status by serology, antral urease test, and antral and corpus histology were used as controls. In addition, some experiments were made on dyspeptic subjects (n = 8) with normal endoscopy and negative *H. pylori* status but mild fasting hypergastrinemia (30-100 pM). Myofibroblasts were obtained from the macroscopically normal gastric corpus mucosa resected at least 3 cm from the tumor margin in patients undergoing surgery for gastric cancer as previously reported (20). The study was approved by the Ethics Committees of the Salford and Trafford Health Authority (Manchester, UK), South Sefton, and Royal Liverpool and Broadgreen University Hospitals National Health Service Trusts. All patients gave informed consent.

Human gastric gland and myofibroblast preparations. Isolated human gastric glands were prepared from endoscopic biopsies of *H. pylori*-negative patients as previously described (33). Human gastric myofibroblasts were prepared as previously described (20, 34).

Animals. The gastric corpus was taken at 3 mo of age from sex-matched FVB/N, gastrin-overexpressing (InsGas), gastrin knockout (GKO), and C57BI/6 [wild type (WT)] mice for protein extraction and histology as previously described (20). Some GKO mice fed ad libitum were treated with G17 (20 nmol ip at 09.00 and 16.00 hours on day 1 and 09.00 hours on day 2), and the gastric corpus was taken for extraction for Western blot analysis 3 h later. Procedures were approved by the Local Animal Welfare Committee and were in accordance to United Kingdom legislations.

Western blot analysis. Protein extracts of human gastric biopsies and gastric myofibroblasts were prepared, and Western blot analysis was performed as previously described (13, 20). Samples were probed with antibodies to MMP-7 (Chemicon, Chandlers Ford, UK),  $\alpha$ -smooth muscle actin ( $\alpha$ -SMA; RDI, Flanders, NJ), phospho-p42/44 MAPK (New England BioLabs, Hertford, UK), and phospho-Akt (Thr<sup>473</sup>) (Cell Signaling Technology), followed by horseradish peroxidase-conjugated secondary antibody and detection by an incubation with SuperSignal West Pico Chemilminescent Substrate (Pierce) and HyperFilm (Amersham) as previously described (28). Samples were reprobed for GAPDH (Biodesign, Saco, MA) or  $\beta$ -actin (Santa Cruz Biotechnology) and total MAPK using anti-ERK1 (BD Transduction Laboratories, Bedford, MA) or anti-Akt antibodies (BD Transduction Laboratories) as appropriate.

*MMP-7 ELISA*. Plasma samples were processed for the detection of MMP-7 using a Quantikine Human MMP-7 (total) Immunoassay kit (R&D Systems, Minneapolis, MN) according to the manufacturer's instructions.

*Gastrin RIA*. Plasma samples from fasted control, MEN-1, and PA patients and from mice were assayed for total amidated gastrin concentrations using antibody L2 (which reacts with G17 and G34 but not progastrin or Gly-gastrins) as previously described (8). The upper

limit of normal fasting gastrin concentrations in this assay was 30 pM (5, 26).

*Fluorogenic substrate assays.* Medium from gastric glands treated with G17 was incubated with specific MMP-7 fluorogenic substrate, and fluorescence was measured using excitation at 280 nm and emission at 360 nm according to the manufacturer's instruction as previously reported (20).

Immunohistochemistry. Formalin-fixed, paraffin-embedded tissue sections were processed for the immunohistochemical detection of MMP-7 (Chemicon) and  $\alpha$ -SMA (RDI) after antigen recovery as previously described (33).

Transfection and luciferase assays. AGS-G<sub>R</sub> cells  $(2 \times 10^5)$  were plated in full medium. The following day, the medium was removed, and cells were cotransfected with MMP-7-luc (1.0 µg/well) and Renilla luciferase (0.1 µg/well; Promega, Madison, WI) using Trans-Fast (Promega) as previously described (33). Medium was then replaced with 2 ml serum-free medium, and cells were incubated with G17 for 8 h. Luciferase activity was measured with Bright-Glo or Dual Glo (Promega) using a LumiCount Platereader (Packard Bio-Science) (33). In addition, to explore the control of MMP-7-luc in primary gastric epithelial cells, we developed the following new method. Primary human gastric glands (33) were transfected using CombiMag (OzBiosciences, Marseille, France) on a magnetic plate according the manufacturer's instructions, and cells were incubated with G17 for 20 h. Luciferase activity was measured by a dualluciferase assay (Promega). Results in both cases are presented as fold increases over the unstimulated control, so 1.0 signifies no change in luciferase activity. The protein concentration was determined when appropriate using a Bio-Rad detergent compatible protein assay (Bio-Rad, Herts, UK) to monitor the plating efficiency.

Cellular targeting of MMP-7-luc in primary adherent glands. To study the cellular targeting of MMP-7-luc expression, transfected human glands were double immunostained with goat anti-luciferase antibody (Rockland Immunochemicals, Gilbertsville, PA) together with one of the following antibodies: rabbit anti-pepsinogen (a gift from Mike Samloff, Center for Ulcer Research, Los Angeles, CA), anti H<sup>+</sup>-K<sup>+</sup>-ATPase (Calbiochem), anti-vesicle monoamine transporter type 2 (VMAT-2) (16), mouse anti-trefoil factor-1 (TFF-1; Dako, Glostrup, Denmark), and mouse anti-TFF-2 (NovoCastra, Newcastle-upon-Tyne, UK) antibodies, with the appropriate FITC- or Texas red-conjugated secondary antibodies, raised in the donkey (Jackson Immunoresearch, Soham, UK), using Vectashield mounting medium with 4',6-diamidino-2-phenylindole (Vector Laboratories, Peterborough, UK) to counterstain nuclei. Slides were examined using a Zeiss Axioplan-2 microscope (Zeiss Vision, Welwyn Garden City, UK), and images were captured using a JVC-3 charge-coupled device camera and KS300 software combined with deconvolution software (Imaging Associates, Oxfordshire, UK). Ten fields of three patients were counted in each case, and results are expressed as percentages of the total cell number as previously reported (33).

Proliferation assays. The incorporation of [<sup>3</sup>H]thymidine into human gastric myofibroblasts was studied using methods previously described (29). Cells (25,000) were cultured in six-well plates in serum-free medium for 48 h and stimulated for 18 h with conditioned medium from G17-treated glands with or without mouse monoclonal MMP-7 neutralizing antibody (4 µg/ml, Chemicon). [<sup>3</sup>H]thymidine (2 µCi/ml) was added for the last 2 h; cells were processed as previously described (29).

Statistics. Results are presented as means  $\pm$  SE; comparisons were made using ANOVA or Student *t*-tests where appropriate and were considered significant at P < 0.05.

#### RESULTS

Increased gastric MMP-7 in both the human and mouse gastric corpus with hypergastrinemia. In initial studies, we found elevated circulating MMP-7, as determined by ELISA,

AJP-Gastrointest Liver Physiol • VOL 292 • APRIL 2007 • www.ajpgi.org

ే

in patients with hypergastrinemia due to PA and MEN-1 compared with control subjects (Fig. 1A). To determine whether the increase of plasma MMP-7 was reflected in the gastric corpus, we then performed Western blot analysis of gastric biopsies from both patient groups. The abundance of MMP-7 was indeed significantly increased in the gastric corpus of both PA and MEN-1 patients compared with control subjects, compatible with the idea that elevated plasma concentrations of gastrin stimulated MMP-7 expression in the stomach (Fig. 1, B and D). Subsets of both PA and MEN-1 patients exhibited ECL cell carcinoid tumors, which are recognized to be gastrin dependent, and when these were separately sampled, we again found increased MMP-7 abundance compared with control subjects (Fig. 1, C and D). Increased gastric MMP-7 was not, however, associated with mild hypergastrinemia, because in biopsies from subjects with fasting plasma gastrin concentrations in the range of 30-100 pM, MMP-7 abundance was not different from controls (gastrin:  $8.7 \pm 1.3$  pM, n = 9, compared with 58.6  $\pm$  8.4 pM, n = 8; control MMP-7 abundance:  $100 \pm 20.2\%$  compared with  $101.6 \pm 21.9\%$  in hypergastrinemic patients).

To explore a causal link between gastrin and MMP-7, we then examined a mouse model of hypergastrinemia (InsGas), in which there are increases in ECL cell markers at 3 mo (6). We found increased abundance of MMP-7 in the stomach of InsGas mice compared with the parental (FVB/N) strain (Fig. 2, A and B). Moreover, treatment of mice null for the gastrin gene with exogenous G17 also increased MMP-7 abundance, whereas there were no significant differences in MMP-7 between GKO mice and their WT counterparts (Fig. 2, C and D).

Gastrin stimulates MMP-7 promoter activity in primary cells. To determine whether gastrin might regulate MMP-7 transcription, we initially studied the response of a promoterluciferase reporter vector (MMP-7-luc) transfected into AGS-G<sub>R</sub> cells; these cells were chosen because they have been widely used in the past as an experimental model in studies (32, 37) of gene expression in ECL cells (Fig. 3A). In response to concentrations of gastrin in the range of 100 pM-10 nM, there was a progressive increase in MMP-7-luc expression; the concentration for half-maximal stimulation was  $\sim 2$  nM. We then asked whether the same stimulation occurs in primary human gastric gland cells. To this end, we developed a novel method for transfection of adherent glands with MMP-7-luc as described in materials and methods. In primary human gastric glands, G17 (1 nM, 20 h) also significantly stimulated MMP-7luc expression (Fig. 3B). Using these methods,  $5.7 \pm 1.2\%$  of all cells expressed the construct. Importantly, expression of MMP-7-luc was targeted to subsets of TFF-1-positive surface epithelial, chief (pepsinogen), and ECL (VMAT-2) cells (Fig. 4, A–I, and Table 1). There was no targeting of the construct to parietal cells (H<sup>+</sup>-K<sup>+</sup>-ATPase) (Fig. 4J) or to TFF-2-positive neck cells (Fig. 4K). The former could, however, be transfected with other constructs, for example, the promoter of the  $\beta$ -subunit of H<sup>+</sup>-K<sup>+</sup>-ATPase coupled with green fluorescent protein (GFP). Moreover, a control GFP-vector was expressed in all cell types in this system (not shown).



Fig. 1. Increased circulating and gastric matrix metalloproteinase (MMP)-7 in hypergastrinemia. A: ELISA showed a modest elevation of circulating MMP-7 in patients with pernicious anemia (PA; n = 6) or multiple endocrine neoplasia type 1 (MEN-1; n = 5) compared with controls (n = 8). B: Western blots of the gastric corpus from PA, MEN-1, and control subjects probed for MMP-7. C: Western blots of gastric carcinoid tumors from PA (N1), MEN-1 (N2), and control (C) subjects probed for MMP-7 and  $\beta$ -actin. D: quantification of Western blots showed an increase in MMP-7 in the corpus of PA patients (n = 9) and MEN-1 patients (n = 7) and in the carcinoid nodules of PA and MEN-1 patients (n = 5). In A and D, plasma gastrin concentrations are shown for the relevant patients used for Western blots or ELISA in each case. \*P < 0.05 vs. control.

#### GASTRIN-REGULATED MMP-7 EXPRESSION IN THE STOMACH

Fig. 2. Increased MMP-7 in mice with hypergastrinemia. A: Western blots of the gastric corpus from gastrin-overexpressing (InsGas) and FVB/N mice probed for MMP-7 and GAPDH. B: quantification of Western blots showed an ~2-fold increase in MMP-7 in the corpus of InsGas (n = 5) compared with FVB/N (n = 7) mice normalized to GAPDH. There was also an  $\sim$ 3-fold increase in plasma gastrin in these mice. \*P < 0.05 vs. FVB/N mice. C: Western blots of the gastric corpus from C57Bl/6 and gastrin knockout (GKO) mice treated with or without G17 and probed for MMP-7 and β-actin. D: quantification of Western blots showed an ~2-fold increase in MMP-7 in the corpus of GKO mice treated with G17 (n = 3) compared with vehicletreated control (n = 3) mice, whereas there were no differences in MMP-7 in the corpus of C57B1/6 mice compared with untreated GKO (n = 3) mice normalized to  $\beta$ -actin. RIA confirmed high circulating plasma gastrin concentrations (~5 nM) in gastrin-treated GKO mice, whereas plasma gastrin concentrations in wild-type mice were normal (56.7  $\pm$  7.2 pM). \*P < 0.05 vs. C57Bl/6 mice.

Increased abundance of the myofibroblast marker  $\alpha$ -SMA in both the human and mouse gastric corpus with hypergastrinemia. Previous work has established that MMP-7 is localized mainly to epithelial cells but not to stromal cells in the gastric mucosa (33). In line with this, in the present study, we found strong immunostaining of MMP-7 in ECL cells in gastric carcinoid tumors (Fig. 5A). In contrast, there was little or no staining of stromal cells. Even so, myofibroblasts identified by staining with antibody to  $\alpha$ -SMA were abundant in gastric ECL cell carcinoid tumors and were often dispersed throughout the tumor (Fig. 5B). There was also increased  $\alpha$ -SMA abundance, as detected by Western blot analysis in the corpus of both PA and MEN-1 patients, including ECL cell nodules, compared with the mucosa from normal subjects (Fig. 6, A and B). Similarly, there was increased MMP-7 and  $\alpha$ -SMA in mice overexpressing gastrin (Fig. 6, C and D).

*MMP-7 released by gastrin from cultured gastric glands stimulates human gastric myofibroblast proliferation.* The presence of abundant myofibroblasts in gastric carcinoid tumors suggested that there might be a gastrin-activated signaling pathway from the epithelium to myofibroblasts. To examine whether MMP-7 might be one such mediator, we first treated gastric gland cultures with G17 and measured MMP-7



Fig. 3. Gastrin regulation of MMP-7-luc in both AGS-G<sub>R</sub> and primary gastric epithelial cells. *A*: in AGS-G<sub>R</sub> cells transiently transfected with MMP-7-luc, G17 (8 h) produced a concentration-dependent increase in luciferase activity. *B*: G17 (1 nM, 20 h) also stimulated MMP-7-luc expression in human primary cultured gastric glands. \*P < 0.05 vs. control.



enzyme activity. G17 (1 nM, 20 h) stimulated MMP-7 enzyme activity by 272.2  $\pm$  36% compared with control (100  $\pm$  19.9%, P < 0.05). Moreover, conditioned medium from cultured human gastric glands treated with G17 stimulated human gastric myofibroblast proliferation, and this was significantly reduced by MMP-7 neutralizing antibodies (Fig. 7A), whereas the proliferative effect of another growth factor, EGF (50 ng/ml), was not affected (EGF: 199.4  $\pm$  23.7%; EGF + MMP-7 antibody: 177.6  $\pm$  15.8%). G17 itself had no effect on myofibroblast proliferation (control: 100  $\pm$  5.56%; 1 nM G17: 89.1  $\pm$  10.3%). These data are, therefore, compatible with the idea that native MMP-7 released from epithelial cells by gastrin stimulates myofibroblast proliferation.

MMP-7 stimulates human gastric myofibroblast proliferation via MAPK and PI3K. To examine the cellular mechanisms that mediated the effects of MMP-7 on myofibroblast proliferation, we then considered the involvement of MAPK and PI3K pathways, as these have been previously implicated in the proliferation of colonic myofibroblasts (13). The MEK inhibitor U-0126 (10 µM) reversed the proliferative effect of recombinant MMP-7 in a concentration (2 µg/ml) that had been previously shown to be optimal for myofibroblast proliferation (20) and so too did the PI3K inhibitor LY-294002 (50  $\mu$ M; Fig. 7B). Compatible with the idea that MMP-7 activates the MAPK pathway, we found increased phosphorylation of p42/44 ERK, which peaked within 10 min of the application of rMMP-7 (Fig. 7C). Similarly, there was also increased phosphorylation of Akt, which is a known target of PI3K; the time course was distinct from that of p42/44 ERK activation with a progressive increase in phosphorylation after up to 240 min of incubation in MMP-7 (Fig. 7D). The activation of ERK and Akt pathways occurred in parallel, since the MEK inhibitor U-0126 inhibited MMP-7-induced p42/44 but not Akt phosphorylation and, vice versa, the PI3K inhibitor LY-294002 inhibited MMP-7-induced Akt but not p42/44 phosphorylation. In contrast, the PKC inhibitor Ro-320432, the EGF receptor tyrosine kinase inhibitor AG-1478, and the Erb-2 receptor tyrosine kinase inhibitor AG-825 had no effect on MMP-7stimulated proliferation, suggesting that these signaling molecules are not implicated in the actions of MMP-7.

#### GASTRIN-REGULATED MMP-7 EXPRESSION IN THE STOMACH

A С В F E D G Н

#### DISCUSSION

The present study was based on the hypothesis that elevated plasma gastrin concentrations stimulate gastric MMP-7 expression, which, in turn, increases myofibroblast proliferation. The main findings of the study were that MMP-7 is increased in ECL cell carcinoid tumors and the gastric corpus of hypergastrinemic patients and that gastrin acts on gastric glands to release MMP-7 in sufficient concentrations to trigger myofibroblast proliferation. ECL tumors provide an interesting

model for studies of epithelial-stromal interactions since their growth is dependent on gastrin, which, in turn, is thought to act exclusively on epithelial cells. When gastrin is removed in PA patients by antrectomy, there is tumor regression, including the stromal component, suggesting that ECL cells release factors regulating stromal cell abundance and deposition of extracellular matrix proteins (14, 15). We suggest now that gastrin activates a sequence of events that includes ECL cell proliferation and increased expression of MMP-7, which then stimu-

AJP-Gastrointest Liver Physiol • VOL 292 • APRIL 2007 • www.ajpgi.org



G1137

Fig. 4. A 2.3-kb portion of the human MMP-7 promoter targets expression to specific cell types in primary human epithelial cells. A-C: MMP-7-luc (green) expression in trefoil factor (TFF)-1 (red)-positive cells.

D-F: MMP-7-luc (green) expression in pepsinogen (red)-positive chief cells. G-I: MMP-7-luc (green) expression in vesicle mono-

amine transporter type 2 (VMAT-2; red)contaning enterochromaffin-like (ECL) cells. J and K: MMP-7-luc (green) expression was not colocalized with  $H^+$ - $K^+$ -ATPase (J; red)

or TFF-2 (K; red).

#### GASTRIN-REGULATED MMP-7 EXPRESSION IN THE STOMACH

Table 1. Percentages of MMP-7-luc-expressing cells that also express TFF-1, VMAT-2, or pepsinogen

|  | TFF-1    | VMAT-2   | Pepsinogen |
|--|----------|----------|------------|
| Percentage of MMP-7-<br>luc-expressing cells | 12.3±0.7 | 36.8±3.9 | 34.4±0.7   |

Values are means  $\pm$  SE. MMP, matrix metalloproteinase; luc, luciferase; TFF, trefoil factor; VMAT, vesicle monoamine transporter. Note that <1% of MMP-7-luc-expressing cells also expressed TFF-2 or H<sup>+</sup>-K<sup>+</sup>-ATPase (n = 3).

lates myofibroblast proliferation, expanding the extracellular matrix around ECL cells.

Previous studies (2, 4, 33) have shown increased gastric epithelial expression of MMP-7 with H. pylori infection. Recently, we (13, 20) reported the consequences of increased epithelial MMP-7 production by H. pylori on the gastric microenvironment, namely, rapid cleavage of IGFBP-5 by secreted MMP-7, followed by increased bioavailability of IGF-II from myofibroblasts. IGF-II stimulated the proliferation and migration of myofibroblasts and increased the proliferation of epithelial cells. The present data suggest that hypergastrinemia is also associated with increased MMP-7 expression. Our data suggest that gastrin increases MMP-7 expression at concentrations only observed in frank hypergastrinemia, since it was observed in groups with plasma gastrin concentrations of >100 pM but not in subjects with fasting plasma gastrin concentrations of 30-100 pM. Moreover, in a transgenic mouse model of hypergastrinemia (31), we also found increased MMP-7. In both humans and mice, circulating gastrin concentrations above 100 pM appeared to be required for increased MMP-7. The available data do not, therefore, suggest that there is physiological control of MMP-7 by gastrin. Importantly, however, patients exhibiting increased MMP-7, i.e., with prolonged plasma gastrin concentrations of >100 pM, include those most at risk of developing ECL cell carcinoid tumors (5). It is possible that the apparent elevation of MMP-7 simply reflects increased ECL cell numbers (3, 18, 23) without changed expression at the level of single cells. Set against this, however, it should be noted that exogenous gastrin increased MMP-7 in mice null for gastrin over a time course that could not be attributed to increased ECL cell numbers, and, in studies of cultured cells, we observed direct stimulation of MMP-7 expression in response to gastrin.

Thus, in addition to describing increased MMP-7 in vivo, we have also shown in an in vitro system that gastrin was able to stimulate the expression of an MMP-7-promoter/luciferase reporter construct at concentrations similar to those in the plasma of hypergastrinemic

patients. For these experiments, we first made use of AGS-GR cells as this system has been intensively used for similar previous studies (32, 37). However, to establish whether primary cells might also express MMP-7-luc, we developed a novel methodology to transfect primary human gastric epithelial cells using cultured gland cells in which all the main cell types are preserved and in which cell-cell junctions are intact. In this system, MIMP-7-luc was expressed in a cell-restricted manner. Importantly, the pattern of expression of luciferase driven from the MMP-7 promoter resembled that of the WT gene (33). In particular, subsets of TFF-1-, pepsinogen-, and VMAT-2-expressing cells all exhibited luciferase activity after transfection, whereas TFF-2- and H+-K+-ATPase-expressing cells did not. Further work will be necessary to define in detail the molecular basis of this expression. It is, however, important to note that luciferase expression in primary cells was increased by gastrin in at concentration of 1 nM. In the future, this method should be generally applicable to studies of other genes normally expressed in cell-restricted patterns in the gastric epithelium. In the present context, the main conclusion was that 2.3 kb of the MIMP-7 promoter is sufficient to determine the physiological pattern of expression in the stomach and to support increased expression in response to gastrin.

It is increasingly clear that myofibroblasts play an important role in determining the organization of epithelia during development, in wound healing, and in cancer. The identification of MMP-7 as a novel target of gastrin in this regard therefore raises new possibilities for manipulating epithelial-stromal interactions both experimentally and therapeutically in ECL cell carcinoid tumors. Our data suggest that MMP-7 is able to stimulate MAPK and PI3K pathways in myofibroblasts, leading to an expansion in cell numbers. These pathways are often activated in inflammation, injury, and cancer due to increased growth factor production or mutations leading to accelerated cell proliferation.

Previous work has established the expression of MMP-7 in gastric cancer and an association with poor outcome (1, 21, 35). The possible cellular mechanisms are still largely unexplored. They may include remodeling of the extracellular matrix leading to fibrosis (36, 38), increased cell invasion, or suppression of apoptosis (10, 25). Our data suggest that there is also increased stimulation of stromal cells, and, because these in turn produce growth factors acting on the epithelium, an aggressive cycle of stimulation may be established. The proliferative responses of myofibroblasts to MMP-7 appear to depend on its proteolytic activity, since we have found that heat-inactivated MMP-7 has no effect (data not shown). While the available data suggest that MMP-7 works via cleavage of

Fig. 5. Immunohistochemical localization of MMP-7 and the myofibroblast marker  $\alpha$ -smooth muscle actin ( $\alpha$ -SMA) in an ECL cell carcinoid tumor from a PA patient. A: MMP-7 immunoreactivity was localized to endocrine cells but not stromal cells in the carcinoid tumor. B:  $\alpha$ -SMA was localized to myofibroblasts that surround tumor cells. Magnification: ×40.



MMP-7



α-smooth muscle actin



Fig. 6. Increased  $\alpha$ -SMA abundance in both humans and mice with hypergastrinemia. A: Western blot of the gastric corpus from both PA and MEN-1 patients and an ECL cell carcinoid biopsy (N) from a PA patient showed increased  $\alpha$ -SMA compared with the control sample. B: quantification of Western blots showed increased  $\alpha$ -SMA in both the gastric corpus of PA (n = 11) and MEN-1 (n = 6) patients and in the ECL cell carcinoid tumors itself (n = 6) compared with controls (n =14) normalized to GAPDH. \*P < 0.05 vs. control. C: Western blot of the gastric corpus from InsGas mice showed increased  $\alpha$ -SMA compared with FVB/N mice. D: quantification of Western blots showed increased  $\alpha$ -SMA in the gastric corpus of InsGas (n = 5) compared with FVB/N (n = 7) mice normalized to GAPDH. \*P < 0.05 vs. FVB/N mice.

IGFBP-5 and liberation of IGF-II, it is worth noting that at least in colonic myofibroblasts, there is some evidence that MMP-7 activates other MMPs that participate in the response (13); further work will be need to determine whether comparable protease interactions occur in the present system. Moreover, the direct evidence for MMP-7 stimulation of myofibroblast cell proliferation comes from studies in vitro, and there is now a need to extend this work to include direct studies of the effects of MMP-7 on myofibroblast cell numbers in vivo.

Taken as a whole, the present data identify MMP-7 as a previously unsuspected target of gastrin that is increased in hypergastrinemic conditions. We also show that in at least one condition, ECL cell carcinoid tumors, MMP-7 is a potential regulator of the stromal compartment by action on myofibroblast cells via activation of MAPK and PI3K pathways. Previous studies (13, 20) have established that MMP-7 degrades

IGFBP-5 released by myofibroblasts, thereby liberating IGF-II. Moreover, there is already evidence that in different cellular systems, gastrin is able to activate distinct paracrine pathways stimulating EGF receptors (29) and FGF receptors (22). However, whereas previous studies were based on a cancer cell line (AGS cells), the present work focused on primary epithelial cells and myofibroblasts. The relative importance of these different pathways in the normal stomach remains to be established; indeed, it should be noted that the gastric myofibroblasts used in our study were obtained from tissue adjacent to gastric tumors and may well not be representative of myofibroblasts in normal gastric tissue. Even so, the present study indicates that, in the future, it will be profitable to determine the precise molecular mechanisms by which MMP-7 contributes to the maintenance of the stromal compartment in epithelial tumors.

> Fig. 7. Conditioned medium (CM) from gastrintreated human gastric glands stimulates proliferation of human gastric myofibroblasts by MMP-7 via MAPK and phosphatidylinositol 3-kinase (PI3K). A: proliferation of human gastric myofibroblasts was stimulated by coculture with CM media from G17 (1 nM, 20 h)-treated gastric glands. Treatment of glands with neutralizing mouse monoclonal MMP-7 antibody (Ab) partially inhibited proliferation. Treatment of myofibroblasts with nonimmune mouse IgG had no effect (n = 3). B: stimulation of [<sup>3</sup>H]thymidine incorporation by recombinant MMP-7 (3 U/ml) in primary human gastrin-treated myofibroblasts was virtually abolished by the MEK inhibitor U-0126 and the PI3K inhibitor LY-294002. C and D: Western blots showed increased abundance of phospho-p42/ 44Erk 10 min after the addition of recombinant MMP-7 (C) and increased abundance of phosphorylated Akt (p-Akt) 240 min after the incubation with recombinant MMP-7 (D).↔, significant differences between MMP-7-treated and control samples. \*Decreases in MMP-7-stimulated proliferation following application of the compound specified.



AJP-Gastrointest Liver Physiol • VOL 292 • APRIL 2007 • www.ajpgi.org

#### GASTRIN-REGULATED MMP-7 EXPRESSION IN THE STOMACH

#### ACKNOWLEDGMENTS

We are grateful to Cath McLean for help with the gastrin assays, Andy Dodson for help with the immonohistochemistry, and Katherine McKittrick for help with the mice. We also thank David Thompson, Gordon Armstrong, and the late Robin Walker for help in patient recruitment, Lynn Matrisian for the MMP-7-luc construct, and Graham Dockray for helpful comments.

#### GRANTS

This work is funded by The North West Cancer Research Fund, The Wellcome Trust, and The Medical Research Council.

#### REFERENCES

- Adachi Y, Itoh F, Yamamoto H, Matsuno K, Arimura Y, Kusano M, Endoh T, Hinoda Y, Oohara M, Hosokawa M, Imai K. Matrix metalloproteinase matrilysin (MMP-7) participates in the progression of human gastric and esophageal cancers. *Int J Oncol* 13: 1031–1035, 1998.
- Bebb JR, Letley DP, Thomas RJ, Aviles F, Collins HM, Watson SA, Hand NM, Zaitoun A, Atherton JC. *Helicobacter pylori* upregulates matrilysin (MMP-7) in epithelial cells in vivo and in vitro in a Cag dependent manner. *Gut* 52: 1408–1413, 2003.
- Bordi C, D'Adda T, Azzoni C, Pilato FP, Caruana P. Hypergastrinemia and gastric enterochromaffin-like cells. Am J Surg Pathol 19, Suppl 1: S8–S19, 1995.
- Crawford HC, Krishna US, Israel DA, Matrisian LM, Washington MK, Peek RM Jr. Helicobacter pylori strain-selective induction of matrix metalloproteinase-7 in vitro and within gastric mucosa. *Gastroenterology* 125: 1125–1136, 2003.
- Dockray GJ. Clinical endocrinology and metabolism. Gastrin Best Pract Res Clin Endocrinol Metab 18: 555–568, 2004.
- Dockray GJ, Bate GW, Hormi K, Wang TC, Varro A, Dimaline R. Genetically engineered animal models: physiological studies with gastrin in transgenic mice. In: *Mechanisms and Consequences of Proton Transport*, edited by Urushidani T, Forte JG, Sachs G. Boston, MA: Kluwer, 2002, p. 159–167.
- 7. Dockray G, Dimaline R, Varro A. Gastrin: old hormone, new functions. *Pflügers Arch* 449: 344–355, 2005.
- Dockray GJ, Hamer C, Evans D, Varro A, Dimaline R. The secretory kinetics of the G cell in omeprazole-treated rats. *Gastroenterology* 100: 1187–1194, 1991.
- Dunsmore SE, Saarialho-Kere UK, Roby JD, Wilson CL, Matrisian LM, Welgus HG, Parks WC. Matrilysin expression and function in airway epithelium. J Clin Invest 102: 1321–1331, 1998.
- 10. Egeblad M, Werb Z. New functions for the matrix metalloproteinases in cancer progression. *Nat Rev Cancer* 2: 161–174, 2002.
- Gibril F, Schumann M, Pace A, Jensen RT. Multiple endocrine neoplasia type 1 and Zollinger-Ellison syndrome: a prospective study of 107 cases and comparison with 1009 cases from the literature. *Medicine* (*Baltimore*) 83: 43–83, 2004.
- 12. Guillem P, Vlaeminck-Guillem V, Leteurtre E, Kornhauser R, Cortot A, Wemeau JL, Triboulet JP. [Fundic endocrine tumors and atrophic gastritis: the value of antrectomy]. *Gastroenterol Clin Biol* 26: 782–785, 2002.
- Hemers E, Duval C, McCaig C, Handley M, Dockray GJ, Varro A. Insulin-like growth factor binding protein-5 is a target of matrix metalloproteinase-7: implications for epithelial-mesenchymal signaling. *Cancer Res* 65: 7363–7369, 2005.
- Higham AD, Dimaline R, Varro A, Attwood S, Armstrong G, Dockray GJ, Thompson DG. Octreotide suppression test predicts beneficial outcome from antrectomy in a patient with gastric carcinoid tumor. *Gastroenterology* 114: 817–822, 1998.
- 15. Hirschowitz BI, Griffith J, Pellegrin D, Cummings OW. Rapid regression of enterochromaffinlike cell gastric carcinoids in pernicious anemia after antrectomy. *Gastroenterology* 102: 1409–1418, 1992.
- Hussain I, Bate GW, Henry J, Djali P, Dimaline R, Dockray GJ, Varro A. Modulation of gastrin processing by vesicular monoamine transporter type 1 (VMAT1) in rat gastrin cells. J Physiol 517: 495–505, 1999.
- Jensen RT. Involvement of cholecystokinin/gastrin-related peptides and their receptors in clinical gastrointestinal disorders. *Pharmacol Toxicol* 91: 333–350, 2002.
- Larsson H, Carlsson E, Mattsson H, Lundell L, Sundler F, Sundell G, Wallmark B, Watanabe T, Hakanson R. Plasma gastrin and gastric enterochromaffinlike cell activation and proliferation. Studies with omeprazole and ranitidine in intact and antrectomized rats. *Gastroenterology* 90: 391–399, 1986.

- Lopez-Boado YS, Wilson CL, Hooper LV, Gordon JI, Hultgren SJ, Parks WC. Bacterial exposure induces and activates matrilysin in mucosal epithelial cells. J Cell Biol 148: 1305–1315, 2000.
- McCaig C, Duval C, Hemers E, Steele E, Pritchard DM, Przemeck S, Dimaline R, Ahmed S, Bodger K, Kerrigan DD, Wang TC, Dockray GJ, Varro A. The role of matrix metalloproteinase (MMP)-7 in redefining the gastric micro-environment in response to *Helicobacter pylori*. *Gastroenterology* 130: 1754–1763, 2006.
- McDonnell S, Navre M, Coffey RJ Jr, Matrisian LM. Expression and localization of the matrix metalloproteinase pump-1 (MMP- 7) in human gastric and colon carcinomas. *Mol Carcinog* 4: 527–533, 1991.
- Noble PJ, Wilde G, White MR, Pennington SR, Dockray GJ, Varro A. Stimulation of gastrin-CCKB receptor promotes migration of gastric AGS cells via multiple paracrine pathways. Am J Physiol Gastrointest Liver Physiol 284: G75–G84, 2003.
- Peghini PL, Annibale B, Azzoni C, Milione M, Corleto VD, Gibril F, Venzon DJ, Delle FG, Bordi C, Jensen RT. Effect of chronic hypergastrinemia on human enterochromaffin-like cells: insights from patients with sporadic gastrinomas. *Gastroenterology* 123: 68-85, 2002.
- 24. Sternlicht MD, Werb Z. How matrix metalloproteinases regulate cell behavior. *Annu Rev Cell Dev Biol* 17: 463–516, 2001.
- Vargo-Gogola T, Fingleton B, Crawford HC, Matrisian LM. Matrilysin (matrix metalloproteinase-7) selects for apoptosis-resistant mammary cells in vivo. *Cancer Res* 62: 5559–5563, 2002.
- Varro A, Ardill JE. Gastrin: an analytical review. Ann Clin Biochem 40: 472–480, 2003.
- Varro A, Hemers E, Archer D, Pagliocca A, Haigh C, Ahmed S, Dimaline R, Dockray GJ. Identification of plasminogen activator inhibitor-2 as a gastrin- regulated gene: role of Rho GTPase and menin. *Gastroenterology* 123: 271–280, 2002.
- Varro A, Noble PJ, Pritchard DM, Kennedy S, Hart CA, Dimaline R, Dockray GJ. *Helicobacter pylori* induces plasminogen activator inhibitor 2 (PAI-2) in gastric epithelial cells through NF-kB and RhoA: implications for invasion and apoptosis. *Cancer Res* 64: 1695–1702, 2004.
- 29. Varro A, Noble PJ, Wroblewski LE, Bishop L, Dockray GJ. Gastrincholecystokinin(B) receptor expression in AGS cells is associated with direct inhibition and indirect stimulation of cell proliferation via paracrine activation of the epidermal growth factor receptor. *Gut* 50: 827–833, 2002.
- Walsh JH. Gastrin. In: *Gut Peptides*, edited by Walsh JH, Dockray GJ. New York: Raven, 1994, p. 75-121.
- Wang TC, Dangler CA, Chen D, Goldenring JR, Koh T, Raychowdhury R, Coffey RJ, Ito S, Varro A, Dockray GJ, Fox JG. Synergistic interaction between hypergastrinemia and *Helicobacter* infection in a mouse model of gastric cancer. *Gastroenterology* 118: 36–47, 2000.
- Watson F, Kiernan RS, Deavall DG, Varro A, Dimaline R. Transcriptional activation of the rat vesicular monoamine transporter 2 promoter in gastric epithelial cells: regulation by gastrin. J Biol Chem 276: 7661–7671, 2001.
- 33. Wroblewski LE, Noble PJ, Pagliocca A, Pritchard DM, Hart CA, Campbell F, Dodson AR, Dockray GJ, Varro A. Stimulation of MMP-7 (matrilysin) by *Helicobacter pylori* in human gastric epithelial cells: role in epithelial cell migration. J Cell Sci 116: 3017–3026, 2003.
- 34. Wu KC, Jackson LM, Galvin AM, Gray T, Hawkey CJ, Mahida YR. Phenotypic and functional characterisation of myofibroblasts, macrophages, and lymphocytes migrating out of the human gastric lamina propria following the loss of epithelial cells. *Gut* 44: 323–330, 1999.
- 35. Yamamoto H, Horiuchi S, Adachi Y, Taniguchi H, Nosho K, Min Y, Imai K. Expression of ets-related transcriptional factor E1AF is associated with tumor progression and over-expression of matrilysin in human gastric cancer. *Carcinogenesis* 25: 325–332, 2004.
- 36. Yu WH, Woessner JF Jr, McNeish JD, Stamenkovic I. CD44 anchors the assembly of matrilysin/MMP-7 with heparin-binding epidermal growth factor precursor and ErbB4 and regulates female reproductive organ remodeling. *Genes Dev* 16: 307–323, 2002.
- Zhang Z, Hocker M, Koh TJ, Wang TC. The human histidine decarboxylase promoter is regulated by gastrin and phorbol 12-myristate 13-acetate through a downstream cis-acting element. J Biol Chem 271: 14188–14197, 1996.
- 38. Zuo F, Kaminski N, Eugui E, Allard J, Yakhini Z, Ben Dor A, Lollini L, Morris D, Kim Y, DeLustro B, Sheppard D, Pardo A, Selman M, Heller RA. Gene expression analysis reveals matrilysin as a key regulator of pulmonary fibrosis in mice and humans. *Proc Natl Acad Sci USA* 99: 6292-6297, 2002.